# FINAL

# SITE INVESTIGATION WORK PLAN PHASE I

# AMES STREET SITE ROCHESTER, NEW YORK

Prepared by:

ABB Environmental Services, Inc. Rochester, New York

February 1996

ABB Environmental Services, Inc.



# AMES STREET SITE SITE INVESTIGATION WORK PLAN - PHASE 1

# TABLE OF CONTENTS

Section	Title	Page No.
1.0 INTR	ODUCTION	
2.0 SITE	HISTORY	
2.1	General History	
2.2	Site Description	
	2.2.1 Site Geology2.2.2 Hydrogeology	
3.0 SUM	MARY OF PREVIOUS ENVIRONMENTAL INVESTIGATION	3-1
3.1	Class 4 Area Investigations (1981-1986)	3-1
3.2	Water Tower Area Investigation (1984-1985)	3-2
3.3	Building 4 Sampling (1987)	
3.4	Building 8 Sampling (1989)	
3.5	Tank Closures	
3.6	Site Investigation (1993)	
3.7	Sub-slab Soil Sampling (1993)	
3.8	Sampling During Demolition (1995)	
4.0 <b>POTE</b>	ENTIAL AREAS OF CONCERN AND CHEMICALS OF CON	CERN 4-1
4.1	Areas of Concern	
4.2	Potential Chemicals of Concern	4-5
5.0 SC	OPE OF WORK	
5.1	AOC Investigations - Phase 1 Soil Boring Program	5-1
	5.1.1 Boring Location and Rationale	5-1
	5.1.2 Soil Sampling	5-4
	5.1.3 Analysis	
5.2	Groundwater Sampling	
	5.2.1 Temporary Well Point Locations	5-7
	5.2.2 Groundwater Sampling and Analysis	5-8

.

# AMES STREET SITE SITE INVESTIGATION WORK PLAN - PHASE 1

# TABLE OF CONTENTS

Section	Title	Page No.
5.3	Risk Assessment and Derivation of Risk-Based Goals	
	5.3.1 Data Evaluation and Summarization	5-10
	5.3.2 Identification of HHRA Chemicals of Potential Concern	5-10
	5.3.3 Exposure Assessment	5-11
	5.3.4 Toxicity Assessment	5-11
	5.3.5 Risk Characterization	5-14
	5.3.6 Use of Risk-based Mercury Screening Concentrations in the	
	Sampling and Analysis Program	5-14
	5.3.6.1 Setting the Mercury Screening Levels for Groundw	ater . 5-14
	5.3.6.2 Setting the Mercury Screening Levels for Soil	5-15
	5.3.7 Development of Risk-Based Goals	5-17
	5.3.7.1 Setting a Site-Specific Risk- and Future Use-based	
	Mercury Groundwater Goal.	5-17
	5.3.7.2 Setting a Site-Specific Risk- and Future Use-based	
	Mercury Soil Goal.	5-17
5.4	Data Quality Objectives	5-23
	5.4.1 Overview	5-23
	5.4.2 Intended Data Uses	5-23
	5.4.3 Chemicals of Concern	5-24
	5.4.4 Sample Quantitation Limits	5-24
5.5	Analytical Program	5-24
5.6	Data Validation	5-27
60 PHASE	T 1 REPORT	6-1
0.0 111/01		
7.0 SCHEI	DULE	7-1

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

-

# AMES STREET SITE SITE INVESTIGATION WORK PLAN - PHASE 1

# LIST OF FIGURES

<u>Figur</u>	Title	Page No.
1-1	Site Location Map	1-3
2-1	Building Locations	2-2
<b>4-1</b> <sup>1</sup>	Areas of Concern	oversize
5-1 5-2 5-3 5-4 5-5	Sampling Locations	oversize 5-19 5-20 5-21 5-26
7-1	Schedule	7-2

# LIST OF TABLES

<u>Table</u>	Title	Page No.
4-1 4-2	Area of Concern Summary	4-2 4-6
5-1 5-2 5-3	Soil Boring SummarySoil Analytical ProgramGroundwater Analytical ProgramSoil Analytical Program	
5-4	Summary of Potential Exposure Pathways and Receptors	5-12

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

7198-23

### **1.0 INTRODUCTION**

ABB Environmental Services (ABB-ES), at the request of Combustion Engineering, Inc. has prepared this Work Plan for subsurface investigations at the former Taylor Instruments facility at 95 Ames Street (Ames Street Site), Rochester, Monroe County, New York (Figure 1-1). The Work Plan presents the technical approach and rationale, as well as the methodologies, that will be used to conduct soil and groundwater sampling at the Ames Street Site. The tasks presented in this plan compose Phase 1 of a planned two-phase voluntary investigation and site restoration.

Phase 1 is designed to identify concentrations of site-related chemicals in overburden soils that represent soil contact hazards or that may act as sources of contamination to groundwater. To accomplish this, soil sampling is targeted to areas of concern (AOCs) across the site. The AOCs were developed based on evaluation of historical records, previous environmental data from the site, and discussions with New York State Department of Environmental Conservation (NYSDEC) and the New York State Department of Health (including meetings on September 13 and October 6 and 30, 1995). Additional site knowledge obtained during observation of the demolition and removal of site structures, foundation slabs, and underground storage tanks (completed in December, 1995) has been incorporated in the development of the AOCs.

Phase 1 will also include perimeter groundwater sampling to assess groundwater quality leaving the site. Three geologic borings extending into bedrock are planned to characterize subsurface geology in addition to data from soil borings at the AOCs.

In general, the potential impact from historic operations will be assessed through analysis of soil and groundwater for three types of chemical compounds:

- 1. volatile organic compounds (VOCs) from petroleum-based products and cleaning/degreasing solvents.
- 2. mercury, used in elemental form in the manufacture of various mercury-filled instruments.
- 3. metals and cyanide used as part of electroplating processes at a specific area of the site.

Phase 1 is designed to provide sufficient information about subsurface conditions at the site to:

- characterize site geology
- complete a risk assessment, setting usage-based soil and groundwater quality goals for comparison to soil and groundwater analytical data

provide sufficient soil data from each AOC for comparison to soil quality goals
 assess groundwater quality along the downgradient perimeter of the site

Soil borings to be completed within the AOCs and geologic borings located apart from potential contaminant sources will provide a comprehensive database to characterize site geology. Soil samples from all borings within the AOCs and groundwater samples from the perimeter borings will be analyzed using a field-based laboratory for target contaminants of concern (COCs) to evaluate impact. Fixed-base (offsite) laboratories will be used to confirm on-site laboratory results and to provide data on mercury speciation needed to assess potential risk to human health.

The Phase 1 Report will present the results of the soil and groundwater analyses and the risk assessment. This information will be used to develop the Phase 2 Work Plan. Depending on the results of Phase 1, Phase 2 may include soil excavation or treatment and/or further groundwater investigations. It is possible that additional work beyond Phase 2 will be required to complete site restoration to site-specific usage-based goals.

The Work Plan is organized with task descriptions in the body of the report and supporting documentation attached as appendices. Section 2 presents general site history and setting. Section 3 summarizes previous investigations and/or sampling conducted at the site. Section 4 summarizes the selection process used to identify areas for investigation in Phase 1 and site-related. Section 5 presents the scope of work to be completed in Phase 1, including a summary of the analytical program. Section 6 introduces the format of the Phase 1 report, and Section 7 presents the project schedule.

Appendix A and Appendix B are the site-specific Quality Assurance Project Plan (QAPjP) and the site-specific Health and Safety Plan (HASP), respectively. In addition, five sections from the Background Documents, provided to NYSDEC on October 17, 1995, have been included in a separately bound volume as appendices: Historical Building Usage Table (Appendix C), First Floor Analytical Data and Map (Appendix D), Potential Areas of Concern Table (Appendix E), Soil Data and Map (Appendix F) and Outline of the External Site Investigation (Appendix G).



-

### 2.0 SITE HISTORY

The following sections summarize the history of industrial operations at the former Taylor Instruments facility on Ames Street.

### 2.1 General History

In 1904, the Taylor Brothers Company purchased a portion of what is now the Ames Street Site, and construction of the original buildings began shortly thereafter. The building numbering convention (e.g., 1, 2, 3) appears to have been used from the beginning, and indicates the general sequence of initial construction. Both the number of buildings and the overall size of the facility expanded from the original Ames Street frontage, replacing houses and small businesses, until the plant property reached to Hague Street on the west and the railroad easement on the north (approximately 1934), and to West Avenue on the south (approximately 1948) (see Figure 2-1). Over the years a number of buildings were erected and later removed, so that although the last building number is 60, only approximately 26 buildings existed as of 1992.

In 1982, Taylor Instruments Company merged with the Ritter Pfaudler Company to form Sybron Corporation (Sybron). Taylor Instruments operated as a division of Sybron until 1983, when it was sold to Combustion Engineering. In January 1990, Asea Brown Boveri, Ltd. (ABB) acquired Combustion Engineering, including the business now known as Kent-Taylor. Title to the property continues to be held by Combustion Engineering, a whollyowned subsidiary of ABB.

In 1991, ABB began construction of a new facility south of Rochester to house the Kent-Taylor manufacturing operations. The transfer of operations began in 1992 and was completed in 1993. Under the direction of the facility environmental manager at the time, a variety of old and unused chemicals and wastes were removed from the 95 Ames Street facility and recycled and/or disposed of off-site. The intent of these "close-up" procedures was to remove everything that was hazardous and not physically part of a structure. Following close-up, ABB's corporate real estate department, through the title holder Combustion Engineering, assumed responsibility for the property.

Once Kent-Taylor made the decision to move to a new facility, ABB's corporate real estate department actively began to market the property. A number of prospective purchasers expressed serious interest in the site but no buyer was found. Prospective purchasers expressed significant concerns about the environmental condition of the buildings and the underlying property.



In late 1993, some initial site characterization work was done. In 1994, ABB-ES, another ABB subsidiary, began the final phase of an asbestos abatement project at the facility and, simultaneously, a characterization of building materials (e.g., concrete, wood) for disposal. After completing this characterization, which involved significant interaction with NYSDEC's Region 8 Hazardous Waste staff, the demolition contract was let and demolition activities were initiated in May 1995.

Currently (February 1996), the demolition project is nearing completion. All buildings (except one that will remain through the site investigation), asbestos, hazardous and non-hazardous building wastes have been completely removed. Installation of asphalt paving over the former building areas will be completed in the spring of 1996.

### 2.1.1 Manufacturing Processes and Building Use History

The original products produced at the facility were mercury-filled thermometers, barometers and related simple instruments. Manufacturing operations changed little during World War I and World War II, as product lines were varied only slightly to meet military needs. For the most part, production continued on fluid-filled instruments until the early 1960's. At that time, a shift away from the fluid-filled instruments towards electronics-based instrumentation occurred. By 1965, all "liquid in glass" (i.e., thermometers and barometers) manufacturing had ceased. With the advent of computers, the various operations required for printed circuit board manufacture and assembly were introduced. By 1992, mercury-filled instruments, and consequently mercury-handling operations, represented a very minor aspect of facility operations.

In addition to mercury, other chemicals and industrial materials considered potentially environmentally significant were also used. Most prominent among these were petroleum fuels and oils, metal-bearing plating solutions and paints and organic solvents. These materials were used in operations by the early 1900's, and their use continued until 1992.

Appendix C provides a summary of known usage information for each building on a floorby-floor basis, as assembled from a number of historical sources. In addition to this summary, the following sections provide some detail relative to the most significant operations at the facility. Also, it is known that the facility was connected to the combined municipal sewer at all times during its history.

Mercury-Filled Instrument Manufacturing and Mercury Use

There were two major types of mercury-containing instruments manufactured at the site: "liquid-in-glass" instruments and "solid-filled" thermometer systems. All liquid-in-glass

ABB	ENVIRONMENTAL	SERVICES,	INC.
-----	---------------	-----------	------

instruments, thermometers, sphygmomanometers and barometers, were manufactured in a similar fashion. Glass tubes were purchased from an outside source, cut to the desired length and fitted with glass "bulbs" at one end. The instrument was then filled with mercury, in some cases under a vacuum. Solid filled thermometer systems typically involved filling narrow stainless steel or other "solid" capillary tubes (some to 200 feet long) with either a varsol/kerosene mixture or mercury, typically under vacuum or pressure. Both types of instruments used elemental mercury exclusively.

Mercury coming into the plant was generally shipped in durable metal containers. From the shipping or initial storage areas, it was brought to a refining operation located for many years on the third floor of Building 2. In this operation, the commercial-grade mercury was further purified to make it suitable for use in instrumentation.

Manufacturing of both types of instruments occasionally resulted in mercury being spilled at the work area. During plant demolition, concrete slabs in some areas were removed as hazardous waste due to their mercury content and several hundred pounds of liquid mercury were removed from between the floorboards of second and third floors of buildings. Overall, the occurrence of mercury in concrete and wood floors in the buildings as observed during demolition is consistent with the known manufacturing history. Buildings 1, 2, 4, 20, 30, 35, 40 and 44 were most impacted by these mercury releases.

Because mercury was a valuable raw material, significant efforts were made to collect the spilled liquid and to recover it from instruments broken or rejected during manufacture. Beginning in approximately 1940 and lasting until the mid 1960's, broken or rejected instruments were routinely collected and brought to a reclamation area located in the northwest corner of Building 40. Reclamation consisted of heating the scrap materials so as to volatilize the mercury, which was then re-condensed and recovered (a process known as "retorting"). Although enough mercury was recovered to make this operation worthwhile, evidently the process was not completely effective in extracting mercury from the small capillaries in the scrap instruments; mercury is still visible in a percentage of the scrap ("glass shards") found in close proximity to this building.

### **Electroplating**

Electroplating operations began at the plant around 1914 to 1915 and are believed to have been conducted in the same area of Building 4 throughout the site's history. Electroplating processes involved various acids, cyanides, alkaline cleaners and metals, including chromium, copper, brass, tin, lead, nickel, silver, cadmium, gold and zinc. Plating tanks and baths were set around the concrete-floored room. In and beneath the floors were a series of troughs and drains carrying discharges from the process, as well as leaks, drips, and

ABB	ENVIR	ONM	ENTAL	SERV	ICES,	INC.
-----	-------	-----	-------	------	-------	------

spills, to the plant sewer system. A series of tunnels beneath the plating area floor served both to convey liquids to the sewer system and to allow fumes to be drawn from the plating area and exhausted from the building. At the time of demolition, most floors and tunnels appeared to be original, although several patched areas were obvious.

In later years, measures to improve the operations included eliminating the use of cyanides and most "hard" (hexavalent) chromium and enhancing management practices to minimize spills and excessive dripping. These measures were implemented primarily in response to increasingly strict effluent discharge requirements, and the efficiency of the drainage systems in conveying process-related liquids to the sewer, which resulted in exceedances of those effluent discharge requirements.

Solvent Degreasing

Major solvent vapor degreasing occurred in two areas, one in Building 4 and the other in Building 48. Both operations used tricholoethene (TCE) and began operation in the mid-1960s. Other smaller desk top-scale degreasing units were also used at various locations at various times.

Each of the two major areas included one vapor degreaser located in a concrete sump designed, in part, to contain any spills. Both sumps exhibited staining. The third unit was a conveyor degreaser set at floor level in Building 48. Nearby this unit was distillation equipment used to reclaim spent TCE. Fresh TCE was stored in Tanks 13 and 14, in an interior "courtyard" outside the Building 48 area, and in Tank 15, behind Building 40. TCE drawn from Tank 15 was dispensed in the Building 40 garage into smaller containers for use in many minor degreasing operations throughout the plant.

Several spills onto the concrete floors of the two vapor degreasing areas have been noted. Both sumps were connected to the plant sewer system.

## Machining

Small parts machining operations occurred in various locations of Buildings 2, 3, 30, 44, 48 and 49 over the years. For the most part machines were self-contained, and the operations used the typical quantities of lubricant and cooling oils and fluids. The lack of significant oil staining on the concrete in most areas indicates the operations were either not very dirty and/or housekeeping was good.

Buildings 48 and 49 housed the most extensive machining operations and were expressly built for that purpose. An advanced (at the time) oiler system consisting of a rooftop filtration

plant, a central collection sump, and a piping network was used to collect and clean used cutting oil and re-distribute it to each machine. Existence of this system is most likely the reason very little evidence of oil releases was noted during demolition of these buildings.

### Chemical and Waste Storage and Handling

Manufacturing operations utilized numerous chemicals and some hazardous materials, and generated a number of hazardous waste streams. Bulk chemicals were stored in underground and aboveground tanks, in drums and other smaller containers. Buildings 12, 34 and 42 were all significant storage areas. However, with the exception of mercury, which was used in relatively large quantities, overall chemical throughput was low, consistent with the manufacture of small precision instruments.

Detailed records relative to waste generation and handling at the facility prior to approximately 1980 are not available. Based on interviews of facility personnel, available documentation, and observations during demolition, materials, such as coal slag, ash, and glass scrap and shards were used as fill for construction of both buildings and utilities. There is no information developed to date suggesting that waste materials other than this "fill" were disposed of at the site.

### 2.2 Site Description

The Ames Street Site covers approximately 13 acres north of West Avenue in the city of Rochester. The site is bounded on the south by West Avenue, the west by Hague Street, east by Ames Street and to the north by Conrail railroad tracks. The site is in the final stages of demolition. Final grading plans, to be completed in the spring, call for asphalt paving over the entire site with a maximum slope of 3 percent. There are no identified wetlands or surface water bodies on-site.

The area within one-half mile of the site is primarily mixed residential and light industrial. Rochester Gas and Electric has a leased facility on the west side of Hague Street. South of West Avenue and east of Ames Street is predominantly residential.

## 2.2.1 Site Geology

The surficial geology of the site area is dominated by Late Pleistocene glacially-deposited stratified sands and silty sands. According to previous investigations, overburden on the site ranges from 14 to 28 feet thick (Lozier, 1983). The overburden consists of 3 to 9 feet of fill overlying the glacial till. The fill material consists largely of clayey silt and includes at various locations coal slag, ash, building debris, and refuse (e.g., bottles, etc.). The till is

composed of dense silt and sand with some gravel and rock fragments. The till is characterized as a red-brown basal till overlying the bedrock and a laterally discontinuous layer of yellow-brown ablation till overlying the basal till in some areas. The ablation till is generally slightly more sandy and less dense than the basal till.

Bedrock underlying the overburden has been identified as the upper Silurian-age Lockport dolomite (Rickard and Fisher, 1970). Regionally this formation consists of flat to very gently dipping medium- to thick-bedded fine-grained dolomite with interbedded shales (Williams, 1990). The highest bedrock elevations were observed in the southeast sloping toward the northwest (Lozier, 1983).

#### 2.2.2 Hydrogeology

Groundwater is present within the overburden between 5 and 10 feet below the ground surface with a 4- to 5-foot seasonal fluctuation (Lozier, 1983). According to the Water Resources Investigations Report -- Groundwater-water Availability in the Genesee River Basin, New York and Pennsylvania, prepared by the United States Department of the Interior Geologic Survey (1986), the overburden yields less than 1 gallon per minute and the underlying bedrock less than 50 gallons per minute.

Overburden groundwater flow is believed to be generally from the south and west to the north and east. This general flow direction was described in the Lozier report and confirmed by water level measurements made on October 25, 1995 by ABB-ES. According to the Lozier report, hydraulic conductivities range from  $1.6 \times 10^{-5}$  to  $8.8 \times 10^{-5}$  centimeters per second (cm/s) with a geometric mean value of  $5.2 \times 10^{-5}$  cm/s. Using Lozier's measured hydraulic gradient of 0.0095 feet per foot and an representative porosity of 0.15, average horizontal groundwater velocity in the overburden is estimated at 3.4 feet per year.

## 3.0 SUMMARY OF PREVIOUS ENVIRONMENTAL INVESTIGATION

Since 1982, several environmental investigation efforts have been undertaken at the Ames Street Site. Soil sampling has also occurred relative to facility maintenance and demolition issues. This section describes the rationale and scope of each known effort. Except for the mercury contamination area north of Buildings 34 and 40, soil sampling locations and analyses are provided on a figure and table in Appendix F. (Sampling locations and results from the area north of Buildings 34 and 40 can be found in the various reports and correspondence submitted to NYSDEC at the time of the investigation.)

In addition to the work described below, a large number of building materials samples (e.g., wood, concrete) were collected during 1993 and 1994 to characterize wastes for disposal during plant demolition. Since all the building materials will have been removed by the time the Phase 1 field work begins, these analytical results will not be discussed in this report. However, these results were evaluated in selecting Areas of Concern (AOCs) for Phase 1 (see Section 4.0), and are included as Appendix D.

### 3.1 CLASS 4 AREA INVESTIGATIONS (1981-1986)

Mercury contamination in the area north of former Buildings 34 and 40, now referred to as the "Class 4 area", was first identified in 1981 (see Figure 2-1). Glass instrument shards, some containing visible mercury, were observed on the surface and in shallow subsurface soil. In late 1981, Taylor's consultant, Lozier, installed borings and wells to provide soil and groundwater samples in the immediate shard areas. Results showed total mercury concentrations up to 52,000 milligrams per kilogram (mg/kg) in shallow soil in the shard areas with much lower levels outside the areas (Lozier, 1983). NYSDEC became involved in early 1982, and another round of investigation, including installation of additional soil borings and shallow monitoring wells and lysimeters, was completed late that year. By the end of this field effort, a thorough characterization of soil within the 1/2 acre area was completed. The investigation also detected mercury in groundwater above the Class GA standards, primarily at the source areas.

NYSDEC subsequently approved installation of asphalt paving over the entire area as a remedial measure, and this was completed in late 1982 and early 1983. Quarterly groundwater monitoring was initiated and continued until September 1986, by which time mercury concentrations had generally fallen to below the Class GA standard. The area continues to be listed as a "Class 4" on the New York Registry of Inactive Hazardous Waste Sites (Registry), indicating the site is properly closed and requires continued management.

## 3.2 WATER TOWER AREA INVESTIGATION (1984-1985)

Discovery of glass shards in the area beneath the former water tower near the center of the property led to a soil investigation in late 1984 and early 1985. Approximately 16 samples from depths of 2 to 10 feet were collected and analyzed for mercury by the EP Toxicity method. Mercury concentrations in the extractions ranged from non-detectable up to 0.0094 milligrams per liter (mg/l). NYSDEC accepted the characterization and approved installation of asphalt paving in order to close the site. This was completed in late 1986, and the location was deleted from the Registry in approximately 1990.

### **3.3 BUILDING 4 SAMPLING (1987)**

According to a 1987 Taylor interoffice memorandum, an "area under the old zinc cyanide tank will have to be excavated because of poor structural integrity." The memo describes "several core samples and the soil underneath" composited and analyzed by the EP Toxicity method for cadmium, chromium, lead and mercury to determine if the material constituted a hazardous waste. The results were below the EP Toxicity limits, and apparently no further sampling was performed.

## 3.4 BUILDING 8 SAMPLING (1989)

According to a 1989 Combustion Engineering interoffice memorandum, a composite sample of soil from several locations beneath Building 8 was collected to characterize soil to be excavated during a construction project. Samples were presumably collected just beneath the floor slab. According to the EP Toxicity results, the soil was determined to be non-hazardous, based on Resource Conservation and Recovery Act metals. The total concentration of each metal ranged from 0.7 mg/kg for chromium to 231 mg/kg for copper.

## 3.5 TANK CLOSURES

<u>Tank 2</u>. In September 1986 this 1000-gallon gasoline underground storage tank (ust), was suspected to have leaked several hundred gallons of gasoline. With NYSDEC's concurrence, Taylor sampled nearby wells and lysimeters that had been installed for the NYSDEC Class 4 area investigation. No evidence of groundwater impact was found, and on this basis NYSDEC allowed the tank to be closed in place with no further groundwater or soil sampling or remediation. The Taylor correspondence file also indicates that "a 12/23/86 inspection of the tank site by the Rochester Fire Department revealed no significant soil contamination", although the scope of this inspection is not discussed.

## ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

Tank 2 was removed during demolition activities in November 1995. The tank had been closed by filling it with concrete. Samples were collected from the excavation and excavated material. Results indicated VOC levels beneath the STARS guidance values for gasoline contaminated soils. However, VOCs were present in a water sample that was collected from standing water in the excavation. The excavation has been backfilled. ABB-ES plans to investigate groundwater in this area as part of the site-wide investigation. It is unclear whether the presence of VOCs is a result of the 1986 release or from other potential sources in the immediate area. SVOCs detected in the soil are believed to be indicative of coal combustion by-products historically used as fill

<u>Tank 16</u>. This 3,000 gallon underground tank used to store paint thinner was removed in early 1993 by ABB Kent-Taylor. The initial set of soil samples from the excavation sidewalls showed levels of toluene, xylene and 4-methyl-2-pentanone above the NYSDEC Spill Technology and Remediation Series guidelines. At NYSDEC's request, additional soil was removed from the excavation and a second set of samples was obtained. These samples indicated the impacted soil had been removed, and tank closure was granted.

<u>Tanks 11/12</u>. These former xylene and toluene tanks were removed in December 1995. After initial field PID screening, unsaturated contaminated soil was excavated. This soil was shipped off site for proper disposal. Analytical results indicate that standing water in the excavation and surrounding soils (bottom and east sidewall) remain impacted. The extent of remaining contamination is unclear. Additional investigation is planned for this AOC. The excavation has been lined with polyethylene plastic sheeting and temporarily backfilled.

<u>Tanks 17/18/19</u>. These three 54-gallon tanks were removed in December 1995. Sampling of sidewalls and bottom of the excavation indicated remaining contamination at the bottom of the excavation. An additional three feet was excavated and the bottom resampled in January 1996. The contaminated soil was shipped off site for proper disposal. Analytical results indicate no remaining contaminated soil. The excavation was backfilled and no further investigation is planned.

### 3.6 SITE INVESTIGATION (1993)

In May 1993, ABB-ES conducted a limited site investigation to establish the general environmental condition of the property through selected sampling of potentially suspect areas. Outside the buildings, ABB-ES used a drill rig to collect soil samples from borings completed near each existing underground tank; the former Building 42 solvent recovery/drum storage area, the former Tank 15 area, and several other locations. Inside the buildings, shallow soils samples were collected by hand from beneath floor slabs in several areas, including the plating and degreasing areas, Building 30 mercury filling room, Building

12 and the former Tank 13 and 14 locations.

Results of field screening, field lab analysis and off-site lab analysis of the samples indicated the presence of mercury, metals and VOCs in some areas, and led to a report to NYSDEC regarding suspected releases from Tank 9 and former Tanks 13/14. The results suggested the need for further work in some areas and formed the basis for selection of some of the AOCs described in Section 4.0 of this workplan.

### 3.7 SUB-SLAB SOIL SAMPLING (1993)

During a September 1993 effort to sample concrete and other materials for waste characterization, several soil samples (that may have contained floor material) from beneath floor slabs were collected to augment the information from the May 1993 work. Samples were analyzed in a field laboratory for mercury only, with results ranging from not detected to 360 mg/kg.

#### 3.8 SAMPLING DURING DEMOLITION (1995)

Plans for plant demolition included the possibility of removing obvious contaminant source, or limited areas of impacted soil, if encountered. In addition, regulatory requirements require sampling during underground tank closures.

Two areas of impacted soil were excavated during demolition activities in 1995. One area is located at the west end of former Building 49, where hydraulic pistons for the loading docks apparently leaked oil into the surrounding soil. The second area included apparently VOC-impacted soil around an old brick structure found several feet below grade between former Buildings 5 and 8. At both locations, ABB-ES collected soil samples from the excavation sidewalls to evaluate whether the impacted material was completely removed. Results indicated that all contaminated soils had been removed at these locations.

In November 1995, a 25,000 gallon underground storage tank was removed during the demolition project. While the tank itself appeared to be intact, visible material that appeared to be fuel oil was observed in the excavation. Notification was provided to NYSDEC. The liquid material was subsequently removed by a vacuum truck. Approximately 150 tons of soil were removed and sent off-site for treatment and/or disposal. The remaining soil was sampled. Analytical results indicate that soil above NYSDEC guidance values was removed. No further investigation is planned for this AOC.

## 4.0 POTENTIAL AREAS OF CONCERN AND CHEMICALS OF CONCERN

This section presents the Phase 1 potential AOCs and COCs identified for investigation at the Ames Street Site and the basis for the identifications.

4.1 AREAS OF CONCERN

ABB-ES reviewed all available site information, historical use information, and previous environmental analytical data to identify areas of the site that would be most likely to exhibit impact from historical plant operations. This information, supplemented by observations of shallow soils during demolition of the plant buildings, were used to develop AOCs that will be investigated during Phase 1.

Types of AOCs include:

- broad areas of potential soil impact (e.g. areas of the site containing glass shards as fill *or* soils beneath a building where soil impact was observed during removal of ground level floors)
- potential point sources such as underground storage tanks
- potential non-point background sources (e.g. soils beneath floors that were removed as hazardous waste due to elevated concentrations of COCs )

Table 4-1 and Figure 4-1 (attached at end of report) summarize the AOCs that will be investigated during Phase 1. A complete listing of all site features evaluated during selection of Phase 1 AOCs is located in Appendix E. Table 4-1 is organized as follows: COLUMN DATA PRESENTED

- 1 Area number and site description
- Historical usage summary based on interviews with current and former employees; fire insurance maps from 1913,1934,1952,1986,1979, and 1988; mercury vapor/mercury problem reports and measurements circa 1946 to 1949 and 1963; Building Floor Area Report, July 1990, Rev 1 (8/14/90); and handwritten notes, circa 1982 to 1983, on "Places Where Mercury Has Been Used and Probably Spilled.
- 3 Results of concrete and wood floor analyses from the overlying first floor collected during four stages of building material characterization (see Appendix D)

# AREA OF CONCERN SUMMARY AMES STREET INVESTIGATION WORK PLAN - PHASE 1

POTENTIAL AREA OF CONCERN HISTORICAL USAGE		OVERLYING FLOOR ANALYTICAL RESULTS	VISUAL OR SCREENING EVIDENCE DURING DEMOLITION	AVAILABLE SOIL ANALYTICAL RESULTS	PHASE   AREA OF CONCERN? [RATIONALE]	PLANNED PHASE I ACTIONS	PHASE I CHEMICALS OF CONCERN	
1	1 Areas underlying floors removed as <i>characteristic</i> hazardous waste due to mercury.		Contained total mercury at concentrations exceeding 550 mg/kg.	Floors largely in good condition - no visual evidence of mercury in underlying soil.	Concentrations of mercury range from ND to 360 mg/kg in several isolated samples.	Yes [Elevated mercury results in floors indicate need to look at underlying soils.]	Shallow Soil Sampling	Mercury
2	Areas underlying Building 5 floors removed as hazardous waste.	Painting operations, paint chemical and waste use/generation.	Total VOC concentrations ranged from ND to 4,000 ug/kg.	Floors in good condition - no visual evidence of VOC impacts in soils.	None	Yes [Potential for VOC concentrations in underlying soil]	Shallow Soil Sampling	VOCs
3	Area underlying former electroplating area floors removed as hazardous waste. Building 4.	Metal plating and degreasing.	Concentration ranges: VOCs 7 to 312 ug/kg Cr <sup>+3</sup> ND - 34 mg/kg Cr <sup>+6</sup> 1.4 - 68 mg/kg Cd 1.2 - 26 mg/kg Pb ND - 0.76 mg/kg CN ND - 98 mg/kg Floors/debris not characteristic hazardous waste based on TCLP results.	Floors in fair condition but evidence of historical deterioration/patching - no visual evidence of releases to soil.	Metal concentration ranges: Cd ND - 260 mg/kg Cr <sup>+3</sup> ND - 6,000 mg/kg Cr <sup>+5</sup> ND - 6,300 mg/kg Pb ND - 500 mg/kg Ni ND - 5,500 mg/kg Zn 40 - 12,000 mg/kg No VOC results available.	Yes [Use history; floor condition may have allowed soil impacts.]	Shallow Soil Sampling	VOCs, Metals and Cyanide
4	Area underlying Building 34	Chemical/waste drum storage (also adjacent to former TCE use/storage area)	Floor materials removed as listed hazardous waste due to low levels of VOCs, CN, metals.	Floors in fair-to-good condition, but staining, drum rings visible. Some shards observed during demolition	Six shallow soil samples showed TCE ranged from ND to 910 ug/kg.	Yes [Presence of VOCs in soil suggests potential impact.]	Shallow Soil Sampling	VOCs, Mercury
5	Area underlying Building 40 garage area.	TCE dispensing/degreasing; garage maintenance operations. Floor drain to sewer present.	Total VOCs ranged from 32 to 9,000 ug/kg.	Floors largely intact, some limited staining; Glass shards observed during removal of floors	None	Yes [Known bulk TCE use]	Shallow Soil Sampling	VOCs, Mercury
6	Area near and underlying former Building 42	1940's-era solvent recovery operations (north end) and waste drum storage. Minimal knowledge of actual operations or locations.	No sampling.	Floors intact;no shards observed during floor removal	Eight soil samples with PCE in one at 280 ug/kg, TCE ranging from 4 to 260 ug/kg and BTEX up to 1,000 ppm	Yes [History of bulk chemical operation and known soil impacts]	Shallow Soil Sampling	VOCs
7	Area underlying Building 12	Oil and chemical storage in subgrade vaulted concrete building. Possible soil floor.	No sampling of concrete.	No visible impacts	Shallow soil samples contained TCE/1,1,1-TCA /PCE ranging from 10 to 200 ug/kg and TEX ranging from 14 to 8,970 ug/kg	Yes [History of bulk chemical storage and existing soil data indicate potential for soil impact]	Shallow Soil Sampling	VOCs
8	Area where glass shards were observed during demolition and in previous studies	Broken instruments observed.	NA - shards apparently deposited prior to floor construction, or no floors over areas.	Shards easily identifiable visually, liquid mercury observed occaisionally Shard density varies from absent to foot+ near-surface fill layer.	1980 Soil samples (from different shard areas) show total mercury concentrations ranging from 20 to 12,800 mg/kg.	Yes [History and existing soil data indicate potential for mercury in associated/underlying soils]	Soil Sampling	Mercury

. ,

# AREA OF CONCERN SUMMARY AMES STREET INVESTIGATION WORK PLAN - PHASE 1

POTENTIAL AREA OF CONCERN HISTORICAL USAGE		HISTORICAL USAGE	OVERLYING FLOOR ANALYTICAL RESULTS	VISUAL OR SCREENING EVIDENCE DURING DEMOLITION	AVAILABLE SOIL ANALYTICAL RESULTS	PHASE I AREA OF CONCERN? [RATIONALE]	PLANNED PHASE I ACTIONS	PHASE I CHEMICALS OF CONCERN
9	Degreasing area, Building 48.	TCE vapor degreasing	TCE from ND - 11,000 ug/kg	Floors in fair-to-good condition but stained; no obvious soil impact.	One soil sample showed TCE 11,000 ug/kg, PCE 97 ug/kg .	Yes [History of bulk chemical use and existing scil data suggest potential soil impact]	Shallow soil sampling	VOCs
10	Area beneath Building 2.	Various mercury-intensive operations	Some overlying floor areas (AOC-1g,1h) contained total mercury above 550 mg/kg.	Floors in good condition. Mercury observed in shallow soils beneath floors, especially associated with subsurface drainlines/air tunnels.	None	Yes mercury present in soils	Shallow soil sampling	Mercury
11	Area beneath Building 44.	Various mercury-intensive operations	Some overlying floor areas contained total mercury above 550 mg/kg (AOCs- 1i,1j,1k,1l,13e)	Floors in good condition. Mercury observed in drainlines beneath floors.	None	Yes mercury observed in drainlines of unknown integrity	Shallow soil sampling	Mercury
12	Areas underlying floors removed as <i>listed</i> hazardous waste due to mercury.	Various mercury-intensive storage and handling areas.	Some overlying floor areas contained total mercury above 550 mg/kg and all overlying areas correspond to NYSDEC listed areas due to mercury operations.	Floors in good condition. No obvious impact to underlying soils	None	Yes Investigate to assess impact beneath a NYSDEC <i>iisted</i> mercury storage area.	Shallow soil sampling	Mercury
Stor	age Tanks			• • • • • • • • • • • • • • • • • • •				
	Tank 1/9 (USTs)	Tank 1:Fuel oil storage (25,000 gal.) Tank 9: Gasoline storage (1,000 gal.) Visually impacted soil around fill port reported to NYSDEC in 1993.	NA	Visually stained soil present around bottom of Tank 1.	Prior samples contained TPH up to 64,000 mg/kg and BTEX between 3 mg/kg and 12 mg/kg. Results during demolition found no residual impact to soils in immediate underlying soils.	No	None	None
	Tank 2 (UST)	Gasoline storage (1,000 gal) adjacent to Tank 15. Tank known to have leaked, cleanup/tank closure done under Fire Dept./ NYSDEC supervision in 1986.	NA	VOCs detected in soils around tank.	See results for Tank 15.	Yes [Existing soil data indicates potential soil impact]	Soil sampling	VOCs
	Tank 15 (AST)	TCE storage (8,000 gal.), adjacent to Tank 2. Removed 1992.	NA .	TBD (slab/foundation to be removed)	Samples adjacent to Tank 15/Tank 2 area showed 0.4 mg/kg to 930 mg/kg TCE, and BTEX from 2 mg/kg to 60 mg/kg.	Yes [Existing soil data indicates potential soil impact]	Soil sampling	VOCs

# AREA OF CONCERN SUMMARY AMES STREET INVESTIGATION WORK PLAN - PHASE 1

POTENTIAL AREA OF CONCERN		HISTORICAL USAGE	OVERLYING FLOOR ANALYTICAL RESULTS	VISUAL OR SCREENING EVIDENCE DURING DEMOLITION	AVAILABLE SOIL ANALYTICAL RESULTS	PHASE I AREA OF CONCERN? [RATIONALE]	PLANNED PHASE I ACTIONS	PHASE I CHEMICALS OF CONCERN
	Tank 10 (AST)	Various organic solvent/paint thinner storage (285 gal.)	NA	Storage vault. No tanks present or staining observed during demolition.	None	No. [Aboveground storage vaults with no tanks present or indications of releases to concrete.]	None	None
	Tanks 13/14	TCE storage (1,500 and 1,000 gal.) Removed 1992.	NA	Some soil staining present.	PCE/TCE/1,2-DCE present in nearby soils from 0.2 mg/kg to 150 mg/kg, BTEX from 70 to 270 mg/kg.	Yes. [Existing soil data indicates potential soil impact]	Soil sampling	VOCs
	Tank 11/12 (USTs)	Toluene/xylene storage (200 gal. each).	NA	Visually stained soils were removed during tank excavation.	No VOCs detected in four samples beneath tanks.	Yes. [Need to determine if soils are contaminated beyond lateral boundary of tank excavation.]	Soil sampling	VOCs
18	Areas underlying hydraulic hoists, elevator pistons, and machine oil sumps, cutting oil storage locations.	As described, locations of bulk oil- containing equipment.	Oils did not contain PCBs, floors managed as non-hazardous solid waste.	Oil-stained soils were removed during demolition from one area (AOC-18a)	Soils results show no residual impact beyond area of removed soils.	No. [No remaining impact ]	None	None
19	Sewerlines and buried pipes	Potential for contamination via floor drains in chemical use areas	None available	Shallow pipes removed during demoition. Mercury observed in shallow soils beneath Building 2 and in pipes beneath Building 44. Some sewers left in place.	Low concentrations of mercury in water in sewers leaving site.	Yes. [AOCs established as footprint of Building 2 (AOC-10) and Building 44 (AOC-11).]	Soil Sampling	Mercury
20	Areas identified during demolition	NA	NA	Soils examined during the removal of ground-level floors and inground pipes resulted in the creation of AOC-10 and AOC-11.	None	Yes. [AOCs 10 and 11 identified during demolition and extent of glass shards in soil (see AOC-8) partially determined.]	Soil Sampling	Mercury

.

Notes: 1) 2) 3)

Only first floor use/condition of floor considered because focus is underlying soil. All mercury results expressed as total mercury. NA = Not available. ND = Not Detected.

4) 5) TBD = To Be Determined.

IRM = Interim Remedial Measure.

4 Observations, both visual and screening by PID and mercury vapor analyzer, made during building demolition and tank and sewer pipe removal (demolition is not yet complete and additions to this data are expected prior to Phase 1 on-site work)

Results of soil analyses from May 1993 site investigation (see Appendix F)

- 6 Findings of the data review, including yes or no decision for Phase 1 sampling and rationale for decision
- 7 COCs detected in flooring or soils in the area that will be analyzed during Phase 1
- 4.2 POTENTIAL CHEMICALS OF CONCERN

5

Potential COCs can be grouped into three categories based on the information developed during the selection of the Phase 1 AOCs: (1)VOCs, (2) mercury, and (3) metals/cyanide.

VOCs of concern are primarily aromatic hydrocarbons associated with petroleum products (fuel oil, gasoline), toluene or xylene (stored in product form), and chlorinated hydrocarbons used in cleaning and degreasing operations (e.g. TCE and PCE). Mercury was used in large quantities over the lifetime of the facility in the manufacture of various mercury-filled instruments throughout large areas of the building complex. Metals and cyanide are COCs at a specific area of the site where electroplating activities were carried out.

The potential COCs are presented in Table 4-2. In addition to identified COCs, selected other samples will be subjected to target compound list analyses as discussed in Section 5.5 to ensure that there are no additional site-related compounds that should be considered COCs.

The COCs listed are site-related compounds and metals detected in floors or soils at the site during previous site sampling events. Vinyl chloride is included as a possible degradation product of chlorinated compounds at the request of NYSDEC. Significant new site-related metals and VOCs identified by off-site laboratory analysis during the Phase 1 field investigations will be added to the calibration standards for the field laboratory in Phase 1.

### ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT TAY\PHASEI\FINAL.DOC

# CHEMICALS OF CONCERN AMES STREET SITE INVESTIGATION WORK PLAN -PHASE 1

Parameter	Analytes
VOCs	Tetrachloroethene
	Trichloroethene
	1,1-Dichloroethene
	1,2-Dichloroethene
	1,1,1-Trichloroethane
	Benzene
	Toluene
	Ethyl Benzene
	Xylenes
	Acetone
	4-Methyl-2-pentanone
	Vinyl Chloride
Metals	Mercury
	Cadmium
	Chromium (trivalent)
	Chromium (hexavalent)
	Lead
	Zinc
	Nickel
Other	Cyanide

## 5.0 SCOPE OF WORK

The following sections describe the activities planned as part of the Phase 1 site investigation. The field investigations will be conducted in accordance with the specifications presented in the QAPP (ABB-ES, August 1995)) and the site-specific QAPjP (Appendix A). Quality Assurance/Quality Control (QA/QC) procedures for sample handling, sample shipment, and data validation are presented in the program QAPP (ABB-ES, August 1995) and QA/QC sample frequencies are presented in the site-specific QAPjP (Appendix A). Health and safety procedures for all on-site activities are presented in Appendix B, the Site-Specific HASP.

## 5.1 AOC INVESTIGATIONS - PHASE 1 SOIL BORING PROGRAM

The scope of work that will be completed during Phase 1 consists principally of a boring program targeted to the identified AOCs. Borings will be drilled at each AOC to collect soil samples for field and laboratory analysis. The primary objective of the soil sampling program is to provide data for comparison to usage-based soil quality goals to identify which AOCs will require further action. To accomplish this, a minimum number of borings will be drilled at each AOC and soil samples from each boring will be analyzed for specific COCs. Target COCs have been identified for each AOC based on AOC type, previous environmental data, and demolition observations as detailed in Table 4-1. Some selected soil samples may also be analyzed for speciation and bioavailability. Additionally, the soil boring observations will be used to characterize the physical setting. Three geologic borings are planned (outside the AOCs) to further characterize the overburden and shallow bedrock at the site.

### 5.1.1 Boring Location and Rationale

The final location and number of borings planned for each AOC will depend on the size of the AOC, site history, and analytical results generated during the field investigation. The AOCs vary in size from the locations of former underground storage tanks to broad areas of potential impact beneath former buildings or related to contaminated fill (glass shards). AOCs targeted for different COCs often overlap or are located in close proximity. For example, AOC-2 (consisting of soils beneath former painting operations) is potentially impacted by VOCs and is located partially within AOC-8, a large area of potentially mercury-contaminated fill soils. Because of this complexity, a minimum number of borings have been assigned specific locations within each AOC to provide an initial understanding of soil concentrations. These assigned borings and the rationale for locations are listed in Table 5-1. Figure 5-1 (attached at end of report) shows the location of each boring.

# TABLE 5-1

# SOIL BORING SUMMARY AMES STREET INVESTIGATION WORK PLAN - PHASE 1

AREA OF	DEDICATED SOIL	
CONCERN	BORINGS	NOTES ON BORING LOCATION/RATIONALE
AOC-1 1a, 1b, 1c	3	These areas are located apart from other mercury AOCs in the southern portion of the site. One initial boring will be drilled at each to determine if mercury is present in soils. Up to three additional borings will be located based on field results to provide additional data on distribution in soils.
AOC-1 1d, 1e, 1n, 1m	4	These four mercury AOCs are located within a large potential area of impact from glass shard fill (AOC-8). One initial boring will be drilled within each AOC. Other dedicated borings will be located nearby to assess the overall impact within AOC-8.
AOC-2 2a, 2b, 2c	8	Eight borings will be drilled to assess impact from VOCs. The borings at AOC-2a and 2b are located at former degreasing operations. Borings at AOC-2c include locations of former drum storage and paint storage areas.
AOC-3 3a, 3b, 3c	7	AOC-3 is an area including former electroplating operations. Seven dedicated borings will collect samples for off-site analysis of metals/CN. Soils will be also analyzed for VOCs from the borings located within AOC- 3b. Boring locations include the former location of a zinc cyanide tank and concrete floor containing elevated levels of metals.
AOC-4	4	Four borings will be drilled initially to assess impact from VOCs/mercury. Up to 2 additional borings will be located downgradient of the AOC based on field results.
AOC-5	2	Two borings will be drilled initially to assess impact from VOCs/mercury. One is located at a former TCE dispensing area.
AOC-6	2	Two borings will be located to determine impact from VOCs at this former 1940s-era solvent recovery/storage area. VOCs will also be screened in adjacent AOC-1n borings.
AOC-7	1	One boring will be drilled to determine impact from VOCs beneath the floor of this former oil/chemical storage area. Downgradient boring(s) in AOC-10 will provide additional information on potential impact.
AOC-8	6	A minimum of 6 borings will be drilled within this large area of suspected impact from glass shards. Three of these borings are at specific locations of shards observed during demolition and three are located to provide spatial information on overall soil impact within the AOC. These borings are in addition to more than 10 borings at smaller AOCs within the boundary of AOC-8 that will include screening for mercury. Up to 4 additional borings may be added based on field results.
AOC-9	1	One initial boring will be drilled at this former degreasing area. Additional borings may be drilled to examine the extent of impact based on field results from this AOC or at adjacent T-13/14
AOC-10 10, 1f, 1g, 1h	4	Four borings will be drilled beneath Building 2 to assess the distribution of mercury, observed in shallow soils during demolition. This AOC includes AOC- 1f, 1g and 1h, areas of mercury-impacted concrete floor.

# TABLE 5-1 (cont'd.)

AREA OF CONCERN	DEDICATED SOIL BORINGS	NOTES ON BORING LOCATION/RATIONALE
AOC-11 11, 1i, 1j 1k, 1l, 12e	7	At least seven borings will be drilled beneath Building 44 to assess impact from mercury. Mercury was observed in excavated drainlines in shallow soils during demolition. This AOC includes smaller mercury AOCs 1i, 1j, 1k, 1l, (impacted concrete floor) and 12e (a NYSDEC listed area for Hg).
AOC-12 12a, 12b, 12c, 12d	6	These AOCs are considered listed areas for mercury under NYSDEC. Two borings will be drilled at AOC-12a, a receiving area for mercury shipments to the site. At least 4 borings will be drilled at AOC-12b, 12c, and 12d in Building 3.
T-1/9	1	One boring will be drilled at the downgradient edge of these former fuel tanks to assess if any VOCs remain in soils below the excavated depth (12 ft bgs)
T-2/15	1	One boring will be drilled downgradient adjacent to the location of these former tanks. Up to two additional borings will be added to assess the extent of VOC-impacted soils, if any.
T-11/12	1	One boring will be drilled at the location of former tanks 11/12 to assess if VOCs remain below the level of excavated soils. The downgradient boring in AOC-1m will provide additional information on potential impact.
T-13/14	1	One initial boring will be drilled at the location of these former TCE storage tanks. Additional borings may be drilled downgradient based on field results from this boring or adjacent AOC-9.
Total	60	Minimum soil borings to be completed during Phase I

Additional borings may be completed, if required to assess the feasibility of remedial measures such as soil excavation or treatment. The number and location of these additional borings will be determined in the field. The project team will continually evaluate the analytical findings generated by the on-site field laboratory and revise the conceptual model for the site accordingly. Factors that will be used to decide on additional sampling include; the nature of contaminants found at an AOC, the distribution and concentration of chemicals vertically within the overburden soils, and the existence of quality goals for a compound. A budget for Phase 1 has been established that will allow the completion of up to 20 additional borings.

Three geologic borings will be drilled to provide additional information about the overburden and bedrock geology at the Ames Street Site. The geologic borings will be drilled in areas near the northern, eastern and south-western areas of the site, respectively, to provide representative coverage. Each boring will be drilled outside an AOC to avoid the potential for contamination of the upper bedrock groundwater system.

## 5.1.2 Soil Sampling

In each soil boring, samples will be collected continuously at 2-foot intervals starting at the top of the original fill and ending at boring refusal or bedrock, whichever comes first. Borings will be advanced using conventional hollow-stem augers. Drilling, sampling, and decontamination procedures are described in the QAPP (ABB-ES, August 1995). In addition to PID screening for VOCs, a mercury vapor analyzer will be used to assist in selecting samples for analysis.

A minimum of two soil samples will be selected from each boring for analysis. The selection criteria will depend on the COCs at each AOC. In general, the sampling is intended to identify soils that may have concentrations of site chemicals above soil quality goals. The following paragraphs describe the COC-specific selection criteria:

<u>Mercury AOCs</u> Samples for mercury analysis will be collected from every split spoon and visually inspected for the presence of free mercury. Samples will be submitted to the field laboratory and analyzed sequentially. If mercury is detected, samples will be run from shallow to deep until concentrations are less than the method detection limit (0.1 parts per million) or until the overburden soil has been fully characterized. If no mercury is detected in the first two shallow soil samples, a third sample will be selected from the surface of the first less permeable layer or the overburden/bedrock interface, whichever is encountered first. This sample is intended to evaluate the potential for elemental mercury to move down through the water/soil column and accumulate on less permeable layers.

### ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

<u>VOC AOCs</u> Samples from each boring will be screened for the presence of VOCs using a PID. A minimum of two samples from each boring will be selected for field analysis; one each from both the saturated and unsaturated zones. As at other AOCs, the unsaturated zone samples are intended to identify soils that may have concentrations of site chemicals above soil quality goals. Saturated zone VOC samples are intended to identify site chemicals that may be migrating via overburden groundwater flow. Samples will be selected from the most likely impacted areas, based on PID meter screening results and visual observations. If there is no visual or PID evidence of VOCs, one sample from the original fill material and one from near the overburden bedrock interface will be selected.

<u>Metals/Cyanide AOCs</u> Two samples will be collected from the unsaturated zone of each boring. One sample will be selected from 2 feet below the top of the original fill in order to avoid the upper layers that may have been disturbed by demolition activities and therefore not representative of the original fill. A second sample will be collected for analysis at the water table or at the base of the fill, whichever is encountered first, to characterize the vertical distribution of metals.

<u>Geologic borings</u>. Soil samples will be collected continuously at 2-foot intervals using either conventional hollow-stem auger or drive and wash drilling techniques. Each boring will be backfilled with a cement/bentonite grout mixture following sampling. The upper ten feet of bedrock will be drilled using "N" size rock core bit and barrel. Overburden and bedrock drilling and split-spoon sampling procedures are outlined in the QAPP (ABB-ES, August 1995).

### 5.1.3 Analysis

Chemical analysis for selected VOCs and mercury will be performed at an on-site field laboratory as a rapid and cost-effective means of data acquisition. Field analysis will provide same-day results to assist in decision making about future boring locations. Samples from each AOC will be analyzed for the COC list chemicals (Table 4-1) targeted for that specific AOC. Details of the field analytical program are provided in Section 5.4. Five percent of the total VOC and mercury samples field analyzed, will be split and sent for off-site analysis. For the VOC splits, the analytes of interest will include all VOCs from the Target Compound List (TCL) for VOCs, plus up to ten library search compounds. The metals analysis will be done at an off-site laboratory for the Target Analyte List (TAL) inorganics and will also include hexavalent and trivalent chromium, which are site-specific COCs. Section 5.4 and the site-specific QAPjP (Appendix A) describe the analytical methodology for both on-site and off-site laboratory analysis. Table 5-2 contains a breakdown of anticipated soil analyses for each AOC.

#### TABLE 5-2

#### SOIL ANALYTICAL PROGRAM AMES STREET SITE INVESTIGATION WORK PLAN - PHASE 1

		FIELD ANALYSIS		OFF-SITE ANALYSIS			MERCURY
AREAS OF CONCERN	COCs	VOCs'	MERCURY	VOCs <sup>2</sup>	MERCURY	METALS	SPEC /BIO4
AOC 1 (a-n)	Mercury		x		5%		TBD
AOC 2 (a-d)	VOCs	X		5%			
AOC 3(a)	Select metals		<u>x</u>			X	
AOC 3(b)	VOCs, Select metals	x	X	5%		x	
AOC <u>3(c)</u>	Select metals		x			<u>X</u>	
AOC 4	VOCs, mercury	x	x	5%			
AOC 5	VOCs, mercury	<u>x</u>	X	<u>5</u> %			
AOC 6	VOCs	x		5%			
AOC 7	VOCs	x		<u>5%</u>			
AOC 8	Mercury		X		5%		TBD
AOC 9	VOCs	x		5%			
AOC 10	Mercury		x		5%		
AOC 11	Mercury		x		5%		_
AOC 12	Mercury		x		5%		
Tank 1/9	VOCs	X		5%			
Tank2	VOCs	x		5%			
Tank 15	VOCs	x		5%			
Tank 13/14	VOCs	x		5%			
Tank 11/12	VOCs	x		5%			

NOTES:

COCs = Chemicals of Concern

TBD = To Be Determined

UNK = Unknown

VOCs = Volatile Organic Compounds

1 - Field laboratory analysis for the following:

TetrachloroetheneTolueneTrichloroetheneEthylbenzene1,1-DichloroetheneXylenes1,2-Dichloroethene4-Methyl-2-Pentanone1,1,1-TrichloroethanAcetoneBenzeneVinyl Chloride

2 = Off-site laboratory analysis for Target Compound List (TCL) VOCs plus up to 10 library search compounds.

3 = Off-site laboratory analysis for Target Analyte List (TAL) inorganics plus trivalent and hexavalent chromium.

4 = Analysis for mercury speciation and bioavailability. Number and location of samples will be dependent on mercury concentrations measured by on-site laboratory.

Additional split samples may be collected for mercury speciation and bioavailability analysis in areas with high mercury concentration. The rationale for conducting these analyses is discussed in Section 5.3.4.

## 5.2 GROUNDWATER SAMPLING

Groundwater sampling will be performed at existing monitoring wells and at temporary well points to be installed along the northern and eastern boundaries of the Ames Street Site. The intent of the groundwater sampling program is to characterize overburden groundwater quality along the down-gradient site perimeter for comparison to groundwater quality goals, and to determine the optimal location for permanent monitoring wells that may be installed during Phase 2. At least eight existing monitoring wells and a minimum of 20 temporary well points will be sampled. The location of each monitoring well and planned temporary well point is shown on Figure 5-1.

### 5.2.1 Temporary Well Point Locations

A total of 20 temporary well points are planned along the northern and eastern boundary of the Ames Street Site as part of the groundwater sampling program to provide a relatively inexpensive means of providing groundwater samples. Additional well points may be added after the initial 21 points have been sampled and analyzed, to trace chemicals detected back to possible on-site sources. The location and distribution of the initial 20 well points is based on:

1) <u>Groundwater flow.</u> A prior investigation conducted within the northwest portion of the Ames Street Site showed an interpreted north and east groundwater flow direction in the overburden groundwater system (see Section 2.2.2). In addition, analytical results from previous groundwater samplings show the presence of chemicals suspected of having migrated toward the northeast from possible sources on-site. Based on the location of the different AOCs at the site relative to the predominant groundwater flow direction, the northern and eastern property boundary are considered to be hydraulically downgradient of the site.

2) <u>Dispersion</u>. Assuming 50 feet as a conservative estimate of average source area width, an average distance to the property boundary of 100 feet, and estimates of flow velocity from previous investigations (Lozier, 1983), a minimum hypothetical plume width of approximately 75 feet is estimated as a result of dispersion mechanisms. Based on this width, spacing of the initial 21 temporary well points was set at 60 foot intervals along the northern and eastern property boundaries (see Figure 5-1).

3) <u>Location of AOCs</u> The location of the AOCs is concentrated within the central and northern sections of the Ames Street Site. The position of all AOCs relative to that of the interpreted groundwater flow, exposes approximately 700 feet of the eastern, and nearly all of the northern boundary of the site, as areas of potential off-site migration. The temporary well points are located in these areas.

# 5.2.2 Groundwater Sampling and Analysis.

Temporary wells will be installed using a conventional augering. Two inch ID PVC with 5-foot wellscreens will be installed in an open auger hole and backfilled with clean sand. The bottom of the screen will be placed at least 5-feet below the top of the water table.
Following installation, the wells will be purged and sampled similar to the existing site monitoring wells. After sampling, the PVC will be removed and the remaining open borehole filled with bentonite grout.

Groundwater samples will also be collected from at least eight existing on-site overburden monitoring wells (see Figure 5-1) in conjunction with the temporary well point sampling. Procedures for groundwater sampling will follow those in the United States Environmental Protection Agency (USEPA) Region I guidance for low-flow purge and sampling (USEPA, 1994) and the "clean-hands technique" for water sampling of mercury. These procedures are outlined in the site-specific QAPjP (Appendix A).

Samples from both temporary well points and existing monitoring wells will be analyzed at the on-site field laboratory for selected VOCs and mercury. All water samples will be split for off-site analysis of Target Compound List VOCs. Ten percent of these will also be analyzed off-site for mercury. Table 5-3 is a summary of the analytical groundwater program. Section 5.5 describes the analytical methodology to be employed for the on-site analysis.

## 5.3 RISK ASSESSMENT AND DERIVATION OF RISK-BASED GOALS

A human health risk assessment (HHRA) will be performed for the Ames Street Site. The HHRA activities will be conducted consistent with NYSDEC guidance and guidance developed for the federal Superfund program. Because the purpose of the HHRA is to provide the information required to develop risk-based soil and groundwater quality goals for the site, the risk assessment will be documented in the Site Investigation Report - Phase 1 but a separate formal baseline risk assessment report will not be prepared. The risk-based soil and groundwater quality goals for the site will be derived based on the site-specific risk assessment information.

## TABLE 5-3

# GROUNDWATER ANALYTICAL PROGRAM AMES STREET SITE INVESTIGATION WORK PLAN - PHASE 1

	NO. OF	FIELD A	NALYSIS	OFF-SITE A	
SAMPLE SOURCE	SAMPLES <sup>1</sup>	VOCs <sup>2</sup>	MERCURY	VOCs1	MERCURY
Temporary Well Points	20	20	20	100 %	10 %
Existing Monitoring Wells	8	8	8	1	1

NOTES:

VOCs = Volatile Organic Compounds

1 = Minimum of 20 temporary well point samples; additional samples may be collected to further delineate the distribution of VOCs and mercury.

2 - Field laboratory analysis for the following:

Tetrachloroethylene	Ethylbenzene
Trichloroethylene	Xylenes
1,1-Dichloroethylene	2-Butanone
1,2-Dichloroethylene	2- Hexanone
1,1,1-Trichloroethane	4-Methyl-2-Pentanone
Benzene	Acetone
Toluene	Vinyl Chloride

3 = Off-site laboratory analysis for Target Compound List (TCL) VOCs and SVOCs (including search for tentatively identified compounds) and Target Analyte List (TAL) metals.

No ecological risk assessment is proposed since the site is and will continue to be an industrial/commercial property without significant habitat, and no surface water or sediments are present at the site.

Risk-based soil and groundwater quality goals at the site will be developed based on the current and foreseeable uses of the property, using a methodology similar to that in NYSDEC's Revised TAGM - Determination of Soil Cleanup Objectives and Cleanup Levels TAGM HWR-94-4046, except that: the human health levels will be based on the planned future use of the site, i.e. as a commercial or an industrial property and on the current and reasonably foreseeable uses of the underlying groundwater; achieving background levels of contaminants will not be a site goal, unless merited by the risk analyses; and no goal will be set lower than the site-specific quantitation level for a contaminant. The future land uses, which are the basis of the goals, will be enforced via deed restrictions and no other land uses will be implemented without further review by NYSDEC.

- The HHRA will include the following tasks:
  - data evaluation and summarization
  - identification of HHRA chemicals of potential concern (CPCs)
  - exposure assessment (for both current and future land uses)
  - toxicity assessment
  - risk characterization
  - derivation of risk-based goals

### 5.3.1 Data Evaluation and Summarization

The Phase 1 analytical results will be compiled and evaluated for suitability of use in the risk assessment. The data will be summarized by media and by separate areas of concern (AOCs). The product of this task will be a series of media-specific data summaries to be used in determining exposure point concentrations.

### 5.3.2 Identification of HHRA Chemicals of Potential Concern

HHRA CPCs are those chemicals which are included in the quantitative risk assessment. CPCs are COCs most likely to be risk contributors based on their toxicity, frequency of occurrence and their concentrations in soil or groundwater.

### 5.3.3 Exposure Assessment

The exposure assessment involves:

- characterization of the exposure setting (including current and future land use)
- identification of exposure pathways (including receptor identification)
- identification of exposure point concentrations
- quantification of exposures (including speciation and bioavailability issues potentially important for mercury or other CPCs)
- summary of exposures by receptor and land use

The exposures will be estimated for reasonable maximum exposure scenarios. Under current land use there is very limited exposure potential and the future use will be industrial/commercial with virtually the entire site being paved or covered by buildings. The only foreseeable potential soil exposures are those associated with construction activities during site redevelopment and more limited periodic below-ground maintenance activities. Exposure to potentially hazardous vapors potentially entering buildings through basement walls and/or floors will also be considered. Table 5-4 presents theoretically possible exposure pathways and identifies those pathways and receptors to be evaluated. A groundwater receptor analysis will be conducted to determine if there are any potential receptors for groundwater by any environmental pathway. Figure 5-2 presents the theoretically possible mercury migration pathways and exposure pathways for the site. The Phase I investigation will determine which of these pathways are in fact complete or likely to be complete given the planned use of the site. In particular, information will be gathered to determine 1) if there is mercury contamination in groundwater, and if so, if it is volatile; 2) if there are any potable or non-potable uses of the groundwater; 3) if there is mercury at depth such that partitioning between soil, soil gas and groundwater is occurring. Only those pathways which are determined to be complete will be evaluated quantitatively.

Quantification of exposures to soil and groundwater may include an assessment of bioavailable mercury as discussed below.

## 5.3.4 Toxicity Assessment

The toxicity assessment will identify available information on the potential toxic effects of CPCs and the relationship between exposure (and dose) and likelihood and/or severity of adverse human health effects associated with potential exposures. The toxic effects associated with each of the CPCs will be summarized. The dose-response values for the CPCs will be obtained from USEPA's Integrated Risk

## ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

# **TABLE 5-4**

# SUMMARY OF POTENTIAL EXPOSURE PATHWAYS AND RECEPTORS AMES STREET SITE INVESTIGATION WORK PLAN - PHASE 1

LAND USE	MEDIUM	RECEPTOR	EXPOSURE ROUTES	SELECTED FOR EVALUATION?	RATIONALE
Current	Soils	Trained site investigators/workers	Ingestion Dermal Inhalation (dust and/or vapors)	No	Current workers are operating under a health and safety plan.
	Soils	Trespassers	Ingestion Dermal Inhalation (dust and/or vapors)	No	Access to the site is limited by a fence and ground surfaces are paved/covered by buildings.
· .	Groundwater	Area residents	Ingestion Dermal Inhalation of vapors	Yes	Neighboring residences are served by remote public water supply. This will only be addressed if a potentially complete exposure pathway is identified.
	Groundwater	Site investigators/workers	Ingestion	No	There is no use of groundwater for potable or non-potable uses on site.
	On-site Surface Water/ Sediment	Trespassers, Site workers	Ingestion Dermal Inhalation	No	No surface water/sediment present on site.

1
## TABLE 5-4 (cont'd)

1

LAND USE	MEDIUM	RECEPTOR	EXPOSURE ROUTES	SELECTED FOR EVALUATION?	RATIONALE
Future	Soils	Residents	Ingestion Dermal Inhalation (dust and/or vapors)	No	Property will be in industrial or commercial use with virtually total paving/building coverage and deed restrictions which prevent exposure to soils.
	Soils	Construction Workers/ Below- Ground Maintenance Workers		Yes	During and after site redevelopment construction workers/ maintenance workers are potentially exposed to impacted soils
	Soils	Industrial/Commercial workers and site visitors	Ingestion Dermal Inhalation (dust and/or vapors indoors and outdoors)	Yes	After redevelopment basement or other sub-grade areas may be periodically occupied.
	Groundwater	On-site residents On-site workers On-site visitors	Ingestion Dermal Inhalation	Νο	Area is served by public water supply. Deed restriction will prohibit use of site groundwater as drinking water.
	Groundwater	Area residents	Ingestion Dermal Inhalation	Yes	Area is served by public water supply. This will only be addressed if a potentially complete exposure pathway is identified.

1

1

TABLE54.DOC

•

Information System (IRIS), Health Effects Assessment Summary Tables, the Agency for Toxic Substance and Disease Registry and other published scientific literature. The doseresponse values used will be Reference Doses (RfD) for non-carcinogenic effects and Cancer Slope Factors (CSFs) for carcinogenic effects. For CPCs that can be present in both bioavailable and non bioavailable forms, published estimates of their bioavailability will generally be used where appropriate. A combination of speciation and in vitro bioavailability testing may be used to determine the relative bioavailability of the mercury which is found in soils at the site.

#### 5.3.5 Risk Characterization

The risk characterization will utilize the toxicity assessment and exposure assessment results to identify the nature and magnitude of risks to human receptors associated with potential exposures to CPCs. The risks will be evaluated by comparing site concentrations to published soil cleanup levels and/or to site-specific risk-based concentration goals. The site-specific risk-based concentration goals will be associated with acceptable levels of non-carcinogenic (hazard quotient) and carcinogenic risks (excess lifetime cancer risk). The identified risks will be put into context by comparison to commonly used risk management criteria. Uncertainties in the risk characterization will be discussed. In particular, the impact of various assumptions and of uncertain values will be discussed.

#### 5.3.6 Use of Risk-based Mercury Screening Concentrations in the Sampling and Analysis Program

Risk-based screening concentrations for total mercury will be used as a component of a decision process to determine the nature and extent of sampling and analyses that will be conducted for mercury in soil, groundwater and possibly soil gas. These conservative screening concentrations, in general, indicate, for a particular exposure pathway, a concentration below which human health risks would not expected to be significant. If concentrations are above these screening values, some additional sampling and analyses may be required to determine the nature and magnitude of the associated site-specific human health risks and to develop site-specific risk-based goals. Risk-based screening levels will be employed for the following exposure pathways: ingestion of groundwater (if receptors are identified), soil ingestion, and inhalation of vapors potentially released from soil and groundwater into buildings. The following sections describe how each screening level has been derived. Section 5.5 describes how the screening levels will be used to adjust the Phase I analytical program.

5.3.6.1 Setting the Mercury Screening Levels for Groundwater. Due to its potential importance at the site, a multi-step goal setting process for mercury in groundwater will be

#### ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

followed. Initially, mercury concentrations in groundwater at the downgradient property boundary will be compared to the New York groundwater standard for mercury in class GA groundwaters (2 ug/l). If concentrations exceed the standard, it will be determined if there are any complete or potentially complete ingestion exposure pathways for the groundwater and additional well points may be installed to trace contaminants back to a source. If there is a complete or potentially complete ingestion exposure pathway, filtered analyses of groundwater from representative wells will be conducted to determine if the mercury is dissolved and therefore actually representative of potential drinking water ingestion exposures. In addition, a groundwater mercury concentration which theoretically might be associated with vapor migration into basements of buildings at a concentration of concern will be identified through transport modeling. This screening concentration is based on the potential for mercury vapors to be released form groundwater into soil gas in the subsurface soil and subsequently, diffusion-based transport of vapors through subsurface basement walls and/or floors of buildings which might be constructed in the future. This screening concentration would be utilized to identify candidate locations for future soil gas or groundwater monitoring activities if they are deemed necessary. This soil gas monitoring would be conducted to determine if indeed soil gas contains mercury vapor at concentrations of potential concern.

5.3.6.2 Setting the Mercury Screening Levels for Soil. Risk-based soil screening levels will be employed for the following exposure pathways: soil ingestion and inhalation of vapors potentially released from soil and groundwater into buildings. These screening levels are discussed in the following paragraphs.

Initially, a default human toxicity-based soil screening level for soil ingestion based on an industrial/commercial land use will be identified. The total mercury levels measured in soil at various AOCs will be compared to this screening level. If, in a given area, no total mercury levels are found above the screening level, no mercury speciation or bioavailability testing will be necessary for soils in that area. However, although not strictly necessary, some samples with total mercury concentrations below the screening level will be evaluated with respect to speciation and bioavailability to provide a complete picture of site conditions. It will be assumed that mercury in that area is not of significance with respect to soil ingestion. Investigations and evaluation of other exposure pathways will also be conducted in these areas. In areas where total mercury levels are above the soil ingestion screening level, mercury speciation and possibly bioavailability analysis will be conducted in order to develop the soil ingestion component of a site-specific, future site use, risk-based goal.

Based upon a literature review, the initial soil ingestion mercury screening concentration will be 200 mg/kg. This number, representing an exposure which is acceptable for a lifetime exposure, was published by USEPA Region III as a screening level for soils in an industrial

ABB	ENV	IRONN	MENTAL	SERVICES,	INC.
-----	-----	-------	--------	-----------	------

setting (USEPA, 1995). This number is based on the reported toxicity of methyl mercury, which is considered the most toxic form of mercury via the ingestion exposure route. It assumes that a 70 kg adult is exposed via incidental ingestion of 100 mg soil per day (half of which comes from the site), 250 days per year for a period of 25 years. The background information supporting this screening level is presented in Appendix H. Given the planned commercial/industrial use of the property and the planned paying of virtually the entire property, ingestion exposures associated with soil concentrations below this screening level should not be a public health concern. As a confirmation of the utility of this screening level for this site, ABB-ES has also calculated a soil ingestion screening concentration specifically for a construction worker. This screening level is 165 mg/kg. It assumes a 70 kg adult is exposed via incidental ingestion of 118 mg soil per day (half of which is derived from the site), 5 days per week for an entire year. The derivation of this screening concentration is shown in Appendix I. A screening concentration for the construction worker dust inhalation scenario will not be incorporated into the decision tree for determining which chemical analyses will be conducted. A preliminary evaluation indicates that such a screening level would be well above the 200 mg/kg soil ingestion screening value, and therefore would not be a factor in determining if further chemical analyses be conducted.

In those areas where no total mercury concentrations are found above the screening level, no further sampling and analysis will be required to determine the human health risk associated with soil ingestion exposure. In that case, the soil ingestion screening level would become the soil ingestion component of the mercury soil goal at the site as discussed below. It should be noted that the total mercury analytical method will have a detection limit of 0.1 mg/kg, even though the soil ingestion screening level is substantially higher than that.

A human toxicity-based soil screening level based on potential inhalation of vapors released from groundwater and subsurface soil into buildings will be identified. This screening level represents a mercury soil concentration which would be protective for two exposure pathways: direct migration of vapors from soil into buildings; and leaching of mercury from soil to groundwater with subsequent partitioning of mercury to soil gas followed by diffusion-based migration into buildings. These pathways are depicted in figure 5-2. The screening concentration will be based on a target indoor air concentration for mercury which is equal to the OSHA Permissible Exposure Limit (PEL) and will utilize EPA approved modeling techniques for leaching to groundwater as well as partitioning of vapors between groundwater, soil gas and buildings. The total mercury levels measured in soil at various AOCs will be compared to this screening level. Areas where this soil screening level is exceeded may be targeted for mercury soil gas analysis to determine if this potential mercury migration pathway is a significant pathway at this site.

#### 5.3.7 Development of Risk-Based Goals

Site-specific risk-based goals will be developed once the Phase I data are obtained.. The current and future uses of the property (with associated potential receptors and institutional controls), the toxicity, fate, and transport characteristics of the CPCs and the speciation and bioavailability of the CPCs will be considered. Direct soil contact, potential leaching of CPCs to groundwater, and inhalation of CPCs potentially released to indoor air (vapors) and outdoor air (vapors and soil particulates) will be considered in developing soil and groundwater goals. The construction worker and maintenance worker exposure scenarios will likely be the basis of the direct soil contact calculations since other exposures to soil will be precluded by industrial/commercial use, paving, and institutional controls. The groundwater into buildings. It should be noted that exceedance of risk-based goals is expected to trigger some action to eliminate or minimize medium-specific and route-specific exposure may be warranted. Such actions may involve removal, treatment, engineering controls, health and safety protection measures, and /or institutional controls.

**5.3.7.1 Setting a Site-Specific Risk- and Future Use-based Mercury Groundwater Goal.** The site-specific risk- and future use-based mercury groundwater goal will be developed in a manner consistent with accepted risk assessment practices. If it is determined that a potentially complete exposure pathway for potable or non-potable use of groundwater exists, one component of the groundwater goal would be the Class GA groundwater standard of 2 ug/liter. Another component of the groundwater site goal could involve a concentration which would be protective of potential vapor migration from the subsurface through basement walls and/or floors if no potable or non-potable use of the groundwater are identified. Potential exposures to vapors released from groundwater may be controlled by means other than reducing concentrations The process which will be employed to derive the mercury groundwater site goal is diagrammed in Figure 5-3.

**5.3.7.2 Setting a Site-Specific Risk- and Future Use-based Mercury Soil Goal.** When total mercury in soil are found above the screening levels, then site-specific goals will be derived through quantification of potential future exposures to mercury in soil and other media potentially impacted by release of mercury from soil. This quantification may include an assessment of speciation and bioavailability of mercury in soils.

The mercury soil goal will be identified through the development of health risk-based concentrations for each potentially complete exposure pathway (e.g. ingestion of soil, inhalation of dust). The risk-based soil concentrations for each of the pathways will be referred to as the "components" of the mercury soil goal. The mercury soil goal will be set

ABB EN	<b>IVIRON</b>	MENTAL ·	SERVICES,	INC.
--------	---------------	----------	-----------	------

at the lowest mercury soil concentration that is represented by one of the components, unless that particular component will be controlled by some engineering or institutional control measures, a process which is shown graphically in Figure 5-4. In such a case, the next lowest concentration among the components would become the mercury soil goal. The four components of the mercury soil goal include:

- 1) soil ingestion by a construction worker;
- 2) inhalation of soil particulates (dust) by a construction worker; 3) inhalation of mercury vapors by a construction worker; and 4) inhalation of vapors by a future on-site worker (mercury released from soil to soil gas and/or potentially leaches from soil to groundwater with subsequent release of vapors to soil gas and into buildings). The allowable risk-based concentration for each of these pathways would be derived using standard risk assessment procedures and readily available, USEPA -endorsed fate and transport modeling techniques. This modeling is required for the estimation of exposures since monitoring of many of these potential exposures is impossible (no construction is ongoing and there are no intact buildings where indoor air monitoring could be conducted).

The Class GA groundwater standard and the vapor migration- based groundwater screening concentration (if soil gas monitoring confirms its validity) will be used to set a site-specific soil goal that is intended to protect the underlying groundwater. The methodology for setting this groundwater protection soil goal will be based upon NYSDEC's revised TAGM HWR-94-4046.







.

Depending on the results of the speciation analysis, it may be necessary to adjust the soil ingestion exposures to coincide with the medium of exposure (soil) and the chemical species present because they are different than the media and mercury species which are involved in the derivation of the oral RfD for mercuric chloride (provided methyl mercury is not present at the site). Appendix A of USEPA's Risk Assessment Guidance for Superfund, Volume I, Human Health Evaluation Manual (Part A), Interim Final (USEPA, 1989) recommends adjustment of exposure estimates if the site-related exposure involves a medium which is different from the medium of exposure assumed by the RfD. Such an adjustment is appropriate here, since the site exposure is to soil while the RfD exposures involve water or mercuric chloride itself. In addition, if the chemical species (or mix of species) is different than the chemical species used in the RfD development (mercuric chloride), it may be more or less bioavailable than the species in the RfD development and therefore may have a significantly different toxicity based on administered dose. In such a case, adjustment of the exposure to reflect the reduced or enhanced bioavailability is appropriate. This adjustment would be accomplished via the application of a bioavailability adjustment factor (BAF) to soil ingestion exposure estimates. The approach for derivation of the BAF for mercury in site soils, including speciation analysis and in vitro bioavailability testing is described in Appendix J.

Mercury speciation analytical results will be reviewed and evaluated to determine if in vitro bioavailability testing will be conducted. In those areas where the speciation techniques clearly indicate the mercury species in soil are insoluble species which are reported in the literature to be not bioavailable via ingestion (such as elemental mercury or mercury sulfides), it will be concluded that for soil ingestion, the mercury is not of toxicological significance. As stated in Casarett and Duoll's Toxicology: The Basic Science of Poisons, Fifth Edition, "metallic mercury, such as may be swallowed from a broken thermometer, is only slowly absorbed by the gastrointestinal tract (0.01 percent) at a rate related to vaporization of the elemental mercury and is generally thought to be of no toxicological consequence." No in vitro bioavailability testing will be conducted for samples taken in such areas.

Speciation and bioavailability testing will not be conducted for any inhalation exposures. There are published inhalation toxicity values for the volatile mercury species. Therefore, no adjustments due to matrix differences or mercury speciation differences will need to be made.

While not planned at this time, laboratory tests to determine site-specific bioavailability of other CPCs may be conducted.

#### ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

## 5.4 DATA QUALITY OBJECTIVES

As stated in Section 5.1, soil sampling will be performed at the various AOCs to characterize the distribution of VOCs, mercury, and/or selected metals in subsurface soils. The intent of the soil sampling program is to provide soil quality data for comparison to usage-based soil quality goals.

Groundwater samples will be collected down-gradient at the property boundary and from available on-site monitoring wells. The purpose of the groundwater evaluation is to characterize groundwater quality on-site and provide data for comparison to usage-based quality goals.

Data quality objectives (DQOs) are qualitative and quantitative statements developed by the data users to specify the quality of data needed from a particular data activity to support specific decisions. DQOs are based on the concept that the intended use of the data determines the quality of the data required. As such, DQOs for this Phase 1 voluntary investigation have been developed to ensure that the data used to support decisions are of sufficient quality for the intended data uses (see Appendix A).

#### 5.4.1 Overview

In accordance with USEPA guidance, the DQOs are developed after considering the intended data uses, the COCs, the detection limits to meet criteria concentrations (e.g., risk-based or regulatory standards), and analytical QA and QC requirements and documentation. DQOs for the site investigation are presented in the site-specific QAPjP in Appendix A.

#### 5.4.2 Intended Data Uses

Phase 1 investigation analytical data collected at the Ames Street Site must be adequate for the following primary data uses:

- to characterize the soils beneath each AOC
- to compare to usage-based soil quality goals
- to evaluate groundwater quality on-site
- to compare to usage-based groundwater quality goals
- to provide useable data to conduct a risk assessment to set soil and groundwater quality site goals

## 5.4.3 Chemicals of Concern

The COCs identified in Section 4.0 are summarized below, and include VOCs, metals, and inorganics:

<u>VOCs</u>

#### Metals and Inorganics

tetrachloroethene (PCE) TCE	mercury (elemental, bioavailable, speciation) cadmium
1,2-dichloroethene (1,2-DCE)	chromium, trivalent
1,1-dichloroethene (1,1-DCE)	chromium, hexavalent
1,1,1-trichloroethane (TCA)	lead
benzène	nickel
toluene	zinc
ethylbenzene	cyanide
xylenes	
acetone	
4-methyl-2-pentanone (methyl is vinyl chloride	sobutyl ketone)

Analytical protocols for identifying and quantifying concentrations of the above mentioned COCs meet the DQOs for this project and are presented in Section 5.4 and Appendix A.

#### 5.4.4 Sample Quantitation Limits

Concentrations detected in the soil and groundwater samples collected as part of the Phase 1 site investigation will be compared to use-based soil and groundwater quality goals established by the Phase 1 risk assessment. Since these quality goals are not yet established, quantitation limits for Phase 1 will be determined by other existing regulatory-based soil and groundwater standards.

#### 5.5 ANALYTICAL PROGRAM

Selection of analytical procedures to be used for this program and the required analytical deliverables were chosen to support the DQOs and to provide technically and legally defensible data.

The detailed sampling program is described in Sections 5.1 (soil) and 5.2 (groundwater), and the associated analytical methods and parameters are described in the site-specific QAPjP (Appendix A). Off-site laboratory chemical analysis will be conducted at a frequency of 5

percent of all soil samples and 10 percent of all groundwater samples provided to a laboratory certified by the New York State Department of Health.

On-site chemical analysis will be conducted at a DQO Level III (consistent with off-site analysis) for selected VOCs by USEPA SW-846 modified methods 8010/8020/8015 and mercury by USEPA SW846 modified methods 7470/7471. Off-site chemical analysis will be performed for the TCL list of SVOCs by method 8270, VOCs by methods 8240/8260, the TAL list of metals by methods 6010/7000s, cyanide by method 9010, and mercury by methods 7470/7471. A few soil samples will be submitted for special mercury analyses being conducted to assess the bioavailability of mercury in soils with respect to potential ingestion of soil. These specialized analytical method are summarized below; details of these methods are provided in the site-specific QAPjP (Appendix A of the Phase I Workplan). The analytical program is also summarized in Table A-2 in the site-specific QAPjP.

The sampling and analysis for mercury in soils will be conducted consistent with the decision tree presented in Figure 5-5. The decision tree presents a tiered approach for determining what analysis for mercury will be conducted. The approach is intended to focus mercury speciation and bioaccessibility analysis on areas where total mercury concentrations (per USEPA SW846 methods 7470/7471) are above a risk-based soil ingestion screening concentration of 200 mg/kg. Given the planned commercial future use of the property and the planned paving of virtually the entire property, ingestion exposures associated with soil concentrations below this screening level should not be a public health concern, and no further analysis will be required for samples in AOCs where concentrations are below the screening concentration.

If, however, the total mercury soil concentration at a given location is greater than the soil ingestion screening level, additional analysis will be conducted to identify and quantify the mercury species which are present. The speciation analysis will be conducted via sequential extractions, electron microprobe analysis and heavy mineral separations and identification and quantification of the various mercury species. Species to be identified and quantified will include, but not be limited to, elemental mercury, mercury oxide, acid soluble mercury (including inorganic mercury, salts), organic mercury, and mercuric sulfide. If the two sequential extraction procedures indicate clearly that only species that are non-bioavailable via ingestion (elemental mercury and/or mercury sulfides) are present in soil, the electron microprobe analysis and the heavy mineral separation techniques may





not be implemented. Since the literature indicates that elemental mercury and mercuric sulfide are very poorly absorbed and have very low toxicity via the oral route, if these species are the only species present, no further analysis will be required and it will be assumed that the exposure associated with ingestion of the relevant soil is not of public health concern. However, if the potentially bioavailable mercury species are present (organic mercury such as methylmercury, inorganic salts other than sulfides), then the total concentration of those species will be compared to the soil ingestion screening level. If the screening level is not exceeded, it will be assumed ingestion exposure to those soils is not of public health concern and no further analysis will be required.

If the combined organic mercury and inorganic mercury (not including sulfides or elemental mercury) exceeds the 200 mg/kg screening value, samples will be submitted for an *in Vitro* bioaccessibility test conducted in a manner similar to the test described by Ruby et al (1993). and described in Appendix J. In this test, relative bioaccessibility of mercury in soil from the property is determined relative to mercuric chloride in a water medium. The relative bioaccessibility of mercury in site soils and the mercuric chloride in water will be used to develop a bioavailability adjustment factor (BAF) which will be used to develop a site-specific mercury soil concentration goal.

#### 5.6 DATA VALIDATION

Both off-site and on-site analytical results will undergo an evaluation to determine data usability. Details of the data validation process are included in the site-specific QAPjP (Appendix A).

#### 6.0 PHASE 1 REPORT

Following completion of all field activities and the receipt of analytical results from the laboratory, ABB-ES will prepare a Site Investigation Phase 1 Report summarizing the data and presenting the findings.

The report will be presented in two, separately-bound volumes. Volume 1 will contain the report and Volume 2 will contain the appendices. The Site Investigation Report will include the following:

EXECUTIVE SUMMARY

- 1.0 **PURPOSE**
- 2.0 BACKGROUND INFORMATION
  - 2.1 SITE DESCRIPTION
  - 2.2 SITE HISTORY
  - 2.3 **PREVIOUS INVESTIGATIONS**
- 3.0 SCOPE OF WORK
  - 3.1 SOIL BORINGS
  - 3.2 **GROUNDWATER SAMPLING**
  - 3.3 **RISK ASSESSMENT**
  - 3.4 LABORATORY ANALYSIS AND DATA VALIDATION
- 4.0 SITE CHARACTERIZATION
  - 4.1 **GEOLOGY AND HYDROGEOLOGY** 
    - 4.1.1 Geology
    - 4.1.2 Hydrogeology
  - 4.2 SUBSURFACE SOIL ANALYTICAL RESULTS
    - 4.2.1 Area of Concern 1
    - 4.2.2 Area of Concern 2
    - 4.2.3 Etc.
  - 4.3 GROUNDWATER ANALYTICAL RESULTS
- 5.0 RISK ASSESSMENT RESULTS
- 6.0 CONCLUSIONS AND RECOMMENDATIONS

#### REFERENCES APPENDICES

The appendices will include the field data sheets, laboratory deliverables, and other supporting documentation.

## ABB ENVIRONMENTAL SERVICES, INC.

G:\T92\KENT\_TAY\PHASEI\FINAL.DOC

7198-23

ABB will provide five copies of the Volume 1 Draft Site Investigation Report - Phase 1 for review by NYSDEC. After incorporation of comments, ABB-ES will issue five copies of the final Site Investigation Report - Phase 1 and three copies of Volume 2 for NYSDEC.

#### 7.0 SCHEDULE

The proposed schedule for the Ames Street Site Investigation is provided in Figure 7-1.

As shown, the on-site work will begin as soon as the Work Plan Addendum is completed. ABB-ES anticipates that the on-site work will take 25 working days to complete; however, activities included in the addendum could increase the length of this task. The final off-site laboratory data should be received within one month of the end of the on-site work. This data will not be available for use until data validation is completed (approximately two weeks after the final data package is received.

At this time, ABB-ES anticipates delivery of the draft Ames Street Site Investigation Report -Phase 1 the last week in March.

	Nov	Dec	Jan	Feb	Mar	Apr	May	June	July	Aug
Develop Phase I Work Plan										
NYSDEC Review										
Revise Phase I Work Plan										
Phase I Site Investigation										
Off-Site Laboratory Analysis										
Data Evaluation						_				
Risk Analysis										
Phase I Report/Risk Assessment										
Develop Phase II Work Plan										
Phase II Investigation/Remedial Action										

FIGURE 7-1 SCHEDULE SITE INVESTIGATION WORK PLAN - PHASE I 95 AMES STREET ROCHESTER, NEW YORK

.

1

Agency for Toxic Substances and Disease Registry, Toxicological profile for MERCURY, Draft, February 19, 1993

EPA Region III Risk-Based Concentration Table, January 31, 1995

Lozier Architects and Engineers, 1983. *Phase 1 Mercury Contamination*, prepared for Taylor Instrument Company.

Rickard, Lawrence V. and Donald W. Fisher, 1970. Geologic Map of New York, Finger Lakes Sheet.

Ruby, M.V., Davis, A., Link, T.E., Schoof, R., Chaney, R. L., Freeman, G.B., and P. Bergstrom, *Development of an in Vitro Screening Test To Evaluate the in Vivo Bioaccessibility of Ingested Mine-Waste Lead*, Environmental Science & Technology, Vol. 27, No. 13, 1993.

Williams, R.E., 1990. Dolomite Products Company, Inc. Quarry-Gates Center, New York, Geologic Report of Proposed Operation for 1990-1993: Dolomite Products Company, Inc., Rochester, New York, February 1990.

United States Department of the Interior Geological Survey, 1986. Water-Resources Investigation Report 86-4048, *Ground-Water Availability in the Genesee River Basin, New York and Pennsylvania*, John C. Kammerer and William A. Hobba, Junior.

ABB	Asea Brown Boveri
ABB-ES	ABB Environmental Services, Inc.
AOCs	areas of concern
BAF	Bioavailability Adjustment Factor
BTEX	benzene, toluene, ethyl benzene, and xylene
Cd	cadmium
Cr	chromium
COCs	Chemicals of Concern
CPCs	Chemicals of Potential Concern
CSFs	Cancer Slope Factors
DCE	dichloroethene
DQOs	Data Quality Objectives
HASP	Health and Safety Plan
HHRA	Human Health Risk Assessment
HQ	Hazard Quotient
IRIS	Integrated Risk Informations System
NYSDEC	New York State Department of Environmental Conservation
PCBs	poly-chlorinated biphenyls
PCE	tetrachloroethene
QA	quality assurance
QAPP	Quality Assurance Program Plan
QAPjP	Quality Assurance Project Plan
QC	quality control
RfD	Reference Dose
Sybron	Sybron Corporation
SVOCs	semivolatile organic compounds
TCA	trichloroethane
TCE	trichloroethene
TPH	total petroleum hydrocarbons

## ABB ENVIRONMENTAL SERVICES, INC.

.

## USEPA United States Environmental Protection Agency USTs underground storage tanks

VOCs volatile organic compounds

#### **APPENDIX A**

## SITE-SPECIFIC QUALITY ASSURANCE PROJECT PLAN

## **ATTACHMENTS:**

- 1 Low Flow Standard Operating Procedures
- 2 Clean Hands Sampling
- 3 ABB-ES' Standard Operating Procedure No. FGC00202
- 4 ABB-ES' Standard Operating Procedure No. FXRFS00102
- 5 Mercury Analyzer Manufacturer Documentation
- 6 Bioavailability Reference

#### ABB ENVIRONMENTAL SERVICES

G:\T92\KENT\_TAY\PHASEI\WORKPLAN..DOC

## QUALITY ASSURANCE PROJECT PLAN AMES STREET SITE INVESTIGATION - PHASE 1

This site-specific Quality Assurance Project Plan (QAPjP) is designed to identify the sections of the "Draft Quality Assurance Program Plan (QAPP), NYSDEC Superfund Standby Contract, Contract Number D002472" (ABB-ES, 1995) that apply to the activities described in the Site Investigation Work Plan - Phase 1. This QAPjP will also describe any variations to those procedures, and specify the analytical data quality objectives (DQOs) for the site and associated analytical methods used for the laboratory analysis of environmental samples.

<u>Project Organization</u>. Figure A-1 presents the Ames Street Site Investigation project team organization. Roles and responsibilities of personnel are those described in the QAPP.

<u>General Procedures and Practices</u>. The general procedures used to conduct the site investigation at the Ames Street Site will be taken from the following sections of the QAPP:

Section 5.0	Sample Custody
Section 6.0	Calibration Procedures
Section 9.0	Internal Quality Control
Section 10.0	Audits
Section 11.0	Preventive Maintenance
Section 12.0	Data Assessment
Section 13.0	Corrective Action

<u>Field Procedures and Sampling</u>. The following field investigation tasks and associated procedures set forth in Section 4.0 of the QAPP will be used at the Ames Street Site.

Subsection 4.1	Sample Labels and Records
Subsection 4.2	Sample Container and Preservation Requirements
Subsection 4.3	Decontamination Procedures
Subsection 4.4.3	Exploratory Drilling
Subsection 4.5.1	General Soil Sampling Methodology
Subsection 4.5.2.2	Groundwater Sampling (using the Low-Flow sampling technique described below)
Subsection 4.6	TerraProbe/Field Laboratory Analysis
Subsection 4.9	Surveys
Subsection 4.10	Management of Investigation-Derived Wastes



.

ı

Ĺ

ł

Low Flow Sampling Technique. The objective of the low flow sampling procedure is the same as the standard groundwater sampling procedure: to collect representative groundwater samples for chemical analysis. Low flow is specifically designed to minimize disturbance of the water column in the well and minimize the stress on the aquifer and drawdown inside the well. Purging and sampling is conducted at low rates (e.g., 0.2 to 0.5 liters per minute) with pump intakes located in the well screen zone in an effort to minimize the entrainment of solids (e.g., silt, clay, and colloids) and minimize mixing of the standing water column inside the well with water from the aquifer.

Bladder pumps approved by USEPA Region I for sampling VOCs will be used, and all sample vessels are filled directly from the pump, eliminating the use of bailers.

The low flow procedure outlined below is based upon the USEPA Region I Draft Final Low Flow SOP, GW 0001 dated August 10, 1994 (Attachment 1), and discussions with the Region I Hydrogeologist. The procedure is limited in scope to purging the well up to the time at which samples are ready to be collected. The procedure for filling sample vessels is covered in the standard groundwater sampling procedure, described in the QAPP.

- 1) Open the well and measure the depth to water through the opening in the well cap to the nearest 0.01 feet. Leave the probe in the well just out of the water.
- 2) Hook the pressurized gas line from the control box to the quick connect on the well cap and begin purging. Adjust the pump settings to control the discharge between 0.2 and 0.5 liters per minute. The low end of this range is preferred to minimize turbidity and drawdown.
- 3) When the discharge rate is stable, record the flow rate and begin measuring and recording time, water level, temperature, pH, specific conductance, dissolved oxygen and turbidity every three to five minutes.
- 4) Parameters are considered stable when the water level has stabilized within +/-0.1 foot, temperature, specific conductance, and dissolved oxygen have stabilized to within +/- 10 percent, pH has stabilized to within +/- 0.2 pH units, and turbidity is less than or equal to 10 NTU or stable to within +/- 10 percent if 10 NTU does not appear to be achievable.
- 5) Purging is complete when parameters are stable as in item 4, and if the volume of water removed due to drawdown is less than 0.3 of the total volume removed during the course of purging the well. If the drawdown volume exceeds 0.3 of the total volume, continue purging until this ratio decreases to less than 0.3, continuing to record all parameters every three to five minutes to assure the stability criteria are still met.

6) Collect samples directly into sample vessels from the pump discharge. When all vessels are filled, shut off the pump and record the time and the total volume purged from the well. Remove the water level probe and disconnect the pressurized gas line. Close and secure the well.

<u>Clean-Hands Sampling Technique</u>. This SOP for collecting water samples for mercury analysis may be followed when performing groundwater sample collection. The technique employs two people, one designated as the "clean-hands person" and the other designated as the "dirty-hands person". This SOP is included as Attachment 2.

<u>Sampling and Analysis Program</u>. The results of the DQO analysis for the Ames Street Site sampling activities are summarized in Table A-1. The analytical data requirements were established using methods described in "Test Methods for Evaluating Solid Waste" (USEPA, 1992). The analytical methods to be used for both on-site and off-site analysis are presented in Table A-2.

On-site volatile organic analysis compound (VOC) analysis will be in accordance with ABB-ES' Standard Operating Procedure No. FGCPT00202, dated 4/14/93 (see Attachment 3). Analysis for VOCs will be performed using a gas chromatograph (GC) equipped with three detectors in series. A photoionization detector (PID) will be used for the identification of selected volatile aromatics, a flame ionization detector (FID) will be used for the non-halogenated volatiles (ketones), and a electrolytic conductivity detector (ELCD) will be used for the identification of the selected halogenated volatiles. The GC will be connected to a purge and trap (P&T) device for sample preparation and instrument delivery. The approximate reporting limits for the selected VOCs are provided in Table A-3. Actual method detection limits will be determined on site at the start of the field program. All quality control protocols described in the SOP will be included as part of the on-site laboratory program so that Level III analytical data quality is achieved. Quality control protocols include routine calibration, method blanks, surrogate standards, matrix spike/matrix spike duplicate samples, laboratory duplicate samples.

Soil samples to be analyzed for mercury will be first analyzed for gross mercury contamination using x-ray fluorescence (XRF). Analytical procedures will be inaccordance with ABB-ES' Standard Operating Procedure No. FXRFS00102 (see Attachment 4). The reporting limit is expected to be approximately 100 mg/kg.

Soil samples with mercury concentrations at or below the reporting limit requiring characterization to levels below 100mg/kg and all groundwater samples will be analyzed analyzed in accordance with instrument manufacturer documentation (provided in Attachment 5) for sample preparation and delivery combined with the cold vapor atomic absorption (CVAA) Method 7470/7471. A Leeman Lab's Automated Mercury Analyzer will be connected to a tabletop atomic absorption (AA) in the field laboratory. The Leeman Mercury Analyzer performs sample preparation and digestion followed by sample introduction to the AA. The reporting limit is expected to be at least 0.1 parts per million

## TABLE A-1

## ANALYTICAL DATA QUALITY OBJECTIVE LEVELS AMES STREET SITE INVESTIGATION WORK PLAN - PHASE 1

PARAMETER	LISE	
pH, turbidity,	To provide background physical data on	Levell
temperature, specific conductivity	liquid and groundwater samples.	
PID screening	To provide qualitative real-time information	Level I
Mercury vapor analyzer screening	on air quality in the breathing zone for health and safety decisions and to identify potential contaminated materials.	
On-Site and Off-site VOCs, metals, and Inorganics	To provide analytical information to evaluate the presence and concentration of organic and inorganic compounds in site media, and compare to applicable state and federal soil quality guidance values.	Level III
Surface Soil/Soil Boring/Groundwater Samples		
Off-site Mercury Speciation/ Bioavailability	To provide analytical data to evaluate mercury speciation and bioaccessibility for risk assessment soil goal.	Level V

NOTES:

Level I: Qualitative information generated on-site using portable equipment for identification of sampling locations and health and safety monitoring.

Level III: These data represent laboratory information generated using USEPA-approved procedures, but using methods other than Contract Laboratory Program protocols. These data are both qualitative and quantitative and are used for site characterization, risk assessment, and engineering studies.

Level V: These data represent fixed base laboratory analysis results using non-standard methods that require development for specific constituents and/or detection limits. Data obtained from these analyses are often used for risk assessment purposes.

#### TABLE A-2

## ANALYTICAL PROGRAM AMES STREET SITE INVESTIGATION WORK PLAN - PHASE 1

MEDIA & PROGRAM	PARAMETER	METHOD <sup>1</sup>
Soil - On-Site Analysis <sup>2</sup>	Selected VOCs <sup>3</sup>	8010/8015/8020 -
		Modified
	Mercury	7471 - Modified
		X-ray fluorescence
Groundwater - On-Site <sup>2</sup>	Selected VOCs <sup>3</sup>	8010/8015/8020 -
Analysis		Modified
	Mercury	7470 - Modified
Soil & Groundwater	TCL SVOCs	8270
Off-Site Analysis <sup>4</sup>	TCL VOCs	8240/8260
	Mercury	7471/7470
	TAL Metals	6010/7000s
	Cyanide	9010
	Lead	7421/7420
Special Hg-ingestion Analysis	Mercury	Special Method⁵

NOTES:

- 1 USEPA, 1986. "Test Methods for Evaluating Solid Waste", SW-846, Third Edition with revisions.
- 2 On-Site analysis includes adherence to all quality control protocols described in ABB-ES' SOP for "Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography" SOP No. FGCPT00202, 4/14/93. On-site mercury analysis follows manufacturer's recommendations and methods stated above. (See Appendix A).
- 3 Selected VOCs include: tetrachloroethene (PCE); trichloroethene (TCE); 1,1-dichloroethene; 1,2-dichloroethene; 1,1,1-trichloroethane; benzene; toluene; ethylbenzene; xylenes; acetone; 4-methyl-2-pentanone.
- 4 Off-site split-sample analysis being conducted at a frequency of 5 percent for soil samples and 10 percent for groundwater samples. Analysis to be conducted at a New York State Department of Health certified laboratory.
- 5 Special Hg-ingestion method being performed for risk assessment purposes. Refer to section 5.5 of text for brief explanation of analyses.

#### **APPENDIX A**

## TABLE A-3

# CHEMICAL OF CONCERN QUANTITATION LEVELS FOR SOIL AMES STREET SITE INVESTIGATION WORK PLAN - PHASE 1

Description		Estimated Instrumental Quantitation	Field Lab Detection Limit	TAGM Recommended Soil Cleanup
Parameter	Analytes		(maximum)	Objectives
VOCs	Tetrachloroethene	5 μg/kg	10 μg/kg	1,400 μg/kg
1	Trichloroethene	5 μg/kg	10 μg/kg	700 μg/kg
}	1,1-Dichloroethene	5 μg/kg	10 μg/kg	200 µg/kg
	1,2-Dichloroethene	5 μg/kg	10 µg/kg	100 μg/kg
	1,1,1-Trichloroethane	5 μg/kg	10 μg/kg	800 µg/kg
	Benzene	5 μg/kg	10 μg/kg	60 µg/kg
	Toluene	5 μg/kg	10 µg/kg	1,500 μg/kg
	Ethyl benzene	5 μg/kg	10 μg/kg	5,500 μg/kg
	Xylenes	5 μg/kg	10 μg/kg	1,200 µg/kg
	Acetone	5 μg/kg	10 µg/kg	200 µg/kg
	4-Methyl-2-pentanone	5 μg/kg	10 μg/kg	1,000 µg/kg
	Vinyl chloride	5 μg/kg	20 µg/kg	200 μg/kg
Metals	Mercury	0.2 μg/kg	NĀ	100 μg/kg
	Cadmium	42 µg/kg	NA	1,000 µg/kg
	Chromium (total)	7 μg/kg	NA	10,000 μg/kg
}	Lead	42 µg/kg	NA	30,000 µg/kg
	Zinc	2 μg/kg	NA	20,000 µg/kg
	Nickel	15 µg/kg	NA	13,000 μg/kg
Other	Cyanide	1,000 μg/kg	NA	Not available

<sup>1</sup> New York State Department of Environmental Conservation, TAGM, "Determination of Soil Cleanup Objectives and Cleanup Levels"; November 16, 1992.

## ABB ENVIRONMENTAL SERVICES

07198.17

(ppm) for soil samples and 0.002 ppm for water samples. Quality control protocols include routine instrument calibration, method blanks, matrix spike samples, and laboratory duplicate samples.

On-site soil and groundwater analytical results will be verified by collecting and submitting split samples to an off-site New York State Department of Health-certified laboratory at a frequency of 5 and 10 percent, respectively. Split samples for groundwater collection is being done at a higher frequency due to the limited number of total groundwater samples planned for collection.

Off-site analyses for routine analysis (VOCs, metals, cyanide, elemental mercury) will be performed by a New York Department of Health certified laboratory using method protocols described in USEPA SW-846 (USEPA, 1992).

Soil samples may also be submitted for mercury speciation and bioavailability testing if the total mercury results are greater than the to-be-determined risk-based screening level. The bioavailability testing is similar to that described by Ruby, et.al. (1993). A copy of the bioavailability testing article (Ruby, 1993) is provided in Attachment 6.

Data Reduction, Validation, and Reporting. Both off-site and on-site analytical results will undergo an evaluation to determine data usability. This evaluation will include the following:

- blank contamination to determine potential for laboratory and/or field sampling contamination
- surrogate standard recoveries for organic analyses for analytical accuracy
- field duplicates to assess sampling precision and environmental matrix heterogeneity
- matrix spike/matrix spike duplicate sample results for analytical precision and accuracy, and to assess any potential for matrix interferences
- initial and continuing calibration for on-site analyses only to assess analytical precision and accuracy

Once the data have been reviewed by a chemist, data flags may be determined appropriate to qualify the data before use. All data, unless rejected during data evaluation, will be used for making decisions. As such, both on-site and off-site analytical data will be generated at DQO Levels III and V.

Data generated in the field will be available to the project team on a real-time basis for decision making. The analytical data from the on-site laboratory are generated using

existing software that produces text-formatting output from the GC and AA. The data are then converted into database-formatted files using ABB-ES-developed software. This reformatting allows for initial editing and evaluation of the data. Data are both reviewed by the site and project chemists before results are considered "final". Preliminary (unevaluated) data are available real time and will be used for interpretive purposes.

The data management program proposed for this site investigation is intended to provide an efficient and cost effective method for production and interpretation of data as it is obtained in the field. This program includes procedures that will ensure consistency and validity of the data generated and interpreted.

<u>Sample Identification</u>. Sample identification will adhere to Section 4.1.1 of the QAPP with the following exceptions:

Digits 1,2 All sample identification will be begin with the AOC number (e.g., 01, 02) for soil samples and "KT" for non-AOC related samples, such as groundwater samples.

Digits 11,12 The event number will be the year the samples are collected (96).

## ATTACHMENT 1

## LOW FLOW STANDARD OPERATING PROCEDURE

## ABB ENVIRONMENTAL SERVICES

G:\T92\KENT\_TAY\PHASEI\WORKPLAN.DOC

07198.17

OPTIONAL FORM 88 (7-99)		
FAX TRANSMIT	TAL For pages = 6	
"hate Hagelin	From Dicks Willey	SOP #: Region
Dopt./Agency ABB	617-573-9839	Revisi
Fax# 207-772-4762	Fax 3	Date: Page 1
NBN 7540-01-317-7368 5099-101	GENERAL SERVICES ADMINISTRATION	

BOP #: GW 0001 Region I Low Flow SOP Revision Number: 0 Date: August 10, 1994 Page 1 of 6

#### U.S. ENVIRONMENTAL PROTECTION AGENCY REGION I

GROUND WATER SAMPLING PROCEDURE LOW FLOW PURGE AND SAMPLING

#### DRAFT FINAL

#### I. SCOPE & APPLICATION

The purpose of this standard operating procedure (SOP) is to provide information on collecting ground water samples using a pump. The procedure will in most situations provide for the collection of ground water samples with minimum turbidity. This procedure is designed to be used in conjunction with the analyses for the most common types of ground water contaminants (volatile and semi-volatile organic compounds, pesticides, PCBs, metals and inorganic compounds).

#### II. EQUIPMENT

- Adjustable rate, positive displacement pumps (e.g., centrifugal or bladder pumps constructed of stainless steel or Teflon). Peristaltic pumps may be used only for inorganic sample collection.
- Clear plastic bottom filling bailer may be used to obtain samples of light or dense non-aqueous liquids (LNAPL, DNAPL).
- Tubing Teflon or Teflon lined polyethylene must be used to collect samples for organic analysis. For samples collected for inorganic analysis, Teflon or Teflon lined polyethylene, PVC, Tygon or polyethylene tubing may be used.
- Water level measuring device, 0.01 foot accuracy, (electronic preferred for tracking water level drawdown during all pumping operations).
- Flow measurement supplies (e.g., graduated cylinder and stop watch).
- Interface probe, if needed.

Use of trademarked names does not imply endorsement by U.S. EPA but is intended only to assist in identification of a specific product.

SOP #: GW 0001 Region I Low Flow SOP Revision Number: 0 Date: August 10, 1994 Page 2 of 6

- Power source (generator, nitrogen tank, etc.).
- Indicator parameter monitoring instruments pH, turbidity, specific conductance, and temperature. Optional Indicators - eH and dissolved oxygen.
- Decontamination supplies.
- Logbook(s).
- Sample Bottles.
- Sample preservation supplies (as required by the analytical methods).
- Sample tags or labels.
- Well construction data, location map, field data from last sampling event.
- Field Sampling Plan.

III. PRELIMINARY SITE ACTIVITIES

- Check well for damage or evidence of tampering, record pertinent observations.
- Lay out sheet of polyethylene for monitoring and sampling equipment.
- Remove well cap.
- Measure VOCs (volatile organic compounds) at the rim of the well with a PID or FID instrument and record the reading in the field logbook.
- If the well casing does not have a reference point (usually a V-cut or indelible mark in the well casing), make one.
- Measure and record the depth to water (to 0.01 ft) in all wells to be sampled before any purging begins. Care should be taken to minimize disturbance of any particulate attached to the sides or at the bottom of the well.
- Measure and record (as appropriate) the depth of any DNAPL's or LNAPL's with an interface probe. Care should be given to minimize disturbance of any sediment which has accumulated at the bottom of the well. If LNAPL's or DNAPL's are present, a decision needs to be made on whether to collect samples of the free phase liquid(s) and/or the dissolved phase.

SOP #: GW 0001 Region I Low Flow SOP Revision Number: 0 Date: August 10, 1994 Page 3 of 6

#### IV. SAMPLING PROCEDURE

A positive displacement type pump will be used to purge and sample monitoring wells which have a 2.0 inch I.D. or greater well casing.

Pump, safety cable, tubing and electrical lines will be lowered slowly into the well to a depth corresponding to the center of the saturated screen section of the well. It is also advisable to keep the pump intake at least two feet above the bottom of the well, to prevent mobilization of any sediment present in the bottom of the well.

Measure the water level again with the pump in well before starting the pump. Start pumping the well at 0.2 to 0.5 liters per minute. Ideally, the pump rate should cause little or no water level drawdown in the well (less than 0.3 ft. and the water level should stabilize). The water level should be monitored every three to five minutes (or as appropriate) during pumping. Care should be taken not to cause pump suction to be broken, or entrainment of air in the sample. Record pumping rate adjustments and depths to water. Pumping rates should, if needed, be reduced to the minimum capabilities of the pump (e.g., 0.1 - 0.2 1/min) to avoid pumping the well dry and/or to ensure stabilization of indicator parameters. If the recharge rate of the well is very low and the well is purged dry, then sampling shall commence as sooh as the well has recharged to a sufficient level to collect the appropriate volume of samples with the pump.

During purging of the well, monitor the field indicator parameters (turbidity, temperature, specific conductance, pH, etc.) every three to five minutes (or as appropriate). The well is considered stabilized and ready for sample collection once all the field indicator parameter values remain within 10% for three consecutive readings. If the parameters have stabilized, but the turbidity is not in the range of the 5 NTU goal the pump flow rate should be decreased, and measurement of the parameters should continue every three to five minutes. Measurements should be obtained using a flow through cell (preferred method) or taken in a clean container (a glass beaker is ideal).

VOCs samples will be collected first and directly into prepreserved sample containers. All sample containers should be filled by allowing the pump discharge to flow gently down the inside of the container with minimal turbulence.

Samples requiring pH adjustment will have their pH checked to assure that the proper pH has been obtained. For VOC samples, this will require that a test sample be collected to determine the amount of preservative that needs to be added to the sample containers prior to sampling.
SOP #: GW 0001 Region I Low Flow SOP Revision Number: 0 Date: August 10, 1994 Page 4 of 6

If filtered metal samples are to be collected, the use of an in-line filter is preferred. A high pressure, in-line 0.45 um particulate filter will be pre-rinsed with approximately 400 ml of deionized water and attached to the discharge end of the pump's tubing. After the sample is filtered, it must be preserved immediately.

As each sample is collected, the sample will be labeled. Those samples (volatile organics, cyanide, etc.) requiring cooling will be placed into an ice cooler for delivery to the laboratory. Metal samples after acidification to a pH less than 2 do not need to be cooled.

After collection of the samples, the pump's tubing may either be dedicated to the well for resampling (by hanging the tubing inside the well), decontaminated or properly discarded.

Measure and record well depth.

Secure the well.

#### V. DECONTAMINATION

Sampling equipment will be decontaminated prior to use and following sampling of each well. Pumps will not be removed between purging and sampling operations. The pump and tubing (including support cable and electrical wires which are in contact with the sample) will be decontaminated by one of the procedures listed below. Alternative procedures must be approved by the Quality Assurance Office prior to sampling event.

Procedure 1

- Steam clean the outside of the submersible pump.
- Pump hot water from the steam cleaner through the inside of the pump. This can be accomplished by placing the pump inside a three or four inch diameter PVC pipe with end cap. Hot water from the steam cleaner jet will be directed inside the PVC pipe and the pump exterior will be cleaned. The hot water from the steam cleaner will then be pumped from the PVC pipe through the pump and collected into another container. Note: additives or solutions should not be added to the steam cleaner.
- Pump five gallons of non-phosphate detergent solution through the inside of the pump.
- Pump tap water through the inside of the pump to remove all of the detergent solution.

SOP #: GW 0001 Region I Low Flow SOP Revision Number: 0 Date: August 10, 1994 Page 5 of 6

Pump distilled/deionized water through the pump.

#### Procedure 2

- The decontaminating solutions can be either be pumped from buckets through the pump or the pump can be disassembled and flushed with the decontaminating solutions. It is recommended that detergent and isopropyl alcohol used in the decontamination process be used sparingly and water flushing steps be extended to ensure that any sediment trapped in the pump is flushed out. The outside of the pump and the electrical wires must be rinsed with the decontaminating solutions, as well. The procedure is as follows:
- Flush the equipment/pump with potable water.
- Flush with non-phosphate detergent solution (five gallons).
- Flush with tap water to remove all of the detergent solution.
- Flush with distilled/deionized water.
- Flush with isopropyl alcohol.
- Flush with distilled/deionized water.

#### VI, FIELD QUALITY CONTROL

Quality control samples are required to verify that the sample collection and handling process has not affected the quality of the ground water samples. All field quality control samples must be prepared exactly as regular investigation samples with regard to sample volume, containers, and preservation. The following quality control samples will be collected for each batch of samples (a batch may not exceed 20 samples). Trip blanks are required for the VOC samples at frequency of one per sample cooler.

- ► Field duplicate.
- Matrix spike.
- Matrix spike duplicate.
- Equipment blank.
- Trip blank (VOCs)

SOP #: GW 0001 Region I Low Flow SOP Revision Number: 0 Date: August 10, 1994 Page 6 of 6

#### VII. FIELD LOGBOOK

· .

A field log must be kept each time ground water monitoring activities are conducted in the field. The field logbook should document the following:

- Well identification.
- > Well depth, and measurement technique.
- Static water level depth, dats, time and measurement technique.
- Presence and thickness of immiscible liquid layers and detection method.
- Collection method for immiscible liquid layers.
- Pumping rate, drawdown, indicator parameters values, and clock time, at three to five minute intervals; calculate or measure total volume pumped.
- Well sampling sequence and time of sample collection.
- Types of sample bottles used and sample identification numbers.
- Preservatives used.
- Parameters requested for analysis.
- Field observations of sampling event.
- Name of sample collector(s).
- Weather conditions.
- QA/QC data for field instruments.

## ATTACHMENT 2

## CLEAN HANDS SAMPLING STANDARD OPERATING PROCEDURE

## ABB ENVIRONMENTAL SERVICES

## STANDARD OPERATING PROCEDURE

## CLEAN-HANDS TECHNIQUE (FOR WATER SAMPLING OF MERCURY) SOP-72

The concentration of many metals in ambient waters is typically very low, and collecting water samples that are representative of ambient conditions requires extreme care to prevent contamination during handling. This SOP outlines procedures for collecting water samples to be analyzed for mercury and its various forms. Water samples are more prone to be contaminated with mercury than other metals and require that specialized clean techniques be used. This SOP is based on a technique used by the University of Wisconsin Trout Lake Station. The following technique is commonly called the "clean-hands technique." It employs two people, one designated as the "clean-hands person" and the other designated as the "dirty-hands person."

## INITIAL STEPS AND PRECAUTIONS

Prior to collecting a water sample, the following steps or precautions must be taken to avoid sample contamination:

- 1. Prior to collecting the sample or handling the sampling equipment or sample bottles, a quick survey of the sampling area should be performed to identify sources of potential contamination to the sample (e.g., sources of dust, engines running, batteries). If there is an obvious source, it should be removed, cleaned, or isolated from the sample handling area.
- 2. Both the clean-hands and dirty-hands persons will wear lint-free suits that reduce the amount of airborne dust in the immediate vicinity of the sample. These suits can be made of nylon, Tyvek, or a plastic-coated material (e.g., Saranex). These suits should be kept isolated from dust and contamination (in plastic bags) until ready for use and discarded after they have been used once. Avoid bumping into potentially contaminated surfaces.

Two types of gloves will be used, an inner polyethylene glove and an outer vinyl glove. The inner polyethylene gloves are shoulder length. These gloves provide the first level of isolation (they are relatively clean compared to bare hands) and allow the sampler to put on the outer gloves without significantly contaminating them. These gloves should be taped to the shoulder of the suit or rubber-banded at the upper arm to prevent them from sliding down while working. The vinyl gloves are powderfree, clean-room gloves (e.g., Oak Class 100, powder-free). These gloves come in vacuum-sealed plastic bags containing 50 pairs. Once opened, the entire pack is potentially exposed to contamination. To minimize this potential, open only the end of the bag at the wrist end of the glove (not the fingertip end), and remove only one pair at a time. Keep the unused gloves in the original bag inside of a large Ziploc<sup>®</sup> bag. When handling the clean gloves, do not touch the fingertips: handle the gloves only around the wrist.

3. At all times, the clean-hands person must avoid touching surfaces that are not known to be clean, including their suit. The clean-hands person should only touch the inner bag and the sample bottle used during sample collection. While sampling, both samplers must be conscious of the potential chains of contamination that can occur. A chain of contamination could involve handling an object that touched another object that touched something contaminated. Unless it is known whether an object is clean (i.e., was cleaned appropriately and isolated from contaminants from the time of cleaning until the time of use), it should be assumed that it is dirty.

## SAMPLE COLLECTION PROCEDURE

After the initial steps have been taken, the clean-hands person and the dirty-hands person will use the following procedure to collect the samples:

- 1. To retrieve the sample bottle, the dirty-hands person will remove the double-bagged bottle from the ice chest and open the outer bag. While the dirty-hands person is holding the outer bag open, the clean-hands person will reach inside, lift the inner bag (but not remove it), open it, remove the sample bottle, and push the inner bag back down inside the outer bag. The dirty-hands person will close the outer bag. See Consideration 1 below.
- 2. While the clean-hands person holds the bottle, the dirty-hands person will use pliers to loosen the cap on the bottle. See Consideration 2 below.
- 3. The clean-hands person will remove the cap and, while filling the bottle, will hold the cap in the upright position so that particles cannot land inside the cap. If the cap must be set down, lay down a clean vinyl glove and place the cap upright on it.
- 4. While the clean-hands person is filling the sample bottle, the dirty-hands person will keep the outer bag closed and prepare the sample tag. When the sample bottle is full, the clean-hands person will replace the cap and the dirty-hands person will use pliers to do the final tightening of the cap.
- 5. The dirty-hands person will open the outer bag, and the clean-hands person will reach inside, lift the inner bag, place the bottle in the inner bag, and

seal it. Before lowering the bagged bottle into the outer bag, the dirtyhands person will place the sample tag on the inner bag. The clean-hands person will then lower the inner bag into the outer bag, and the dirty-hands person will close the outer bag and place the bagged bottle in the ice chest.

## ADDITIONAL CONSIDERATIONS

- 1. Sample bottles for mercury are made of FEP Teflon<sup>®</sup> and are doublebagged in Ziploc<sup>®</sup> bags. Because labels do not stick to Teflon<sup>®</sup>, each sample bottle has a unique identifier etched on the outside. It is assumed that the exterior of the outer bag is contaminated and that its contents are clean. Therefore, only the dirty-hands person is allowed to handle the outer bag, and only the clean-hands person its contents.
- 2. It is common for the caps on Teflon<sup>®</sup> bottles not to seal well, and they will frequently leak if tightened only by hand. Procedures used at the University of Wisconsin Trout Lake Station call for using pliers to tighten the cap. However, the pliers can be another source of metal contamination, and, if possible, should be avoided. If pliers are necessary, put them inside a couple of pairs of gloves to keep metal from contacting the bottle. Avoid tearing the gloves. For larger bottles, the cap diameter is large enough to get a good grip on it, and tightening can be done by hand. For sample bottles with small caps, pliers should be used.
- 3. Rain has elevated concentrations of mercury relative to most surface water samples and may contaminate a sample. Sampling in the rain should be avoided. If it is necessary to sample in the rain, be careful that rain that contacts surfaces does not get into the sample bottle (e.g., rain that contacts the suit does not splash into or run down your sleeve into the bottle).
- 4. If there is any question as to whether gloves are clean, change gloves.

#### ATTACHMENT 3

## ABB-ES' STANDARD OPERATING PROCEDURE NO. FGCP00202

### ABB ENVIRONMENTAL SERVICES

G:\T92\KENT\_TAY\PHASEI\WORKPLAN.DOC

07198.17

.

.

#### STANDARD OPERATING PROCEDURE

Author: Advanced Technology Team Reviewed by: Issuing Unit: ABB-ES SOP No.: FGCPT00202 Date: 04/14/93

m Jumes 1 SENIOR CHEMIST

(Name & Title) (Functional Area)

Approved by: Tennel (Name & Title)

(Functional Area)

TITLE: Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

PURPOSE: To establish standard operating protocols for the analysis of Volatile Organic Compounds by Purge and Trap Field Gas Chromatography (GC).

SCOPE: This procedure describes the preparation and analysis of soil and water samples for Volatile Organic Compounds using Purge and Trap concentration followed by GC analysis.

**REQUIREMENTS:** Requirements for the method include either a fully operational Purge and Trap/GC system or the ability to assemble one shipped from the vendor. The SOP also requires that the operator understand basic GC operation (e.g., troubleshooting, data handling systems, interferences, etc.) and have familiarity with the operation of Purge and Trap concentration devices.

The critical components needed for the assembly and operation of the Purge and Trap/GC system are listed. Additional items such as syringes, glassware, ventilation hoods, etc. are not included as they pertain to general GC operation and set-up.

**Purge & Trap Device.** A Tekmar LSC 2000 or equivalent is recommended for most applications. The Purge and Trap (P&T) is a concentration device for volatiles in both soil and water. A sparger vessel, a removable glass tube, is used to hold either the water or soil sample. A polymer trap is used to trap the volatiles as they are purged from the sample. After the volatiles are purged from the sample, the trap is flash heated and the volatiles are transferred to the GC for separation, identification, and quantitation.

Purge & Trap Run Conditions. Run conditions may be altered in project specific work plans, however, the following settings are recommended for most operations.

- o Polymer trap: Supelco (# 3) or as required by the project.
- o Purge pressure: 30 psi  $\pm$  5 psi.
- Purge: 6 minutes at 30 mL/minute  $\pm$  5 mL/minute with a temperature of < 40°C.
- o Desorb: 3 minutes at 220°C.
- o Bake: 5 minutes at 225°C.
- o Mount temperature: off.
- o Line and valve temperatures: 180°C.
- o Bake gas delay: on for 2 minutes.

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 2 of 17

Gas Chromatograph. A wide range of temperature programmable gas chromatographs may be used for this method. For most field operations a Hewlett Packard 5890 Series II (or equivalent) temperature programmable GC is recommended. The instrument is durable, easy to assemble, and uses keypad input for operation.

GC Run Conditions. Run conditions may be altered in project specific work plans, however, the following settings are recommended for most operations (e.g., fuels: benzene, toluene, ethylbenzene, xylenes; solvents: VC, DCE, TCE, PCE, etc.)

- Carrier gas: Helium at 99.999% purity with a flow of 10 mL/minute ± 2 mL/minute.
- o Injector temperature: 225°C.
- o Initial temperature: 40°C.
- o Rate: 8°C/minute.
- o Final temperature: 180°C.
- o Final time: 0 to 15 minutes depending on the retention time of the final peak.

Columns. A number of columns are available for volatile analysis, however, unless otherwise stated in a project specific work plan a DB-624 0.53mm ID megabore column or equivalent will be used. The column may be 30, 60, 75, or 120 meters in length as long as the information is recorded.

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 3 of 17

**Detectors.** A variety range of detectors are available for GC operation and specific compound detection. The appropriate selection of detectors is project specific and depends on the project objectives and Data Quality Objectives (DQO's) of the program.

#### Electron Capture Detector (ECD):

- o Make-up Gas: Helium at 99.999% purity with a flow of 35 mL/minute ± 5 mL/minute.
- o Ionization Gas: Ultra-pure P-5 (95% Argon, 5% Methane) or nitrogen with a flow of 35 mL/minute ± 5 mL/minute.
- o Detector Temperature: 325°C.
- o Special Requirements: Set GC switch to either the argon-methane or nitrogen mode.

Flame Ionization Detector (FID):

- Make-up Gas: Helium at 99,999% purity with a flow of 35 mL/minute ± 5 mL/minute.
- o Combustion Gases: hydrogen at 99.999% purity with a flow of 25 mL/minute  $\pm$  5 mL/minute and 0.1 grade air with a flow of 400  $\pm$  150 mL/minute.
- o Detector Temperature: 275°C.

#### ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES

- SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography
- Revision 2, 4/14/93 Page 4 of 17

Electrolytic Conductivity Detector (ELCD):

=

- Make-up Gas: Helium at 99.999% purity with a flow of 35 mL/minute ± 5 mL/minute.
- Combustion Gases: Hydrogen at 99.999% purity with a flow of 100 mL/minute ± 20 mL/minute.
- o Reactor Temperature:  $900^{\circ}C \pm 25^{\circ}C$ .
- o Solvent Flow: n-Propanol flow at  $35\% \pm 5\%$ , this is equivalent to 3.5 turns  $\pm 0.5$  turns.

#### Photoionization Detector (PID):

- o Make-up Gas: Helium at 99,999% purity with a flow of 35 mL/minute ± 5 mL/minute.
- o Detector Temperature: 275°C.
- Special Requirements: The operating range of the lamp is a power setting between 1 and 10 with normal operation at 5.

ABB Environmental Services, Inc.

## STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 5 of 17

**PROCEDURE:** The procedure can be broken down into nine categories: conventions. calibration, blanks, sample preparation, analysis, identification, quantification, quality control, and reporting.

Conventions. Conventions are established to ensure that documentation and analysis are performed in a consistent manner from operator to operator and project to project. Conventions are established for coding standards, recording logbook entries, performing calculations, and analyzing quality control samples.

Stock Standards. Stock chemical standards will be purchased from Supelco. Inc.. Chem Service, Inc., or an equivalent supplier. All standard information will be logged in a bound logbook with the pages sequentially numbered. This information will include, at a minimum:

- o Vendor name supplying standards.
- o Name and concentration of the standard (this can be done by attaching the label from the standard to the logbook).
- o Lot number of the standard.
- o Expiration date of the standard, if listed.

<u>Working Standards</u>. Working chemical standards may either be purchased through an approved vendor or prepared from stock standards. Working standards will be made by diluting neat or prepared stock standards. For standards made from neat solutions, the compound density or weight will be used in calculating the appropriate amount of compound and solvent to be combined. All working standards will be labeled with an identification code (see below), compound or mix name, and concentration. The associated logbook will contain the following information at a minimum:

- o The Stock solution used to make the working standard, including either the information required for logging the stock solution or a reference to where that information can be found.
- o **Both** a written description and calculation of how the working standard was prepared.

ABB Environmental Services, Inc.

#### STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 6 of 17

- o The solvent(s) and associated lot number(s) used in preparing the standards.
- o The final concentration of the working standard.
- o The GC code associated with the standard.
- o Standard and stock solutions will be kept refrigerated at less than 10°C when not in use.
- o The surrogate standard is made separately from the working standards.

<u>GC coding system</u>. The GC coding system is used to trace all standards back to the vendor. All standard are required to have the appropriate code assigned when they are prepared and must be labeled accordingly. The GC code will follow the format:

- o FGCXXXXXXWWYZPPPP Where:
- XXXXXX is the month, day, and year that the standard was made, i.e.,
  041693 = April 16,1993.
- o WW is the page in the logbook where the standard can be found, i.e., 01 to 99.
- o Y is where the standard fell chronologically on the page, i.e., A through Z.
- o Z is the logbook number where the standard can be found, i.e., 1 through 9.
- o **PPPP** is the project identifier, i.e., ABB = Asea Brown Boveri, a short character identification of the project from 1 to 4 characters in length.

### STANDARD OPERATING PROCEDURES

ABB Environmental Services, Inc.

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 7 of 17

<u>Logbook entries</u>. While the style and specific requirements for logbook entries will vary between operators and projects, certain information is required for all projects. At a minimum this will include:

- o A table of contents listing what and where specific information is located.
- A listing of P&T and GC run conditions and set points
- o A chemical standards preparation section containing the required standards information.
- O A run log section containing at a minimum: sample identification, run number or computer file identification, sample amount (weight or volume), spiked surrogate amount and % recovery, standard code for each standard used, amount of standard(s) used, an example calculation for any calculations performed, dilution factor, and a remarks column with any pertinent information (e.g., unusual sample color or odor, unusable blanks, failed standards, coelutions, reruns, etc.).

 $Surrogate \ recovery = \left(\frac{calc. \ amt. \ surrogate}{amt. \ surrogate \ added}\right) + 100$ 

o An equipment maintenance section containing both routine and equipment malfunction maintenance (e.g., column replacement, electronic parts replacement, lamp replacement or cleaning, etc.)

Calibration. The calibration process for GC involves two steps, the first is compound identification and second is quantitation. Identification is accomplished by matching retention times of peaks. Quantitation for purge and trap analysis will be performed using an external calibration technique. The calibration method will use either a single-level or multilevel external calibration technique depending on the DQOs of the project.

ABB Environmental Services, Inc.

### STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 8 of 17

<u>Identification</u>. Compounds are identified by matching the retention times of peaks. The retention time is defined as the period of time from the injection of a mixture onto a column to the elution of the peak of interest. For proper compound identification, a known standard is analyzed on the GC and the compounds of interest retention times are noted. A three percent (note: the 3% retention time window is normalized to the surrogate to adjust for injection variation) retention time window will be used for compound identification.

<u>Single Level Calibration</u>. The single-level external calibration technique requires that a standard containing a known concentration of selected compounds is analyzed. The retention time for each target compound is identified and used for the identification of sample peaks. The peak areas are then used to compute the response factors for the compounds of interest.

<u>Multilevel Calibration</u>. In the multilevel calibration, a minimum of three levels (concentrations) of standards are analyzed. The standard concentrations are selected to cover the concentration range of contaminants expected. This includes a low concentration standard at or near the practical quantitation limit (PQL). In addition, the upper level standard needs to be within the working range of both the detector and the column.

o Approximate working range, ECD: PQL to 40 ppb.

- o Approximate working range, FID: PQL to 200 ppb.
- o Approximate working range, ELCD: PQL to 40 ppb.
- Approximate working range, PID: PQL to 200 ppb.

ABB Environmental Services, Inc.

#### STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 9 of 17

<u>POL</u>. PQL's are determined on an annual basis or as required by project workplans. This is accomplished by analyzing seven replicate low concentration standards near the expected method detection limit (MDL). From these analysis a standard deviation is calculated and multiplied by 3.14 to establish the MDL. This correlates to a student's t of 99% for (n-1). The PQL's are set above the MDL's to present uniform reporting limits (e.g., the MDL's may be calculated to be 0.23, 0.67, 0.45, etc., with the PQL's set at 1.0). Note: The various detectors are sensitive to different amounts (e.g., ng, pcg,  $\mu$ g, etc.) of specific compounds. The detection limits at different concentrations (e.g.,  $\mu$ g/L, mg/L, etc.) are dependent on the technique used to achieve them (i.e., whether or not a 5 mL or 25 mL purge is used).

<u>Quantitation</u>. When the quantitation requirements are not defined in the project workplans, the operator can choose (based on operator experience, the selected detectors, and the compounds of interest) one of three options for quantitating sample results. These options include using the average response factor, (the response factor (RF) is calculated by dividing the area counts of a compound by the amount of compound in the standard, Ac/amt), linear regression, or point to point calibration.

The average response factor may be used when the relative standard deviation (RSD) of the RF's is less than 30 percent.

$$\Re RSD = \left(\frac{S}{\overline{x}}\right) * 100$$

where: S = standard deviation of response factors x = mean of response factors

When this technique is chosen, the average RF is used for calculating sample results.

Linear regression may be used when a correlation coefficient  $(r^2)$  of 0.95 or greater is achieved.

ABB Environmental Services, Inc.

## STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 10 of 17

A point to point calibration may be used for non-linear calibrations (i.e., the RSD of the average response factor exceeds 30 percent or the  $r^2$  for linear regression exceeds 0.95). Care must be taken so that the response for the upper calibration point is not equal to or less than the response for the previous calibration point (i.e., the curve may flatten or curve back toward the baseline). Note: If this technique is chosen, then all analytes must be calculated using this technique.

Once a calibration technique is chosen it must be used for all compounds associated with each detector for the length of the project. The initial calibration is valid for twenty-four hours beginning when the last calibration standard run is complete.

Independent Check Standard Verification. After the first initial calibration for a project, an independent check standard will be analyzed when required by project DQO's. The check sample will be obtained from a different source than the stock solution and working standards. The check standard is used to verify the accuracy of the working standards. A percent difference of less than 30 percent is considered acceptable to confirm standard accuracy.

<u>Continuing Calibration</u>. Prior to daily sample analysis, a continuing calibration check standard will be analyzed at or near the mid-level of the current calibration curve. The target analytes must have percent differences of less than 30 percent when compared to the initial calibration.

Samples may be analyzed only if no more than one compound per detector, or up to 10 percent of the target compounds, exceed the percent difference criteria of 30 percent. If the above criteria are not met, a second standard is analyzed for the compounds that failed during the initial analysis. If the second standard is unacceptable, a new calibration curve will be prepared. Following analysis of an acceptable continuing calibration standard, samples can be analyzed for a period of 24 hours from the time the standard run was completed.

ABB Environmental Services, Inc.

## STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 11 of 17

Blanks. There are three types of banks associated with purge and trap analysis: low level method blanks, medium level method blanks, and cleaning blanks.

Low Level Method Blanks. For routine analysis, a low level method blank must be analyzed before samples are analyzed. A low level method blank consist of 5 mL of analyte free water that has a surrogate added. Method blanks are acceptable if no target compounds are present above the PQL. Samples should not be analyzed until an acceptable method blank is run demonstrating that the instrument is free of interferences. If an acceptable method blank can not be obtained then all associated results will be flagged with a (B).

<u>Medium Level Method Blanks</u>. A medium level blank will be analyzed when medium level extracts are analyzed (see below). The medium level blank consists of 5.0 mL of reagent water, surrogate, and 100 microliters of methanol. The methanol will be from the same source as that used for the medium level extractions. If an acceptable method blank can not be obtained then all associated results will be flagged with a (B).

<u>Cleaning Blanks</u>. A cleaning blank is 5 mL of reagent water only. Blanks will be analyzed after any high level sample to ensure that carryover is not occurring. A high level sample is defined as having a concentration 5 times higher than the highest calibration point. Cleaning blanks will be analyzed until the analysis of further samples will not be affected by carryover. (based on experience, the concentration in the following samples, and the DQO's of the project).

#### ABB Environmental Services, Inc.

## STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 12 of 17

Sample Preparation. Sample analysis and preparation techniques have been adapted from protocols outlined in SW-846 3rd ed. USEPA Purge and Trap methods 8010, 8015, 8020. and 8240 (USEPA 1986).

<u>Water Samples</u>. Rinse a 5 mL syringe with one volume of sample. Draw the sample into the syringe, invert and remove all air adjusting the final volume to 5.0 mL. Add the appropriate amount of surrogate directly to the sample. Deliver the sample to the sparger and start the run. After the desorb and bake cycle, remove spent sample, rinse the sparger with reagent water prior to analyzing subsequent samples.

If high concentrations of VOC's are suspected (indicated by PID headspace results or odor), samples should be diluted to bring the target compound concentrations within the instrument calibration range. Sample volumes of 1 mL (dilutions of 1 to 5) or more may be measured in the 5 mL syringe and reagent water drawn directly into the syringe for a final volume of 5.0 mL. To dilute sample requiring a volume of less than 1 mL add 5.0 mL reagent water to the syringe. Then remove a volume equal to the volume of sample to be added. Add the sample volume and surrogate to the syringe using an appropriate size syringe.

The results of diluted samples must be multiplied by a dilution factor to correct for volume change. Calculate the dilution factor as follows:

DF = 5 mL/X

where X =sample volume, mL

If the sample results are still outside the calibration range the sample may be further diluted or flagged with an E(E = estimated).

<u>Soil Samples</u>. Weigh  $5 g \pm 0.5 g$  into the sparging vessel and attach to the purge & trap. Add surrogate to 5.0 mL reagent water and add into the sparging vessel and start the purge cycle. After the desorb and bake cycle, remove spent sample, rinse the sparger with reagent water prior to analyzing subsequent samples.

<u>Reagent water</u>. Reagent water or locally purchased distilled or spring water that has been demonstrated to be free of target analytes (i.e, QC analysis) below the PQL's. This is checked daily with method blanks.

ABB Environmental Services, Inc.

### STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 13 of 17

<u>Balance check</u>. The field balance will be checked daily with a standard certified weight and the weight recorded. The balance must be accurate to  $\pm 0.1$  g. If the balance fails this criteria the balance either needs to be serviced or a new balance obtained.

<u>Medium level soil samples</u>. If the concentration of target compounds and/or other hydrocarbons is suspected to be high, soils will be extracted with methanol. Weigh 4  $g \pm 0.4$  g of sample into a VOA vial and add 10 mL  $\pm 0.5$  mL of methanol (alternately, 2.0  $g \pm 0.2$  g. in 5 mL  $\pm 0.2$  mL may be used). Cap the tube and shake for approximately 1 minute until the soil and methanol are completely mixed. Centrifuge if required and withdraw 100  $\mu$ L of supermatant. Add to 5.0 mL of reagent water and surrogate in a 5 mL syringe and proceed with analysis according to the procedure for water.



Where X = the amount of soil extracted (g) Y = the amount of methanol used for extraction (mL)

For highly contaminated soils less than 100  $\mu$ L of extract may be used. If the amount of extract needed to bring the concentration of contaminants within the instrument calibration range is less than 2  $\mu$ L, the extract will be further diluted with methanol.

Medium level blanks are required each day that medium level soil samples are analyzed

ABB Environmental Services, Inc.

## STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 14 of 17

<u>Dry weight</u>. Project specifications may require that VOC concentrations be based on dry weight. Weigh sample drying pan to the nearest 0.1 g. Weigh approximately 10 g of sample. Dry in an oven at 110°C,  $\pm$  5°C for approximately 1 hour. Temperature of the oven will be checked at the beginning of each project and the proper setting set. Calculate the solids content from:

$$\$$$
Solids =  $\left(\frac{A}{B}\right)$ \* 100

where

A = dry weight of sample, g B = wet weight of sample, g

Dry weight calculations are performed on all samples when required.



Analysis. The standards and samples are analyzed as outlined in the previous sections. The routine order of analysis is:

o Calibration standard(s) at the start of the project and a continuing calibration standard for each subsequent day (note: a new calibration may have to be run each day if the continuing calibration fails criteria, see calibration section).

o Independent check to verify calibration at the beginning of the project, if required (note: this standard is only analyzed once during the project).

o Method blank and samples.

o Matrix spike and matrix spike duplicate (MS/MSD) samples if required.

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 15 of 17

Quality Control. The following procedures will be implemented by the field chemist to insure standardization of the operating procedures:

<u>Review</u>. The field chemist will review each chromatogram before analyzing the next sample. The review will include the calculation of surrogate recoveries, comparison of surrogate and target compound retention times to calibration standards, and the evaluation of carryover potential.

<u>Surrogate recoveries</u>. Surrogate recoveries for water samples must be greater than 50 percent and less than 150 percent on at least one detector if dual detector analysis is performed. Water samples with surrogate recoveries of less than 50 percent must be reanalyzed to confirm matrix interferences. Water samples with recoveries greater than 150 percent will be qualified with an (S).

Surrogate recoveries for soil samples must be greater than 30 percent and less than 200 percent on at least one detector if dual detector analysis is performed. Soil surrogate recoveries of less than 30 percent must be reanalyzed to confirm matrix interferences. Soil surrogate recoveries of greater than 200 percent will be qualified with an (S).

Matrix spike. Matrix spike quality control samples will be analyzed when required by project workplans. These are field samples to which target compounds at the midcalibration concentration are added. The percent recovery of target compounds will be recorded.

Matrix spikes and matrix spike duplicates are analyzed after twenty samples or as required by the project workplans. Matrix spikes are prepared as either water or soil samples with a mid-level standard added immediately after the surrogate.

ABB Environmental Services, Inc.

## STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

Revision 2, 4/14/93 Page 16 of 17

Matrix spike and matrix spike duplicate results will be calculated and recorded in the GC logbook. Unless specified in the project workplans there will be no action criteria for matrix spike results, the results will be advisory only.

<u>RPD</u>. The relative percent difference (RPD) for the matrix spike and matrix spike duplicates will be calculated and recorded in the project logbook.



Reporting. Data from all sample analyses and relevant calibration and blank analyses will be documented in the project GC run logbook. A quality control summary as outlined in project specific workplans will be generated at the completion of the project and will include some or all of the following as required: initial calibrations, continuing calibrations, surrogate recoveries, matrix spikes, matrix spike duplicates, method blanks, dilutions, reanalyses, observations of the field chemist, problems and fixes, unknown peaks, raw data, etc..

#### **REFERENCE:**

U.S. Environmental Protection Agency, 1986, "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods; SW-846"; Office of Solid Waste and Emergency Response, Washington, D.C.

STANDARD OPERATING PROCEDURES

SOP FGCPT00202, Purge and Trap Analysis of Volatile Organic Compounds by Field Gas Chromatography

ABB Environmental Services, Inc.

Revision 2, 4/14/93 Page 17 of 17

## ATTACHMENT 4

## ABB-ES' STANDARD OPERATING PROCEDURE NO. FXRFS00102

#### ABB ENVIRONMENTAL SERVICES

#### STANDARD OPERATING PROCEDURE

Author: Advanced Technology Team Reviewed by:

Issuing Unit: ABB-ES SOP No.: FXRFS00102 Date: 05/12/94

CHEMIS SENID? (Name & Title)

(Functional Area)

Approved by: €U) Juge Team (e (Name & Title)

(Functional Area)

TITLE:

Field Analysis of Metals by X-ray Fluorescence

PURPOSE: To establish standard operating protocols for the analysis of metals using the Spectrace 9000 X-ray Fluorescence Analyzer (XRF).

SCOPE: This procedure describes the preparation and analysis of soil samples for metals using XRF.

**REQUIREMENTS:** Requirements for the method include a Spectrace 9000 XRF. The critical components needed for the assembly and operation of the system are listed. Additional items such as gloves, oven, ventilation hoods, etc. are not included as they pertain to general XRF operation.

Instrumentation. Spectrace 9000 Field Portable X-ray Fluorescence Analyzer. The Spectrace is fully self-contained and only requires additional sample cups, mylar, and the independent check standard for operation.

**Run Conditions.** The following settings are used for routine sample analysis. Some parameters will be changed during daily calibration checks. Allow the XRF to warm up for approximately 30 minutes before use.

- o Soils mode
- o Acquisition Times: Cd-109 at 300 seconds Fe-55 at 20 seconds Am-241 at 20 seconds

**PROCEDURE:** The procedure can be broken down into six categories: conventions, calibration, blanks, sample preparation, analysis, and reporting.

**Conventions.** Conventions are established to ensure that documentation and analysis are performed in a consistent manner from operator to operator and project to project. Conventions are established for coding standards, recording logbook entries, performing calculations, and analyzing quality control samples.

<u>Standards</u>. Soil reference standards will be purchased through USGS, NIST, or equivalent. All standard information will be logged in a bound logbook with the pages sequentially numbered. This information will include, at a minimum:

- o Vendor name supplying standards.
- o Name and concentration of the standard

o Lot number of the standard.

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES SOP FXRFS00102, Field Analysis of Metals by X-Ray Fluorescence Revision 2, 5/12/94 Page 2 of 5 Logbook entries. While the style and specific requirements for logbook entries will vary between operators and projects, certain information is required for all projects. At a minimum this will include:

- A table of contents listing what and where specific information is located. 0
- A listing of XRF run conditions and calibration check results. 0
- A run log section containing at a minimum; sample identification, run number, 0 an example calculation for any calculations performed, and a remarks column with any pertinent information (e.g., unusual sample color or odor, unusable blanks, failed standards, reruns, etc.).
- 0 An equipment maintenance section containing both routine and equipment malfunction maintenance.

Calibration. The Spectrace is calibrated at the factory. However, three checks are required to ensure that the instrument is operating properly. These checks are performed daily prior to sample analysis.

Energy Calibration Check. Place the safety shield on the probe. Set the acquisition time for each source to 60 sec (180 sec total). Upon completion save results and examine spectra. For each source locate and record the KeV for the major peaks as listed below (use the x12 horizontal magnification).

Source Peak	Range, KeV
Cd-109 Pb L-alpha Pb L-beta Pb L-gamma Emission	$\begin{array}{r} 10.54 \pm 0.040 \\ 12.61 \pm 0.040 \\ 14.76 \pm 0.040 \\ 22.10 \pm 0.040 \end{array}$
Fe-55 S K-alpha	$231 \pm 0.010$

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES SOP FXRFS00102, Field Analysis of Metals by X-Ray Fluorescence Revision 2, 5/12/94 Page 3 of 5

ABB-ES

	Emission	5.89 <u>+</u> 0.010
Am-241	Pb L-alpha Pb L-beta Emission	$\begin{array}{r} 10.54 \pm 0.050 \\ 12.61 \pm 0.050 \\ 59.50 \pm 0.20 \end{array}$

Perform an energy calibration if any peaks do not meet the requirements and repeat the energy calibration check.

<u>Resolution Check</u>. Set the Cd-109 source acquisition time to 60 sec and the Fe-55 and Am-241 times to 20 sec and measure the pure iron standard supplied with the XRF. Save and examine the Cd-109 spectrum. Record the KeV values measured at the full width at half height. Subtract the left side value from that of the right; the difference must be less than 0.300 KeV. Contact the manufacturer if the resolution is out of specification.

<u>Element X-ray Response Check</u>. The relative intensities of Fe, Mn, and Co are compared using the spectrum obtained in the resolution check (above). From the raw data verify that the Fe intensity is greater than 0.96 and those for Mn and Co are less than 0.003. If the intensities are out of range ensure that the window is clean, reposition the standard and reanalyze.

<u>Target Element Response Check</u>. Standards containing the metals of interest are analyzed using the sample analysis application and acquisition times (see below). The concentrations should be at low, mid, and high levels for some or all of the target elements. The low level standard should be near the instrument detection limit; mid level near the project action level, and the high level near the maximum concentration of the metals on-site.

<u>Continuing Calibration</u>. This is the mid-level standard at or near the action level. This check will be run after every 10 samples.

Blanks. A teflon blank is supplied with the XRF. This must be analyzed under the same application and source acquisition times as in sample analysis (see below). All elements should be at concentrations at or less than 3 times their standard deviation. Reanalyze if any elements are out of range. Check probe window and teflon blank for cleanliness if any are still out and run the Acquire Background program if needed.

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES SOP FXRFS00102, Field Analysis of Metals by X-Ray Fluorescence Revision 2, 5/12/94 Page 4 of 5 Sample Preparation. A representative portion of the sample, approximately 20 to 30 g, is placed in an appropriate container until dry. The sample is then sieved through a 30 to 60 mesh screen and homogenized. Pour enough sample into the sample cup to fill at least halfway; assemble mylar and ring.

Analysis. Lead and copper are both measured using the Cd-109 source. The acquisition times are set at 300 sec for the Cd-109 source and 20 sec for Fe-55 and Am-241.

**Reporting.** Data from all sample analyses and relevant calibration and blank analyses will be documented in the project XRF logbook. A quality control summary as outlined in project specific workplans will be generated at the completion of the project and will include some or all of the following as required: daily calibration checks, continuing calibrations, reanalyses, observations of the field chemist, problems and fixes, etc..

#### **REFERENCE:**

Spectrace 9000 Portable X-ray Fluorescence Analyzer Operating Instructions, Revision 03, January 1992.

ABB Environmental Services, Inc.

STANDARD OPERATING PROCEDURES SOP FXRFS00102, Field Analysis of Metals by X-Ray Fluorescence Revision 2, 5/12/94 Page 5 of 5

### ATTACHMENT5

## MERCURY ANALYZER MANUFACTURER DOCUMENTATION

#### ABB ENVIRONMENTAL SERVICES

.

9



he Leeman Labs' Model PS200 Automated Mercury Analyzer offers exceptional value to any laboratory measuring mercury. Everything is included: a random access PS Series autosampler. an IBM-compatible PC. and Leeman Labs' powerful PS Series software complement an instrument specifically designed for mercury determinations.

The PS200 is an energy-efficient double beam photometer with parts per trillion (ppt) sensitivity. Energy efficiency is achieved by eliminating the monochromator used in AAs. The monochromator serves no function in a dedicated mercury analyzer. In fact, it causes a large energy loss by using narrow slits and by diffracting energy into unused spectral orders.

Since the optical cell need not fit the physical constraints of an AA burner or furnace cavity, it has been optimized for cold vapor determinations. The system has a wide dynamic measuring range, and can be used for determining mercury under USEPA CLP and TCLP protocols. Low ppt sensitivity makes it the ideal companion to our ICP/Echelle Spectrometers.

# Unattended Operation

The PS200 completely automates the analysis of prepared samples, freeing time for other laboratory tasks. Up to 88 samples, 6 calibration standards, and 7 QC check standards and/or spiked samples can all be loaded into the autosampler. Even baseline correction is performed automatically. Large reservoirs hold the rinse, reductant and blank solutions.

Autosampler functions are defined through the PS Series software, so that operation can be customized to your needs. You can even interrupt an autosampler batch to run a "rush" sample. Results are easily transmitted to a LIMS.

# User-Defined Automation

The PS200 software contains a library of routines known as "macros." These routines not only allow you to customize operation of your instrument, they also simplify training, reduce errors, and enhance consistency of results.

Use macros to reduce multiple steps to a single keystroke. Have the system make intelligent decisions and act on them automatically – saving time and increasing sample throughput. For example:

- Standardize start-up and shut-down procedures for consistent operator-to-operator performance.
- Build in as many standards and control samples as required for your QA/QC needs.
- Set your own limits for out-of-spec readings on control/check samples, and have the system automatically recalibrate and repeat sample determinations as needed.
- Calculate matrix spike recoveries and apply those recovery factors to subsequent samples if desired.
- Calculate for the method of standard additions.

b' 5

W hile analytical testing for mercury has become increasingly automated and efficient, sample preparation has remained a tedious and time-consuming procedure. The Leeman Labs AP200 Automated Mercury Sample Preparation System performs all sample preparation steps for mercury determination by cold vapor AA. Sample and standard preparation for the most demanding QA/QC protocol is made simple and easy with the AP200.

Samples and standards are digested, and reagents are precisely added for consistent results. Potassium permanganate additions are made as required under control of a "Complete Oxidation Probe" (COP).

## **Consistent Results**

Leeman Labs AP200 Automated Mercury Preparation System ensures that all samples and standards are treated identically, sample-tosample and operator-to-operator. for consistent, reliable results.

The AP200 is capable of preparing up to 88 samples, 6 calibration standards, and 7 check standards at one time without operator intervention. The samples and standards are placed directly in autosampler cups in a rack that mounts in the enclosed temperature-controlled bath for reagent additions and digestion. Reagent volumes and digestion times are programmed to meet specific sample requirements. COP, a colorimetric detector, ensures that samples receive sufficient potassium permanganate and retain color for a specified time period. This assures complete oxidation of organics in the sample, eliminating their interference with mercury determinations.

## Maximum Productivity

The AP200 significantly improves the quality of results and reduces man-hours required for mercury sample preparation. When combined with the PS200 Automated Mercury Analyzer, you have the complete "walk-away" answer for mercury determinations.

# Cost Savings

The AP200 not only saves time and labor, it saves operating costs as well. Only 8 mL of sample are required, along with correspondingly reduced amounts of reagents, so reagent costs are 10 times lower than conventional methods. This also means 10 times less hazardous waste for disposal. Also, because of the high degree of reliability associated with automation, fewer samples will need to be reworked or reanalyzed due to errors in preparation.



#### **Principles of Operation**

The AP200 uses established methodology for the preparation of mercury samples for cold vapor AA. The required 8 mL of sample are placed directly in autosampler cups. (After preparation the racks transfer directly to the autosampler of the PS200 Automated Mercury Analyzer.) Sulfuric acid, nitric acid, and potassium permanganate are precisely added, mixed, and sparged.

The Complete Oxidation Probe examines each cup to determine if the purple color persists for the specified time period. Additional potassium permanganate is added if necessary. Should the purple color continue to disappear, the sample is flagged for operator attention. Potassium persulfate is then added, and the samples are digested for 2 hours at 95° C. After cooling, sodium chloride-hydroxylamine sulfate is added to reduce the excess permanganate. The samples and standards are then ready for Hg determination by cold vapor AA, which commences with the addition of stannous chloride.





# Early Warning of High Mercury Levels

One of the problems in mercury analysis occurs when a sample with high mercury content is measured. Adsorption of these high vapor levels results in slow elution from internal surfaces, characterized by long baseline return times. Once a system has been exposed. it can take hours to clean out, during which time the instrument is unavailable for use.

- The PS200 detects high mercury levels early before the vapor has a chance to saturate the system. When the detector senses a rapid rate of signal increase, the system automatically shuts off the flow of carrier gas (argon or nitrogen) to prevent more mercury vapor from flowing into the system. Immediately, tinse solution starts flowing through the liquid sample path to wash out the sample. Throughout this process, the system monitors the detector signal, and waits until it has sufficiently decreased before running the next sample.
  - The entire procedure typically takes 5 minutes saving you hours of system flushing, and minimizing downtime.

## Analytical Reliability

The PS200 Automated Mercury Analyzer is designed and supported to give you maximum confidence in your results. Leeman Labs offers everything you need for accurate, reliable mercury determinations. including:

- A completely automated system at significantly less cost than an automated cold vapor AA:
- Double beam measurement for accuracy and sensitivity.
- PLASMA-PURE calibration standards.
- Optional pre-shipment testing to specific protocols, including CLP and TCLP.
- Regionally located Sales/Support Centers for quick response.
# ATTACHMENT 6

# BIOAVAILABILITY REFERENCE

## ABB ENVIRONMENTAL SERVICES

# Development of an *in Vitro* Screening Test To Evaluate the *in Vivo* Bioaccessibility of Ingested Mine-Waste Lead

Michael V. Ruby, <sup>\*,†</sup> Andy Davis,<sup>†</sup> Timothy E. Link,<sup>†</sup> Rosalind Schoof,<sup>‡</sup> Rufus L. Chaney,<sup>§</sup> Gary B. Freeman,<sup>†</sup> and Paul Bergstrom<sup>o</sup>

PTI Environmental Services, 2995 Baseline Road, Suite 202, Boulder, Colorado 80303, PTI Environmental Services, 15375 SE 30th Place, Suite 250, Bellevue, Washington 98007, Soil Microbial Systems Laboratory, USDA Agricultural Research Service, Building 318, BARC West, Beltsville, Maryland 20705, Battelle Columbus Laboratories, 505 King Avenue, Columbus, Ohio 43201, and ARCO, 555 Seventeenth Street, Denver, Colorado 80202

A screening-level in vitro test was developed to evaluate the relative solubility of ingested lead (Pb) from different mine wastes in the gastrointestinal (GI) tract. The in vitro method, modeled after assay methods for available iron from food, used a laboratory digestion procedure designed to reproduce GI tract chemistry and function. The *in vitro* method was independently calibrated against a rabbit feeding study, demonstrating that only 1-6% of the total Pb in four mine-waste samples with disparate Pb mineralogy was bioaccessible. In vitro method development tests indicated that H<sup>+</sup> concentration and Cl<sup>-</sup> complexation control dissolution of Pb-bearing minerals in the stomach and that both GI tract enzymes and organic acids are necessary to maintain Pb in the soluble form on entering the small intestine. The experimental results indicate that ingestion of Pb-bearing mine wastes results in limited Pb solubility and that the *in vitro* test provides a screening-level estimate of the maximum available Pb from mine wastes.

#### Introduction

HAR BEEN

REAL PROPERTY OF LOUGH STREET

When assessing risks associated with lead (Pb)-contaminated soils, one exposure pathway typically evaluated is soil ingestion by children. Standard procedures recommended by the U.S. Environmental Protection Agency (EPA) for estimating soil Pb exposures using the Uptake Biokinetic Model assume that a typical child will ingest 100 mg of soil/day and that 30% of the ingested Pb will be absorbed into the systemic circulation (i.e., will be bioavailable) (1). However, recently completed animal studies indicate that lead from mine waste is much less bioavailable than lead from other sources (2-4). Our previous research has demonstrated that Pb-bearing minerals dissolve slowly and incompletely in the gastrointestinal (GI) tract, due to a variety of geochemical factors (5) resulting in Pb forms that are bioaccessible. The limited dissolution of Pb-bearing minerals in the GI tract is most likely responsible for the low bioavailability observed in the animal studies, because Pb absorption has been shown to occur from the soluble phase in the small intestine (6, 7).

The geochemical factors controlling dissolution of metals from mine waste include the mineral composition, the degree of encapsulation, and the presence of alteration rinds. Because the solubility of Pb from mine waste is

Table I.	Occurrence	of Pb	Minerals	in	Mine-Waste
Samples					

	sample no.			
Pb phase	BMW-1	BMW-2	BMW-3	BMW-4
anglesite (PbSO4) (%)	59	4	10	
galena (PbS) (%)	26		1	5
lead phosphates <sup>a</sup> (%)	4	7	37	39
manganese-lead oxides <sup>b</sup> (%)		31	7	26
iron-lead oxides <sup>c</sup> (%)	2	29	6	
iron-lead sulfates <sup>d</sup> (%)	2	24	39	
lead oxides [PbO and Pb(OH) <sub>2</sub> ] (%)	1	3		17
lead metallics <sup>e</sup> (%)				2
lead silicates (PbSiO <sub>3</sub> and PbSiO <sub>4</sub> ) (%)		2		11
lead barite [(Ba-Pb)SO <sub>4</sub> ] (%)	3			
cerusaite (PbCO <sub>3</sub> ) (%)	3			
Pb concn (mg/kg)	3900	1030	5820	17 <b>90</b>
As concn (mg/kg)s	1380	620	1180	420
Fe concn (%)#	6. <del>9</del>	12.3	5.0	5.4
Mn concn (mg/kg)	2200	2000	1100	1200
Zn concn (mg/kg)4	6040	NA	NA	1600
Ca concn (%)#	0.7	2.9	1.2	0.8
pH <sup>h</sup>	3.7	2.8	3.8	2.6

<sup>a</sup> Lead phosphate grains contain variable compositions of Pb, PO<sub>4</sub>, SO<sub>4</sub>, and halogens. <sup>b</sup> Manganese-lead oxide particles contain more Mn than Pb, with variable compositions of constituents. <sup>c</sup> Iron-lead oxide particles contain more Fe than Pb, with variable compositions of constituents. <sup>d</sup> Iron-lead sulfate grains contain more Fe than Pb, with variable compositions of constituents. <sup>c</sup> Lead metallics contain elemental Pb with Fe, Mn, As, Cu, and Zn in varying proportions. <sup>f</sup> Determined by digestion (31) and atomic absorption spectroscopy (31). <sup>d</sup> Determined by X-ray fluorescence. <sup>h</sup> Determined by the saturated paste method (32). <sup>i</sup> NA, not analyzed.

constrained by the composition of the mineral assemblage, site-specific mineralogical data are necessary to characterize the availability of Pb (2, 5). Two dominant Pb mineral assemblages-sulfide/sulfate and oxide/phosphate—have been identified in residential soils containing mine waste (8). The sulfide/sulfate assemblage consists of galena (PbS) grains altering to the oxidation product anglesite (PbSO<sub>4</sub>) or enclosed in pyrite (FeS<sub>2</sub>) or silicate  $(SiO_2)$  matrices, making them unavailable for alteration or dissolution. Precipitation of jarosite  $[KFe_3(SO_4)_2(OH_6)]$ rinds on Pb-bearing particles was also observed and would retard Pb dissolution, both physically by reduction in the exposed surface area and chemically due to the insoluble nature of jarosite in acidic (pH < 4) media (9). The oxide/ phosphate assemblage consists of lead phosphates of varying compositions, lead ferromanganese oxides, ironlead sulfates that are similar in composition to the mineral lead jarosite  $[PbFe_6(SO_4)_4(OH)_{12}]$ , and lead oxides [PbOand Pb(OH)<sub>2</sub>, Table I]. These Pb minerals occur in mine waste and soils as a complex set of alteration products

and a survey of the factor of the same

Author to whom correspondence should be addressed.

<sup>&</sup>lt;sup>†</sup> PTI Environmental Services, Boulder.

<sup>&</sup>lt;sup>1</sup> PTI Environmental Services, Bellevue.

USDA Agricultural Research Service.

<sup>&</sup>lt;sup>I</sup> Battelle Columbus Laboratories.

<sup>●</sup> ARCO.

that are weathering toward more stable lead phosphates. The geochemical controls on Pb dissolution from one of the mine-waste samples (BMW-1) used in this study have been discussed in detail in previous papers (5, 10).

The mass of Pb dissolved is also controlled by physiological factors, including gastric pH, the rate of Pb dissolution relative to the residence time of Pb-bearing mine waste in the GI tract, and by *in vivo* sorption and precipitation reactions that may limit dissolved Pb concentrations. Gastric pH, which in humans ranges from a basal level of 1-2 to 4-6 after ingestion of food, with its attendant buffering capacity (11, 12), appears to be one of the most important physiological factors in determining the mass of Pb solubilized from a mine waste. Lead that

enters the fluid phase in a fasting stomach (assumed pH
= 1.3 and high chloride activity) will be present at approximately equimolar concentrations of Pb<sup>2+</sup> and PbCl<sup>+</sup> (13). Another critical factor is the residence time in the stomach. The gastric contents are emptied completely into the small intestine within approximately 2 h in humans when fed various test meals (14).

On entering the duodenum, NaHCO<sub>3</sub> excreted with the pancreatic juices is mixed with the intestinal chyme, resulting in an increase in pH to approximately 7 in humans (15). Consequently, dissolved Pb concentrations are likely to decrease in the small intestine due to adsorption to mineral and food-particle surfaces with increasing pH and as a result of precipitation reactions (16). For the purpose of this discussion, the fluid or soluble phase within the small intestine is operationally defined as the fraction that can be separated from the solid fraction by relatively low-speed centrifugation, as defined in the methods section. Lead in the soluble small intestinal phase will be distributed between fractions bound by proteins and enzymes

- (17, 18); complexed by amino acids (19, 20), low molecular weight carboxylic acids (21), and tannins and humic acids released from ingested soil and food, respectively (16); adsorbed to suspended particulates in the fluid phase; or
- present as free lead cations and hydroxide complexes. The distribution of Pb between these forms will control the
  solubility and, consequently, the mass available for absorption across the intestinal epithelium.

Because of the complexity of the geochemical and physiological factors controlling dissolution of lead from mine waste, it was determined that a rapid, inexpensive in vitro method was needed to investigate interactions of the various factors and to compare the relative solubility

- of Pb from solid sources. The *in vitro* method for estimating Pb availability presented in this paper was modeled after an *in vitro* assay method for available Fe in foodstuffs (22-26). The Fe availability method sim-
- ulates the stomach and small intestinal phases of digestion using solutions of specific pH that contain digestive enzymes (pepsin in the stomach, pancreatic enzymes and bile acids in the small intestine) mixed with the test
- substrate to reproduce GI tract function and chemistry. The pH values of 1.3 and 7.0, selected for the gastric

and small intestinal incubations, respectively, were based
on measurements in fasting rabbit stomachs and small intestines. The strongly acidic stomach solution selected for the *in vitro* method is representative of a fasting child and maximizes Pb dissolution to provide an upper-bound

 estimate of available Pb that would apply to ingestion of small particles due to mouthing behavior by children several hours after food ingestion or under fasting conditions. Selection of appropriate concentrations for digestive enzymes is problematic because concentrations are highly variable in the human system. The enzyme concentrations selected for the *in vitro* test were those used in refs 23 and 24 for *in vitro* estimation of Fe from food. The method developed herein included the addition of small quantities of organic acids that were determined to be present in the rabbit GI tract.

Gastric mixing rate (the rate at which ingested material is mixed with fluid in the stomach due to peristalsis) will also affect mineral dissolution kinetics and influence the concentrations of Pb solubilized in the stomach. Therefore, mixing of the *in vitro* flask contents employed a mixing rate designed to replicate *in vivo* mixing, as previously determined in Pb dissolution rate experiments (9). The test material mass (4 g) and fluid volume (40 mL) for the *in vitro* experiments were based on those found in the stomachs and small intestines of 33 New Zealand White rabbits. The rabbits had been dosed with 2.0 g of mine waste/kg of body weight and weighed, on average, 2.1 kg, resulting in 4.2 g of ingested mine waste (16.4 mg of Pb).

To test the validity of the in vitro model, an in vivo experiment was conducted to assess the dissolution of Pb from mine waste during passage through the GI tract of New Zealand White rabbits. A mine-waste dose of 2.0 g/kg of body weight was selected to represent the worstcase scenario of a child with pica-for-soil (e.g., a child who intentionally ingests soil), who may ingest up to 10 g of soil/day (27, 28). Although differences in digestive anatomy and function between humans and rabbits, such as coprophagy, biliary excretion, and development of Pb absorption mechanisms during growth, must be considered when assessing Pb bioavailability, the primary factors controlling Pb dissolution (i.e., pH, mixing, and transit time) are comparable for humans and rabbits (5, 11, 12, 14, 29). Therefore, the rabbit GI tract provides an appropriate model to assess Pb bioaccessibility from a mine waste on ingestion by a child. Values for several important parameters in the in vitro model were therefore based on observed values in rabbits, as well as humans.

#### Methods

Mine-Waste Collection and Characterization. The mine waste used for both the *in vivo* experiment and lnvitro method development (BMW-1) was blended using five mine-waste samples (waste rock and mine overburdøn material) collected from one mining site to achieve a desired Pb content (3900 mg/kg of Pb). Before blending, each ofthe five mine-waste samples was air-dried and sieved 10  $<250 \ \mu m$  using a mechanical sieve shaker, because predominantly smaller particles adhere to children's hands and may be ingested (30). The remaining samples (BMW-2-4) were composite samples collected from individual mine-waste piles and treated in an identical manner  $\omega$ BMW-1. Bulk Pb concentrations were determined for each sample by digestion in HNO<sub>3</sub>/H<sub>2</sub>O<sub>2</sub>, followed by atomic absorption spectroscopy (31). Bulk arsenic (Aa), Fe, manganese (Mn), zinc (Zn), and calcium (Ca) concentrations were determined by X-ray fluorescence. Sample pH was determined by the saturated paste method

(32). Lead minerals in the mine-waste samples were identified by electron microprobe (JEOL 8600) at the Department of Geological Sciences, University of Colorado at Boulder (Table I). A description of microprobe methods, and more detailed descriptions of the Pb mineralogy, may be found in refs 5 and 8.

In Vivo Experiment. Thirty-six female New Zealand White rabbits, weighing an average of 2.1 kg at 12 weeks, were used in the in vivo study undertaken at Battelle Columbus Laboratories (Columbus, OH) and performed in compliance with Good Laboratory Practice Regulations (33). Animals were caged in individual stainless steel cages with mesh bottoms. To avoid interactions between the rabbit chow and the mine waste, all of the rabbits were fasted for 16 h prior to dosing and 4 h after. During nonfasting periods, Purina high-fiber rabbit chow was provided ad libitum. Deionized (DI) water was available at all times. Twenty-seven rabbits were dosed with  $2.0 \pm$ 0.02 g of BMW-1 (in gelatin capsules)/kg of body weight, and three rabbits each were killed at 0.5, 1, 1.5, 3, 6, 8, 16, 24, and 36 h after dosing. One group of three rabbits was given a mass of soluble Pb salts equal to that given to the mine-waste-dosed animals (14.3 mg of lead acetate [Pb-(CH<sub>3</sub>CO<sub>2</sub>)<sub>2</sub>·3H<sub>2</sub>O]/kg of body weight) and killed 1 h after dosing. Two groups of three rabbits each served as controls and were killed at 1 and 36 h after dosing.

Whole blood was collected from all rabbits via cardiac puncture after sedation with sodium pentobarbital. Subsequently, a lethal injection of pentobarbital was administered, the stomach and intestines were exposed, and the region below the pyloric valve was clamped to prevent passage of the stomach contents into the duodenum. Measurements of pH were performed using an Orion portable pH meter in conjunction with a Ross semimicro pH electrode. The electrode was inserted through an incision in the stomach wall for measurement of pH in the fundic and pyloric regions of the stomach. These measurements were also taken in the duodenum and ileum by removing the small intestine and gently inserting the probe into either end of the segment. Large intestine measurements were made in the middle of the cecum.

Samples of the contents of the stomach, small intestine. and large intestine were collected from each animal in 50-mL Corning polypropylene centrifuge tubes and centrifuged at approximately 2100g for 25 min in a Beckman Model TJ6 centrifuge. Approximately 30-40 mL of stomach material was collected with a spoon through a slit in the stomach wall, and any remaining material was removed from the stomach with a spoon, collected into a disposable beaker, and weighed. The volume of material in the fasted rabbit stomachs was approximately 30 mL, while the volume in fed rabbit stomachs was approximately 60 mL. Entire small intestinal contents (10-20 mL) were extruded into single centrifuge tubes. Approximately 40 mL of large intestinal material was collected with a spoon through a slit in the middle portion of the organ, and the remaining material was collected with a spoon into a disposable beaker and weighed. The fluid phase was decanted, and reagent-grade HNO<sub>3</sub> (17N) was used to acidify the fluid samples (1% v/v). Solid samples were dried to determine percent moisture and digested in HNO<sub>3</sub>/ H<sub>2</sub>O<sub>2</sub> (31). A Thermo Jarrel Ash, Video 11E atomic absorption spectrophotometer, equipped with a Model CTF 188 graphite furnace and Smith-Hieftje background correction, was used for analysis of samples with < 2 mg/L

王 建金油

of Pb, and the same instrument, equipped for flame analysis, was used for samples with  $\geq 2 \text{ mg/L}$  of Pb. Organic acid analyses were performed by capillary electrophoresis at Galbraith Labs (Knoxville, TN).

Analytical quality assurance and quality control (QA/QC) samples consisted of 1 in 20 samples analyzed as matrix spikes and duplicates, calibration verification, continuing calibration verification and blanks, and the use of the method of standard additions when recalcitrant matrices were encountered. Based on quality control limits for acceptable analytical results (34), all data were accurate and precise.

In Vitro Experiments. A mine-waste/solution ratio of 1:10 (4.0  $\pm$  0.01 g of mine waste and 40 mL of fluid) was selected for the *in vitro* method because each rabbit was dosed with 2 g of BMW-1/kg of body weight (average of 4.2 g of BMW-1/rabbit), and the 1:10 solid-to-fluid ratio was the average ratio observed in the 36 rabbit stomachs and small intestines. The *in vivo* experiment was designed to minimize interactions between the mine waste and rabbit chow; however, because rabbits practice coprophagy (35), some interaction between mine waste and food is inevitable. Therefore, in vitro tests were conducted with both mine waste alone (in triplicate) and a mine-waste/ rabbit chow mixture (4.0 g of BMW-1, 1.0 g of rabbit chow, 40 mL of fluid), in duplicate. The in vitro test was also conducted with an equivalent mass of Pb as lead acetate, in both the presence and absence of Pb-free mine waste, to allow calculation of Pb solubility from BMW-1 relative to lead acetate. Subsequent to method development using BMW-1, three additional mine-waste samples (BMW-2-4, Table I) were subjected to the in vitro test to examine Pb availability from mine waste with disparate Pb mineralogies.

A total of 40 mL of type I DI water was adjusted to pH 1.3 with HCl. Mine waste (4 g), pepsin (50 mg, activity of 800-2500 units/mg, Sigma Chemical Co.), and organic acids (acetate [0.5 mL], citrate [0.5 g], lactate [0.42 mL], and malate [0.5 g]) were added, and the flask was placed on a wrist-action shaker in a water bath at 37 °C, representative of GI tract temperature (23). The pH was checked at 10-min intervals during the first half hour and at half-hour intervals thereafter, and measured volumes of 12N HCl were added to maintain a pH of 1.3. Aliquots (2 mL) were removed from the reaction flask at 0.5, 1.0, 1.5, and 2.0 h after initiation of the reaction and centrifuged immediately at approximately 2100g for 25 min, and the liquid fraction was decanted for Pb analysis. After 2 h, the reaction vessel was titrated to pH 7.0  $\pm$  0.2 by the addition of a dialysis bag (8000 MWCO, Spectra/Por cellulose ester tubing) containing 3 mL of DI and NaHCO<sub>3</sub> of equivalent molarity to the calculated acidity (HCl) in the flask. The system required approximately 30 min to reach a pH of 7.0  $\pm$  0.2, after which the dialysis bag was removed and the contents were emptied into the reaction flask. Pancreatin (20 mg, activity equivalent to  $4 \times U.S$ . Pharmacopeia specifications, Sigma Chemical Co.) and bile extract (70 mg) were added, and the flask was returned to the water bath. Two hours were allowed for the flask contents to reach equilibrium, the fluid volume was measured, and a 10-mL fluid sample was collected by centrifugation for Pb analysis. Fluids were analyzed for Pb using the methods and instrumentation described in the in vivo experimental section. GI tract enzymes and



**Figure 1.** pH variations in rabbit GI tract as a function of time and location: ( $\blacksquare$ ) fundle stomach, ( $\square$ ) pyloric stomach, ( $\blacktriangle$ ) duodenum, (+) ileum, (×) cecum. Error bars indicate standard deviation for the measured values in three rabbits.

acids were obtained from Sigma Chemical Co. (St. Louis, MO).

- Method development testing for the *in vitro* test also included omission of the organic acids, omission of the GI tract enzymes, and omission of both the acids and enzymes (each test in duplicate). A parallel procedure was carried
- out (in duplicate) with an equimolar amount of a soluble lead acetate [Pb(CH<sub>3</sub>CO<sub>2</sub>)<sub>2</sub>·3H<sub>2</sub>O, 24.5 mg] to evaluate Pb availability from a soluble salt with no matrix present. Finally, the *in vitro* test was conducted with lead acetate (in duplicate) in reaction flasks containing 4 g of low-Pb mine waste (100 mg/kg of Pb). This mine waste was also subjected to the *in vitro* procedure (in duplicate) to determine available Pb. An experimental blank (no solid material in flask) was carried through the procedure during each experiment.
- Results

Mineralogy, Metal Concentrations, and Sample pH. Lead mineralogy in samples BMW-1-4 (Table I) was calculated as percent occurrence of Pb-bearing grains and indicates a diverse Pb mineralogy consisting of both sulfate/sulfide and oxide/phosphate phases, consistent with Pb mineralogy observed during a comprehensive study at this site (8). Bulk metals data are presented (Table I) for comparison of these mine-waste samples with samples that may be analyzed by the in vitro method in the future. Although the bulk concentrations of As, Fe, Mn, Zn, and Ca will have a minimal effect on the dissolution of Pb-bearing phases, the presence of soluble Fe and Ca will decrease the absorption of Pb across the intestinal epithelium (16). The acidic pH values determined in these samples (2.6-3.8) are consistent with the high Fe concentrations (5.0-12.3 wt %) and the presence of pyrite and suggest that these materials will have a limited capacity to buffer stomach acids on ingestion.

In Vivo pH. The pH measurements collected in the rabbit GI tracts (36 rabbits) indicate that a pH in the range of 1.0-1.3 exists in the stomach under fasting conditions, increasing to 2.8 and 4.1, respectively, in the fundic and pyloric regions of the stomach on ingestion of food (Figure 1). Small and large intestinal pH values were independent of nutritional status and averaged 6.9 in the duodenum, 7.4 in the ileum, and 5.9 in the cecum. These

# Table II. Concentrations of Organic Acids in Rabbit GI Tract

organic acid	pK.	lead-acid log K <sub>ML</sub> ,• ionic strength 1.0	stomach concnª (mg/L)	small intestine <sup>a</sup> (mg/L)
acetate	4.70	2.1	189	212
butyrate	4.94	2.1	<10	<10
citrate	pK <sub>1</sub> 3.1 <sup>b</sup>	4.1	534	<10
	pK <sub>2</sub> 4.8 <sup>b</sup>			
lactate	· 3.8*	2.0	174	972
malate	$pK_1 2.8^d$	2.4	1302	100
	pK2 5.74			
succinate	p <i>K</i> ₁ 4.2 <sup>d</sup>	2.4	64	<10
	$pK_{2} 5.6^{d}$			
tartrate	pK1 3.1*	2.6	<10	<10
	pK <sub>2</sub> 4.8 <sup>6</sup>			

<sup>a</sup> Determined by capillary electrophoresis. <sup>b</sup> Ref 40. <sup>c</sup> Ref 41. <sup>d</sup> Ref 42. <sup>e</sup> Ref 36.

values are in agreement with the basal gastric pH of 1.0-2.0(11, 12) and with the small intestinal pH of 4.7-7.2(15) for human subjects, validating the use of female New Zealand White rabbits as a GI model for humans with respect to pH. Based on observations in New Zealand White rabbits, pH levels of 1.3 and 7.0 were selected to represent the stomach and small intestinal phases, respectively, in the *in vitro* model.

Organic Acids. Organic acids are known to chelate Pb (21) and may increase solubilized Pb concentrations in the small intestine, promoting transport of Pb across the intestinal epithelium (6). Therefore, concentrations of organic acids were measured by capillary electrophoresis (Table II) in the stomach and small intestinal fluids of a rabbit from the 6-h post-dose group. The  $pK_a$  values for the first proton released from all of the carboxylic acids for which analyses were performed are  $\geq 2.8$  (Table II); therefore, these acids will be protonated during fasting stomach conditions (pH 1-2) but may chelate Pb in the stomach in the presence of food (pH 4-6) and in the small intestine (pH 7). Comparison of stability constants (determined at 25 °C and solution ionic strength of 1.0 (36) for the lead carboxylate complexes) indicates that citrate, if present, will form the most stable complex with Pb (log  $K_{ML} = 4.1$ ), while the remaining acids will form complexes of decreasing stability in the order tartrate > malate, succinate > acetate, butyrate > lactate. The ability of organic acids to serve as Pb chelators depends on the ability of the soluble chelator to complex Pb (which is pH dependent due to proton competition for the ligand), the strength of Pb<sup>2+</sup> binding to individual ligands (e.g., selectivity for Pb), and the activity of competing cations (e.g.,  $Ca^{2+}$ ,  $Zn^{2+}$ , or  $Cu^{2+}$ ) that may displace Pb. Therefore, in the small intestine at pH 7, the  $K_{\rm ML}$  values for organic acids (Table II) indicate that nearly all Pb complexed by organic acids would be present as lead citrate complexes; however, this does not eliminate the possibility that Pb would be preferentially complexed by stronger chelators (e.g., enzymes and amino acids) or displaced by competing cations.

Malate and citrate were the most concentrated carboxylic acids in the rabbit stomachs, whereas lactate and acetate were most concentrated in the small intestine (Table II). The decrease in citrate and malate concentrations between the stomach and small intestinal fluid may be due to the absorption of these species in the small intestine or to the formation of stable complexes that are

time (h)	stomach fluid <sup>o</sup>	small int. fluid <sup>o</sup>	large int. fluid <sup>o</sup>	bloodd	stomach solids	small int. solids	large int. solids	% total Pb in stomach in fluid phase <sup>n</sup> (%)
0.5	$230 \pm 170$	$6.9 \pm 6.7$	3.0°	$2.0 \pm 0.4$	$6980 \pm 2290$	$38 \pm 20$	$170 \pm 60$	$3 \pm 2$
1	$260 \pm 140$	$28 \pm 34$	$3.8 \pm 2.4^{\circ}$	$1.5 \pm 0.5$	$6670 \pm 2820$	$480 \pm 710$	$130 \pm 47$	$4 \pm 2$
1.5	$620 \pm 250$	140 ± 27•	4	$1.7 \pm 0.4$	$6210 \pm 2070$	$1230 \pm 630$	$690 \pm 760$	9 ± 4
3	$62 \pm 53$	$250 \pm 220$	$32 \pm 25$	$5.4 \pm 2.8$	$850 \pm 300$	$4580 \pm 4830$	$550 \pm 530$	. 7 ± 6
6	$5.8 \pm 3.8$	$6.1 \pm 6.7$	$53 \pm 18$	$7.1\pm2.1$	$62 \pm 80$	$160 \pm 230$	$1170 \pm 300$	9 ± 6
8	$1.6 \pm 0.5$	$1.6 \pm 1.0$	<b>46 ±</b> 20	$7.7 \pm 2.8$	NA	NA	NA	
16	$5.3 \pm 1.2$	$15 \pm 6.1$	$7.2 \pm 0.4^{\circ}$	$3.9 \pm 0.8$	NA	NA	NA	
24	$3.4 \pm 2.0$	5.3 ± 0.4*	8.2 ± 1.0 <sup>e</sup>	$5.3 \pm 1.4$	NA	NA	NA	
36	$3.4 \pm 2.4$	$3.3 \pm 2.1$	$2.9 \pm 0.2^{e}$	$6.4 \pm 2.5$	NA	NA	NA	
I (soluble salt dose)#	$2000 \pm 610$	$810 \pm 390$	620 ± 58°	$14 \pm 7.9$	$3400 \pm 2300$	$1150 \pm 300$	$530 \pm 400$	$37 \pm 11$
1 (blank)	$1.4 \pm 0.2$	$0.6 \pm 0.4$	$10 \pm 1.4^{\circ}$	$0.8 \pm 0.4$	$120 \pm 44$	$33 \pm 28$	$180 \pm 31$	$1 \pm 0.1$
36 (blank)	$2.5 \pm 1.7$	$28 \pm 0.6^{\circ}$	$0.7 \pm 0.2^{e}$	$2.9 \pm 0.4$	NA	NA	NA	

<sup>a</sup> Each rabbit dosed with 2.0 g of BMW-1/kg of body weight (average animal weight of 2.1 kg) for an average dose of 4.2 g of BMW-1 (16.4 mg of Pb). <sup>b</sup> Corrected for percent moisture in solids. <sup>c</sup> No error is presented, because only one measurement was obtained. <sup>d</sup> Assuming average rabbit blood volume of 56 mL/kg of body weight (43). <sup>e</sup> Error based on two measurements. <sup>f</sup> No fluid available after centrifugation. <sup>d</sup> One hour between dosing and sample collection. <sup>h</sup> Value calculated by dividing the Pb mass in stomach fluid by total Pb mass in stomach. Associated error calculated as standard deviation of Pb mass in stomach fluid divided by total Pb mass in stomach. <sup>i</sup> NA, not analyzed.

not resolved during capillary electrophoresis. Lactic acid, which may be produced in the stomach (37), has been observed previously in the small intestines of rabbits (38). Because the concentrations of acetate, citrate, lactate, and malate were variable throughout the rabbit GI tract, an average concentration of 500 mg/L for each acid was used during the *in vitro* experiments.

Lead in in Vivo Fluid and Solid Samples. Lead analyses were performed on all of the fluid samples and on the solid samples collected from the 0.5-6-h post-dose groups. Solid samples from the 8-36-h post-dose groups were not analyzed, because the solid Pb mass had traversed the small intestine and was no longer available for dissolution and absorption (Table III). Reported data were corrected for percent moisture in the solids and are reported as mass ( $\mu$ g) rather than concentration to account for the varying fluid volumes and solid masses present in each rabbit (Table III). Data are presented with standard deviations for each set of three rabbits and indicate substantial interanimal variability (Table III); observed standard deviations are generally in the range of 30-50%. Lead masses in the 1- and 36-h control rabbits are less than or equivalent to any of the dosed rabbits, indicating that background Pb masses are 1–10 and 30–180  $\mu$ g in GI tract fluids and solids, respectively (Table III), probably due to low concentrations of Pb (0.75 mg/kg) present in the rabbit feed.

Acidification of the small intestinal fluid with HNO<sub>3</sub>, for preservation, resulted in the formation of a flocculent precipitate (believed to be an agglomeration of proteins and enzymes that are present in small intestinal fluid). which was readily resuspended when the pH was raised to >5.0 using NaOH. To evaluate the banding of Pb by the precipitate, three small intestinal fluids were analyzed for Pb concentration, with the precipitate both solubilized and precipitated and removed by centrifugation. Comparison of the Pb concentrations in the fluid samples after solubilization of the precipitate indicates that  $63 \pm 4\%$  of the total Pb in solution is removed by this precipitation reaction. Furthermore, a dialysis experiment using 12000 MWCO tubing indicated that 93% of Pb in one small intestinal fluid sample was incapable of diffusing out of the dialysis bag in 24 h. These results suggest that a

substantial portion (approximately 60-95%) of solubilized small intestinal Pb is bound to a protein or enzyme that is >12000 MW.

Mean solubilized Pb masses from BMW-1 as a function of time and GI tract compartment exhibit the expected profile (Table III), with maximum solubilization in the stomach, small intestine, and large intestine occurring at approximately 1.5, 3, and 6 h, respectively. Maximum solubilized Pb masses decrease in each successive segment of the GI tract, with stomach, small intestinal, and large intestinal masses of 0.62, 0.25, and 0.053 mg recovered, respectively. The decrease in Pb mass in solution between stomach and small intestine is probably due to precipitation or sorption reactions that occur as the pH is increased from 1.3 to 7, whereas the decrease in Pb mass between small and large intestine is most likely due to Pb absorption across the intestinal epithelium. Mean Pb masses in GI tract solids are consistent with the pattern observed in the fluid phase (Table III). Emptying of solid Pb from the stomach into the small intestine is complete within 3 h, during which time Pb in the small intestinal solids reaches a maximum (Table III).

When considering available Pb based on either in vivo or in vitro results, it should be recognized that absorption of Pb in the small intestine is a nonequilibrium process. Four pools of Pb are present in the small intestine: (1) solubilized Pb, either ionic or bound by small organic ligands (Pb<sub>sol</sub>); (2) solubilized Pb that is bound to large MW entities (Pb<sub>bnd</sub>); (3) Pb that was solubilized in the stomach but has sorbed to solid surfaces on entering the small intestine (Pb<sub>sorb</sub>); and (4) Pb that was insoluble in the stomach (Pbinsol). Because Pbinsol is unlikely to be solubilized during passage through the small intestine, this fraction of ingested Pb is not bioaccessible. The other three forms of Pb are all available for absorption. Pb<sub>sol</sub> is immediately available, while Pbbnd and Pboorb must become unbound or desorbed, respectively, prior to absorption in the small intestine. However, as Pbsol is absorbed, Pb<sub>bnd</sub> and Pb<sub>sorb</sub> could be released into solution to regain equilibrium partitioning of Pb between the fluid and solid phases. Therefore, assuming that the binding or adsorption mechanisms are reversible, the entire pool of Pb<sub>sol</sub>, Pb<sub>bad</sub>, and Pb<sub>sorb</sub> [which is equal to the total Pb





in the stomach fluid phase, because surface adsorption of Pb will be negligible at the acidic pH of the stomach (39)] is theoretically available for absorption, if the kinetics of Pb absorption across the intestinal epithelium are sufficiently rapid with respect to intestinal transit time and to the kinetics of Pb release or desorption. The mass of Pb

 present in the Pb<sub>bnd</sub> and Pb<sub>sorb</sub> pools and the relative release and desorption kinetics were not evaluated during this study. Consequently, the maximum mass of solubilized Pb in the stomach compared to the mass that is insoluble

 (e.g., the partitioning ratio from solid to fluid) represents a conservative (upper) estimate of available Pb.

Based on the above reasoning, calculation of *in vivo* Pb solubility from BMW-1 (mass of Pb solubilized divided by total Pb present in the stomach) shows that the solubilized Pb fraction increased from an initial 3% to a maximum of 9% at the 1.5-h time point (Table III). The limited Pb dissolution from mine waste is due to the low

- solubility of Pb-bearing minerals, kinetic dissolution limitations (10), and encapsulation of Pb phases by inert matrices (e.g., silicates, pyrite, and jarosite) (5).
- A total of 37% of the recovered Pb from Pb(OAc)<sub>2</sub> is present in the stomach fluids after 1 h, suggesting that the balance is present in the stomach solids. It is unclear why
- the bulk of Pb from Pb(OAc)<sub>2</sub> was present in the stomach solids. Blood-lead response 1 h after ingestion of Pb(OAc)<sub>2</sub> was 9.4 times larger than from an equivalent mass of Pb contained in BMW-1, supporting maximum *in vivo* mine waste-lead solubility, relative to Pb(OAc)<sub>2</sub>, of approxi-
- mately 10%. Lead in *in Vitro* Fluids. Reproducibility of the *in*

*vitro* method was tested by conducting triplicate experiments using BMW-1, for which the maximum Pb mass solubilized (2-h sample) was  $0.65 \pm 0.04$  mg. The method development tests, consisting of the addition of rabbit

- chow, the omission of organic acids, the omission of GI tract enzymes, and the omission of both acids and enzymes (each performed in duplicate), resulted in maximum solubilized Pb masses at the 2-h time point of 0.58 ± 0.04,
- 0.55 ± 0.01, 0.52 ± 0.02, and 0.52 ± 0.02 mg of Pb, respectively (Figure 2). The blank flasks were below the instrument detection limit (IDL, 0.10 mg/L) in each
- experiment. The method detection limits (IDL × dilution factor × volume in flask) in the mine-waste and method development flasks and the Pb(OAc)<sub>2</sub> flask were 0.020 and 0.60 mg, respectively.
- The method development experiments indicate that the addition of rabbit chow to the reaction flask causes a slight

decrease in the mass of Pb solubilized in the stomach from  $0.65 \pm 0.04$  to  $0.58 \pm 0.04$  mg, while the mass of Pb in the soluble phase during the small intestinal incubation was increased slightly from  $0.11 \pm 0.04$  to  $0.18 \pm 0.03$  mg (Figure 2). The reason for a decrease in stomach-solubilized Pb in the presence of rabbit chow is unknown, while the increase in intestinal solubilized Pb is most likely due to a solubilized component of rabbit chow, probably citrate, amino acids, or suspended organic matter, that is capable of binding Pb and retaining it in solution. The absence of either organic acids or GI tract enzymes, or both components together, results in a 20% decrease in maximum stomach Pb solubility  $(0.65 \pm 0.04 \text{ to } 0.52 \pm 0.02)$ mg), consistent with the observation that the presence of organic acids in solution may increase the dissolution of Pb-bearing phases (38). In addition, both acids and enzymes are necessary to retain Pb in solution during the small intestinal phase (Figure 2). These data suggest that organic acids and GI tract enzymes in the pH 7 environment of the small intestine either bind Pb or inhibit the formation of Pb-bearing precipitates.

Based on the reasoning presented in the *in vivo* results section, the fraction of available Pb in vitro from BMW-1  $(4 \pm 0.2\%)$ , determined by dividing the average dissolved Pb mass at 2.0 hr (0.65 mg) by the mass of Pb in the flask (15.6 mg) (Table IV), was in good agreement with the in vivo result  $(9 \pm 4\%)$ . The comparison of in vivo and in vitro solubilized Pb masses (1.5 h for in vivo and 2.0 h for in vitro) were nearly identical: 0.62 versus 0.65 mg for the in vivo and in vitro systems, respectively (Tables III and IV). The overall rate of Pb dissolution in the stomach (the rate of change in solubilized Pb with time in Table III versus Table IV) was similar in the in vivo and in vitro systems, although the *in vitro* dissolution rate was more constant (less variability in the rate of Pb dissolution), indicating that dissolution kinetics are important in controlling Pb bioaccessibility from BMW-1.

The Pb(OAc)<sub>2</sub> results indicate that, after correction for Pb emanating from the blank mine waste (100 mg/kg of bulk Pb concentration resulted in 0.07 mg in the in vitro test), 68% of the Pb from  $Pb(OAc)_2$  is bioaccessible in vitro in the presence of a mine-waste matrix, while 76% is available when no matrix is present (Table IV). The discrepancy between available Pb from Pb(OAc)<sub>2</sub> during in vitro (68 or 76%, depending on the matrix present) and in vivo (37%) testing could be due to a variety of factors, including (1) pH differences between the systems, (2) lower concentrations of organic acids in vivo than used in vitro, or (3) incomplete recovery of ingested material during the in vivo study. Soluble Pb mass from Pb(OAc)<sub>2</sub> in the absence of mine waste decreased by a factor of 2 in the small intestinal simulation (Table IV), while in the presence of mine waste, soluble Pb decreased by a factor of 9. These results suggest that Pb absorption to the minewaste surface may reduce Pb solubility in the small intestine.

The mine-waste samples BMW-2-4, consisting predominantly of lead phosphates, manganese-lead oxides, iron-lead oxides, and iron-lead sulfates, also resulted in limited bioaccessible Pb when tested by the *in vitro* method (0.5-6%, Table IV). BMW-2, which consists of ferromanganese lead oxides and iron-lead sulfate, produced only 0.5% available Pb during the *in vitro* test and reached equilibrium dissolved Pb concentration prior to collection of the initial sample (0.5 h, Table IV). BMW-3 and BMW-4 

	Pb mass (mg) solubilized from mine waste during in other test					
time (h)	BMW-1	BMW-2	BMW-3	BMW-4	Pb(OAc)2 no mine waste	Pb(OAc)2 blank mine wast
0.5	$0.43 \pm 0.04$	0.02ª	0.94ª.»	0.063	$11.0 \pm 0.2$	$10.6 \pm 0.2$
1.0	$0.50 \pm 0.02$	0.02	1.09*	0.09	$11.9 \pm 0.2$	$10.4 \pm 0.3$
1.5	$0.59 \pm 0.06$	0.02	1.300	0.11	$11.8 \pm 0.3$	$10.2 \pm 0.1$
2.0	$0.65 \pm 0.04$	0.02	1.260	0.15	$11.4 \pm 0.1$	$10.2 \pm 0.1$
mass of Pb (mg) solubilized in small intestine simulation	$0.11 \pm 0.04$	<0.008ª	0.03	0.04	$5.3 \pm 0.2$	$1.0 \pm 0.1$
mass of Pb added to flask (mg)	15.6	4.1	23.3	7.2	15.6	15.6
fraction available Pb in vitro (%)	4 <sup>d</sup>	0.5ª	6ª	2ª	76ª	68 <sup>¢</sup>
fraction available Pb in vivoc (%)	9e				37"	

<sup>a</sup> No standard error available because only one test was conducted. <sup>b</sup> Based on detection limit of 0.1 mg/L of Pb by direct-aspiration atomic absorption spectrometry. <sup>c</sup> Method for establishing maximum available Pb *in vivo* same as *in vitro*, using data from Table II. <sup>d</sup> Fraction of bulk Pb mass solubilized during *in vitro* test (4.0 g substrate/40 mL of fluid). <sup>c</sup> Fraction of bulk Pb mass solubilized during *in vivo* feeding study.

(6% and 2% available Pb, respectively) contain greater percentages of Pb minerals such as lead oxides, lead silicates, and anglesite (Table I), which appear to be more soluble than the lead phosphates, ferromanganese lead oxides, and iron-lead sulfates. In addition, BMW-3 reached equilibrium concentration after only 1.5 h, while BMW-4 did not reach equilibrium during the 2-h stomach incubation. These data indicate that dissolution kinetics limit Pb bioaccessibility from the suite of Pb minerals present in BMW-1 and BMW-4 but not from BMW-2 and BMW-3. No discrete Pb mineral phase appears consistently responsible for the available Pb from these mine wastes, a result that was expected, due to the highly variable Pb mineral composition. Factors such as mineral associations, the presence and type of alteration and precipitation rinds, encapsulation of Pb phases in inert minerals (e.g., silicates), and the presence of surfaceadsorbed Pb are likely to be of equal or greater importance in controlling Pb bioaccessibility in the GI tract.

#### Conclusions

This study has demonstrated that an *in vitro* method can be an effective tool in evaluating the geochemical and physiological factors controlling the dissolution of minewaste Pb in the GI tract.

The rabbit was determined to be an appropriate model for human GI tract conditions, with respect to pH and transit times. In a validation study using female New Zealand White rabbits, average pH values were 1.3 and 3-4 in the stomach under fasting and fed conditions, respectively, and approximately 7 in the small intestine, similar to pH values reported in humans. Acetate, citrate, lactate, and malate are all present in varying concentrations in both human and rabbit GI tracts and may chelate Pb with relative complex formation strengths of citrate > malate > acetate > lactate, thereby increasing Pb solubility and subsequent absorption across the intestinal epithelium by passive diffusion. However, a variety of potential chelating agents (organic acids, amino acids, proteins, enzymes, and tannins) will compete for soluble Pb, and on the basis of two preliminary tests, 60-95% of solubilized Pb in the small intestine appears to be bound to a protein or enzyme >12000 MW. Although a stomach pH of 1.3 was used in the *in vitro* test to maximize Pb dissolution, this stomach acidity would apply only to fasting children. A stomach pH of 3-4, typical when food is in the stomach. would be expected to further reduce Pb bioaccessibility by decreasing the extent of Pb mineral dissolution.

Acknowledgmentsf mine-f mine-The authors wish to acknowledge the assistance of ChrisremodelSellstone and Maria Montour for analytical support; RickDH andNelson for editing; Joyce Teter, Barbara Kyvik, PatMadow, and Lucy Norman for secretarial assistance, and1.3 andJohn Drexler for mineralogical analyses. We appreciateditions,the review of this manuscript and comments provided bywestine,two anonymous reviewers. One of us (A.D.), is indebtedcitrate,to Kathy Davis for child care over the duration of thisresearch.ate Pbrate >Literature Citedubility(1) EPA User's Guide for Pb: A PC Software Application of

mine wastes.

 EPA. User's Guide for Pb: A PC Software Application of the Uptake/Biokinetic Model Version 5.0; Department of Research and Development, U.S. Environmental Protection Agency; Washington, DC, 1990.

The *in vitro* beaker assay, modeled after an assay method used to determine available Fe from food, was successful

in replicating Pb dissolution during the feeding study.

Four mine-waste samples with highly variable mineral

composition (BMW-1-4) resulted in only 4, 0.5, 6, and 2%

bioaccessible Pb, respectively, suggesting that geochemical

and physiological factors limit the solubility and, therefore,

the availability of Pb from mine waste, regardless of whether the bulk Pb mineralogy consists of the sulfate/

sulfide or the oxide/phosphate assemblage. The 4% available Pb from BMW-1 is in good agreement with a

30-day feeding study in Sprague–Dawley rats, using this

mine waste, which produced only 3% absolute Pb bio-

availability based on blood-lead data (3). In addition, method development testing indicates that although HCl

concentration is the most important GI component

controlling Pb dissolution in the stomach, both organic acids and enzymes are necessary to retain Pb in solution

during the small intestinal incubation. Based on these

results, the assay appears to provide a useful, rapid, screening-level test to predict maximum available Pb from

- (2) Freeman, G. B.; Johnson, J. D.; Killinger, J. M.; Liao, S. C.; Feder, P. I.; Davis, A. O.; Ruby, M. V.; Chaney, R. L.; Lovre, S. C.; Bergstrom, P. D. Fund. Appl. Toxicol. 1992, 19, 388-398.
- (3) Freeman, G.; Johnson, J.; Liao, S.; Schoof, R.; Bergstrom, P. Toxicologist 1993, 13 (1), 301.
- (4) Dieter, M. P.; Matthews, H. B.; Jeffcoat, R. A.; Museman, R. F. J. Toxicol. Environ. Health 1993, 39, 79-93.
- (5) Davis, A.; Ruby, M. V.; Bergstrom, P. D. Environ. Sci. Technol. 1992, 26 (3), 461-468.

- (6) Blair, J. A.; Coleman, I. L.; Hilburn, M. E. J. Physiol. 1979, 286, 343–350.
- (7) Conrad, M. E.; Barton, J. C. Gastroenterology 1978, 74, 731-740.
- (8) Davis, A.; Drexler, J. W.; Ruby, M. V.; Nicholson, A. Environ. Sci. Technol. 1993, 27 (7), 1415–1425.
- (9) Vlek, P. L. G.; Blom, J. M.; Beek, J.; Lindsay, W. L. Soil Sci. Soc. Am. Proc. 1974, 38, 429.
- (10) Ruby, M. V.; Davis, A. O.; Kempton, J. H.; Drexler, J. W.; Bergstrom, P. D. Environ. Sci. Technol. 1992, 26 (6), 1242– 1248.
- (11) Malagelada, J. R.; Longstreth, G. F.; Summerskill, W. H. J.; Go, V. L. W. Gastroenterology 1976, 70, 203-210.
- (12) Malagelada, J. R.; Go, V. L. W.; Summerskill, W. H. J. Dig. Dis. Sci. 1979, 24, 101-110.
- (13) Hemphill, C. P.; Ruby, M. V.; Beck, B. D.; Davis, A.; Bergstrom, P. D. Chem. Speciation Bioavailability 1991, 3 (3/4), 135-148.
- (14) Hunt, J. N.; Spurrel, W. R. J. Physiol. 1951, 113, 157-168.
- (15) Murthy, S. N. S.; Kostman, J.; Dinoso, V. P. Dig. Dis. Sci. 1980, 25, 289-294.
- (16) Chaney, R. L.; Mielke, H. W.; Sterret, S. B. Environ. Geochem. Health 1989, 11, 105-129.
- (17) Klotz, I. M.; Urquhart, J. M.; Fiess, H. A. J. Am. Chem. Soc. 1952, 74, 5537-5538.
- (18) Richardt, G.; Federolf, G.; Habermann, E. Biochem. Pharmacol. 1986, 35 (8), 1331-1335.
- (19) Li, N. C.; Manning, R. A. J. Am. Chem. Soc. 1955, 77, 5225– 5228.
- (20) Quarterman, J.; Humphries, W. R.; Morrison, J. N.; Morrison, E. Environ. Res. 1980, 23, 54-67.
- (21) Jugo, S.; Maljkovic, T.; Kostial, K. Toxicol. Appl. Pharmacol. 1975, 34, 259-263.
- (22) Narasinga Rao, B. S.; Proabhavati, T. Am. J. Clin. Nutr. 1978, 31, 169-175.
- (23) Miller, D. D.; Schricker, B. R.; Rasmussen, R. R.; Van Campen, D. Am. J. Clin. Nutr. 1981, 34, 2248-2256.
- (24) Miller, D. D.; Schricker, B. R. In In Vitro Estimation of Food Iron Bioavailability; Kies, C., Ed.; ACS Symposium Series 203; American Chemical Society: Washington, DC, 1982; pp 11-24.
- (25) Reddy, M. B.; Chidambaram, M. V.; Fonseca, J.; Bates, G. W. Clin. Physiol. Biochem. 1986, 4, 78-86.
- (26) Reddy, M. B.; Browder, E. J.; Bates, G. W. In Essential and Toxic Trace Elements in Human Health and Disease; Prasad, A., Ed.; Alan Liss Inc.: New York, 1988.
- (27) Calabrese, E. J.; Barnes, R.; Stanek, E. J., III; Pastides, J.; Gilbert, C. E.; Veneman, P.; Wang, X.; Laszitity, A.; Kos-

tecki, P. T. Regul. Toxicol. Pharmacol. 1989, 10, 123-137.

- (28) Kimbrough, R. D.; Falk, H.: Stehr, P.; Fries, G. J. Toxicol. Environ. Health 1984, 14, 47-93.
- (29) Auvergne, A.; Bouyssou, T.; Pairet, M.; Bouillier-Oudot, M.; Ruckebusch; Y.; Candau, M. Reprod. Nutr. Dev. 1987, 27 (4), 755-768.
- (30) Duggan, M. J.; Inskip, M. J.; Rundle, S. A.; Moorcroft, J. S. Atmos. Environ. 1985, 44, 65-79.
- (31) EPA. Test Methods for Evaluating Solid Waste. Volume 1A: Laboratory Manual, Physical/Chemical Methods; Office of Solid Waste and Emergency Response, U.S. Environmental Protection Agency: Washington, DC, 1986,p 222.
- (32) ASA. Methods for soil analysis. Part 2: Chemical and microbiological properties; Black, C. A., Ed.; American Society for Agronomy, Inc.: Madison, WI, 1965; pp 914.
- (33) EPA. Toxic Substances Control Act, Good Laboratory Practice Standards; U.S. Environmental Protection Agency, 1989, 40 CFR, Part 792.
- (34) EPA. Laboratory Data Validation Functional Guidelines for Evaluating Inorganics Analyses; Hazardous Site Evaluation Division, U.S. Environmental Protection Agency: Washington, DC, 1988, p 19.
- (35) Soave, O.; Brand, C. D. Cornell Vet. 1991, 81, 359-364.
- (36) Martell, A. E.; Smith, R. M. In Critical Stability Constants; Plenum Press: New York, 1977; Vol. 3.
- (37) Argenzio, R. A.; Southworth, M. Am. J. Physiol. 1974, 228
   (2), 454-460.
- (38) Marty, J.; Raynaud, P. Arch. Sci. Physiol. 1966, 20 (4), 515– 524.
- (39) Salim, R.; Cooksey, B. G. Plant Soil 1980, 54, 399-417.
- (40) The Merck Index, 11th ed.; Budavari, S., Ed.; Merck and Co., Inc.: Rahway, NJ, 1989.
- (41) Kemp, D. S.; Vellacio, F. In Organic Chemistry; Poulson, S., Ed.; Worth Publishers, Inc.: New York, 1980: pp 327-330.
- (42) Streitwieser, A.; Heathcock, C. H. In Introduction to Organic Chemistry, 3rd ed.; Macmillan Publishing Co., New York, 1985; p 864.
- (43) Armin, J.; Grant, R. T.; Pels, H.; Reeve, E. B. J. Physiol. 1952, 116, 59-73.

Received for review May 3, 1993. Revised manuscript received August 3, 1993. Accepted August 11, 1993.•

Abstract published in Advance ACS Abstracts, October 1, 1993.

# **APPENDIX B**

# HEALTH AND SAFETY PLAN PART I

H&SPLAN.REV

# HEALTH AND SAFETY PLAN PART I

# AMES STREET FACILITY ROCHESTER, NEW YORK

Prepared by: ABB ENVIRONMENTAL SERVICES, INC. ROCHESTER, NEW YORK

> APRIL 1993 Revision 1: July 1995 Revision 2: November, 1995

> > .

.

H&SPLAN.REV

# HEALTH AND SAFETY PLAN PART I

# TABLE OF CONTENTS

Section	Title	Page No.
10 000		
1.0 GEN	ERAL	I-I
1.1	SCOPE AND PURPOSE	1-1
1.2	2 PROJECT PERSONNEL	1-1
	1.2.1 Project Manager	1-1
	1.2.2 Field Operations Leader	1-1
	1.2.3 Health and Safety Officer	1-2
	1.2.4 Other ABB Environmental Personnel	1-2
	1.2.5 ABB Environmental Subcontractor Personnel	1-2
	1.2.6 Other Site Visitors	1-3
1.3	3 TRAINING	1-3
1.4	4 MEDICAL SURVEILLANCE	1-4
2.0 SITE	CHARACTERIZATION AND ANALYSIS	2-1
2.	1 SITE NAME, LOCATION, AND SIZE	2-1
2.2	2 SITE HISTORY AND LAYOUT	2-1
2.3	3 SCOPE OF WORK (WORK PLAN)	2-1
3.0 TASE	( ANALYSIS	3-1
3.2	1 CATEGORY A - INVASIVE SAMPLING-TYPE ACTIVITIES	3-1
	3.1.1 Hazardous Substances	3-1
	3.1.2 Site Risks	3-1
	3.1.3 Protective Measures	3-5
	3.1.4 Monitoring	3-7
3.2	2 CATEGORY B - NON-INVASIVE ACTIVITIES	3-9
	3.2.1 Hazardous Substances	3-9
	3.2.2 Site Risks	

i

.

# HEALTH AND SAFETY PLAN PART 1

# TABLE OF CONTENTS (continued)

Section	Title	Page No.
3.2	.3 Protective Measures	
3.2	.4 Monitoring	
3.3 CATEC	FORY C - SUBCONTRACTOR OVERSIGHT	
3.3	.1 Hazardous Substances	
3.3	.2 Site Risks	
. 3.3	.3 Protective Measures	
3.3	.4 Monitoring	
3.4 CATEC	FORY D - VISITOR TOURS AND MEETINGS	
3.4	.1 Hazardous Substances	
3.4	.2 Site Risks	
3.4	.3 Protective Measures	
3.4	.4 Monitoring	
3.5 SPECIA	AL PROCEDURES FOR SITE VISITORS	
3.5	.1 Visitor Entry Forms	
3.5	.2 Visitor Entry Procedures	
4.0 DATA SHEE	TS	4-1
5.0 SITE CONTR	ROL	5-1
5.1 ZONA	ΓΙΟΝ	5-1
5.2 Comm	IUNICATIONS	5-1
5.3 WORK	PRACTICES	5-1

6.0 L	DECO	NTAMINATION/DISPOSAL	6-1
	6.1	PERSONNEL DECONTAMINATION	. 6-1
	6.2	SMALL EQUIPMENT DECONTAMINATION	6-2
	6.3	HEAVY EQUIPMENT DECONTAMINATION	.6-3
	6.4	DISPOSAL OF DECONTAMINATED MATERIALS	.6-3

H&SPW.REV

.

# HEALTH AND SAFETY PLAN PART 1

# TABLE OF CONTENTS (continued)

Section	Title	Page No.
7.0 EMER	RGENCY/CONTINGENCY PLAN	7-1
7.1	PERSONNEL ROLES, LINES OF AUTHORITY, AND COMMUNICATION	7-1
7.2	EVACUATION	7-1
	7.2.1 Withdrawal Upwind	7-1
	7.2.2 Site Evacuation	7-2
	7.2.3 Evacuation of Surrounding Area	7-2
7.3	EMERGENCY MEDICAL TREATMENT/FIRST AID	
8.0 ADM	INISTRATION	8-1
8.1	Personnel Authorized Downrange	8-1
8.2	MEDICAL DATA SHEET	8-1
8.3	EMERGENCY TELEPHONE NUMBERS	8-3
8.4	ROUTES TO EMERGENCY MEDICAL FACILITIES	
8.5	HASP APPROVALS	8-7

H&SPW.REV

7198-17

•

.

# HEALTH AND SAFETY PLAN PART II

# TABLE OF CONTENTS (continued)

## **APPENDICES**, Part 2

The following appendices are applicable for the work anticipated at the site (see Health and Safety Plan, Part II):

- X A. AUTHORITY AND RESPONSIBILITY OF HEALTH AND SAFETY PERSONNEL
- X B. TRAINING PROGRAM
- X C. MEDICAL SURVEILLANCE PROGRAM
- X D. ENGINEERING CONTROLS
- <u>X</u> E. PERSONAL PROTECTIVE LEVEL DETERMINATION
- X F. MONITORING EQUIPMENT
- X G. ZONATION
- X H. WORK PRACTICES
- \_ I. CONFINED SPACE ENTRY PROCEDURES
- \_ J. EXCAVATION AND TRENCHING
- \_ K. TEMPERATURE EXTREMES
  - HEAT STRESS
  - \_ COLD STRESS

H&SPLAN.REV

# HEALTH AND SAFETY PLAN PART II

# TABLE OF CONTENTS (continued)

#### **APPENDICES**, Part 2

- X L. DECONTAMINATION
- X M. EMERGENCY PLANNING
- X N. HEALTH AND SAFETY FORMS AND DATA SHEETS
  - X HEALTH AND SAFETY AUDIT FORM
  - X ACCIDENT REPORT FORM
  - <u>X</u> HSO CHECKLIST FOR FIELD OPERATIONS
  - \_\_\_\_ MATERIAL SAFETY DATA SHEETS
  - X LIQUI-NOX
  - \_ ETHYL ALCOHOL (denatured)
  - \_\_\_\_ TRISODIUM PHOSPHATE
  - X OSHA POSTER
  - X DAILY HEALTH AND SAFETY AUDIT FORM
- X O. RESPIRATORY PROTECTION PROGRAM
- X P. OTHER
  - \_ ILLUMINATION
  - X SANITATION
  - <u>X</u> HEALTH AND SAFETY AUDIT PROCEDURES
  - Q. STANDARD OPERATING PROCEDURES
    - \_\_\_\_ STANDARD OPERATING PROCEDURES FOR THE USE OF EXPLOSIVES IN SEISMIC REFRACTION SURVEYS

# HEALTH AND SAFETY PLAN PLAN I

# LIST OF FIGURES

<u>Figure</u>	Title	Page No.
2-1	Site Location Map	2-2
2-2	Site Layout Map	2-3
8-1 8-2	Primary Hospital Route Map Secondary Hospital Route Map	8-5 8-6

# HEALTH AND SAFETY PLAN PLAN I

# LIST OF TABLES

<u>Table</u>		Title	Page No.
			_
3-1	Chemicals of Concern	•••••••••••••••••••••••••••••••••••••••	

H&SPLAN.REV

7198-17

.

## **1.0 GENERAL**

#### **1.1 SCOPE AND PURPOSE**

This Health and Safety Plan (HASP) has been prepared in conformance with the ABB Environmental Services, Inc. (ABB Environmental) Health and Safety Program and is intended to meet the requirements of 29 CFR 1910.120. As such, the HASP addresses those activities associated with field and other operations for this project. Compliance with this HASP is required for all ABB Environmental personnel. Contractor personnel entering the site will be supplied with a copy of this HASP for informational purposes.

#### **1.2 PROJECT PERSONNEL**

#### **1.2.1 Project Manager**

The project manager (PM) is the individual with overall project management responsibilities. Those responsibilities as they relate to health and safety include provision for the development of this site-specific HASP; the necessary resources to meet requirements of this HASP; the coordination of staff assignments to ensure that personnel assigned to the project meet medical and training requirements; and the means and materials necessary to resolve any health and safety issues that are identified or that developed on the project.

As of November, 1995, the PM is Geoff Knight from ABB-ES' Rochester, NY office.

#### 1.2.2 Field Operations Leader/Construction Manager

The Resident Engineer or Field Operations Leader (FOL) is the PM's designee who is onsite and vested with the authority by the PM to carry out day-to-day site operations, including interfacing with the site Health and Safety Officer (HSO).

As of November, 1995, Tony Delano, P.E. from ABB-ES' Rochester office is the FOL.

#### ABB Environmental Services, Inc.

# 1.2.3 Health and Safety Officer

The HSO for each field activity is designated by the PM with concurrence of the Health and Safety Supervisor (HSS) or Health and Safety Manager (HSM). The HSO will have at least an indirect line of reporting to the HSM through the HSS for the duration of his/her assignment as project HSO. The HSO is responsible for developing and implementing this site-specific HASP and any activity-specific procedures in accordance with the ABB Environmental Health and Safety Program. The HSO will investigate all accidents, illnesses, and incidents occurring on-site. The HSO will also conduct safety briefings and site-specific training for on-site personnel. The HSO, in consultation with the PM, HSS or HSM, is responsible for updating and modifying this HASP as site or environmental conditions change.

# **1.2.4** Other ABB Environmental Personnel

Depending on the specific activity being conducted (e.g., soil investigation, demolition, construction, regulatory agency tour), different ABB Environmental personnel may visit or work at the Ames Street site on a one-time, occasional or frequent basis. Regardless of the activity, all ABB Environmental personnel entering the site are subject to the provisions of this HASP.

When a particular site activity may potentially require greater than Level D protection, and for any activity involving sample collection, the "Medical Data Sheet" forms found in Section 8.0 of this HASP must be completed. Site visits, tours or other activities by ABB Environmental personnel for which only Level D protection is required do not require that this form be completed. Prior to each visit, however, the PM or HSO will brief these persons on the HASP components as they may apply to the specific activity or area(s) to be visited as described in Section 3.4.

## **1.2.5 ABB Environmental Subcontractor Personnel**

ABB Environmental subcontractors are generally required under their contract terms to devise a health and safety program for their employees who will enter or work at the site. This program, at a minimum, must comply with local, state and federal requirements (particularly 29 CFR 1910.120) and utilize measures which are at least as protective as those in ABB Environmental's HASP (i.e., this document).

# ABB Environmental Services, Inc.

Unless specified in the subcontract agreement, ABB Environmental is not responsible for monitoring a subcontractor's compliance with the subcontractor's or ABB Environmental's HASP or local, state or federal requirements. ABB Environmental does, however, have the contractual right to control subcontractors whom they believe may not be in compliance, through such measures as denying site access, issuing stop work orders, etc. Such measures will be applied at the discretion of the FOL, HSO, PM and other responsible ABB Environmental personnel.

## **1.2.6** Other Site Visitors

Visitors to the Ames Street site other than ABB Environmental personnel and subcontractors may include:

- Representatives of Combustion Engineering (ABB Environmental's client and the site owner).
- Combustion Engineering contractors (e.g., Nixon, Hargrave, Devans and Doyle [environmental attorneys], Eric Mower and Associates [public relations]).
- Representatives of ABB Kent-Taylor (the former site owner)
- · Potential vendors invited to job walks, etc.
- Regulatory agency, City and State government and local utility personnel.

Although ABB Environmental is not strictly speaking responsible for these visitors' health and safety, as agents of Combustion Engineering and professionals knowledgeable of site conditions, ABB Environmental personnel have a responsibility to provide for visitor's health and safety. As described in Section 3.0, a set of procedures has been devised to provide information and guidance to site visitors.

## ABB Environmental Services, Inc.

#### **1.3 TRAINING**

Training is defined under the ABB Environmental Health and Safety Program, and all ABB Environmental personnel entering potentially contaminated areas of this site must meet the requirements of 29 CFR 1910.120. Personnel without the required training will not be permitted in any area with potential for exposure to toxic substances or harmful physical agents (i.e., downrange). Refer to Appendix B for further information.

#### **1.4 MEDICAL SURVEILLANCE**

All ABB Environmental personnel entering potentially contaminated areas of the Ames Street Site will be medically qualified for site assignment through a medical surveillance program outlined in the ABB Environmental Health and Safety Program. Personnel who have not received medical clearance will not be permitted in any area with potential for exposure to toxic substances or harmful physical agents (i.e., downrange). Refer to Appendix C for further information.

## **ABB Environmental Services, Inc.**

#### 2.0 SITE CHARACTERIZATION AND ANALYSIS

#### 2.1 SITE NAME, LOCATION, AND SIZE

The Ames Street site, the subject of the investigation, is located at 95 Ames Street in Rochester, N.Y. (Figure 2-1). The Site encompasses approximately 14 acres of land and one significant building (Figure 2-2).

#### 2.2 SITE HISTORY AND LAYOUT

The Ames Street facility was part of the Rochester operations of ABB Kent-Taylor. The property was purchased by Taylor Brothers Company (a predecessor to ABB Kent-Taylor) in 1904; the company occupied the site in 1905 or 1906. Industrial operations at the site, conducted by ABB Kent-Taylor or its predecessor companies, have included the manufacture of thermometers, barometers, compasses, altimeters, process automation equipment, and other related products. The property is currently held under the Combustion Engineering name, with site closure activities being coordinated by the ABB, Inc., Real Estate Director in Norwalk, CT.

As of November, 1995 all structures with the exception of Building 60, at the northeast corner of the property, have been removed and the site graded flat. Site operations are conducted from office trailers located at the Hague Street entrance to the site. Telephone and electric power is available at the trailers, but there is no running water at the site unless arrangements are made with the City of Rochester to utilize fire hydrants in the Hague Street or Ames Street sidewalks. Building 60 is used for storage of field equipment.

#### 2.3 SCOPE OF WORK (WORK PLAN)

ABB Environmental will be performing a number of tasks at the site throughout what is expected to be a project lasting several years. For the most part, ABB Environmental's activities will fall into one of the following categories:

#### ABB Environmental Services, Inc.



1

-

\_

<u>CATEGORY</u>	ACTIVITIES
Α	Sampling building materials (e.g., wood, concrete) and/or surface and subsurface soil and groundwater potentially contaminated with mercury, solvents, hydrocarbon fuels or other industrial materials.
В	Other investigation-related activities which do not involve direct contact with potentially contaminated materials. Examples: surveying and mapping, geophysical investigation.
С	Overseeing subcontractor activities associated with remediation of soil or groundwater, and routine site security, maintenance and other services.
D	Attending and/or leading site tours and meetings involving the client, client subcontractors, regulatory agencies, government officials, community members, etc.

The Task Analysis section of this HASP (Section 3.0) analyzes each of the above Task Categories, which provides general information regarding potential health and safety hazards and protective measures. Because of the wide variety of activities likely to occur at this site, analysis of specific tasks within each class is not provided. Instead, the FOL/HSO for each field activity will undertake a specific task analysis prior to field work using the following process:

- 1. Review this entire HASP.
- 2. Using the forms on the following pages, analyze all contemplated field activities likely for the particular task. The general information in Section 3.0 of this HASP will in many cases provide sufficient information to complete the analysis and select protective measures. The task analysis is to be approved by the PM and HSS.

#### ABB Environmental Services, Inc.

H&SPLAN.REV

- 3. Provide the HASP and completed task analysis forms to the field team for review and signature.
- 4. Submit a copy of the approved task analysis to the PM. It will be added to the site HASP "library", where it can be readily referred to for future task analyses.

Remember, each separate field task must be separately analyzed and protective measures approved by the HSS. By itself, this HASP is NOT sufficient for performing field work!

# ABB Environmental Services, Inc.

## 3.0 TASK ANALYSIS

The following sections provide a general analysis of each task category.

It is important to remember that every individual task analysis should also consider the impact of other tasks or operations being simultaneously conducted. Particularly with respect to Categories C, D and E, which are generally conducted at Level D protection, the presence of other operations on the site may restrict both work in or even access to particular areas.

#### 3.1 CATEGORY A - INVASIVE SAMPLING-TYPE ACTIVITIES

This category clearly has the highest risk potential of the four categories. All contemplated tasks within this category must be carefully analyzed and planned for.

#### **3.1.1 Hazardous Substances**

Field investigation activities all have a potential for site personnel to be exposed to low concentrations of organic and inorganic compounds.

Table 3-1 identifies the compounds that are known or suspected to be present on-site, along with the established exposure limits for those substances.

#### 3.1.2 Site Risks

#### Health Hazards

General observations from the three rounds of sampling conducted at the site as of July, 1995, and plant demolition, are summarized below. A complete list of potential chemical hazards by area at the site is contained in the "Investigation Workplan, Phase 1, Former Taylor Instruments Facility."

Visible or non-visible liquid mercury: Significant concentrations of mercury

## ABB Environmental Services, Inc.

#### TABLE 3-1

# CHEMICALS OF CONCERN AMES STREET SITE INVESTIGATION WORKPLAN - PHASE 1

Compound	Exposure PEL/TWA or TLV/TWA	LIMITS STEL/CEIL(c) (ppm)
Mercury	0.05 mg/m <sup>3</sup> [skin]	-
Gasoline	300 ppm	-
Toluene	100 ppm	-
Trichloroethylene	50 ppm	200
Naphtha	100 ppm	-
Xylene	100 ppm	150
No. 6 Fuel Oil	-	
Cadmium	0.005 mg/m <sup>3</sup>	-
Chromium	0.5 mg/m <sup>3</sup>	
Cyanide	5 mg/m <sup>3</sup>	-
perchloroethylene	25 ppm	-
1,1,1-Trichloroethane	350 ppm	450 ppm
Ethylbenzene	100 ppm	125 ppm
Benzene	1 ppm	5 ppm
Nickel	1 mg/m <sup>3</sup>	-
Asbestos	02 f/cc	1 f/cc
PCB	0.5 - 1 mg/m <sup>3</sup>	-
Oils/Motor	-	-

PEL/TW	A=	Permissible Exposure Limit - Tir	ne Weighted Average			
TLV/TW/	4=	Threshold Limit Value - Time W	eighted Average			
STEL	=	Short-term Exposure Limit - The 15-minute time - weighted average which should not be exceeded at any time during the working day.				
CEIL	=	The concentration that should not be exceeded during any part of the working exposure.				
ppm	z	parts per million				
[skin]	2	The potential significant contribution to the overall exposure by the cutaneous route, by direct skin contact with the substance.				
mg/m <sup>3</sup>	=	milligrams per cubic meter				
f/cc	=	fibers	per	cubic	centimeter	

ABB Environmental Services, Inc.

H&SPLAN.REV

7198-17

.

may be present on-site. Liquid mercury has been noted at several locations in soil. Soil concentrations of mercury up to 13,000 mg/kg have been detected, and are expected particularly in association with areas where broken instruments ("glass shards") occur in the site soils. Lower levels of mercury can be expected in soils in many other areas.

Elemental mercury, which was the material used by the manufacturing processes and is the form most likely to occur at the Ames Street site, is primarily a health hazard via the inhalation pathway. Mercury vapor monitoring by ABB Environmental and its subcontractors using the Jerome analyzer has found that excavation of the mercury-bearing soils will frequently produce mercury vapor levels above the action level of 0.025 mg/m<sup>3</sup>. Installing soil borings, where the exposed surface area of soil is lower, is generally less likely to cause action level exceedances. Holding the Jerome analyzer directly over thermometer shards or highly contaminated soil frequently produces a reading, however, indicating that personnel should continue monitoring when working closely with these materials.

Pre- and post-work urine samples have been collected and analyzed from a number of onsite workers at various times. As of November, 1995, only one individual has not been within normal limits. Although the reason for this exception is unclear (the worker was generally engaged in a high exposure-potential activity) and may not definitely be linked to the individual's work at the Ames Street site, continued caution is certainly warranted.

Solvents and hydrocarbons: Several areas in which chlorinated solvent degreasing activity or storage occurred may contain residual TCE and/or PCE. Previous sampling operations in these areas frequently produced PID readings above the 10 ppm level requiring upgrade to Level C, and brief excursions above the 175 ppm limit for Level B upgrade occurred. Odors were very noticeable.

Sampling near the former Tanks 2/15 and 16/17, using hand augers and the TerraProbe<sup>SM</sup>, produced PID readings generally below the 10 ppm upgrade level. Odors were still noticeable, however, and significant concentrations of these two solvents are present in soil. Fuel-related hydrocarbons are also present near the former Building 40 and 42 areas, but TerraProbe<sup>SM</sup> sampling activities produced no PID readings which required Level C. Sampling in Building 12 produced no PID levels above background, and only very low levels of soil contamination.

#### ABB Environmental Services, Inc.

H&SPLAN.REV

*Metals (including cyanide)*: The occurrence of heavy metals at the site is believed to be limited to the former electroplating area in former Building 4. Cyanide compounds may also be present in this area. No real-time techniques have been used during previous sampling of these areas, which has been conducted at Level C.

Other materials: Various oils and greases (e.g., machine oils, hydraulic fluids) may be present in soil in several areas. These should not constitute a significant health hazard under normal conditions, but dermal/respiratory protection needs to be considered during invasive activity.

#### Safety Hazards

In the absence of significant site activity, such as building demolition or soil excavation, there should be very few safety hazards outside of those normally posed by working with drilling rigs and other heavy equipment used in site investigation. Due to the lack of a convenient power source, the use of hand tools, i.e., jackhammers, generators, etc. should be also be considered in the hazard analysis.

"Slip, trip and fall" hazards may be numerous at the Ames Street site. A maintenance contractor is available for debris removal or other duties if necessary to minimize such hazards for a particular task or area.

The locations of underground utilities and structures are believed to be well known, and maps are available from the PM. Although caution should still be exercised, there are no known active electric, water or gas lines within the fenced area of the site.

Cold stress in the winter is a distinct possibility, as is heat stress during warm months when Level C or greater protection is needed.

#### ABB Environmental Services, Inc.

## Conclusion/Risk Assessment

In general, the overall health risks for Category A activities will be low to moderate, and due primarily to the potential presence of mercury and solvent/hydrocarbon vapors. The overall safety risk is moderate and due primarily to the hazards associated with working with heavy equipment. Work conducted in proximity to other site activities, however, can significantly alter the nature of or increase the level of hazard. Cold and heat stress also need to be considered for certain operations.

#### **3.1.3 Protective Measures**

Protective measures will be undertaken to minimize the potential health and safety risks for field personnel engaged in Category A activities.

#### Engineering Controls

It is anticipated that engineered controls will rarely be used. Fans or blowers may be useful for vapor control in some instances; opening windows has also proven to be easy and effective. Use of water to control dust must be carefully evaluated, due to the potential need to containerize runoff, and is not recommended. Prior to subsurface drilling, boring locations should be compared to available site utility maps and reviewed by the PM.

#### Levels of Protection

The following paragraphs describe typical levels of protection utilized for various Category A activities, for each type of known contaminant.

*Mercury and metals*: To prevent dermal exposure and spreading of mercury via foot traffic, the immediate areas where visible mercury is present, and where significant concentrations are known to exist, should not be entered unless, at a minimum, boot covers are worn. Invasive work requires Level C, or at a minimum modified Level D, protection.

*Solvents/hydrocarbons*: For soil/groundwater sampling, Level C protection will probably not be necessary unless large-diameter boreholes are opened or bulk soil excavation occurs. Highly invasive activities, such as excavation, will probably require Level C, and potentially Level B, protection.

## ABB Environmental Services, Inc.

H&SPLAN.REV

Typical levels of personal protection for Category A activities are as follows:

	TYPICAL
ACTIVITY	LEVEL OF PROTECTION
Surface Soil Sampling	Modified Level D
TerraProbe Activities	Level D/Modified Level D
Well Sampling	Modified Level D

Modified Level D consists of Level D personal protective equipment, boot covers and Tyvek coveralls. Levels of protection may be changed at the discretion of the HSO, as warranted by a change in site conditions.

Respirator cartridges, if Level C becomes necessary, will be approved for respiratory protection against organic vapors (magenta/yellow cartridge) or mercury vapors with high efficiency filters (part #466204 and #492924). Refer to Appendix E for details of protection and personal protective equipment and Appendix O for further information on respiratory protection. Task activities may require upgrade per assigned action levels (see below).

#### 3.1.4 Monitoring

Monitoring of the work environment will be undertaken to ensure that Immediately Dangerous to Life or Health (IDLH) or other dangerous conditions are identified. At a minimum, this monitoring will include evaluations for combustible atmospheres, and hazardous concentrations of airborne contaminants.

#### **ABB Environmental Services, Inc.**

# Air Sampling

To the extent feasible, the presence of airborne contaminants will be evaluated through the use of direct reading instrumentation. Information gathered will be used to ensure the adequacy of the levels of protection being used and may be used as the basis for upgrading or downgrading the levels of protection in conformance with action levels provided in this HASP and at the direction of the task HSO.

In general, due to the prevalence of mercury at the Ames Street site, the Jerome analyzer should be used to monitor for mercury vapors during the initial stages of any invasive sampling at any location at the Ames Street site. Subsequent use of the analyzer can depend on the initial readings, visual observations, and the specific area being investigated.

Depending on the task, the following sampling equipment will be used. Refer to Appendix F for information on the calibration and maintenance of the equipment.

- 1. Jerome Mercury Vapor Analyzer
- 2. Photoionization Detector (PID)
- 3. Draeger Tubes benzene 5/c
- 4. Explosimeter

For solvents/hydrocarbons: Monitor the breathing zone with the PID. If levels steadily exceed background, monitor for benzene using the 5/c benzene Draeger Tube. If benzene levels exceed 0.5 ppm, upgrade to Level C. If benzene levels are  $\geq 10$  ppm, upgrade to Level B. If benzene levels are  $\leq 0.5$  ppm, continue work at Level D/Modified Level D until the PID reads  $\geq 10$  ppm then upgrade to Level C. If PID  $\geq 175$  ppm, upgrade to Level B.

Monitor the borehole opening with explosimeter.

For mercury vapors: Monitor continuously with the Jerome Mercury Vapor Analyzer (MVA) and upgrade accordingly.

# ABB Environmental Services, Inc.

H&SPLAN.REV

The action levels for upgrade are as follows:

Level D/Modified Level D acceptable if:

- PID reads < 10 ppm; and
- Benzene 5/c Draeger Tube reads < 0.5 ppm; or
- MVA reads  $< 0.025 \text{ mg/m}^3$

Level C required if:

- PID reads between 10 and 175 ppm; and/or
- MVA reads  $\geq 0.025$  mg/m<sup>3</sup>; and/or
- Benzene 5/c Draeger Tube reads < 10 ppm.

Level B required if:

- PID reads  $\geq$  175 ppm; or
- $\geq$  2.5 mg/m<sup>3</sup> on the MVA; and/or
- Benzene 5/c Draeger Tube reads  $\geq$  10 ppm.

If the explosimeter reads 10%, use non-sparking tools. If the explosimeter reads 20%, stop work, eliminate all ignition sources, and evacuate the area.

## Personal Monitoring

Personal air sampling for mercury has been conducted to some extent by the plant demolition contractor. Results indicated that exposure above the PEL was likely; however, the activities monitored involved significant handling of liquid mercury which are not expected to be repeated again at the site. In general, personal air sampling will not be necessary, however, the HSM should be consulted relative to the specifics of the particular task and make the final evaluation.

Urine monitoring for mercury should be considered for Category A activities which involve sampling or working in suspect or known mercury-containing areas for extended periods. The need for this type of personal monitoring should be discussed with the HSM on a task-by-task basis.

## ABB Environmental Services, Inc.

# 3.2 CATEGORY B - NON-INVASIVE ACTIVITIES

This category involves generally more limited risk potential than Category A, principally because the risk for chemical exposure is lower. Because Category B activities are frequently performed concurrently with Category A activities, however, a task analysis should be made so that personnel involved in the non-invasive activity do not work within the exclusion zone for invasive activity.

# 3.2.1 Hazardous Substances

Category B activities generally pose little potential for chemical exposure, because they are by definition non-invasive. Following complete paving of the site, this work is very unlikely to pose an exposure hazard.

3.2.2 Site Risks

# Health Hazards

*Visible or non-visible liquid mercury*: As described in Section 3.1, mercury may pose risks to work in certain areas. Mercury vapors should not pose a problem for any Category B activity.

Solvents and hydrocarbons: These should not pose a hazard during Category B activities except.

Metals (including cyanide): Similarly, metals exposure should not be an issue.

*Other materials*: For in-building work, the hazards posed by asbestos, PCBs and various oils and greases should be considered if activities occur in areas where they are present. For most Category B activities in the buildings, however, and all outdoor activities, these materials will pose little if any hazard.

# ABB Environmental Services, Inc.

H&SPLAN.REV

## Safety Hazards

In the absence of significant site activity, such as soil sampling or excavation, there should be very few safety hazards outside of those normally posed by working with the particular tools required for the task. As previously indicated, however, "slip, trip and fall" hazards are present and numerous. Although heat stress is unlikely to be a problem, cold stress may during the winter.

#### Conclusion/Risk Assessment

In general, the overall health risks for Category B activities will be low, based on the limited potential presence for chemical exposure. Work conducted in proximity to other site activities (particularly Category A), however, can significantly alter the nature of or increase the level of hazard.

#### 3.2.3 Protective Measures

Protective measures will be undertaken to minimize the potential health and safety risks for field personnel engaged in Category A activities.

## **Engineering Controls**

Engineering controls will rarely be useful for strictly Category B activities. They may, however, be useful if Category A and B activities are being performed in close proximity and simultaneously.

## Levels of Protection

*Mercury/metals*: The areas where significant mercury or metals is present should not be entered unless, at a minimum, boot covers are used. Depending on the activity, gloves and coated Tyvek may be added. Protection or decontamination of non-disposable equipment also needs to be considered.

Typical levels of personal protection for Category B activities are as follows:

## ABB Environmental Services, Inc.

H&SPLAN.REV
ACTIVITY	<u>TYPICAL</u> Level of Protection
Activities in paved areas	Level D
Simple entry into areas of possible mercury/metals contamination	Level D w/boot covers
Work in areas of known mercury/metals contamination	Modified Level D*

<sup>\*</sup>Modified Level D consists of Level D personal protective equipment, Tyvek coveralls and boot covers. Levels of protection may be changed at the discretion of the HSO, as warranted by a change in site conditions.

#### 3.2.4 Monitoring

When working at or near areas with possibly significant mercury or solvent contamination, initial monitoring of work areas should be conducted. Monitoring can probably be performed only occasionally, based on expected low readings. Action levels and upgrade criteria should duplicate those described in Section 3.1.

Monitoring is not required for Category B activities involving only work in paved areas. Personal air or other sampling is unlikely to be warranted for any Category B activity.

#### 3.3 CATEGORY C - SUBCONTRACTOR OVERSIGHT

Risk potential for this category of activity is highly variable based on the exact nature of the subcontractor's activities. Thorough evaluation of the subcontractor's scope of work and ABB Environmental personnel oversight responsibilities, and pre-work consultation with the subcontractor, are necessary to provide adequate protection. Entry into the exclusion zone while work is being performed generally will require protection equivalent to the subcontractor personnel actually performing the work. Contractor activities may include soil removal and construction. During these activities, ABB Environmental personnel may engage in the following:

#### ABB Environmental Services, Inc.

- video tape and photograph activities;
- evaluate dust and dust control procedures;
- evaluate soil protection measures;
- observe soil/waste removal;
- evaluate other environmental issues; and
- observe job progress

When conducting these activities, ABB Environmental Services personnel will follow certain procedures which are described below.

#### 3.3.1 Hazardous Substances

Because a number of subcontractor activities will consist of actual removal of materials containing or contaminated with hazardous substances, there is a potentially significant risk of exposure to both contractor and ABB Environmental oversight personnel. Due to the disturbance required to remove or abate hazardous materials, significant airborne or dermal hazards may develop.

#### 3.3.2 Site Risks

#### Health Hazards

*Visible mercury*: Contractor operations which involve remediation/removal of mercury (visible or not) clearly pose risks to oversight personnel. Risk will probably be dependent primarily on proximity to the work area, and whether significant dust is produced. Prework inspections of these areas may pose a dermal contact hazard.

*Mercury vapors*: Although mercury vapor levels both in ambient air, and during environmental sampling, have been relatively low, remediation or removal operations in mercury-contaminated areas must be considered likely to produce significant vapor concentrations.

*Solvents and hydrocarbons*: Similar to mercury, work in these areas during remediation/removal may produce vapor levels requiring respiratory protection, and are also likely to produce significant dust. Detailed pre-work inspections may pose dermal or inhalation hazards.

#### ABB Environmental Services, Inc.

H&SPLAN.REV

*Metals (including cyanide)*: Again, potentially significant dermal and/or inhalation hazards may be generated by remediation/removal work.

*Other materials*: Due to oils' and greases' typically low volatility and low contact hazard, remediation/removal of these materials should pose little hazard to oversight personnel.

#### Safety Hazards

Category C activities pose potentially significant safety hazards due to the heavy equipment typically utilized, and the expected nature of the activity (e.g., building demolition). Vehicle/equipment traffic, falling objects, "slip, trip and fall", loud/sustained noise, high-intensity light (e.g., metal cutting), poorly lit areas in building and other hazards associated with heavy construction activity can be expected.

The following general procedures should be followed when ABB Environmental personnel are to conduct inspection activities:

- 1) ABB Environmental personnel must notify the work area foreman of their presence when approaching the work area.
- 2) The foreman will indicate where it is safe for ABB Environmental personnel to conduct inspection activities.
- 3) At all times during inspection, ABB Environmental personnel should be visible to the work area foreman.
- 4) The foreman should be notified when ABB Environmental personnel are to leave the area.

#### Conclusion/Risk Assessment

In general, the overall health and safety risks for Category C activities will be moderate. Although significant contaminant disturbance and construction-related safety hazards will be prominent, ABB Environmental personnel in an oversight/observation role will often be able to reduce the level of hazard without significant effect on their job performance (e.g., by simply moving away and making the foreman aware of their planned activities). However, the level of hazard will also depend significantly on subcontractor work practices, which are typically less easily controlled than our own.

#### **ABB Environmental Services, Inc.**

H&SPLAN.REV

#### 3.3.3 Protective Measures

Protective measures will be undertaken to minimize the potential health and safety risks for field personnel engaged in Category C activities.

#### Engineering Controls

Engineering controls may frequently be useful for Category C activities, but will typically be implemented by the subcontractor. For this reason, it is important for each task analysis to include evaluation of the subcontractor's scope of work and planned control measures. If ABB Environmental is to implement controls, compatibility with the subcontractor's operations should also be considered.

The simplest method of controlling hazard exposure to ABB Environmental personnel is to simply design and conduct oversight/observation activities in order to minimize it. For many operations, it may not be truly necessary for ABB Environmental to be in the active work area or exclusion zone to provide adequate general oversight. Observation can also be limited to the extent that it is truly necessary, rather than being a casual or "general interest" activity. The HSO for each task should initiate discussion between the PM, CM and/or field staff to devise an approach that balances health and safety concerns with oversight requirements.

#### Levels of Protection

Typical levels of personal protection for Category C activities are as follows:

#### **ACTIVITY**

Oversight of activities not involving remediation/removal of hazardous materials

Oversight of hazardous material remediation/removal

#### TYPICAL LEVEL OF PROTECTION

Level D

Modified Level D

#### ABB Environmental Services, Inc.

H&SPLAN.REV

Modified Level D for Category C activities consists of Level D personal protective equipment, Tyvek coveralls and boot covers. Levels of protection may be changed at the discretion of the HSO, as warranted by a change in site conditions.

If vapor levels exceed criteria for upgrade to Level B, and engineering control is not possible, the need for direct oversight/observation should be re-evaluated.

#### 3.3.4 Monitoring

Monitoring of the work environment should be performed whenever remediation/removal of hazardous materials is occurring. Such monitoring can be coordinated or supplied by the subcontractor, but action criteria must, at a minimum, remain consistent with that presented in Section 3.1

Because Category C activities are frequently of significant duration (i.e., weeks or months), the need for personal air monitoring, or urine monitoring for mercury, should be discussed with the HSO and HSM as part of the specific task analysis.

#### 3.4 CATEGORY D - VISITOR TOURS AND MEETINGS

The health and safety risk potential for ABB Environmental personnel engaged in Category D activities is typically low, and is largely based on the presence of concurrent site operations which visitors may want to observe. Nevertheless, Category D activities require a significant focus on health and safety for another reason: based on ABB Environmental's position as general contractor for the site and the persons generally most knowledgeable of site conditions, our personnel have a duty to inform visitors of site hazards and assist in avoiding them.

Procedures designed to ensure visitors are duly informed of potential hazards are presented following the Category D hazard analysis. ABB Environmental personnel must be careful, however, to evaluate the activity visitors are to engage in. For purposes of this HASP, Category D is intended to cover walking tours for general observation, potential subcontractor jobwalks, and other activities which <u>do not</u> involve observation, inspection or other direct involvement with site operations. For example, visitors wishing to observe

#### ABB Environmental Services, Inc.

active remediation/removal of hazardous materials are engaging in Category C activity, for which the guidance of Section 3.3 must be followed.

#### **3.4.1 Hazardous Substances**

In general, Category D activities have limited potential for personnel to be exposed to low concentrations of inorganic compounds.

#### 3.4.2 Site Risks

#### Health Hazards

*Visible mercury*: A frequent target of site tours are areas at which liquid mercury or glass shards are visible. These areas clearly present a dermal and/or inhalation exposure hazard.

*Metals (including cyanide)*: Contact with potentially impacted soil is clearly a possible hazard.

Solvents and hydrocarbons: A dermal or inhalation hazard may exist in the exclusion zone.

#### Safety Hazards

Category D activities will pose few safety hazards to ABB Environmental personnel not already covered in previous sections. Again, "slip, trip and fall" hazards are of greatest concern. Because visits will generally be planned to avoid active site operations, these operations should pose little hazard. Site visitors must follow the same procedures as those described for category C activities for entering work areas.

#### Conclusion/Risk Assessment

In general, the overall health and safety risks to ABB Environmental personnel for Category D activities will be low.

#### **ABB Environmental Services, Inc.**

#### 3.4.3 Protective Measures

Protective measures will be undertaken to minimize the potential health and safety risks for field personnel engaged in Category D activities.

#### Engineering Controls

Engineering controls will rarely be useful or necessary.

Levels of Protection

Typical levels of personal protection for Category D activities are as follows:

# ACTIVITYTYPICAL<br/>LEVEL OF PROTECTIONWalking tours, meetings, other activities in<br/>which areas containing hazardous materials<br/>are not entered.Level DEntering areas containing hazardous<br/>materials. NOTE: ENTRY INTO THESE<br/>AREAS BY VISITORS SHOULD BE<br/>CAREFULLY CONTROLLED.As per activity

#### 3.4.4 Monitoring

Monitoring of the work environment should not be necessary, based on the site knowledge developed to date and the nature of Category D activities.

Because Category D activities are frequently of very short duration, personal monitoring is neither necessary or, in general, feasible.

#### ABB Environmental Services, Inc.

H&SPLAN.REV

3-17

#### 3.5 SPECIAL PROCEDURES FOR SITE VISITORS

This section describes administrative procedures established for site visitors which Combustion Engineering, ABB Environmental and their subcontractors should implement.

[NOTE: As of November, 1995, the Ames Street site is not subject to requirements of 29 CFR 1910.120, the OSHA hazardous waste site operations standard. The standard's requirements, for example training requirements for persons entering "exclusion zones" or "contaminated" areas, are therefore not applicable. Should the Ames Street site become "recognized" as a hazardous waste site by a governmental body, the standard would apply and the procedures described below may require modification.]

#### 3.5.1 Visitor Entry Forms

The following three pages contain forms on which visitor-related procedures are based. ABB Environmental personnel should utilize these forms as follows.

#### "Safety Information and Procedures for Visitors to the Ames Street Site"

This form is intended to provide visitors with basic information about potential health and safety hazards at the site, and provide rules/procedures that visitors are to follow.

It is preferable that visitors review this form prior to arrival at the site. ABB Environmental personnel inviting or scheduling visitors (e.g., potential vendors or subcontractors) should ensure the two-page form is sent/FAXed to them in advance.

Because ABB Environmental personnel will often lead site tours or walkthroughs of specific areas, it is important that the Safety Procedures are known and understood. Of particular importance are rules regarding safety equipment (hard hat, sturdy shoes) and the need for escort at all times.

#### "Visitor Log and Signature Form"

This form serves as a permanent record that visitors were informed of potential hazards, and should be completed and signed by anyone entering the site who is not an ABB Environmental or Combustion Engineering employee or subcontractor.

#### ABB Environmental Services, Inc.

When possible, the "Additional Hazard Information" section should be completed in advance by ABB Environmental, noting hazards due to current site operations and specific to areas the visitors will access. For example, if a tour of glass shard areas is planned, an appropriate note under "Summary of other potential hazards..." would be "Significant concentrations of mercury may be present in the shard areas; dermal protection (boot covers and gloves) required."

#### 3.5.2 Visitor Entry Procedures

Specific procedures for bringing visitors to the site are described below. Compliance with procedures can be verified through use of the "Site Visitor Checklist".

1. Notify visitors in advance of the requirements for site entry, i.e., personal protective gear, attending the safety briefing, traveling with an escort. Send or FAX the "Safety Information..." form (2 pages) prior to their visit.

As representative of the site owner, ABB Environmental may deny entry to persons who are unauthorized, uninvited, or who cannot or refuse to follow safety procedures. Remember, Combustion Engineering and ABB Environmental could be held liable for injuries which occur on the site, regardless of who's at fault! If you are unsure whether to allow entry, contact the PM, CM or HSO.

- 2. Prior to visitor arrival, plan their route through the facility. To the extent possible, plan a route which avoids known or potential hazards. Walk the route yourself and note the location(s) of water, debris and other "slip, trip and fall" hazards. Refamiliarize yourself with areas which may contain hazardous materials and appropriate protective measures.
- 3. Upon arrival, confirm visitors have reviewed the "Safety Information..." and have brought their own personal protective gear.
- 4. Deliver a safety briefing to the visitor(s):
  - a) Briefly review the "Safety Information..." sheet, with particular emphasis on the Safety Procedures.

#### ABB Environmental Services, Inc.

- b) Describe health and safety hazards specific to the area(s) which will be visited. Using the site plan, show the route which will be traveled, noting any physical hazards (dim lighting, water, etc.) Describe hazardous materials which may be present.
- c) Review the MSDSs for chemicals of concern (the MSDS for mercury should ALWAYS be reviewed.) Point out the health hazards and appropriate protective measures.
- d) Describe current site operations, locations and potential hazards. Emphasize that visitors must not interfere in any way with these operations.
- e) *Review any additional protective measures which will be used*. If measures or equipment (e.g., booties, gloves) beyond the standard hard hat and safety shoes are needed to access certain areas, describe their use. Include how to don, how to remove, where to dispose of.
- 5. Have all visitors complete and sign the "Visitor Log and Signature Form".
- 6. Escort visitors at all times. [NOTE: Under certain conditions, visitors can be allowed into the site without an escort. Unescorted visitors must be approved by the PM, CM or HSO.]

#### ABB Environmental Services, Inc.

#### 4.0 DATA SHEETS

### ABB Environmental Services, Inc.

H&SPLAN.REV

.

#### **Sposure** Risk

used for hardening metals such id lead, and as a doping agent in products of silicon and germasalts are used in making hereidenticides, in semiconductor, echnics.

#### roperties

metals; tarnish when exposed in several allotropic modificai50°C; bp 1380°C; sublimes.

#### \_ard

allotropic forms, yellow arsenic human poison. The fatal dose :: -2 mg/kg body weight. All arseinds are toxic, the toxicity varioxidation state of the metal and es. Thus the trivalent inorganic of arsenic, such as arsenic arsenic trioxide, and arsine. 'oxic---more poisonous than the spentavalent salts. The organic inpound Lewisite is a severe blist that can penetrate the skin and ge at the point of exposure. -s used as a poison gas in World

s soluble arsenic sulfide exhibits the toxicity. is absorbed into the body

acute symptoms include fever. al disturbances, irritation of -ry tract, ulceration of the nasal id dermatitis. Chronic exposure

pigmentation of the skin. per ropathy, and degeneration of idneys.

c effects of arsenic are attributed ng properties with sulfur. In mexes with coenzymes. This in-

production of adenosine triphos-"), which is essential for energy abolism. 2,3-Dimercapto-1-prol'antidote against acute intoxicabithioerythritol is reported to be ctive and less toxic antidote 1989). Arsenic is carcinogenic to humans. Instion by an oral route has caused an incased incidence of tumors in the liver, hod, and lungs.

#### twosure Limits

LV-TWA 0.2 mg(As)/m<sup>3</sup> (ACGIH), 0.5 (As)/m<sup>3</sup> (MSHA), 0.01 mg(As)/m<sup>3</sup> (SHA); ceiling 0.002 mg(As)/m<sup>3</sup>/15 min NOSH); carcinogenicity: Human Suffiint Evidence (IARC).

#### **9.6 CHROMIUM**

Symbol Cr; at. wt. 52.00; at. no. 24; CAS [7440-47-3]; a transition metal, contains partially filled d orbitals; exhibits valences from 1 to 6, commonly occurs in +3 and +6 oxidation states

#### **Eus and Exposure Risk**

Chromium is used in the manufacture of its aloys, such as chrome-steel or chromeickel-steel. It is also used for chromeplating of other metals, for tanning leather, and in catalysts. It occurs in chromite ores  $FeO \cdot Cr_2O_3$ ).

#### hysical Properties

Grayish, hard, lustrous metal; density 7.14; welts at 1900°C; boils at 2642°C; reacts with Gute HCl and  $H_2SO_4$ ; attacked by alkalies.

#### lealth Hazard

The toxicity of chromium alloys and com-Nounds varies significantly. Chromium Tetal does not exhibit toxicity. Divalent and trivalent compounds of chromium have a low order of toxicity. Exposure to be dusts of chromite or ferrochrome albys may cause lung diseases, including neumoconiosis and pulmonary fibrosis.

Among all chromium compounds only the hexavalent salts are a prime health hazrd. Occupational exposure to these communds can produce skin ulceration, dermatitis, perforation of the nasal septa, and kidney damage. The examples of hexavalent salts are the chromates and dichromates of sodium, potassium, and other metals. The water-soluble hexavalent chromium salts are absorbed into the bloodstream through inhalation. Many chromium(VI) compounds are carcinogenic, causing lung cancers in animals and humans.

#### **Exposure** Limits

TLV-TWA: chromium metal 0.5 mg/m<sup>3</sup> (ACGIH and MSHA), 1 mg/m<sup>3</sup> (OSHA); Cr(II) and Cr(III) compounds 0.5 mg/m<sup>3</sup> (ACGIH); Cr(VI) compounds, water soluble and certain water insoluble, 0.05 mg/m<sup>3</sup> (ACGIH).

#### 34.7 SELENIUM

Symbol Se; at. wt. 78.96; at. no. 34; valences 2, 4, and 6; CAS [7782-49-2]; a Group VIA metal

#### Uses and Exposure Risk

Selenium is used in the manufacture of colored glass, in photocells, in semiconductors, as a rectifier in radio and television sets, and as a vulcanizing agent in the manufacture of rubber.

#### **Physical Properties**

Dark red to bluish-black amorphous solid, or dark red or gray crystals; exists in several allotropic forms; density 4.26–4.28 for amorphous, 4.26–4.81 for crystals; mp 170 to 217°C (crystals); bp 685°C; amorphous form becomes elastic at 70°C.

#### Health Hazard

The toxicity of selenium and its compounds varies substantially. Sodium selenite is highly toxic; many sulfur compounds of selenium are much less toxic. The target organs are the respiratory tract, liver, kidneys, blood, skin, and eyes. The sign of acute poisoning is a garlic-like odor in the CAC500 CADIA DEL PERRO

Arsenic and its compounds are on the Community Right To Know List.

THR: Poison by most routes. See also ARSENIC COM-POUNDS and SULFIDES. Dangerous fire hazard when exposed to heat or by spontaneous chemical reaction, i.e., in air. Vigorous reaction with oxidizing materials. When heated to decomposition it emits toxic fumes of As.

#### CAC500 **CADIA DEL PERRO**

HR: 3

NIOSH: EU 9500000

26.251.80 11,233,82 158,614,78

193.592.62

28,147,74

PROP: Aqueous extract from the dried leaves of the plant (JNCIAM 46,1131,71).

SYNS: K. IXINA

KRAMERIA IXINA

TOXICITY DATA:	CODEN:
scu-rat TDLo: 300 mg/kg/1Y-1:	JNCIAM 46,1131,71
NEO	
ims-rat TDLo:45 g/kg/1Y-I:ETA	JNCIAM 46,1131,71
skn-ham TDLo: 53950 mg/kg/	JNCIAM 53,1259,74
65W-1:CAR	
scu-rat TD :990 mg/kg/55W-I:	JNCIAM 52,1579,74
NEO	

THR: An experimental carcinogen, tumorigen and neoplastigen. When heated to decomposition it emits acrid smoke and fumes.

#### HR: 3 CAD000 **CADMIUM**

NIOSH: EU 9800000 CAS: 7440-43-9 mf: Cd mw: 112.40

PROP: Hexagonal crystals, silver-white, malleable metal. Mp: 320.9°, bp: 767  $\pm$  2°, d: 8.642, vap press: 1 mm @ 394°.

SYNS:	
C.I. 77180	KADMIUM (GERMAN)
COLLOIDAL CADMIUM	

TOXICITY DATA:	CODEN:
cyt-ham:ovr 1 µmol/L	CGCGBR 26,251,80
orl-rat TDLo: 220 mg/kg (1-22D	TOLED5 11,233,82
preg): REP, TER	
orl-rat TDLo:23 mg/kg (1-22D	PSEBAA 158,614,7
preg):REP,TER	
ivn-rat TDLo: 1250 µg/kg (14D	JJATDK 1,264,81
preg): REP, TER	
ims-rat TDLo:40 mg/kg/4W-I:	JEPTDQ 1(1),51,77
CAR	
ims-rat TD :70 mg/kg:ETA	BJCAAI 18,124,64
ims-rat TD :63 mg/kg:ETA	NATUAS 193,592,6
ims-rat TD:45 mg/kg/4W-I:	NCIUS* PH-43-64-
NEO	886,SEPT,71
ihl-man TCLo:88 μg/m <sup>3</sup> /8.6Y:	AEHLAU 28,147,74
KID	
ihl-hmn LCLo:39 mg/m <sup>3</sup> /20M	AIHAAP 31,180,70
unk-man LDLo:15 mg/kg	85DCAI 2,73,70

orl-rat LD50:225 mg/kg ipr-rat LD50:4 mg/kg scu-rat LD50:9 mg/kg ivn-rat LD50:1800 µg/kg unr-rat LD50:1140 mg/kg orl-mus LD50:890 mg/kg ihl-mus LCLo: 170 mg/m<sup>3</sup> unr-mus LD50:890 mg/kg orl-rbt LDLo:70 mg/kg scu-rbt LDLo:6 mg/kg ivn-rbt LDLo:5 mg/kg ims-ham LDLo:25 mg/kg

TXAPA9 41.667,77 TXAPA9 41,667,77 TXAPA9 41,667,77 JJATDK 1,264,81 GTPZAB 22(5),6,78 41HTAH -,14,78 NT1S\*\* PB158-508 GTPZAB 22(5),6,78 AMPMAR 34,127,73 PROTA\* -,-,55 JOGBAS 35,693,28 NCIUS\* PH-43-64-886

IARC Cancer Review: Animal Sufficient Evidence IMEMDT 11,39,76; IMEMDT 2,74,73. Cadmium and its compounds are on the Community Right To Know List Reported in EPA TSCA Inventory. EPA Genetic Toxicoj. ogy Program.

OSHA PEL: TWA 0.1 mg(Cd)/m<sup>3</sup>; CL 0.6 mg(Cd)/m<sup>3</sup> (fume)

ACGIH TLV: TWA 0.01 mg(Cd)/m<sup>3</sup> (dust). Human car. cinogen

DFG BAT: Blood 1.5 µg/dL; Urine 15 µg/dL

NIOSH REL: (Cadmium) Reduce to lowest feasible level

THR: A human poison by inhalation and possibly other routes. Poison experimentally by ingestion, inhalation. intraperitoneal, subcutaneous, intramuscular, and intravenous routes. In humans inhalation causes an excess of protein in the urine. An experimental carcinogen, tumorigen, neoplastigen, and teratogen. Experimental reproductive effects. Mutagenic data. The dust ignites spontaneously in air and is flammable and explosive when exposed to heat, flame. or by chemical reaction with oxidizing agents; metals; HN3. Zn; Se; and Te. Explodes on contact with hydrazoic acid. Violent or explosive reaction when heated with ammonium nitrate. Vigorous reaction when heated with nitryl fluoride. When heated strongly it emits toxic fumes of Cd. See also CADMIUM COMPOUNDS. For further information. see Vol. 3, No. 5 of DPIM Report.

CAD250	HR: 3
CADMIUM(II) ACETAT	<b>E</b>
CAS: 543-90-8	NIOSH: EU 9810000
DOT: 2570	
mf: $C_2H_4O_2 \cdot 1/2Cd$ m	w: 116.25
PROP: Monoclinic, colorle Mp: 256°, bp: decomp. d:	ess crystals; odor of acetic acid. 2.341.
SYNS:	
ACETIC ACID, CADMIUM SALT	CADMIUM DIACETATE
BIS(ACETOXY)CADMIUM	C.I. 77185
CADMIUM ACETATE (DOT)	
TOXICITY DATA:	CODEN:
cyt-hmn:lym 10 nmol/L	MUREAV 85,236.81
otr-ham: emb 1 µmol/L	CNREA8 39,193.79
dnd-ham:emb 1 µmol/L	CNREA8 39.193.79

664

# GASOLINES: CASINGHEAD

#### GCS

	Floats on wat produ	Floats on water. Flammable, imitating vapor is produced.	
Stop discha Shut off ign Stay upwind Isolate and Notify local	rge if possible. Keep people tition sources and call fire de a and use water spray to kn remove discharged material health and pollution control	away parment lock down'' vapor agencies	
Fire	FLAMMABLE. Flashback along vspor trail may occur. Vspor may explode if gnited in an enclosed area. Extinguish with dry chemical, loam o'r caroon dioxide Water may be ineffective on fire Cool exposed containers with water		
Exposure	CALL FOR MEDICAL AID VAPOR Initialing to eyes, nose and throat. If inhibited will cause dizzness, headache, difficult breathing or loss of cause dizzness. Neadache, difficult breathing or loss of cause dizzness. Nove to treah air ' breathing has stobed, give antificial respiration ' breathing is bitlicul, every everygen LIQUID Irritating to skin and eyes. If swallowed, will Cause nauses or vomtino Fush offected areas will beinit of water Fush SYES, hold evens open and fluon with blenity of water F SVALLOWED and victim is CONSCIOUS have victim unnik water or mik DO NOT 'NDUCE VOMITING		
Water Poilution	HARMFUL TO AQUATIC Fouling to shoreline. May be dangerous if it en Notify local health and w	LIFE IN VERY LOW CONCENTRATIONS.	
(See Respons issue warn Evacuate si Disperse an	e Methoda Handbook) ng-nigh flammability 198 Id flush	2.1 Category: Flammable liquid 2.2 Clase: 3	
3. CHEM 3.1 CG Compatibil Hydrocarbo 3.2 Formula: Not p	ICAL DESIGNATIONS ity Class: Miscellaneous n Mixtures entiment nation: 3.1/1257 57 No: Data not evaluable	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Celor: Coloriess 4.3 Odor: Gasoline	
3.4 DOT ID No.: 12 3.5 CAS Registry	i		

----

	6. FIRE HAZARDS		10. HAZARD ASSESSMENT COOF	
5.1	Fish Point <0"E O C	G	See Hazard Assessment Handbool	k)
5.2	Flammable Limits in Air: 1.3%-7.1%		A-T-U-V-W	· /
1.3	Fire Extinguishing Agents: Dry chemical,			
	foam, or carbon dioxide			
	Used: Water may be ineffective			
8.5	Special Hazards of Combustion		11. HAZARD CLASSIFICATIONS	
	Producta: None	11.1	Code of Federal Regulations:	
5.6	Behavior in Fire: Vapor is heavier than air	11.2	Flammable liquid NAS Mayand Bating for Bulk Wate	
	and may travel a considerable distance to a source of gotton and flesh back		Transportation:	
8.7	Ignition Temperature: Data not available		Category Ri	ating
5.8	Electrical Hazard: Class I, group D		Fire	4
B.9	Burning Rate: 4 mm/mih.		Vapor Imtant	
a. 10	Data not available		Liquid or Solid Imtant	0
6.11	Stoichiometric Air to Fuel Ratio:		Porsons	1
	Data not available		Water Polution	
0.12	Fiame Temperature: Data not aveilable		Aquatic Toxicity	2
			Aesthetic Effect	1
	7. CHEMICAL REACTIVITY		Reactivity	
7.1	Reactivity With Water: No reaction		Water	0
7.2	Reactivity with Common Materials: No		Self Reaction	õ
7.3	Stability During Transport: Stable	11.3	NFPA Hazard Classification:	
7.4	Neutralizing Agents for Acids and		Category Classific	ation
	Cauatics: Not pertinent		Flammability (Red)	1
7.5 7 #	Polymerization: Not pertinent		Reactivity (Yellow)	0
	Not pertinent			•
7.7	Molar Ratic (Reactant to			
	Product): Data not available			1
7.6	Reactivity Group: 33			
		12	BUVELERAL AND CUEMICAL DOON	
		12.	PHISICAL AND CHEMICAL PROP	EKLIES
		12.1	Physical State at 15°C and 1 stm	
		12.2	Molecular Weight: Not pertnent	
		12.3	Bolling Point at 1 atm:	
			58-275°F = 14-135°C = 28	7-408°K
		12.4	Freezing Point: Not pertinent	
	8. WATER POLLUTION	12.5	Critical Pressure: Not pertinent	u
6.1	Aquatic Toxicity:	12.7	Specific Gravity:	
	90 ppm/24 hr/juvenile American		0.871 at 15*C (liquid)	
	shed/TL_/fresh water	12.8	Liquid Surface Tension:	
	shed/TL_/salt water		= 0.019—0.023 N/m at 20°C	
8.2	Watertowi Toxicity: Data not available	12.9	Liquid Water Interfacial Tension:	
8.3	Biological Oxygen Demand (BOD):		49-51 dynes/cm = 0.049-0.0	051 N/m
• •	8%, 5 days Food Chain Concentration Retentiat	12 10	at 20°C	
	None	12.10	Ratio of Specific Heats of Vapor	(Gas):
	ļ		Not pertinent	,
		12.12	Latent Heat of Vaporization:	
			130150 Btu/lb = 7181 call → 3.03.4 × 10 <sup>a</sup> J/kg	/9
		12,13	Heat of Combustion:	u/lb =
			-10,400 cal/g = -435.4 × 10	P J∕kg
		12.14	Heat of Decomposition: Not pertin	nent
	9. SHIPPING INFORMATION	12.15	Heat of Polymerization: Not certin	nent
9.1	Grades of Purity: Composition depends	12.25	Heat of Fusion: Data not available	
	on location of oil well Storage Temperature: Artheot	12.26	Limiting Value: Data not available	
1.3	Inert Atmosphere: No requirement	12.27	Reid Vapor Pressure: Data not av	allable
9.4	Venting: Open (flame arrester) or			
	pressure-vacuum			
	NG	TES		

Mercury and its compounds are on the Community Right To Know List.

NIOSH REL: (To Inorganic Mercury) TWA 0.05 mg(Hg)/m<sup>3</sup>

THR: Poison by intravenous and intramuscular routes. When heated to decomposition it emits very toxic fumes of  $NO_x$ ,  $Na_2O$  and Hg vapors. See also MERCURY COM-POUNDS.

<i>MCV750</i> MERCUROPHYLLINE	HR: 3
CAS: 8012-34-8	NIOSH: OV 8650000

SYNS: mercupurin

MERCUZANTHIN

TOXICITY DATA:	CODEN:
ivn-hmn TDLo:28 mg/kg:CVS	JAMAAP 117,1806,41
ipr-rat LD50:121 mg/kg	THERAP 10,936,55
scu-mus LD50:163 mg(Hg)/kg	JPETAB 105,336,52
ivn-mus LD50:1410 mg/kg	JPETAB 99,149,50
ivn-cat LDLo: 250 mg/kg	JPETAB 99,149,50
ivn-rbt LDLo: 177 mg/kg	JPETAB 99,149,50

Mercury and its compounds are on the Community Right To Know List.

ACGIH TLV: TWA 0.1 mg(Hg)/m<sup>3</sup> (skin)

NIOSH REL: (To Inorganic Mercury) TWA 0.05 mg(Hg)/m<sup>3</sup>

THR: Poison by subcutaneous, intraperitoneal and intravenous routes. Human systemic effects by intravenous route: cardiac arrythmias. When heated to decomposition it emits toxic fumes of Hg. See also MERCURY COMPOUNDS.

MCW000		HR: 3
MERCUROU	JS CHLORIDE	
CAS: 7546-30	)-7	NIOSH: OV 8750000
mf: Cl <sub>2</sub> Hg <sub>2</sub>	mw: 472.09	

PROP: White, odorless, tasteless, heavy powder or crystals. Sublimes @ 400°, d: 7.150. Insol in H<sub>2</sub>O, alc and ether. Protect from light. Sunlight causes it to decomp into mercuric chloride and metallic Hg.

SYNS:	
CALOGREEN	MERCUROCHLORIDE (DUTCH)
CALOMEL	MERCURY(I) CHLORIDE
CALOMELANO (ITALIAN)	MERCURY MONOCHLORIDE
CALOSAN	MERCURY PROTOCHLORIDE
CHLORURE MERCUREUX	MILD MERCURY CHLORIDE
(FRENCH)	PRECIPITE BLANC
C.I. 77764	QUECKSILBER(I)-CHLORID (GER-
CLORURO MERCUROSO (ITAL-	MAN)
LAN)	OUECKSILBER CHLORUER (GER-
CYCLOSAN	MAN)
KALOMEL (GERMAN)	SUBCHLORIDE OF MERCURY
TOXICITY DATA	CODEN

mrc-bcs 50 mmol/L sce-ham:ovr 3200 nmol/L orl-rat LD50: 166 mg/kg CODEN: MUREAV 77,109,80 ENMUDM 7,381,85 GTPZAB 25(7),27,81

skn-rat LD50:1500 mg/kg	GTPZAB 25(7),27,81
orl-mus LD50:180 mg/kg	GTPZAB 25(7),27,81
ipr-mus LD50:10 mg/kg	GTPZAB 25(7),27,81

Mercury and its compounds are on the Community Right To Know List. EPA Genetic Toxicology Program. Reported in EPA TSCA Inventory.

ACGIH TLV: TWA 0.1 mg(Hg)/m<sup>3</sup> (skin) NIOSH REL: (To Inorganic Mercury) TWA 0.05 mg(Hg)/m<sup>3</sup>

THR: Poison by ingestion and intraperitoneal routes. Moderately toxic by skin contact. Mutagenic data. A fungicide. An FDA over-the-counter drug. Incompatible with bromides; iodides; alkali chlorides; sulfates; sulfites; carbonates; hydroxides; lime water; ammonia; golden antimony sulfide; cyanides; copper salts; hydrogen peroxide; iodine; iodoform; lead salts; silver salts; sulfides. When heated to decomposition it emits very toxic fumes of  $CI^-$  and Hg. See also MERCURY COMPOUNDS.

MCW250	HR: 3
MERCURY	
CAS: 7439-97-6	NIOSH: OV 4550000
DOT: 2809	
af: Hg aw: 200.59	

PROP: Silvery, heavy, mobile liquid. A liquid metallic element. Mp:  $-38.89^{\circ}$ , bp:  $356.9^{\circ}$ , d:  $13.534 @ 25^{\circ}$ , vap press:  $2 \times 10^{-3}$  mm @ 25^{\circ}. Solid: tin-white, ductile, malleable mass which can be cut with a knife.

#### SYNS:

COLLOIDAL MERCURY	NCI-C60399
KWIK (DUTCH)	QUECKSILBER (GERMAN)
MERCURE (FRENCH)	QUICK SILVER
MERCURIO (ITALIAN)	RCRA WASTE NUMBER U151
MERCURY, METALLIC (DOT)	RTEC (POLISH)
TOXICITY DATA:	CODEN:
cyt-man-unr 150 µg/m <sup>3</sup>	AEHLAU 34,461,79
ihl-rat TCLo: 890 ng/m <sup>3</sup> /24H	GISAAA 45(3),72,80
(16W male):REP	
ihl-rat TCLo:7440 ng/m <sup>3</sup> /24H	GISAAA 45(3),72,80
(16W male):REP	
ipr-rat TDLo:400 mg/kg/14D-I:	ZEKBAI 61,511,57
ETA	
ihl-man TDL0:44300 µg/m <sup>3</sup> /8H:	JOCMA7 20,532,78
CNS,LIV,MET	
ihl-wmn TCLo:150 $\mu$ g/m <sup>3</sup> /46D:	AEHLAU 33,186,78
CNS,GIT	
skn-man TDLo:129 mg/kg/5H-	DERAAC 172,48,86
C:EAR,CNS,SKN	
ihl-rbt LCLo:29 mg/m <sup>3</sup> /30H	AMIHBC 7,19,53

Mercury and its compounds are on the Community Right To Know List.

OSHA PEL: CL 0.1 mg/m<sup>3</sup> ACGIH TLV: TWA 0.05 mg(Hg)/m<sup>3</sup> (vapor, skin) DFG MAK: 0.01 ppm (0.1 mg/m<sup>3</sup>) NIOSH REL: (To Inorganic Mercury) TWA 0.05 mg(Hg)/m<sup>3</sup>

#### DOT Classification: Corrosive Material; Label: Corrosive

THR: Poison by inhalation. An experimental tumorigen. Corrosive to skin, eyes and mucous membranes. Human systemic effects by inhalation: wakefulness, muscle weakness, anorexia, headache, tinnitus, hypermotility, diarrhea, liver changes, dermatitis, fever. An experimental teratogen. Experimental reproductive effects. Human mutagenic data. Used in dental applications, electronics, and chemical synthesis.

May explode on contact with 3-bromopropyne: alkynes + silver perchlorate; ethylene oxide; lithium; methylsilane + oxygen (explodes when shaken); peroxyformic acid; chlorine dioxide; tetracarbonylnickel + oxygen. May react with ammonia to form an explosive product. Mixtures with methyl azide are shock- and spark-sensitive explosives. The vapor ignites on contact with boron diiodophosphide. Reacts violently with acetylenic compounds (e.g., acetylene, sodium acetylide, 2-butyne-1,4-diol + acid); metals (e.g., aluminum; calcium; potassium; sodium; rubidium; exothermic formation of amalgams);  $Cl_2$ ;  $ClO_2$ ;  $CH_3N_3$ ;  $Na_2C_2$ ; nitromethane. Incompatible with methyl azide: oxidants. When heated to decomposition it emits toxic fumes of Hg. See also MERCURY COMPOUNDS. For further information, see Vol. 1, No. 3 of DPIM Report.

# MCW350 H MERCURY(II) ACETYLIDE CAS: 37297-87-3 mf: C<sub>2</sub>Hg mw: 224.61

Mercury and its compounds are on the Community Right To Know List.

THR: A shock- and heat-sensitive explosive. Upon decomposition it emits toxic fumes of Hg. See also MERCURY COMPOUNDS and ACETYLIDES.

# MCW400 HR: 3 MERCURY(II) aci-DINITROMETHANIDE mm: 410.65

Mercury and its compounds are on the Community Right To Know List.

THR: An explosive detonator. Upon decomposition it emits toxic fumes of Hg and NO<sub>x</sub>. See also MERCURY COM-POUNDS.

#### MCW500 HR: 3 MERCURY AMIDE CHLORIDE

CAS: 10124-48-8 NIOSH: OV 7020000 DOT: 1630

mf:  $ClH_2HgN$  mw: 252.07

#### H<sub>2</sub>NHgCl

PROP: White, pulverized lumps or powder.

#### SYNS:

AMINOMERCURIC CHLORIDE AMMONIATED MERCURY MERCURIC AMMONIUM CHLO-RIDE, SOLID MERCURIC CHLORIDE, AMMONI-ATED MERCURY AMINE CHLORIDE MERCURY AMMONIATED WHITE MERCURY PRECIPITATED WHITE PRECIPITATE

Mercury and its compounds are on the Community Right To Know List. Reported in EPA TSCA Inventory.

ACGIH TLV: TWA 0.1 mg(Hg)/m<sup>3</sup> (skin) NIOSH REL: (To Inorganic Mercury) TWA 0.05 mg(Hg)/m<sup>3</sup>

DOT Classification: Poison B, Label: Poison

THR: A poison. Explosive reaction with halogens or amine metal salts. When heated to decomposition it emits very toxic fumes of  $C1^-$ ,  $NO_x$ , and Hg. See also MERCURY COMPOUNDS.

#### MCW509 MERCURY(II) ORTHOARSENATE

CAS: 7784-37-4 NIOSH: OV 7040000 mf: AsHO<sub>4</sub>•Hg mw: 340.52 DOT: 1623

SYN: MERCURIC ARSENATE (DOT)

Arsenic and its compounds, as well as mercury and its compounds, are on the Community Right To Know List.

OSHA PEL: TWA 0.01 mg(As)/m<sup>3</sup> ACGIH TLV: TWA 0.1 mg(Hg)/m<sup>3</sup> (skin) NIOSH REL: (To Inorganic Mercury) TWA 0.05 mg(Hg)/ m<sup>3</sup>; (To Inorganic Arsenic) CL 0.002 mg(As)/m<sup>3</sup>/15M

DOT Classification: Poison B; Label: Poison

THR: Poison. When heated to decomposition it emits toxic fumes of Hg and As. See also MERCURY COMPOUNDS and ARSENIC COMPOUNDS.

MCX000 MERCURY(I) AZIDE CAS: 38232-63-2

#### HR: 3

HR: 3

AZIDE 3-2 N

NIOSH: OV 7045000

mf: Hg<sub>2</sub>N<sub>6</sub> mw: 485.22

SYNS: mercurous azide (dot)

MERCURY AZIDE

Mercury and its compounds are on the Community Right To Know List.

ACGIH TLV: TWA 0.1 mg(Hg)/m<sup>3</sup> (skin)

#### DOT Classification: Forbidden

THR: Poison. Explodes on heating in air. When heated to decomposition it emits very toxic fumes of  $NO_x$  and Hg. See also AZIDES and MERCURY COMPOUNDS.

HR: 3

## NAPHTHA: VM & P

#### NVM

Light naphtha Painter's nanhtha	ļ		6.1 Flat 6.2 Flat
Fauricer S hapricia	Floats on water.	Flammable, imitating vapor is produced.	6.3 Fire
			d
Stop discha	arge if possible. Keep people av	way	
Shut off ign	ition sources and call fire depaid d and use water soray to "knoc	rtment k down vapor	6.5 Spe
Avoid conta	ect with liquid and vapor		
Notify local	health and pollution control ag	encies.	
	FLAMMABLE.		6.6 Elec
	Vapor may explode if ignite	d in an enclosed area.	6.9 Bur
	Cool exposed containers w	ith water	6.10 Add
Fire			6.11 Sta
			6.12 File
	CALL FOR MEDICAL AID		
	Imitating to eyes, nose and	throat.	7.1 Real
	IT INDEED, will cause dizzine Move to tresh air	na or 1038 OT CONSCIOUSNESS.	
	If breathing has stopped, g If breathing is difficult, give	ive aruncia: respiration oxygen	7.3 Stat
	LIQUID		7.4 Neut
<b>F</b>	imitating to skin and eyes. If swallowed, will cause nau	usea or vomiting.	7.5 Poly
Exposure	Remove contaminated clott Flush affected areas with p	ning and shoes. Jenty of water	7.6 inhii
	IF IN EYES, hold evelids of IF SWALLOWED and victim	pen and "ush with plenty of water n is CONSCIOUS, have victim drink water	7.7 Mot
	OF THIK DO NOT INDUCE VOMITIN	4G	
			/.º Hee
	Effect of low concentrations	s on aquabc life is unknown.	
Water	May be dangerous if it ente	rs water intakes.	
Pollution	Notity local health and wild Notity operators of nearoy	life officials water intakes	
1 0500		2   6851	┥┟───
1. KESPU	e Methods Handhook)	2. LAGEL 2.1 Category: Flammable (incert	
	no-high flammability		8.2 Wat
ISSUE WEITS		2.2 CARGE 3	
Evacuate a	rea		8.3 Biok
Evacuate an Disperse an	rea nd flush	2.2 Classe: 3	8.3 Biok
Evecuate a Disperse an	rea Ind flush	2.2 Class. 3	8.3 Biok
Evacuate al			8.3 Biok
Sue warm Evacuate a Disperse an 3. CHEM	rea nd flush ICAL DESIGNATIONS	4. OBSERVABLE CHARACTERISTICS	8.3 Biok (8.4 Foor (8.4 Foor)
3.1 CG competibil	rea d flush ICAL DESIGNATIONS Hty Class: Miscellaneous o Matures	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Colories	8.3 Biok ( 8.4 Foor )
3. CHEM 3. CHEM 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not e	rea diffush ICAL DESIGNATIONS Itty Class: Miscallaneous Applicable	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odor: Like gasoline	8.3 Biok 0 8.4 Foor N
3. CHEM 3. CHEM 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not 8 3.3 MBO/UN Desig 4.4 DOT ID No. 12	rea di flush ICAL DESIGNATIONS Hty Class: Miscallaneous n Motures applicable nation: 3.2/1255 255	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odor: Like gasoline	8.3 Biok C 8.4 Foor
3. CHEM Evecuses a Disperse ar 3. CHEM 11 CG Competibil Hydrocarbo 3.2 Formula: Not 8 3.3 HbO/UN Desig 3.4 DDT ID No.: 12 3.5 CAS Registry i	rea Nd flush ICAL DESIGNATIONS Hty Class: Miscellaneous n Motures spplicable nation: 3.2/1255 255 No.: Data not available	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odor: Like gasoline	8.3 Biok C 8.4 Fool N
3. CHEM Evecuste a Disperse ar 3. CHEM 3.1 CG Compatibil Hydrocarbo 3.2 Formula: Not 8 3.1 MO/UN Desig 3.4 DOT ID Not: 12 3.5 CAS Regiatry 1	rea dd flush ICAL DESIGNATIONS Ity Clase: Miscellaneous n Motures sppicable nation: 3.21255 255 No: Data not available	4. DBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odor: Like gasoline	8.3 Biok C 8.4 Fool P
3. CHEM Evacuate a Disperse ar 3. CHEM 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not a 3.3 HMO/UN Desig 3.4 DOT ID No.: 12 3.5 CAS Registry i 5.1 Personal Pro-	rea dd flush ICAL DESIGNATIONS Hty Clase: Miscellaneous n Motures spokcable nation: 3.2/1255 255 No: Data not available S. HEA tective Equipment: Gooples or	4. OBSERVABLE CHARACTERISTICS 4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odor: Like gasoline LTH HAZARDS face sheld (as for gasoline).	8.3 Biok C 8.4 Food P
3. CHEM Evecusite a Disperse ar 3. CHEM 11 CG Competibil Hydrocarbo 3.2 Formula: Not a 3.1 MAO/UN Desig 3.4 DOT ID No.: 12 3.5 CAS Registry 1 5.1 Personal Prot 5.2 Symptoms Fo	rea Ad flush ICAL DESIGNATIONS Hty Clase: Miscellaneous n Motures spokcable mation: 3.2/1255 255 No.: Data not available 5. HEA tective Equipment: Goggles or palowing Exposure: Vapo rmta	4. OBSERVABLE CHARACTERISTICS 4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odor: Like gasoline 4.3 Odor: Like gasoline LTH HAZARDS face shield (as for gasoline). tes respiratory tract, causes coughing and mid	8.3 Biok C 8.4 Fool 8 9.1 Gree ((
3. CHEMI Evecusite a Disperse ar 3. CHEMI 3.1 CG Compatibil Hydrocarbo 3.2 Formula: Not a 3.1 M60/UN Desig 3.4 DOT ID No:: 12 3.5 CAS Regiatry i 5.1 Personal Proj 5.2 Symptoms Fo depression.	rea Ad flush ICAL DESIGNATIONS Hty Clase: Miscellaneous n Motures spokable nation: 3.2/1255 255 No.: Data not available 5. HEA tective Equipment: Goggles or slowing Exposure: Vapor mits Asprabon causes severe lung:	4. DBSERVABLE CHARACTERISTICS     4.1 Physical State (as shipped): Liquid     4.2 Color: Coloriess     4.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mid     mitation with coughing, gagging, and rapidly     learning to stortect.	8.3 Biok 0 8.4 Fool 9 9.1 Gree (1) 0 0 0 0 0 0 0 0 0 0 0 0 0
3. CHEMI 3. CHEMI 3. CHEMI 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not 8 3.1 MO/UN Desig 3.4 DOT ID Not: 12 3.5 CAS Registry 1 5.1 Personal Prof 5.2 Symptoms Fo depression. developing 1 swelling of 1	rea Ad flush ICAL DESIGNATIONS Ity Clase: Miscellaneous n Motures spplicable nation: 3.2/1255 255 No.: Data not available S. HEA tective Equipment: Goggles or slowing Exposure: Vapor imta Aspreson causes severe lung pulmonary edema. Ingestion imt Aspreson causes severe lung	4. OBSERVABLE CHARACTERISTICS     4. OBSERVABLE CHARACTERISTICS     4.1 Physical State (as shipped): Liquid     4.2 Color: Coloriess     4.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tises respiratory tract, causes coughing and mid emission with coughing, gagging, and rapidly tates mouth and stomach, causes nausea, vomiting,	8.3 Biok 2.8.4 Food 8.4 Food 9.1 Grac (1) 9.1 Grac (2) 9.2 Stor 9.3 Insert
3. CHEM Evecuses a Disperse ar 3. CHEM 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not 8 3.1 M60/UN Desig 3.4 DOT ID Not: 12 3.5 CAS Registry 1 5.1 Personal Prot 5.2 Symptoms Fo depression. developing 1 swelling of 1 S.3 Treetment of	rea to d flush ICAL DESIGNATIONS Ity Ctase: Miscallaneous n Morures spplicable nation: 3.2/1255 255 No: Data not evalable 5. HEA tective Equipment: Goggles or allowing Exposure: Vapor Mageston um Asprabon causes severe lung pulmonary edema. Ingeston um Exposure: INHALATION: main Exposure: INHALATION: main	4. OBSERVABLE CHARACTERISTICS     4. Physical State (as shipped): Liquid     4.2 Color: Coloriess     4.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mid emitation with coughing, gagging, and rapidly tates mouth and stomach, causes nauses, vombing, taten respiration if required. INGESTION: do NOT	8.3 Biok C 8.4 Foor P 9.1 Gree (( 9.2 Stor 9.3 Inser 9.4 Vent
3. CHEM Evecuses a Disperse ar 3. CHEM 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not a 3.1 M60/UN Desig 3.4 DOT ID No:: 12 3.5 CAS Registry i 5.1 Personal Prot 5.2 Symptoms Fo depression. developing of swelling of i swelling of i S.3 Treetment of mduce vom	rea to d flush ICAL DESIGNATIONS INY Class: Miscellaneous n Motures applicable nation: 3.2/1255 255 No: Data not available 5. HEA sective Equipment: Goggles or allowing Exposure: Vapor enta Asprabo causes severe lung pulmonary edema. Ingestion ren bidomen, caracta strifythmasa. Exposure: INHALATION: main tang; observe for pneumonic ta k enforce rest; administer own	4. OBSERVABLE CHARACTERISTICS     4. Physical State (as shipped); Liquid     4.2 Color: Coloriess     4.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mid     mitabon with coughing, gagging, and rapidly     tates mouth and stomach, causes nauses, vomiting,     tatan respiration if required. INGESTION: do NOT     pport in central nervous system depression occurs.     pen; call a doctor. EVES: float with weller for at least	8.3 Biok C 8.4 Foor P 9.1 Grac (1) 9.2 Stor 9.3 Linar 9.4 Vent 9.4 Vent
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Compatibil Hydrocarbo 3.2 Formula: Not e 3.3 HBO/UN Desig 3.4 DOT 10 No.: 13 3.5 CAS Registry 1 5.1 Personal Prot 5.2 Symptoms Fc depression. developing of swelling of i 5.3 Treetment of mduce vom ASPIRATIO 15 mn. Ski	rea and flush ICAL DESIGNATIONS INY Class: Miscallaneous n Motures applicable nation: 3.2/1255 255 No: Data not available 5. HEA tective Equipment: Goggles or silowing Exposure: Vapor mita abdomen, cardae antrythmas. Exposure: INHALATION: main abdomen, cardae antrythmas. Exposure: INHALATION: main ting; observe for presummers as netrors rest; administer oxy; N: wipe off, waan with soap and	4. OBSERVABLE CHARACTERISTICS     4. OBSERVABLE CHARACTERISTICS     4.1 Physical State (as shipped); Liquid     4.2 Color: Coloriess     4.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mid     mitabon with coughing, gagging, and rapidly     tates mouth and stomach, causes nauses, vomiting,     tates mouth and stomach, causes nauses, vomiting,     tates mouth and stomach, causes nauses, vomiting,     tates respiratory tract, causes coughing and mid     mitabon with coughing, gagging, and rapidly     tates mouth and stomach, causes nauses, vomiting,     tates respiratory tract, causes coughing and mid     mitabon with coughing, gagging, and rapidly     tates mouth and stomach, causes nauses, vomiting,     tates mouth,	8.3 Biok C 8.4 Foor 8.4 9.0 9.1 Grac (1) 9.3 Stor 9.3 Inert 9.4 Vert 9.4 Vert 9.4 Vert
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Compatibility Hydrocarbo 3.2 Formula: Not a 3.3 NBO/UN Desig 3.4 DOT ID No.: 12 3.5 CAS Registry i 5.1 Personal Proj 5.2 Symptoma Fc depression. developing i swelling of i 5.3 Treetment of nduce vom ASPIRATIO 15 mm. SKI 5.4 Threshold Lin	rea diffush ICAL DESIGNATIONS ItY Class: Miscellaneous n Motures applicable nation: 3.2/1255 255 No:: Data not available S. HEA tective Equipment: Googles or slowing Exposure: Vapor mits abdomen, cardae arrhythmes. Represent: INHALATION: main rang: observe for pneumona; a N: enforce rest; edminister oxy; enforce rest; edminister oxy; infly care rest; edminister oxy; infly wap off, waan with soap and ntf Value: Data.eot.available	Classe: 3     A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mid     mitaon with oughing, gagoing, and rapidly     tates mouth and stomach, causes nausea, vombing,     tates respirator if required. INGESTION: do NOT     pport if central nervous system depression occurs.     jen; call a doctor. EYES: flush with water for at least     dismon	8.3 Biok C 8.4 Foor 9.1 Grac (1 9.2 Stor 9.3 Inert 9.4 Vert 9.4 Vert 9.7 Vert
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Compatibility Hydrocarbo 3.2 Formula: Not a 3.3 HHO/UN Desig 3.4 DOT ID No.: 12 3.5 CAS Registry i 5.1 Personal Prof. 5.2 Symptoms FC depression. developing i swelling of 0 15 Treetment of mduce vom ASPIRATIO 15 Short Term if 5.5 Short Term if 6.6 Toxicitry by in	rea diffush ICAL DESIGNATIONS Hty Class: Miscellaneous n Motures spplicable nation: 3.2/1255 255 No.: Data not available S. HEA tective Equipment: Goggles or slowing Exposure: Vapor mis- abdomen, cardac arrhythmea. N: enforce rest, edminister oxy; N: enforce rest, edminister oxy; hilde: Data.not.available malation Limits; 500 ppm for 3 gestion: Grade 1: LDac = 5 to	Classe: 3     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     Ise respiratory tract, causes coughing and mid     mitation with coughing, gasging, and rapidly     tates mouth and stomach, causes nausea, vombing,     to only the cause of	8.3 Biok (8.4 Foor 8.4 Foor 9.1 Grac (1) 9.2 Stor 9.3 Inert 9.4 Vert 9.4 Vert 9.4 Vert
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Compatibil Hydrocarbo 3.2 Formula: Not a 3.3 HMO/UN Desig 3.4 DOT ID No12 3.5 CAS Registry i 5.1 Personal Prof 5.2 Symptoms FC depression. developing i swelling of i swelling of i swelling of i S.3 Treetment of mduce vom ASPIRATIOL 15.5 Short Term in 5.6 Toxicity by in 5.7 Late Toxicity by in	rea diffush ICAL DESIGNATIONS Ity Class: Miscallaneous n Motures spplicable nation: 3.2/1255 255 No:: Data not available <b>5.</b> HEA tective Equipment: Goggles or slowing Exposure: Vapor mits abdomen, cardac arrhythmea. Exposure: Index shorthythmea baborne, cardac arrhythmea N: enforce rest; edminister oxp an mit Value: Data.not.available walation Limits: 500 ppm for 3 gestion: Grade 1: LDae = 5 to None	Class: 3     A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face stiveid (as for gasoline).     Ise resperatory tract, causes coughing and mid     mitabon with coughing, gagging, and rapidly     taser mouth and stomach, causes auseau vomiting,     taser respiration if required. INGESTION: do NOT     apport if central nervous system depression occurs.     ent. all a doctor. EYES: flush with water for at least     di water.         3000 ppm         15 g/kg	8.3 Biok (8.4 Foor 8.4 Foor 9.1 Gree (1) 9.2 Stor 9.3 Inert 9.4 Vent 9.4 Vent
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Competibil Hydrocarbo 3.2 Formula: Not a 3.3 HMO/UN Desig 3.4 DDT ID No.: 12 3.5 CAS Registry i 5.1 Personal Prof 5.2 Symptoms FC depression. developing i swelling of d swelling of d nduce vom ASPIRAT.02 15.5 Short Term (r 5.6 Toxicity by In 5.7 Late Toxicity by In 5.7 Late Toxicity by In 5.9 Short Cash () Support Cas	rea diffush ICAL DESIGNATIONS Ity Class: Miscallaneous n Motures splicable nation: 3.2/1255 255 No:: Data not available <b>5.</b> HEA tective Equipment: Goggles or slowing Espoaurs: Vapor mits abdomen, cardac arrhytmisa. N: enforce rest; edminister oxy; numorary adema. Ingestion im abdomen, cardac arrhytmisa. N: enforce rest; edminister oxy; walation Limits: 500 ppm for 3 gestion: Grade 1: Lbis = 5 to : None rritant Characteristics: Vapors	Class: 3     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes resperatory tract, causes coupling and med     imitation with coupling, gagging, and rapidly     tates mouth and stomach, causes nueses ausedu     trates mouth and stomach, causes nueses withouth     to an stomach, causes nueses over the store of the set of water.         30 on, EYES: flush with water for at least         d water.         30 on, ppm         15 g/kg         reause a slight smarting of the eyes or respiratory         be affect is temporary.	8.3 Biok (8.4 Foor 8.4 Foor 9.1 Grac (1) 9.2 Stor 9.3 Inser 9.4 Vent 9.4 Vent 0
3. CHEM Evecusite a Disperse ar Japarse ar Japarse ar Compatibility Hydrocarbo Japarse ar Japarse a	rea diffush ICAL DESIGNATIONS Ity Class: Miscallaneous n Motures splicable nation: 3.2/1255 255 No: Data not available <b>5.</b> HEA tective Equipment: Goggles or slowing Exposure: Vapor mits abdomen, cardac arrhytmisa. Represure: Vapor mits abdomen, cardac arrhytmisa. N: enforce rest; edminister oxyg multimolary edema. Ingestion int abdomen, cardac arrhytmisa. N: enforce rest; edminister oxyg multimolary edema. Ingestion int abdomen, cardac arrhytmisa. N: enforce rest; edminister oxyg multimolary edema. Ingestion int ang: observe for pneumona; sa N: enforce rest; edminister oxyg meator in the concentrations. Th d Irritant Characteristics: Mano	Class: 3     A. OBSERVABLE CHARACTERISTICS     A. Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes resperatory tract, causes coughing and med     imitabon with coughing, gagging, and rapidly     tates mouth and stomach, causes nueses nueses. vomting,     taan resperatory tract, causes coughing and med     imitabon with coughing, gagging, and rapidly     taan resperatory tract, causes coughing and med     imitabon with coughing, gagging, and rapidly     taan resperatory tract, causes coughing and med     imitabon with coughing, gagging, and rapidly     taan resperatory for a loss of the set of water.     3000 ppm     15 g/kg     icause a slight smarting of the eyes or respiratory     we affect is temporary.     imum hazard, if spiled on clothing and allowed to	8.3 Biok ( 8.4 Foor 8.4 Foor 9.3 Inset 9.2 Stor 9.3 Inset 9.4 Vent D
3. CHEM     3. CHEM     3. CHEM     3.1 CG Competibil     Hydrocarbo     3.2 Formula: Not a     3.3 HMO/UN Desig     3.4 DOT ID Not. 12     5.1 Personal Prof     5.2 Symptoms Fc     depression.     devices on depression.     Solution of the set of the depression.     Solution of the depression.	rea diffush ICAL DESIGNATIONS Ity Class: Miscellaneous in Motures splicable pation: 3.2/1255 255 No:: Data not available S. HEA tective Equipment: Goggles or slowing Esposure: Vapor mita Asprabon causes severe lung pulmonary adema. Ingestion im abdomen, cardac arrhytmisa. Ris enforce rest; administre oxy N: wipe off, waan with soap and N: enforce rest; administre oxy N: wipe off, waan with soap and N: who off, waan with soap and sector in high concentrations. Th d Inftant Characteristics: Manor r cause smarting and reddening ski: Data not evaluable	Class: 3     A. OBSERVABLE CHARACTERISTICS     A. Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mad     mitation with coughing, gagoing, and rapidly     tates mouth and stomach, causes neuses, vorming,     tates a tenta inservour system depression occurs,     port (if certai nervour system depression occurs,     to valer,	8.3 Biok ( 8.4 Foo 8.4 Foo 9.1 Gree () 9.2 Stor 9.3 Iner 9.4 Vent 9.4 Vent 9.4 Vent
3. CHEM     3. CHEM     3. CHEM     3. CHEM     3.1 CG Competibil     Hydricarbo     3.2 Formula: Not a     3.3 HMO/UN Dealg     3.4 DDT ID No12     3.5 CAS Registry i     5.1 Personal Proi     5.2 Symptoms FC     depression.     developing i     swelling of i     swelling o	rea and flush ICAL DESIGNATIONS Ity Class: Miscallaneous n Motures spplicable nation: 3.2/1255 255 No:: Data not available <b>5.</b> HEA tective Equipment: Goggles or slowing Exposure: Vapor mits abdomen, cardac arrhytmisa. Represure: Vapor mits abdomen, cardac arrhytmisa. N: enforce rest; edminister oxp an mit Value: Data-not-available wealston Limits: 500 ppm for 3 (sinone mit Value: Data-not-available is none mit take: Characteristics: Vapor sent in mpt concentrations. Th di Initiant Characteristics: Mini r cause sinarting and reddening ski: Data not evailable 0.000 ppm	Class: 3     A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline),     tes resperatory tract, causes coughing and mid     imitation with coughing, gagging, and rapidly     tates mouth and stomach, causes nueses, working,     tash respiration if required. INGESTION: do NOT     upport if central nervous system depression occurs,     en; call a doctor. EYES: flush with water for at least     di water.         30 mp         15 g/kg     icause a slight smarting of the eyes or respiratory     we affect is temporary,     mum hazard. If spiled on clothing and allowed to     of the skin.	8.3 Biok ( 8.4 Foor 8.4 Foor 9.1 Gree () 9.1 Gree () 9.2 Stor 9.3 Iner 9.4 Vent 9.4 Vent
3. CHEM     3. CHEM     3. CHEM     3. CHEM     3. CHEM     4. CG Competible     Hydrocarbo     3.2 Formula: Not a     3.3 HMO/UN Design     4.4 DOT ID No.: 12     3.5 CAS Registry i     5.1 Personal Proi     5.2 Symptoms FC     depression.     developing i     swelling of c     swelling to c	rea and flush ICAL DESIGNATIONS ItY Class: Miscellaneous in Motures applicable nation: 3.2/1255 255 No:: Data not available <b>5.</b> HEA tective Equipment: Goggles or slowing Exposure: Vapor mits abdomen, cardae arrhythmes. Exposure: INHALATION: main abdomen, cardae arrhythmest N: enforce rest; administer oxyg petitonary edema. Ingestion im abdomestics i administer oxyg petitonary ender i administer oxyg neatoon Limits: 500 ppm for 3 gestion: Grade 1: LDice = 5 to : None mitant Characteristics: Vapors esent in high concentrations. Tr d Initiant Characteristics: Mini- reause size and and reddening bid: Data not available 0,000 ppm	Classe: 3     A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     Ise respiratory tract, causes coughing and mid     mitation with oughing, agging, and rapidly     takes mouth and stomach, causes nausea, vombing,     to cause a slight simple of the eyes or respiratory     the affect is temporary,     mum hazard, if spilled on clothing and allowed to     of the skin.	8.3 Biok ( 8.4 Foor 8.4 Foor 9.1 Grac () 9.1 Grac () 9.2 Stor 9.3 Inert 9.4 Vert p
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Compatibility Hydrocarbo 3.2 Formula: Not e 3.3 HMO/UN Desig 3.4 DOT ID No:: 12 3.5 CAS Registry 1 5.1 Personal Proto devisioning of 3.5 CAS Registry 1 5.2 Symptoms FC devisioning of 3.5 Treetment of mduce vom ASPIRATIO 15 Threetment of mduce vom ASPIRATIO 15 Short Term (r 5.6 Toxicity by In 5.7 Late Toxicity; 5.8 Vapor (Gae) I system (r) 5.9 Liquid or Soli remain, may 5.10 Odor Threetho 5.11 IDLH Value: 1	rea rea diffush ICAL DESIGNATIONS Ity Class: Miscallaneous n Morures applicable nation: 3.2/1255 255 No: Data not available <b>5.</b> HEA tective Equipment: Googles or allowing Exposurs: Vapor inta abdomen, carbas severe iurg pulmonary edema. Ingestion im- bidomen, carbas severe iurg pulmonary edema. Ingestion im- bidomen, carbas severe iurg pulmonary edema. Ingestion im- stance i administer oxy N: wipe off, waan with soap and mit Value: Data not available Neators Grade 1: LDss = 5 to None values in high concentrations. The d Initiant Characteristics: Vapors esent in high concentrations. The d Initiant Characteristics: Mapor cause smaring and reddening bid: Data not available 0,000 ppm	A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A. Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline).     Iser respiratory tract, causes coughing and mid     mitason with coughing, pagging, and rapidly     tates mouth and stomach, causes nausea, voming,     tates,     addition decore, EYES; flush with water for at least     di water,     addition,     bits g/kg	8.3 Biok C 8.4 Foor N 9.1 Grac (( 9.3 Inert 9.3 Inert 9.4 Vert 9.4 Vert
3. CHEM     Subset water     Evecute a     Disperse ar     Disperse ar     Subset and a subset of the subset	rea rea diffush ICAL DESIGNATIONS Ity Class: Miscellaneous n Mortures applicable nation: 3.2/1255 255 No: Data not available S. HEA tective Equipment: Gogoles or allowing Exposure: Vapor anta Asprabon causes severe lung pulmonary edema. Ingestion im domain, cause severe lung pulmonary edema. Ingestion im tang: observe for pneumona: sa Exposure: INHALATION: main tang: observe for pneumona: sa Ke enforce reat: administer oxy N: wipe off, waan with soap are mit Value: Data not-available is None mit Value: Data not-available is none mit cause smarting and reddening bit: Data not available 0.000 ppm	A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A. Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     face shield (as for gasoline).     tes respiratory tract, causes coughing and mid     emission with coughing, gagging, and rapidly     taites mouth and stomach, causes nausea. vomting,     tain respiratory tract, causes coughing and mid     emission with coughing, gagging, and rapidly     taites mouth and stomach, causes nausea. vomting,     tain respiratory tract, causes coughing and mid     emission with coughing, gagging, and rapidly     taites mouth and stomach, causes nausea. vomting,     taites mouth and stomach, causes nausea. vomting,     taites doctor. EYES: flush with water for at least     givater.     30 mm.     15 g/kg     is gause a slight smarting of the eyes or respiratory     the affect is temporary.     mum hazard, if spiled on clothing and allowed to     of the slun.	8.3 Biok C 8.4 Foor P 9.1 Gree (( 9.2 Stor 9.3 Inser 9.4 Vent p
3. CHEM Evecusite a Disperse ar 3. CHEM 3.1 CG Compatibilit Hydrocarbo 3.2 Formula: Not e 3.3 HBO/UN Desig 4. DOT 10 No.3 13 5.5 CAS Registry 1 5.1 Personal Prot 5.2 Symptoms Fc depression. developing of 0 3.5 CAS Registry 1 5.7 Treetment of mduce vom ASPIRATIO 15 min. Ski 5.7 Late Taxicity; 5.8 Yapor (Gas) 1 5.9 Liquid or Soli remain, may 5.10 Odor Thresho 5.11 IDLH Value: 1	rea Ad flush ICAL DESIGNATIONS INY Class: Miscellaneous n Mortures applicable nation: 3.2/1255 255 No: Data not available <b>5.</b> HEA sective Equipment: Googles or allowing Exposure: Vapor enta abdomen, carbas severe lung pulmonary edema. Ingestion en- tabodome, carbas severe lung pulmonary edema. Ingestion en- ting: observe for pneumona; sa Exposure: INHALATION: main stag; observe for pneumona; sa Ke enforce rest; administer oxy N: wipe off, waan with soap and mit Value: Data not available None water that: Soo poin for 3 sesent in high concentrations. Th d Irritant Characteristica: Vapors esent in high concentrations. The d Irritant Characteristica: Vapors esent in high concentrations. The d Irritant Characteristica: Vapors esent in high concentrations. The d Irritant Characteristica: Mapor (Jung) (Jung) (Jung) (Jung) (Jung) Michael (Jung) (Jung) (Jung) (Jung) Michael (Jung) (Jung) (Jung) (Jung) (Jung) Michael (Jung) (Jun	A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A.1 Physical State (as shipped): Liquid     A.2 Color: Coloriess     A.3 Odor: Like gasoline     LTH HAZARDS     If a color: Colories coughing and mid     mitation with coughing, gagging, and rapidly     taites mouth and stomach, causes nausea. vomting,     tain respiration if required. INGESTION: do NOT     toport if central nervous system depression occurs.     Jen; call a doctor. EYES: flush with water for at least     di water.     30 mm.     115 g/kg     is cause a slight smarting of the eyes or respiratory     be effect is temporary.     mum hazard, if splied on clothing and allowed to     of the skin.	8.3 Biok C 8.4 Foor 8.4 Foor 9.1 Grae (( 9.2 Stor 9.3 Iner 9.4 Vent 9.4 Vent 9.4 Vent 9.4 Vent

E HAZAROS	10. HAZARD ASSESSMENT CODE
20-55°F C.C.	(See Hazard Assessment Handbook)
imits in Air: 0.9%-6.7%	A-T-U-V-W
shing Agenta: Foam, carbon	
ary chemical shing Agents Not to be	
e may be ineffective	11. HAZARD CLASSIFICATIONS
s of Combustion	11.1 Code of Easteral Regulations:
	Fiammable liquid
av long distances to a source	11.2 NAS Hazard Rating for Bulk Water
and flash back.	Transportation: Not issed
perature: 450°F	Category Classification
r: 4 mm/mm.	Health Hazard (Blue) 1
ime Temperatura:	Flemmability (Red)
variable	Heactivity (Yellow)
vallable	
erature: Data not available	
ICAL REACTIVITY	
th Water: No reaction	
h Common Materials: No	
Transmert Stable	
Agenta for Acids and	
Not pertinent	
n: Not pertinent	1
olymerization: Int	
Reactant to	
Data not available	
oup: 33	
	12 PRISICAL AND CHEMICAL PROPERTIES
	12.1 Physical State at 15°C and 1 stm:
	12.2 Molecular Weight; Not pertinent
	12.3 Boiling Point at 1 atm:
	200-300°F = 93-149°C =
	366-422"K 12.4 Freezing Point: Not pertnent
ER POLLUTION	12.5 Critical Temperature: Not perment
city: Data not available	12.6 Critical Pressure: Not persnent
sticity: Data not available	12.7 Specific Gravity: 0.75 at 2010 (ligarit)
valabie	12.8 Liquid Surface Tension:
Concentration Potential:	19-23 dynes/cm = 0.019-0.023 N/m
	at 20°C
	39-51 dynes/cm = 0.0390.051 N/m
	at 20°C.
	12.10 Vapor (Gas) Specific Gravity:
	Uata not available 12.11 Ratio of Specific Heats of Venor (Gen)
	(est.) 1.030
	12.12 Latent Heat of Veporization:
	130-150 Btu/lb = 7161 ca//o = 3.03.4 X 104 1/kg
<u> </u>	12.13 Heat of Combustion: (est.)
ING INFORMATION	
rity: Petroleum hydrocarbons	424 X 10* J/kg 12.14 Heat of Decomposition: Not parament
aromatic hydrocarbons such	12.15 Heat of Solution: Not pertnent
and tokuene (10%)	12.16 Heat of Polymerization: Not pertinent
nereture: Ambient here: No requirement	12.25 Heat of Fusion: Data not available
(flame arrester) or	12.27 Reid Vapor Pressure: 0.12 pea
icuum	
	//23

# OILS, FUEL: 6

C.00	mmon Synony	ms	Thick liquid	Black Ter odor		6. FIRE HAZARDS	10. HAZARD ASSESSMENT CODE
Bunker ( Residual	Coll Ifueloi No.6		Usually floats on v	nater.	6.1 6.2 6.3	Flesh Point: > 150°F C.C. Flemmable Limits in Alt: 1%-5% Fire Extinguishing Agents: Dry chemical, foam, or cerbon dioxide Fire Extinguishing Agents had to be	(See Hazard Assessment Handbook) A-T-U
	Stop discharg Call fire depa Avoid contac leolate and ri Notify local h	ge if possit artment. It with liquid emove disc lealth and (	ole. J. charged matenal. poliution control age	nces.	67 64	Used: Water may be ineffective Special Hiszards of Combustion Products: Not pertinent Behavior in Fine: Not pertinent ignition Temperature: 785'F Electrical Hiszard: Not pertinent	11. HAZARD CLASSIFICATIONS     11.1 Code of Federal Regulations: Combustble liquid     11.2 NAS Hazard Rating for Bulk Water Transportation: Not listed     11.3 MERA hazard Castleration
	Fire	Combust Extingui Water m Cool ext	ble. ah with dry chemical ary be ineffective on posed containers wit	l, foam or carbon deoxide. , fre th water,	6.9 6.1 6.1	Burning Rate 4 mm/mm. 9 Addebatic Flores Temperature: Data not available 1 Stoichloometric Air to Fuel Ratio: Data not available 2 Flores Temperature: Data not available	Category Cleastification Health Hazard (Blue)
Exp	oosure'	CALL FC LIQUID Initiating Harmful Faush at IF IN EY IF SWAI OF NOT	DR MEDICAL AID. to skin and eyes. If swalkowed. to contaminated cloth fected areas with pi ceck, hold eyends op LLOWED and victim mik. T INDUCE VOMITING	ing and shoes. enty of water, en and flugh with bienty of weter, is CONSCIOUS, have victim drink water G.	7.1 7.2 7.3 7.4 7.5 7.6 7.7 7.8	7. CHEMICAL REACTIVITY Reactivity With Water: No reaction Reactivity with Common Materials: No reaction Stability During Transport: Stable Neutralizing Agents for Acids and Cassifics: Not partment Polymertsation: Not partment Inhibitor of Polymertsation: Not partment Moter Ratio (Reactant to Product): Data not evaluable Reactivity Group: 33	· · · ·
W Pol	ater lution	Dangero Fouling t May be o Notify lo	us to aquatic life in h o shoreline. Jangarous if it entan ical health and wildli	high concentrations.			12. PHYSICAL AND CHEMICAL PROPERTIES     12.1 Physical State at 15°C and 1 stm: Liquid     12.2 Molecular Weight: Not pertnemt     12.3 Boiling Point at 1 atm: 415—>> 1083°F
(1	1. RESPON See Response Mechanical of Should be rer Chemical and	ISE TO DIS Methode ( ontainment moved ( physical to	parators of nearby w iCHARGE tandbook)	2. LABEL 2.1 Category: None 2.2 Class: Not pertinent	6.1 6.2 8.3	WATER POLLUTION     Aquatic Texticity:     2400 ppm/48 hr/juvanile American     shed/TL_/tesh vester     2417 mg/l/48 hr/juvanile American     shed/TL_/calt vester     Waterhowi Texticity: Data not available     Biological Oxygen Demand (BOD);	= 212->>588'C = 485->>861'K 12.4 Freezing Point: 25 to 55'F = -4 to + 13'C = 289 to 286'K 12.5 Critical Temperature: Not perment 12.5 Critical Present: Not perment 12.7 Specific Gravity: 0.95 (approx.) at 20'C (aust) 12.8 Liquid Surface Tension: (est.) 25 dynes/cm = 0.025 N/m at 20'C
3.1 C 3.2 Fe 3.3 W 3.4 D 3.5 C	3. CHENIC G Competibility Hydrocarbon onmula: Not list SO/UH Designi OT ID No.: 122 AS Registry N	AL DESIGN y Class: M mature ad atton: 3.3/1 3 5.: Data no	LATIONS Iscalianeous 1223 4 available	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liqud 4.2 Celor: Dark 4.3 Odor: Tany: Bie karosene		Data not available Food Chein Concentration Potantiat: Data not available	12.3         Liquid Water Interfacial Tension: (est) 50 dynas/on: 0.05 N/m at 20°C           12.10         Vapor (Ges) Specific Gravity: Not partment           12.11         Ratio of Specific Heats of Vapor (Ges): Not partment           12.12         Latent Heat of Vaportzation: Not partment           12.13         Heat of Combustion: -18.000 Bb//b = 10.000 Bb//b =
L1 F 6.2 S L3 T 6.6 S 6.5 T L7 L 6.6 V 6.9 L 6.11 H	Personal Prote symptoms Foll initiation is no freatment of E probably not i chest x-tays; copious quant forestold Lind found for a solid remain, may coll diar Threaches in Jajuid or Solid remain, may coll DLH Value; Der	ctive Equil owing Exp rmally mini- isposure: I required; di consider private ti Value: De aston: Gin estion: Gin estion: Gin estion: Chari infant Ch	5. HEAL present: Protective gl leasane: INGESTION: mai but may become NGESTION: do NOT naywel development ophylactic antibiotic , SKIR: wipsoff an sta not available mits: Not pertinent ade 1: LDes = 5 to allable acteristics: None arracteristics: Mone arracteristics: Mone arracteristics: Mone arracteristics: Mone arracteristics: Mone arracteristics: Mone arracteristics: Mone arracteristics: Mone	TH KAZARDS overs: goggles or face sheld. : gastroinlestinal initiation. ASPIRATION: pulmonary , a more severe several hours after exposure. I levage or induce vonting. ASPIRATION: tratement of pulmonary initiation can be detected by serial regime if condition warranta. EYES: weak with d weak with scap and water. 15 g/kg num hazard. If spilled on clothing and allowed to of the skin.	8.1 12 8.3 9.4	9. SHIPPING INFORMATION Oradee of Purky: Commercial Storage Temperature: Elevated Inert Atmosphere: No requirement Venting: Open (flame arrester)	12.14 Heat of Decomposition: Not partinent 12.15 Heat of Solution: Not partnent 12.16 Heat of Solution: Not partnent 12.26 Heat of Pusitor: Data not available 12.27 Reid Vapor Pressure: Data not available 12.27 Reid Vapor Pressure: Data not available

--- .

# TOLUENE

Toluci	nyme Watery kquid	Colorless Pleasant odor		
Methylbenzené Methylbenzól	Floats on water.	Flammable, irritating vapor is produced.	ຍ.2 ເມ	
<b>.</b>		· · ·		
Shut off ign Shut off ign	or possible. Read people aw bon sources and call five depar and use writer spray to "knoci	ray. traent_ k.down'' vapor.		
Avoid conta Isolate and Notify local	ct with liquid and vapor remove discharged matenal. health and pollution control age	incies.	6.0	
	Flashback along vapor trail Vapor may explode if ignite	mey occur. d in an enclosed area.	6.7	
Fire	Extinguish with dry chemical Water may be ineffective or Cool exposed containers with	(Foam, or working expension). Foam, or ambon demose, fire, th water.	6.1	
	CALL FOR MEDICAL AID.			
	Imtating to eyes, nose and If inhaled, will cause nause difficult breathing, or in	throat. a, vomtong, headache, dizziness, and consciousiones.	7.1	
	Move to fresh air. If breathing has stopped, gr	ve artificial respiration.	7.3	
	LIQUID	13-11.	7.6	
Exposure	If swallowed, will cause naw Remove containinated cloth	uses, vorwting or loss of consciousness. Ing and shoes.	7.	
	F IN EYES, hold evends on F SWALLOWED and victor	ency or water. en and flush with plenty of water. I is CONSCIOUS, have victim drink water	7.1	
	DO NOT INDUCE VOMITIN	G.	7.1	
Water Pollution	Dangerous to aquatic life in Fouling to shoreline. May be dangerous if it ente Notify local health and wildl	I high concentrations. In water intakes.		
1. RESPO	L NORTH ODERATORS OF MELETRY V	2. LABEL		
(See Respone issue warm Evacuate al	e läethods Handbook) ng-high flammability let	2.1 Category: Flammable Kquid 2.2 Class: 3	8.1 6.2 6.3	
3. CHEM	ICAL DESIGNATIONS	4. OBSERVABLE CHARACTERISTICS	"	
3.1 CG Competibi	ity Cines: Aromatic	4.1 Physical State (as shipped): Liquid		
3.2 Formula: CeHs 3.3 IMO/UN Denim		4.3 Odor: Pungent; aromatic, benzene-like; distinct, pleasant		
3.4 DOT ID No.: 17 3.5 CAS Registry	294 No.: 108-88-3			
	5. HEA	LTH HAZARDS		
	recuve Equipment: Air-subblied	I mask; goggies or tace shield; plastic gloves.	9.1	
5.1 Personal Prot 5.2 Symptoms Fo	blowing Exposure: Vapors init	ate eyes and upper respiratory tract; cause dizzmass.		
5.1 Personal Prof 5.2 Symptoms For hesdache, a asprated. c	blowing Exposure: Vapors init inesthesis, respiratory arrest. U auses coughing, gagging, distra uses commitme crimine client	ate eyes and upper respiratory tract; cause dizzmess, iquid imtates eyes and causes drying of ston. If ess, and rapidly developing pulmonary edems. If depressed respiration.		
5.1 Personal Prof. 5.2 Symptoms Follower followe	Nowing Exposure: Vepors infl anesthesia, respiratory artest. Li auses coughing, gagging, distre uses vomiting, griping, diantea, Exposure: INHALATION: remo; a doctor, INGESTION: dn MO	ate eyes and upper respiratory tract; cause dizzness. quid intrates eyes and causes drying of stor. If ass. and rappid; developing pulmonary edems. If depressed respiration. we to fresh er; give attribuel respiration and oxygen if Triduce vomenic; call a doctor. EVES: Nain with		
5.1 Personal Prof 5.2 Symptoms For headactre, ( aspirated, c ingested ca 5.3 Treatment of needed; cal water for at 5.4 Tyreathing Lin	showing Exposure: Vapori inti inesthesia, respiratory arrest. Li auses coording, gagging, distre, lesse voniting, groping, distree, Exposure: INHALATION: remo a doctor. INGESTION: do NO' least 15 min. SKIN: wipe off, winh Vable: 100 pom	ate eyes and upper respiratory tract; cause dizzmess, quid imates eyes and causes drying of ston. If iss, and rapidly developing pulmonary edema, if depressed respiration. we to fresh ar, give artificial respiration and oxygen if T induce vometing; call a cloctor. EYES: flush with rish with soap and water.	12 13	
5.1 Personal Prof 5.2 Symptoms FC headache, / asprated, c ingested cla 5.3 Treatment of needec, cal wster for at 5.4 Threahold Lin 5.5 Short Term li 5.5 Toricitari kor lin	where the second	ate eyes and upper respiratory tract; cause dizzness. quid intrates eyes and causes drying of ston. If sa, and rapidly developing pulmonery edema. If depressed respiration. we to fresh ar, give atrifical respiration and oxygen if T induce vornsing; call a doctor. EYES: flush with resh with scep and water. 30 min. to 5 g/kg	8.2 9.3 8.4	
5.1 Personal Prof 5.2 Symptoms Fro headache, / asprated. c ingested ca water for at 5.4 Threathold Lin 5.5 Short Term if 5.5 Taxiolty by in 5.7 Late Taxiolty 5.4 Versue (Jenn)	Nowing Exposure: Vapors infl Insethesia, respiratory arrest. U auses coupling, opping, distri- auses vomiting, orping, diarthas, Exposure: INHALATION: remo a coctor. INGESTION: do NO' least 15 mm. SKIN: wipe off, w nH Value: 100 ppm while the transformatic states of the values of the states of the states of the states of the values of the states of the states of the states of the values of the states of the states of the states of the values of the states of the states of the states of the values of the states of the states of the states of the states of the values of the states of the states of the states of the states of the values of the states of the states of the states of the states of the values of the states of the states of the states of the states of the values of the states of the states of the states of the states of the values of the states of the states of the states of the states of the values of the states of the values of the states of t	ate eyes and upper respiratory tract; cause dizzness, quid intrates eyes and causes drying of ston. If ass, and rappid developing pulmonary adems. If depressed respiration, we to fresh ar, give attificial respiration and oxygen if i fudue vormining; call a doctor. EYES: flush with rish with soap and water. 30 min. to 5 g/kg follow ingestion. cause a sight smarting of the eves or respiratory.	82 83 84	
5.1 Personal Prof. 5.2 Symptoms FC headache, i aspratied, c ingested ca systement of needec, cal water for at 5.3 Traditive by 5.5 Short Term if 5.5 Short Term if 5.5 Taxicity by in 5.5 Vapor (Ges) 5 system if or 5.1 Light or Sult	Nowing Exposure: Vapors infl unesthesia, respratory arrest. U auses coughing, gagging, distries, Exposure: INHALATION: remo a doctor. INGESTION: do NO <sup>0</sup> a doctor. INGESTION: do NO <sup>0</sup> assist 15 ms. SKIN: where off, w net Value: 100 ppm whateton Limits: 600 ppm for 3 spestion: Grade 2; LD is = 0.5 Kidney and liver damage may mitant Characteristics: Vapors seent in high concentrations. The	ate eyes and upper respiratory tract; cause dizoness, liquid trates eyes and causes drying of stor. If execution of the eyes and causes drying of stor. If depresed respiration, we to fresh arr, give antificial respiration and oxygen if induce vornating; call a doctor. EVES: flush with rish with scap and water. 30 min. to 5 g/kg follow regestion, cause a slight smarting of the eyes or respiratory te effect is temporary. mum haged. If splited on clothing and ellowed to	92 93 94	
5.1 Personal Prof. 5.2 Symptoms Pc headecne. ( asprated. c ingested ca s.3 Treativent of needec cal water for at 5.4 Threathold Lin 5.5 Short Term if 5.5 Taxicity by in 5.5 Layed or Soli reman, may 10 Odger Theathol	Nowing Exposure: Vapors intr unesthesia, respratory arrest. Li auses coughing, gagging, distria esse vonting, griping, diamea, Exposure: INHALATION: remo e a doctor. INGESTION: do NOT esset is nim. SKIN: wipe off, wi nikalation Limita: 600 ppm for ( gestion: Grade 2; LDa o = 0.5; Kidney and liver damage may rithant Characteristics: Vapor esent in high concentrations. Ti d Infrant Characteristics: Minu cause smarting and reddening at: 0.17 opm	ate eyes and upper respiratory tract; cause dizzness. quid intrates eyes and causes drying of ston. If ss. and rapidly developing pulmonary edems. If depressed respiration. we to fresh ar; give athficial respiration and oxygen if T induce vomiting; call a doctor. EYES: flush with mash with scap and water. 30 min. to 5 g/kg follow ingestion. cause a slight smarting of the eyes or respiratory to effect is temporary. mum hazard. If spilled on clothing and allowed to of the skin.	11 13 14	
<ul> <li>5.1 Personal Prof.</li> <li>6.2 Symptoms FG</li> <li>headache, i</li> <li>asprated.c</li> <li>ingested ca</li> <li>water for at</li> <li>5.3 Treatment of</li> <li>needed: cal</li> <li>water for at</li> <li>5.4 Threehold Lin</li> <li>5.5 Short Term if</li> <li>6.5 Toxicity by in</li> <li>5.7 Late Toxicity</li> <li>5.8 Vapor (Gas.)</li> <li>aystem if pn</li> <li>5.9 Liquid or Solid</li> <li>remain, may</li> <li>5.10 Odor Threehol</li> <li>5.11 IDLH Value: 2</li> </ul>	Nowing Exposure: Vapors infl Insethesia, respratory arrest. U auses coupling, gagging, distra saes vomiting, griping, diarthas, Exposure: INHALATION: remo a coctor. INGESTION: do NO' least 15 mm. SKIN: wipe off, w int Value: 100 ppm with Value: 100 ppm withing Characteristics: Vapor seart an high concentrations. Th d inflamt Characteristics: Mini- cause smarting and reddening with 0.17 ppm .000 ppm	ate eyes and upper respiratory tract; cause dizzness, quid intrates eyes and causes drying of ston. If ass, and rapidly developing pulmonary adems. If depressed respiration, we to fresh ar, give attificial respiration and oxygen if i fudue vormising; call a doctor. EYES: flush with rish with soap and water. 30 min. to 5 g/kg follow ingestion, cause a sight smarting of the eyes or respiratory te effect is temporary. mum hazerd. If spilled on clothing and allowed to of the skin.	813 813 814	
<ol> <li>Personal Prof.</li> <li>Symptoms Fr.</li> <li>headache. ( asprated, c ingested ca</li> <li>Trestivent of needet, cal water for at</li> <li>Trestivent of needet, cal water for at</li> <li>Trestivent Ling</li> <li>Short Term if</li> <li>Taxicity by in</li> <li>Taxicity by in</li> <li>Taxicity by in</li> <li>Short Term if</li> <li>Lique (or Solid reman, may</li> <li>Odor Threach</li> <li>IDLH Value: 2</li> </ol>	Nowing Esposure: Vapors infl Insethesia, respiratory arrest. U Lasses couphing, gapging, distri- uses vomiting, griping, diarthea, Exposure: INHALATION: remo a doctor. INGESTION: do NO least 15 mm. SKIN: wipe off, w nit Value: 100 ppm for 3 gestion: Grade 2: LDus = 0.5 Kidney and liver damage may inhant Characteristics: Vapors seart in high concentrations. Ti d Initiant Characteristics: Mining austice and the concentrations and d Initiant Characteristics: Mining add 0.17 ppm (200 ppm)	ate eyes and upper respiratory tract; cause dizoness, quid trates eyes and causes drying of ston. If ass, and rappid developing publiconary edema, If depresed respiration, we to fresh arr, give artificial respiration and oxygen if induce vorneting; call a doctor. EVES: flush with rish with scep and water. 30 min. to 5 g/kg follow ingestion. cause a slight smarting of the eyes or respiratory se effect is temporary, mum hazard. If spilled on clothing and allowed to of the skin.	8.1 8.1	
<ul> <li>5.1 Personal Prof.</li> <li>5.2 Symptoms Pc.</li> <li>headsche. (</li> <li>asprated. c</li> <li>ingested. c</li> <li>asprated. c</li> <li>asprated. c</li> <li>tratitionent of</li> <li>Teationent of</li> <li>Texterione to a</li> <li>the Toxicity by in</li> <li>5.5 Short Term in</li> <li>5.5 Color Term in</li> <li>system if pr</li> <li>5.9 Liquid or Solid reman, may</li> <li>5.10 Odor Threaded</li> <li>5.11 IDLH Value: 2</li> </ul>	Nowing Esposure: Vapors infl unesthesis, respratory arrest. Li auses coughing, gagging, distre lasses vonting, griping, diarmea, Exposure: INHALATION: remo is a cotor. INGESTION: do NOT lasst 15 mn. SKIN: wep off, wi helastion Limita: 600 ppm for 3 gestion: Grade 2; LDxe = 0.5 i: kidney and liver damage may initiant Characteristics: Vapora esent in high concentrations. Ti d initiant Characteristics: Mini- cause smarting and raddening dit 0.17 ppm .000 ppm	ate eyes and upper respiratory tract; cause dizzness, quid intrates eyes and causes drying of ston. If ass, and rappid developing pulmonary edems. If depressed respiration, we to fresh ar, give attificial respiration and oxygen if i fudue vorning; call a doctor. EYES: flush with rash with soap and water. 30 min. to 5 g/kg follow ingestion. . cause a slight smarting of the eyes or respiratory to effect is temporary. mum hazard. If spilled on clothing and allowed to of the skin.	9.2 9.3 9.4	
<ul> <li>5.1 Personal Prof.</li> <li>5.2 Symptoms FG</li> <li>headache, i</li> <li>asprated, c</li> <li>ingested ca</li> <li>ingested ca</li> <li>ingested ca</li> <li>treatment of</li> <li>needec, cal</li> <li>water for at</li> <li>5.5 Short Tet III</li> <li>5.7 Late Toxichy</li> <li>5.4 Physenold ta</li> <li>5.7 Late Toxichy</li> <li>5.8 Liquid or Statisty</li> <li>5.10 Odor Threath</li> <li>5.11 IDLH Value: 2</li> </ul>	Nowing Exposure: Vapors infl Insethesia, respiratory arrest. Li auses couphing, graping, diarti- saes vomiting, graping, diarti- saes vomiting, graping, diarti- saes vomiting, graping, diarti- ta doctor. INGESTION: do NO least 15 mm. SKIN: wipe off, w net Values: 100 ppm for 3 whet dia the same same same values on Limits: 600 ppm for 5 Nothey and liver damage may whet characteristics: Vapors seart in high concentrations. Th d inftant Characteristics: Mini- cause smarting and reddening whet 0.17 ppm .000 ppm	ate eyes and upper respiratory tract; cause dizzness, quid trates eyes and causes drying of ston. If ass, and rappid; developing publiconery edems, if depressed respiration, we to freen ar, give attificial respiration and oxygen if i induce vornising; call a doctor. EVES: flush with rish with soap and water. 30 min. to 5 g/kg follow rigestion, cause a sight smarting of the eyes or respiratory re effect is temporary. mum hazard. If spilled on clothing and ellowed to of the skin.	9.2 9.3 9.4 9.1 9.1	
<ul> <li>5.1 Personal Prof.</li> <li>5.2 Symptoms Prof.</li> <li>4.2 Symptoms Prof.</li> <li>4.3 Trestwart of needed: cal.</li> <li>4.4 Threshold Lin</li> <li>5.5 Short Term if</li> <li>6.5 Short Term if</li> <li>6.5 Taxiotity by it</li> <li>6.5 Vapor (Ges) is system of Point</li> <li>7. Late Taxiotity:</li> <li>8.5 Vapor (Ges) is system of Point</li> <li>9.10 Odor Threach</li> <li>5.11 IDLH Value: 2</li> </ul>	Nowing Esposure: Vapors infl Institues, respiratory arrest. U Lasse coupling, gapging, distri- uses vomiting, griping, diarthea, Exposure: INHALATION: remo a doctor. INGESTION: do NO least 15 mm. SKIN: wipe off, w and Value: 100 ppm for 3 gestion: Grade 2: LDus = 0.5 Kidney and liver damage may inhant Characteristics: Vapor seart in high concentrations. Th d Initiant Characteristics: Minin cause smarting and reddening add 0.17 ppm .000 ppm	ate eyes and upper respiratory tract; cause dizoness, quid trates eyes and causes drying of stor. If ass, and report developing publicities expiration and oxygen if those vorning; call a doctor. EVES: flush with rish with scep and water. 30 min. to 5 g/kg follow ingestion. cause a slight smarting of the eyes or respiratory select is temporary. mum hazard. If spilled on clothing and allowed to of the skin.	8.2 9.3 8.4 9.1	

1

	·
6. FIRE HAZARDS	10. HAZARD ASSESSMENT CODE
Test Point: 40°F C.C.; 55°F O.C.	(See Hazard Assessment Handbook)
The Extinguishing Agents: Carbon dioxide	A-1-U
or dry chemical for small fires, ordinary	
roam for large tires. Fire Extinguishing Agents Not to be	
Used: Water may be ineffective	11. HAZARD CLASSIFICATIONS
Products: Not pertinent	Flammable iguid
Sehavior in Fire: Vapor is heavier than air	11.2 NAS Hazard Rating for Bulk Water
and may travel a considerable distance to	Transportation:
a source of ignition and flash back. gnition Temperature: 997°F	Fire
Electrical Hazant: Class I, Group D	Health
Suming Rate: 5.7 mm/mm.	Liqued or Solid Imtant
Data not available	Poisons
	Water Polution
(Continued)	Aquatic Toxicity
	Aesthetic Effect 2
7. CHEMICAL REACTIVITY	Reactivity Other Chemicals 1
esclivity with Water: No reaction	Water 0
reaction	Self Reaction
tability During Transport: Stable	Category Classification
Caustics: Not pertinent	Health Hazard (Blue) 2
olymerization: Not pertinent	Flammability (Red)
Not pertment	
olar Ratio (Reactant to	
Product): Data not available	
NICHVITY GYDUP: 32	
	12. PHYSICAL AND CHEMICAL PROPERTIES
	12.1 Physical State at 15°C and 1 atm:
	Liquid
	12.2 Molecular Weight: 92.14
	231.1°F ≈ 110.8°C = 383.8°K
	12.4 Freezing Point:
8. WATER POLLUTION	
quatic Toxicity:	605.4°F = 318.8°C = 591.8°K
1160 mg/l/96 hr/sunfish/TL_/fresh water	12.8 Critical Pressure:
aterfowi Toxicity: Data not available	MN/m <sup>4</sup>
clogical Oxygen Demand (BOD):	12.7 Specific Gravity:
bod Chain Concentration Potential:	12.4 Liquid Surface Tension:
None	29.0 dynes/cm = 0.0290 N/m at 20°C
	12.8 Liquid Water Interfacial Tension:
	12.10 Vapor (Gas) Specific Gravity:
	Not perbnent
	12.11 Ratio of Specific Heats of Vapor (Ges); 1.089
	12.12 Latent Heat of Vaporization:
	155 Bhu/kb = 86.1 cas/g ⇒ 3.61 x 108 1//-
	12.13 Heat of Combustion:
wine of Busine Descent comments	=9688 cal/g =405.5 X 10* J/kg
ntration-all 99.8 + %; industnal:	72.14 Heat of Decomposition: Not permant 12.15 Heat of Solution: Not common
contains 94 + %, with 5% xylene and	12.16 Heat of Polymerization: Not pertinent
smail amounts of benzene and	12.25 Heat of Fusion: 17.17 cal/g
noneromatic hydrocarbons; 90/120:	12.26 Limiting Value: Data not available
1995 pure than industrial.	12.27 Reid Vapor Pressure: 1.1 psia
eraye Imperature: Ambient	
Dissum-vaciam	
6. FIRE HAZAI	RDS (Continued)
Noichiometric Air to Fuel Ratio: Date not a	valable
•	

# TRICHLOROETHYLENE

Chlorylen Semalgene Frethylene Frichloran, Trilene	Sinks in water. Ii	ritigiting vapor is produced.
Stop discha Avoid conta Cail fire der Isolate and Notify local	rge if possible Keep people aw ca with liquid and vapor, sartment, remove discharged material, health and pollution control age	ay.
Fire	Combustible. POISONOUS GASES ARE Wear goggles and self-cont Extinguish with dry chemica	PRODUCED IN FIRE. amed breathing apparatus. I, carbon pioxide, or foam
Exposure	CALL FOR MEDICAL AID VAPOR Initiating to eyes, nose and If inhaled, wil cause nause or loss of conclousme Move to fresh air. If breathing is difficult, give LICQUID Initiating to skin and eyes, if eveloped, wil cause na- or loss of consciousne Henove contaminated clott Fluos affected areas with o IF IN EYES, not eyends on IF SWALLOWED and victim or SWALLOWED and victim VULSIONS, do nothing	throat. s, vorming, difficult breathing, is: ve artificial resolvation oxygen ises, vorming, difficult breathing, is, ing and shoes. enty of water is CONSCIOUS, have victim drink water indice verming. Is UNCENSCIOUS OR HAVING CON- except is see victim watm.
Water Pollution	Effect of low concentrations May be dangerous if it enter Notify local health and wild Notify operators of nearby to	i on aquebo life is unknown. I water intakes. Ne officials. Refer intakes
1. RESP( (Bee Respons Shouid be / Chemical a	NYSE TO DISCHARGE e Methode Handbook) emoved nd physical treatment	2. LABEL 2.1 Category: None 2.2 Class: Not pertinent
3. CHEM 3.1 CG Competible hydrocarbo 3.2 Pointule: CHCI 3.3 IMO/UN Desig 3.4 DOT ID No.1 3.5 CAS Registry	CAL DESIGNATIONS Hy Class: Halogenated - CCI: netion: 9.0/1710 710 No.: 79-01-6	4. OBSERV/ "LE CHARACTERISTICS 4.1 Physical Bate (as shipped): Liquid 4.2 Color: Coloriess 4.3 Odior: Chioroform-Risc; ethereel
E.1 Personal Pro- apparatus is neoprens 5: 39mptoms Fri throat to na nervous syn HKGESTICK slightly ims 5.3 Treatment of cases of ov respiration 1 repast three water. SKIN 6.4 Threethold Line 5.5 Short Term is 6.5 Toziotty by in 6.7 Late Toxiotty by in 6.7 Late Toxiotty.	5. HEA sective Equipment: Organi. va or emergencies; naoprene or vis lefy shoes; naoprene or vis lefy shoes; naoprene suit or a lefwing Exposure: INHALATIC uses, an attitude of irresponsibilities rangeosaris: InHALATICON: remo exclosure: Do NOT administer exposure: DN NOT administer exposure: INHALATICON: remo nal/or administer oxygen. INGE bimas; then give 1 tablespoon whatetion Limite: 200 ppm for 3 gestion: Grade 3; LDse = 50 Data: not available rithert Characteristics: Vapora	LTH HAZARDS on-acid gas canister, self-contained breathing ht gloves: chemical safety goggles; face-shekd; iron for splesh protection. N: symptoms range from initiation of the nose and ity, burned vesion, and finally disturbance of central Chronic apposure may cause organic injury. SKIN: diversiting action can cause dermastise. EYES: adrenatin or epinephrine; get medical attention for all we vicim to freeh air; if necessary, apply stiffical STION: have vicitm divis water and induce voriting; apaom safe in vesior. EYES: flush thoroughly with d warm weser. 0 min. to 500 mg/kg cause a slight emerting of the systs or respiratory re effect a temporary.

.

6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 7.1 7.2	6. FIRE HAZARDS Flash Point: 90°F C.C.; practically nonflammable Flanmable Limits in Air: 8.0%-10.5% Fire Extinguishing Agents: Water log Fire Extinguishing Agents Not to be Uesd: Not perment Special Hazards of Combuetton Products: Toxic and initiating gases are produced in fire stuations. Behavior in Fire: Not pertinent lightion Temperature: 770°F Electrical Hazard: Not pertinent Burning Rate: Not pertinent Addisabit: Flams Temperature: Data not available Stoichiometric Air to Fuel Ratio: Data not available Flame Temperature: Data not available Flame Temperature: Data not available Resctivity With Wate: No reaction Resctivity With Wate: No reaction Resctivity with Common Materials: No	10. HAZARD ASSESSMENT CODE (See Hazard Assessment Handbook) A-X-Y         11. HAZARD CLASSIFICATIONS         11.1 Code of Federal Regulations: ORM-A         11.2 NAS Hazard Reting for Bulk Water Transportation: Category Rating Fire
7.3	Stability During Transport: Stable	11.3 NFPA Hazard Classification:
7.4	Neutralizing Agents for Acids and	Category Classification Health Hazard (Blue)
7.5	Polymerization: Not pertinent	Flammability (Red)
7.9	Inhibitor of Polymerization: Not pertnent	neecuvity (Tenow)
7.7	Moler Retio (Reactant to	· · · · ·
7.8	Product: Usta not available Reactivity Group: 36	
		12. PHYSICAL AND CHEMICAL PROPERTIES
		12.1 Physical State at 15°C and 1 atm:
		Liquid 12.2 Molecular Weight: 131.39
		12.3 Bolling Point at 1 atm: 18915 - 8710 - 2801K
		12.4 Freezing Point:
	3. WATER POLLUTION	-123.5°F =88.4°C = 186.8°K
		12.5 Critical Temperature: Not pertinent
6.1	Aquatic Toxicity:	12.5 Critical Temperature: Not pertinent 12.6 Critical Pressure: Not pertinent
6.1	Aquatic Toxicity: 660 mg/l/40 hr/daphnia/kill/tresh water	12.5         Critical Temperature: Not pertinent           12.6         Critical Pressure: Not pertinent           12.7         Specific Gravity;           1.46 at 20°C (liquid)
6.1 8.2 8.3	Aquetic Toxicity: 660 mg/i/40 hr/daphnia/kill/tresh water Waterfowi Toxicity: Data not evalable Biological Oxygen Demend (BOD):	12.5     Critical Temperature: Not pertinent       12.6     Critical Pressure: Not pertinent       12.7     Specific Gravity:       1.46 at 20°C (liquid)       12.8     Liquid Surface Temelon:       29.3 dynes/end m = 0.0269 N/m at 20°C
6.1 8.2 8.3	Aquetic Toxichty: 660 mg/l/40 hr/daphnia/kill/tresh water Watertowi Toxichty: Data not available Biological Oxygen Demand (BOO): Data not available Evod Chain Consentation Britantich	12.5     Critical Pressure: Not pertinent       12.6     Critical Pressure: Not perinent       12.7     Specific Gravity:       1.46 at 20°C (liquid)       12.8     Liquid Surface Tension:       2.9.3 dynes/rm = 0.0263 N/m at 20°C       12.9     Liquid Water Interfactal Tension:       2.8     Liquid Water Interfactal Tension:
6.1 8.2 8.3 8.4	Aquistic Toxichty: 660 mg/l/40 hr/daphnia/kill/tresh water Waterfowt Toxichty: Data not available Biological Ozygen Demand (BOD): Data not available Food Chain Concentration Potential: None	12.5     Critical Pressure: Not pertinent       12.6     Critical Pressure: Not perinent       12.7     Specific Grewity:       1.46 at 20°C (liquid)       12.8     Liquid Surface Tension:       29.3 dynes/cm = 0.0283 N/m at 20°C       12.6     Liquid Water Interfacial Tension:       3.4.5 dynes/cm = 0.0345 N/m at 20°C       12.10     Vapor (Gea) Specific Gravity: 4.5
6.1 8.2 8.3 8.4	Aquistic Toxichiy: 660 mg/l/40 hr/daphnia/kill/tresh water Waterfowi Toxichiy: Data not available Biological Grygen Demand (BOD): Data not available Food Chain Concentration Potential: None	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grewity: <ol> <li>1.46 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Surface Tension: 29.3 dynes/cm = 0.0283 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension: 34.5 dynes/cm = 0.0345 N/m at 24°C</li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ol> <li>1.116</li> </ol> </li> </ul>
6.1 8.2 8.3 8.4	Aquatic Toxicity: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfoor Toxicity: Data not available Biological Oxygen Demand (BOD): Data not available Pood Chain Concentration Potential: None	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity; <ol> <li>1.46 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Surface Tension: <ol> <li>20.3 dynes/cm = 0.0283 N/m at 20°C</li> </ol> </li> <li>12.9 Liquid Water Interfacial Tension: <ol> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ol> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Retio of Specific Heats of Vapor (Gas): <ol> <li>1.116</li> </ol> </li> <li>12.12 Latent Heat of Vaportzation: <ol> <li>103 Blu/b = 57.2 cal/g = <ol> <li>4.4 x 10<sup>4</sup> J/kg</li> </ol> </li> </ol></li></ul>
6.1 8.2 8.3 8.4	Aquatic Toxicity: 660 mg/l/40 hr/daphnia/kill/tresh water Waterfoor Toxicity: Data not available Biological Oxygen Demand (BOD): Data not available Pood Chain Concentration Potential: None	<ul> <li>12.5 Critical Temperature: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: 1.46 at 20°C (liquid)</li> <li>12.8 Liquid Surface Temelor: 29.3 dynes/cm = 0.0280 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension: 34.5 dynes/cm = 0.0345 N/m at 20°C</li> <li>12.10 Vepor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Gravity: 4.5</li> <li>12.12 Latant Heat of Veportzation: 103 Blu/lb = 57.2 cal/g = 2.4 X 10° J/kg</li> <li>12.13 Heat of Combustion: Not pertnent</li> <li>12.14 Heat of Combustion: Not pertnent</li> </ul>
6.1 8.2 8.3 8.4	Aquatic Toxicity: 660 mg/l/40 hr/daphnia/kill/tresh water Waterfoor Toxicity: Data not available Biological Oxygen Demand (BOO): Data not available Pood Chain Concentration Potential: None	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ol> <li>1.48 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Surface Tension: <ol> <li>2.9 dynes/cm = 0.0280 N/m at 20°C</li> </ol> </li> <li>12.9 Liquid Water Interfacial Tension: <ol> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ol> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Hests of Vapor (Gas): <ol> <li>1.116</li> </ol> </li> <li>12.12 Latant Hest of Vaportzation: <ol> <li>103 Bhi/lb = 57.2 cal/g = <ol> <li>4.4 X 10<sup>4</sup> J/kg</li> </ol> </li> <li>12.13 Hest of Combustion: Not pertnent</li> <li>12.14 Hest of Decompetitor: Not pertnent</li> <li>12.15 Hest of Decomposition: Not pertnent</li> </ol> </li> </ul>
6.1 4.2 8.3 8.4	Aquetic Toxichiy: 660 mg/1/40 hr/daphnia/kill/tresh water Watertowi Toxichiy: Data not available Biological Oxygen Demand (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION	<ul> <li>12.5 Critical Pressure: Not partment</li> <li>12.6 Critical Pressure: Not partment</li> <li>12.7 Specific Gravity: <ol> <li>1.46 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Surface Tension: <ol> <li>2.9 dynes/cm = 0.0280 N/m at 20°C</li> </ol> </li> <li>12.9 Liquid Water Interfactal Tension: <ol> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ol> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ol> <li>1.116</li> </ol> </li> <li>12.12 Laternt Heat of Vaporization: <ol> <li>103 Btu/lb = 57.2 cal/g = <ol> <li>2.4 X 10<sup>4</sup> J/kg</li> </ol> </li> <li>13 Heat of Decomposition: Not periment</li> <li>12.16 Heat of Decomposition: Not periment</li> <li>12.16 Heat of Decomposition: Not periment</li> <li>12.16 Heat of Polymetization: Not periment</li> <li>12.17 Heat of Polymetization: Not periment</li> <li>12.18 Heat of Polymetization: Not periment</li> <li>12.19 Heat of Fusion: Data not available</li> </ol> </li> </ul>
6.1 8.2 8.3 8.4	Aquetic Toxichty: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfowt Toxichty: Data not evailable Bological Oxygen Demand (BOD): Data not evailable Food Chain Concentration Potentia: None 9. SHIPPING INFORMATION Grades of Purthy: Technical; dry cleaning;	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: 1.48 at 20°C (liquid)</li> <li>12.8 Liquid Surface Tension: 2.9 dynes/cm = 0.0263 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension: 34.5 dynes/cm = 0.0345 N/m at 24°C</li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): 1.116</li> <li>12.12 Learnt Heat of Vaporization: 103 Blu//b = 57.2 cal/g = 2.4 X 10° J/kg</li> <li>13 Heat of Solution: Not pertinent</li> <li>12.16 Heat of Decomposition: Not pertinent</li> <li>12.16 Heat of Dolution: Not pertinent</li> <li>12.16 Heat of Polymeization: Not pertinent</li> <li>12.17 Heat of Polymeization: Not pertinent</li> <li>12.18 Heat of Polymeization: Not pertinent</li> <li>12.19 Heat of Polymeization: Not pertinent</li> <li>12.21 Heat of Polymeization: Not pertinent</li> <li>12.31 Heat of Polymeization: Not pertinent</li> <li>12.4 Heat of Polymeization: Not pertinent</li> <li>12.5 Heat of Polymeization: Not pertinent</li> <li>12.6 Heat of Polymeization: Not pertinent</li> <li>12.6 Heat of Polymeization: Not pertinent</li> <li>12.6 Heat of Polymeization: Dista not available</li> <li>12.47 Reid Vapor Peasure: 2.5 Dea</li> </ul>
6.1 8.2 8.3 8.4 8.1 8.1	Aquetic Toxichty: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfowit Toxichty: Data not evailable Biological Oxygen Demand (BOD): Data not evailable Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purthy: Technical; dky cleaning: degressing: estisction Stargent Temparature, Archivert	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Surface Tension: <ul> <li>2.9 dynes/cm = 0.0263 N/m at 20°C</li> </ul> </li> <li>12.9 Liquid Water Interfactal Tension: <ul> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Gravity: 4.5</li> <li>12.12 Litent Heat of Vaporization: <ul> <li>103 Bit//b = 57.2 cal/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> </ul> </li> <li>12.14 Heat of Decomposition: Not pertinent</li> <li>12.15 Heat of Solution: Not pertinent</li> <li>12.16 Heat of Polymerization: Not pertinent</li> <li>12.17 Heat of Polymerization: Not pertinent</li> <li>12.18 Heat of Polymerization: Not pertinent</li> <li>12.19 Heat of Polymerization: Not pertinent</li> <li>12.19 Heat of Polymerization: Not pertinent</li> <li>12.26 Heat of Polymerization: Not pertinent</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul>
6.1 8.2 8.3 8.4 8.1 8.2 8.3	Aquetic Toxichty: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfowit Toxichty: Data not evailable Biological Oxygen Demand (BOD): Data not evailable Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dxy cleaning; degressing: extraction Startege Temperature: Ambient Inert Atmosphere: No requirement	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Surface Tension: <ul> <li>2.0 dynes/cm = 0.0283 N/m at 20°C</li> </ul> </li> <li>12.9 Liquid Water Interfactal Tension: <ul> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.16</li> <li>12.12 Latent Heat of Vaporization: <ul> <li>103 Blu//b = 57.2 ca/g =</li> <li>2.4 X 10° J/kg</li> </ul> </li> <li>12.13 Heat of Sociation: Not pertnent</li> <li>12.16 Heat of Sociation: Not pertnent</li> <li>12.16 Heat of Polymerization: Not pertnent</li> <li>12.17 Heat of Polymerization: Not pertnent</li> <li>12.18 Heat of Polymerization: Not pertnent</li> <li>12.26 Heat of Polymerization: Not pertnent</li> <li>12.27 Reid Vapor Pressure: 25 pas</li> </ul></li></ul>
6.1 8.2 8.3 8.4 8.1 8.2 8.3 8.4	Aquetic Toxichty: 660 mg/l/40 hr/daphnia/kill/tresh water Waterfowi Toxichty: Data not available Biological Orygen Demand (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning; degressing: straction Storage Temperature: Ambient Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Surface Tension: <ul> <li>20.3 dynes/cm = 0.0283 N/m at 20°C</li> </ul> </li> <li>12.9 Liquid Water Interfactal Tension: <ul> <li>34.5 dynes/cm = 0.0345 N/m at 24°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.16</li> </ul> </li> <li>12.12 Latent Heat of Vaporization: <ul> <li>103 Blu/b = 57.2 ca/g = <ul> <li>2.4 X 10<sup>s</sup> J/kg</li> </ul> </li> <li>12.13 Heat of Sociation: Not pertnent</li> <li>12.14 Heat of Decomposition: Not pertnent</li> <li>12.15 Heat of Polymerization: Not pertnent</li> <li>12.16 Heat of Polymerization: Not pertnent</li> <li>12.16 Heat of Polymerization: Not pertnent</li> <li>12.17 Heat of Solition: Data not evalable</li> <li>12.27 Reid Vapor Pressure: 2.5 psa</li> </ul></li></ul>
6.1 8.2 8.3 8.4 8.4 8.3 8.4	Aquetic Toxichiy: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfowi Toxichiy: Data not available Biological Grygen Demand (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning: degreean; extraction Storage Temperature: Ambient Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ol> <li>1.48 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Surface Tension: <ol> <li>29.3 dynes/cm = 0.0283 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Retio of Specific Heats of Vapor (Gas): <ol> <li>1.16</li> </ol> </li> <li>12.12 Latant Heat of Vaporization: <ol> <li>38.07 log</li> <li>1.16</li> </ol> </li> <li>12.13 Heat of Combustion: Not pertinent</li> <li>12.14 Heat of Combustion: Not pertinent</li> <li>12.15 Heat of Combustion: Not pertinent</li> <li>12.16 Heat of Fusion: Not pertinent</li> <li>12.17 Heat of Fusion: Not pertinent</li> <li>12.18 Heat of Fusion: Not an or available</li> <li>12.27 Reid Vapor Pressure: 2.5 pase</li> </ol></li></ul>
6.1 8.2 8.3 8.4 8.1 8.2 8.3 8.4	Aquatic Toxichiy: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfoort Toxichiy: Data not available Biological Grygen Demand (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; div cleaning: degreasing: extraction Starage Temperature: Antiient Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grevity; <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Water Internation Tension: <ul> <li>29.3 dynes/cm = 0.0280 N/m at 20°C</li> <li>12.9 Liquid Water Internation = 0.0345 N/m at 20°C</li> <li>12.10 Vapor (Gas) Specific Gravity; 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.116</li> </ul> </li> <li>12.12 Latent Heat of Vaporization: <ul> <li>103 Bhu/lo = 57.2 car/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> </ul> </li> <li>12.13 Heat of Combustion: Not pertnent</li> <li>12.16 Heat of Decomposition: Not pertnent</li> <li>12.16 Heat of Solution: Not pertnent</li> <li>12.16 Heat of Solution: Not pertnent</li> <li>12.16 Heat of Fusion: Not pertnent</li> <li>12.17 Heat of Solution: Not pertnent</li> <li>12.18 Heat of Fusion: Data not available</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul></li></ul>
6.1 8.2 8.3 8.4 8.4 8.4 8.3 8.4	Aquatic Toxichiy: d60 mg/l/40 hr/daphnia/kill/tresh water Weterfoor Toxichiy: Data not available Biological Oxygen Demend (BOD): Data not available Pood Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; day cleaning: degressing: extraction Storage Tempersture: Antient Isert Atmosphere: No requirement Venting: Preseure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grevity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Water Internation: <ul> <li>29.3 dynes/cm = 0.0283 N/m at 20°C</li> <li>12.8 Liquid Water Internation:</li> <li>29.4 5 dynes/cm = 0.0345 N/m at 24°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5 <ul> <li>1.116</li> </ul> </li> <li>12.12 Letent Hest of Vaportation: <ul> <li>10.3 Blu/lb = 57.2 ca/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> </ul> </li> <li>12.13 Hest of Combustion: Not pertnent</li> <li>12.16 Hest of Decomposition: Not pertnent</li> <li>12.16 Hest of Polymerization: Not pertnent</li> <li>12.16 Hest of Solution: Not pertnent</li> <li>12.16 Hest of Solution: Not pertnent</li> <li>12.17 Hest of Solution: Not pertnent</li> <li>12.18 Hest of Solution: Not pertnent</li> <li>12.19 Hest of Solution: Not pertnent</li> <li>12.21 Hest of Solution: Not pertnent</li> <li>12.22 Limiting Value: Oata not evalable</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul>
6.1 8.2 8.3 8.4 8.4 8.3 8.4 8.3 8.4	Aquatic Toxicity: d60 mg/l/40 hr/daphnia/kill/tresh water Weterfoor Toxicity: Data not available Biological Oxygen Demand (BOD): Data not available Pood Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning; degressing: extraction Storage Temperature: Antoient Isert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity;</li> <li>1.48 at 20°C (liquid)</li> <li>12.8 Liquid Surface Tension:</li> <li>2.9 dynes/cm = 0.0280 N/m at 20°C</li> <li>12.9 Liquid Water Interfactal Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 24°C</li> <li>12.10 Vapor (Gas) Specific Gravity; 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas):</li> <li>1.116</li> <li>12.12 Latent Heat of Vaporization:</li> <li>103 Blu//b = 57.2 cal/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> <li>12.15 Heat of Solution: Not perinent</li> <li>12.16 Heat of Solution: Not perinent</li> <li>12.16 Heat of Polymetation: Not perinent</li> <li>12.17 Heat of Polymotation: Not perinent</li> <li>12.18 Heat of Polymotation: Not perinent</li> <li>12.19 Heat of Polymotation: Not perinent</li> <li>12.14 Heat of Polymotation: Not perinent</li> <li>12.15 Heat of Polymotation: Not perinent</li> <li>12.16 Heat of Polymotation: Not perinent</li> <li>12.17 Reid Polymotation: Not perinent</li> <li>12.28 Heat of Polymotation: Not perinent</li> <li>12.29 Reid Vapor Pressure: 2.5 pse</li> </ul>
6.1 8.2 8.3 8.4 8.4 8.4 8.2 8.3 8.4	Aquetic Toxichiy: 660 mg/l/40 hr/daphnia/kill/tresh water Waterfowi Toxichiy: Data not available Biological Orygen Demand (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning; degressing: extraction Storage Tempersture: Antient Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity;</li> <li>1.48 at 20°C (liquid)</li> <li>12.8 Liquid Surface Tension:</li> <li>20.3 dynes/cm = 0.0263 N/m at 20°C</li> <li>12.9 Liquid Water Interfactal Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 24°C</li> <li>12.10 Vapor (Gas) Specific Gravity; 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas):</li> <li>1.16</li> <li>12.12 Latent Heat of Vaporization:</li> <li>103 Blu//b = 57.2 ca/g =</li> <li>2.4 X 10° J/kg</li> <li>12.13 Heat of Sociation: Not pertnent</li> <li>12.16 Heat of Sociation: Not pertnent</li> <li>12.16 Heat of Sociation: Not pertnent</li> <li>12.16 Heat of Polymerization: Not pertnent</li> <li>12.16 Heat of Polymerization: Not pertnent</li> <li>12.17 Heat of Polymerization: Not pertnent</li> <li>12.26 Heat of Polymerization: Not pertnent</li> <li>12.37 Reid Vapor Pressure: 2.5 pss</li> </ul>
6.1 8.2 8.3 8.4 8.1 8.2 8.3 8.4	Aquetic Toxichiy: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfowi Toxichiy: Data not available Biological Grygen Demand (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning: degreesing: extraction Storage Temperature: Ambient Inert Atmosphere: No requirement Venting: Pressure-vecuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Water Interfacial Tension: <ul> <li>20.3 dynes/cm = 0.0283 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension: <ul> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.116</li> </ul> </li> <li>12.12 Latent Heat of Vaporization: <ul> <li>103 Bit/lb = 57.2 ca/g = <ul> <li>2.4 X 10<sup>4</sup> J/kg</li> </ul> </li> <li>12.13 Heat of Combustion: Not pertinent</li> <li>12.16 Heat of Docomposition: Not pertinent</li> <li>12.17 Heat of Combustion: Not pertinent</li> <li>12.18 Heat of Polymerization: Not pertinent</li> <li>12.19 Heat of Polymerization: Not pertinent</li> <li>12.19 Heat of Polymerization: Not pertinent</li> <li>12.27 Reid Vapor Pressure: 2.5 psis</li> </ul></li></ul></li></ul>
6.1 8.2 8.4 8.4 8.4 8.4 8.2 8.3 8.4	Aquetic Toxichiy: 660 mg/i/40 hr/daphnia/kill/fresh water Weterfowi Toxichiy: Data not available Biological Grygen Demand (BOD): Data not available Pood Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning: degraem; extraction Stange Temperature: Ambient Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Gravity: <ul> <li>1.46 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Water Interfacial Tension: <ul> <li>28.3 dynes/cm = 0.0269 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension:</li> <li>34.5 dynes/cm = 0.0364 N/m at 24°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.116</li> </ul> </li> <li>12.12 Latant Heat of Vaporization: <ul> <li>103 Blu/lb = 57.2 ca/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> </ul> </li> <li>12.13 Heat of Combustion: Not pertinent</li> <li>12.16 Heat of Decomposition: Not pertinent</li> <li>12.16 Heat of Polymerization: Not pertinent</li> <li>12.17 Heat of Polymerization: Not pertinent</li> <li>12.18 Heat of Polymerization: Not pertinent</li> <li>12.19 Heat of Polymerization: Not pertinent</li> <li>12.26 Heat of Polymerization: Not pertinent</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul>
6.1 8.2 8.4 8.4 8.4 8.2 8.3 8.4	Aquetic Toxichiy: 660 mg/l/40 hr/daphnia/kill/tresh water Weterfowi Toxichiy: Data not available Biological Grygen Demand (BOD): Data not available Pood Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; dry cleaning: degraem; stratcion Storage Temperature: Ambient Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grewity: <ol> <li>1.46 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Water Interfacial Tension: <ol> <li>29.3 dynes/cm = 0.0283 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> </ol> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Retio of Specific Heats of Vapor (Gas): <ol> <li>1.116</li> </ol> </li> <li>12.12 Latant Heat of Vaporization: <ol> <li>103 Bhu/b = 57.2 ca/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> </ol> </li> <li>12.13 Heat of Combustion: Not pertinent</li> <li>12.16 Heat of Combustion: Not pertinent</li> <li>12.16 Heat of Combustion: Not pertinent</li> <li>12.17 Heat of Decomposition: Not pertinent</li> <li>12.18 Heat of Fusion: Oat not evalable</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul>
6.1 6.2 8.3 8.4 8.1 8.2 8.3 8.4	Aquatic Toxichy: 600 mg/l/40 hr/daphnia/kill/tresh water Weterfour Toxichy: Data not available Biological Grygen Damand (BOD): Data not available Pood Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purthy: Technical; dry cleaning: degreasing: entraction Starage Temperature: Antilent Inert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grevity; <ol> <li>1.48 at 20°C (liquid)</li> </ol> </li> <li>12.8 Liquid Water Interfacial Tension: <ol> <li>29.3 dynes/cm = 0.0345 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 24°C</li> </ol> </li> <li>12.10 Vapor (dea) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (des): <ol> <li>1.16</li> </ol> </li> <li>12.12 Latant Heat of Vaporization: <ol> <li>103 Bhu/b = 57.2 ca/g = <ol> <li>4.X 10<sup>4</sup> J/kg</li> </ol> </li> <li>12.13 Heat of Combustion: Not pertnent</li> <li>12.16 Heat of Decomposition: Not pertnent</li> <li>12.16 Heat of Combustion: Not pertnent</li> <li>12.16 Heat of Fusion: Data not evalable</li> <li>12.27 Reid Vapor Pressure: 2.5 pas</li> </ol> </li> </ul>
6.1 8.2 8.3 8.4 8.4 8.4 8.4	Aquatic Toxichy: d60 mg/l/40 hr/daphnia/kill/mesh water Weterfoor Toxichy: Data not available Biological Oxygen Demand (BOD): Data not available Food Chain Concentration Potential: None 3. SHIPPING INFORMATION Grades of Purity: Technical; day cleaning; degressing: estracton Storage Temperature: Anbient Inert Atmosphere: No requirement Venting: Pressure-vectum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grevity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Water Interfacial Tension: <ul> <li>29.3 dynes/cm = 0.0328 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 20°C</li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.116</li> </ul> </li> <li>12.12 Latent Heat of Vaporization: <ul> <li>103 Blu/lb = 57.2 car/g =</li> <li>2.4 X 10<sup>4</sup> J/kg</li> </ul> </li> <li>12.13 Heat of Combustion: Not pertnent</li> <li>12.14 Heat of Combustion: Not pertnent</li> <li>12.15 Heat of Solution: Not pertnent</li> <li>12.16 Heat of Fusion: Data not evalable</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul></li></ul>
6.1 6.2 8.3 8.4 8.4 8.4 8.4 8.4	Aquatic Toxichy: d60 mg/l/40 hr/daphnia/kill/mesh water Weterfoor Toxichy: Data not available Biological Oxygen Demend (BOD): Data not available Food Chain Concentration Potential: None 9. SHIPPING INFORMATION Grades of Purity: Technical; day cleaning: degressing: extraction Storage Temperature: Antoient Isert Atmosphere: No requirement Venting: Pressure-vacuum	<ul> <li>12.5 Critical Pressure: Not pertinent</li> <li>12.6 Critical Pressure: Not pertinent</li> <li>12.7 Specific Grevity: <ul> <li>1.48 at 20°C (liquid)</li> </ul> </li> <li>12.8 Liquid Water Interfacial Tension: <ul> <li>29.3 dynes/cm = 0.0328 N/m at 20°C</li> <li>12.9 Liquid Water Interfacial Tension:</li> <li>34.5 dynes/cm = 0.0345 N/m at 24°C</li> </ul> </li> <li>12.10 Vapor (Gas) Specific Gravity: 4.5</li> <li>12.11 Ratio of Specific Heats of Vapor (Gas): <ul> <li>1.116</li> </ul> </li> <li>12.12 Latent Heat of Vaportation: <ul> <li>103 Blu/lb = 57.2 ca/g =</li> <li>2.4 X 10<sup>6</sup> J/kg</li> </ul> </li> <li>12.13 Heat of Combustion: Not pertnent</li> <li>12.16 Heat of Decomposition: Not pertnent</li> <li>12.16 Heat of Solution: Not pertnent</li> <li>12.17 Heat of Solution: Not pertnent</li> <li>12.18 Heat of Fusion: Data not available</li> <li>12.27 Reid Vapor Pressure: 2.5 pse</li> </ul>

# m-XYLENE

#### XLM

Common : 3-Dimethylbe	Synonyme I Inzene	Watery liquid	Coloriess	Sweet odor
yloi		Floata on water. f	Flammable, initating va	apor la producad.
Stop dr. Call fire Avoid c Isolate Notify I	scherge if possible department, contact with liquid and remove disct local health and p	e. Keep people away and vapor. harged matenal, ioliution control agen	y.	
Fire	FLAMM/ Fisshbac Vapor m Wear self Extinguisi Water m Cool exp	ABLE ck along vapor trait r lay explode if ignted t-contained breathing in with foam, dry che by be ineffective on losed containers with	may occur. I in an enclosed area. 3 spearatus. mincal, or carbon dioxic fire. 3 water	de.
Exposu	CALL FO VAPOR Imitanie Com It inter- Move to It breathin It breathin It breathin It breathin It breathin It breathin It breathin It swallo Remove Filush aft IF IN EY IF SWAL O' IT DO NOT	R MEDICAL AID. to eyes, nose, and i d, will cause headed scousness. Iresh sr. ng has stopped, gwe ng is difficult, gwe or to skin and eyes. to skin and eyes. to skin and eyes. to skin and eyes to skin and eyes	throat. ine, difficuit breathing, i artificial respiration, sygen, sign, conting, or loss i rg and shoes, ni and fligan with plent is CONSCIOUS, have ;	or loss of of consciousness. y of water victum dimik water
Water Pollutio	n Notify or Notify or	UL TO AQUATIC LIF to shorekne, dangerous if it enter cel health and wildlifk verators of nearby w/	E IN VERY LOW CO 3 water intakes. e officials. ater intakes.	NCENTRATIONS.
1. F (See Res Isaue Evecu Shouk Chem	IESPONSE TO DIS uponae Methode i warning-high flamu jate area d be removed wcal and physical i	SCHARGE Hendbook) mability treatment	2. LABEL 2.1 Category: F 2.2 Class: 3	Rammable liquid
3. ( 3.1 CG Comy Hydro 3.2 Formula: 3.3 IMO/UN : 3.4 DOT ID # 3.5 CAS Reg	HEMICAL DESIGI setibility Cleas: A carbon m-CaH4(CH3)s Designation: 3.2/ No.: 1307 Jatry No.: 108-38	NATIONS romatic 11307 -3	4. OBSER 4.1 Physical St 4.2 Color: Color 4.3 Odor: Like I	VABLE CHARACTERISTICS ate (se shipped): Liquid rises benzene; charactenetic aromatic
5.1 Persone plastic 5.2 Symptol skin, i edenv	A Protective Equi ; gloves and boots me Following Ex; if taken into lungs, a, if ingested, cau	5. HEAL Ipment: Approved or a. posure: Vapors caus , causes severe coup see nauses, vombrir	TH HAZARDS anister or ur-supplied i se headache and duzzi ghing, distress, and raj g, cramps, headache, .	mask; goggles or face shield; nesa. Liquid irtitates eyes and pidly developing pulmonary and coma; can be fatal. Kidney
and iv L3 Treatme oxyger flush v 5.4 Threaho 5.5 Short Te 5.6 Toxichy 5.7 Late To: 5.8 Vapor (C system 5.9 Liquid o nemar 5.10 Odor Th 5.11 IDLH Ve	er damage can or er damage can or er da Exposure: 1 vith water for at le de Limit Value: 10 by Ingestion: Gr iscity: Kdiney and dea) Initiant Cher in I present in higi ir Solid Initiant Ch in may cause small weshold: 0.05 pp sker: 10,000 ppm	ccur. INHALATION: nemove ast 15 min. SKIN: w X0 ppm mits: 300 ppm for 33 ade 3; LDss = 50 tc ilver damage. secteristics: Vapors h concentrations. Th haracteristics: Minir riting and reddening m	e to fresh ar; edminis 4: do NOT induce von npe off, wash with soa 2 mm. 5 500 g/kg Cause a sight smartin e effect is temporary. num hazard. If spilled of the skin.	iter artificial respiration and hiting; call a doctor. EYES: Ip and water. Ig of the eyes or respiratory on clothing and allowed to
reman 1.10 Odor Th 1.11 IDLH Ve	, may cause smar reshold: 0.05 ppr lue: 10,000 ppm	ting and reddening o n	of the skin.	

X,

_			
	6. FIRE HAZARDS		10. HAZARD ASSESSMENT CODE
6.1	Flash Point: 84*F C.C.	Ģ	See Hazard Assessment Handbook)
6.2 - 1	Flammable Limits in Air: 1.1%-6.4%		A-T-U
8	chemical, or carbon dioxide		
6,4	Fire Extinguishing Agents Not to be lineart Water may be ineffective.		
6.5	Special Hazards of Combustion		11. HAZAKU LLASSIFILATIONS
	Products: Not pertinent	11.1	Code of Federal Regulations: Stammable locat
6.6	and may travel considerable distance to a	11.2	NAS Hazard Rating for Bulk Water
	source of ignition and flash back.		Transportation:
6.7	Ignition Temperature: 966°F		Category Hating
6.9	Surning Rate: 5.8 mm/mm.		Health
6,10	Adlabatic Flame Temperature:		Vapor Imtant
6.11	Data not available Stoichiometric Air to Fuel Ratio:		Liquid or Solid Imaint
6.12	Data not available Fisme Temperature: Data not available		Water Polution Human Toxicity
			Aquabo Toxicity
	7. CHEMICAL REACTIVITY		Reactivity
7.1	Reactivity With Water: No reaction		Uther Chemicals
·	reaction		Self Reaction 0
7.3	Stability During Transport: Stable	11.3	NFPA Hazard Classification:
7.4	Neutralizing Agents for Acids and Caustics: Not partnert		Health Hazard (Sive)
7.5	Polymerization: Not pertinent		Flammability (Red)
7.6	Inhibitor of Polymerization:		Heactivity (Yellow)
7.7	Not perthent Notar Ratio (Reactant to		
••	Product): Data not evalable	]	
	Harcovity Group: 32		
		12.	PHYSICAL AND CHEMICAL PROPERTIES
		12.1	Physical State at 15°C and 1 atric
		12.2	Liquid Molecular Weight: 106.18
		12.3	Bolling Point at 1 stm:
		12.4	269.4°F = 131.9°C = 405.1°K Freezing Point:
	8. WATER POLLUTION	17.6	54.2°F =47.9°C = 225.3°K
8.1	Aquatic Toxicity:	12.0	650.8"F = 343.8"C ~ 617.0"K
	22 ppm/96 hr/bluegit/TL_/trash water	12.5	Critical Pressure:
	Biological Oxygen Demand (BOD):		MN/m <sup>a</sup>
••	0 lb/ib, 5 days; 0% (theor.), 8 days	12.7	Specific Gravity:
6.4	Dista not available	12.8	0.864 at 20°C (liquid) Liquid Surface Tension:
			28.6 dynes/cm = 0.0286 N/m at 20°C
		12.9	36.4 dynes/cm = 0.0364 N/m at 30°C
		12.10	Vapor (Gas) Specific Gravity:
		12.11	Ratio of Specific Heats of Vapor (Gas);
		12,12	1.071 Latent Heat of Vaporization:
			147 Btu/lb = 81.9 cal/g =
	9. SHIPPING INFORMATION	12, 13	Heat of Combustion: 17.554 Btu/Ib
9.1	Grades of Purity: Research: 99.99%;	12 14	-9752.4 cal/g =408.31 X 10 <sup>4</sup> J/kg
	Pure: 99.9%; Technical: 99.2%	12.15	Heat of Solution: Not persnent
9.2	Storage Temperature: Ambient	12.16	Heat of Polymerization: Not pertinent
8.4	Venting: Open (flame arrester) or	12.25	Uniting Value: Data not avaiable
	pressure-vacuum	12.27	Reid Vepor Pressure: 0.34 psa
	NO	TES	

# POLYCHLORINATED BIPHENYL

Notified reals and blacks office agreeds       11.3         Fire       Demostration         Fire       Cult. FOR MEDICAL ALD.         Update State of the S	10. HAZARD ASSESSMENT CODE (See Hazard Assessment Handbook) II 11. HAZARD CLASSIFICATIONS 11.1 Code of Federal Regulations:	10. HAZJ (See Hazar 11. HAZ 11.1 Cods of 1	IRE HAZARDS t: Jantis in Air: available uishing Agents: Water, Ioam, scal, or carbon dioxide uishing Agents Not to be of pertinent zards of Combustion	<ol> <li>FIRE HAZ</li> <li>Flash Point: &gt;286°</li> <li>Flammable Limits = Data not avalable</li> <li>Fire Extinguishing dry chemical, or of Fire Extinguishing Used: Not perfunce</li> <li>Special Hazards of</li> </ol>	6.1 6.2 6.3 6.4 8.5	Light yellow liquid, or Weak odor white powder	nyma Oity liquid to solid powder Sinks in water. Sinks in water. irge I possible. Keep people awi tot with liquid and solid. partment. remove discharged material.	Common Synon PCB Chionnated biphenyl Arochior Halogenated waxes Potychloropolyphenyls Stop dischai Avoid conta Call fre deg Isolate and
Cut: FON MEDICAL AD Drawing by Sale of Print Pictor Sole of a start of Print Pictor Sole of a start of print of water Print Pictor Sole of a start of print of water Print Pictor Sole of a start water relation Point of the sole of the of the sole Point of the sole of the sol	UHM-E 11.2 NAS Hazard Rating for Bulk Water Transportation: Not listed 11.3 NFPA Hazard Classification: Not listed	a: Infating gases are generated in Firs: Not pertinent imperature: Data not available fazard: Not pertinent tis: Data not available Flame Temperature: available stric Air to Fuel Ratio: available sperature: Data not available	Products: Instain in fires. Behavior in Fire: N Ignition Temperatu Electrical Hazard: I Burning Rate: Data 0 Adiabatic Fiame Tr Data not available 1 Stolchiometic Air Data not available 2 Fiame Temperature	6.6 6.7 6.8 6.9 6.10 6.11 6.12	rcies.	Nealth and pollution control age	Fire	
Water Pollution       HARMFUL TO AQUATE LIFE IN VERY LOW CONCENTRATIONS. May be dragedout if a times water indukes. Notify Contense of Industry water indukes. Notify Contense of Industry water indukes.       12. Fit         1. RESPONSE TO DISCHAREE (See Response Methods headbook) Issue water part ordination and upper induces. Commod and physical treatment       2. LABLE       12. Catagory: None 2.1 Second 3.1 Catagory: None 3.1 C			MICAL REACTIVITY With Water: No reaction with Common Materials: No ring Transport: Stable Agents for Acids and : Not pertnent lon: Not pertnent Polymerization: nent (Reactant to : Data not available iroup: Data not available	7. CHEMICAL R Reactivity With Wat Reactivity with Com reaction Stability During Trai Caustica: Not per Polymerization: Not Inhibitor of Polymer Not pertnent Molar Ratio (React Product): Data n Reactivity Group: D	7.1 7.2 7.3 7.4 7.5 7.6 7.7 7.8	nty of water in and liush with pienty of water.	CALL FOR MEDICAL AID. LIQUID OR SOLID Imating to skin and eyes. Flush affected areas with ple IF IN EYES, hold eyends ope	Exposure
1. RESPONSE TO DICCHARGE       2. LABL       12. 6 Cr         (See Response Methods Handbook)       2.1 Category: Noce       2.2 Date and containment       0.278 ppm/56 mVblogil/TL_//sen       12.8 Cr         Should be enrowed       2.1 Category: Noce       2.2 Date: Not pertinent       0.278 ppm/56 mVblogil/TL_//sen       12.8 Cr         3. CHEMICAL DESIGNATIONS       4. OBSERVABLE CHARACTERISTICS       1.1 Physical State (as shipped): Lopid       0.278 ppm/56 mVblogil/TL_//sen       12.8 Cr         3. CHEMICAL DESIGNATIONS       4. OBSERVABLE CHARACTERISTICS       1.1 Physical State (as shipped): Lopid       0.28 Food Chain Concentration Potentiat       12.1 Re         3. 10C/T M Designation: Not isted       4.2 Coort: Plais yellow (liquid): coortess       1.3 Short Fine: State (as antipped): Lopid       1.2 Sime Fine: State (as shipped): Lopid       1.2 Sime Fine: State (as shipped): Lopid         3.1 GCA Compatibility Class: Not isted       4.2 Coort: Plais yellow (liquid): coortess       1.3 Short Fine: State (as antipped): Lopid       1.3 Sime Fine: State (as shipped): Lopid         3.1 BOT/D ND Esegortize: Not issid       4.2 Coort: Plais yellow (liquid): coortess       1.3 Create of Purity: 11 grades (some loquid): Some solds) which dive wellow (liquid): coortess       1.3 Short Fine: State of wellow (liquid): coortess         5.1 Personal Protective Equipment: Gloves and motion:       1.3 Create of Purity: 11 grades (some loquid): Some solds) which dive contenet (locortes)       1.3 Short Fine: State of	<ol> <li>PHYSICAL ANO CHEMICAL PROPERTING</li> <li>Physical State at 15°C and 1 atm: Solid</li> <li>Molecular Weight: Not pertnent</li> <li>Boiling Point at 1 atm: Very high</li> <li>Freezing Point: Not pertnent</li> </ol>	12. PHYSICA 12.1 Physical Solid 12.2 Molecula 12.3 Boiling P 12.4 Freezing				E IN VERY LOW CONCENTRATIONS. water intakes. e officials ater intakes.	HARMFUL TO AQUATIC LIF May be dangerous if it enters Notity local health and wild/if Notity operators of nearby w	Water Pollution
3. CHEMICAL DESIGNATIONS       4. OBSERVABLE CHARACTERISTICS         3.1 GC Compatibility Class: Not listed       4. Observable Characteristics         3.2 Formula: (C):H1:s, C),       4. Observable (I):C, C): C, C):C, C, C	12.5     Critical Temperature: Not pertinent       12.6     Critical Pressure: Not pertinent       12.7     Specific Gravity: 1.3—1.8 at 20°C (liquid)       12.8     Liquid Surface Tension: Not pertinent       12.9     Liquid Water Interfacial Tansion: Not pertinent       12.10     Vapor (Gas) Specific Gravity: Not pertinent       12.11     Ballo of Specific Hasts of Vapor (G	12.5 Critical T 12.6 Critical F 12.7 Specific 1.3	TER POLLUTION Icity: m/96 hr/bluegill/TL_/fresh m/336-1080 //TL_/salt water oxicity: LD== 2000 ppm duck)	<ol> <li>WATER POL</li> <li>Aquatic Toxicity:</li> <li>0.278 ppm/96 hr/ water</li> <li>0.005 ppm/336-11</li> <li>N/pinfish/TL_/sa</li> <li>Waterfowl Toxicity:</li> <li>(mailard duck)</li> </ol>	8.1 8.2	2. LABEL 2.1 Category: None 2.2 Class: Not pertinent	NSE TO DISCHARGE Methods Handbook) g-water contaminant emoved d physical treatment	1. RESPO! (See Response Issue warrun Should be re Chemical and
5. HEALTH HAZARDS       9. SHIPPING INFORMATION         5.1 Personal Protective Equipment: Gloves and protective gaments.       9. SHIPPING INFORMATION         5.2 Symptoma Following Exposure: Acre from skin contact.       9. Gradea of Purity: 11 grades (some liquid, some solids) which differ primarily in their chlorine content (20%-68% by weight)         5.1 Totacity by Ingestor: Grade 2: oral rat LDue = 3980 mg/kg       9. Stortfor: Gause chromosome in rats, birth defects in birds         5.3 Vapor (Gas) Irritant Charecteriaties: Vapor cause avere inflation of eyes and throat and cause eye and lung inpury. They cannot be tolerated even at low concentrations.       9.4 Venting: Open         5.11 IDLH Value: 5 to 10 mg/m <sup>3</sup> NOTES	Not pertinent 12.12 Latent Heat of Vaporization: Not pertinent 12.13 Heat of Combustion: Not pertinent 12.14 Heat of Decomposition: Not pertinent 12.15 Heat of Solution: Not pertinent 12.16 Heat of Polymerization: Not pertinent 12.25 Heat of Fusion: Data not available 12.26 Limiting Value: Data not available	Not p 12.12 Latent H Not p 12.13 Heat of ( 12.14 Heat of ( 12.15 Heat of ( 12.15 Heat of ( 12.25 Heat of ( 12.26 Limiting	xygen Demand (BOD): Concentration Potential:	Biological Oxygen E Very low Food Chain Concen High	8.4	<ol> <li>OBSERVABLE CHARACTERISTICS</li> <li>Physical Stata (as shipped): Liquid or solid</li> <li>Color: Pale yellow (liquid); coloriess (solid)</li> <li>Odor: Practically odoriess</li> </ol>	CAL DESIGNATIONS Ity Class: Not listed http://class: http://class: http://classical.	3. CHEMIC 3.1 CG Compatibili 3.2 Formula: (Ciali 3.3 IMO/UN Design 3.4 DOT ID No.: 23 3.5 CAS Regiatry N
NOTES	12.37 Reid Vapor Pressure: Data not avail	12.27 Reid Ver	PPING INFORMATION writy: 11 grades (some liquid, ids) which differ primarity in wine content (20%-58% by npersture: Ambient phers: No requirement en	9. SHIPPING IN Grades of Purity: 11 some solids) white their chlorine con weight) Storage Temperatur Inert Atmosphere: N Vanting: Open	9.1 9.2 9.3 9.4	TH HAZARDS protective garments. kin contact up and water. able = 3980 mg/kg lities in rats, birth defects in birds cause severe irritation of eyes and throat and plerated even at low concentrations. lact with skin may cause irritation.	5. HEAL tective Equipment: Gloves and slowing Exposure: Acne from Exposure: SKIN: wash with so nit Value: 0.5 to 1.0 mg/m <sup>3</sup> histation Limits: Data not avail gestion: Grade 2; oral rat LD+o thatistich Limits: Data not avail gestion: Grade 2; oral rat LD+o recurso thromosomal abnorms rithant Charecteriatics: Vapors and lung inpury. They cannot be t d Irritant Charecteriatics: Con bid: Data not available 5 to 10 mg/m <sup>9</sup>	<ul> <li>5.1 Personal Prote</li> <li>5.2 Symptoma Foi</li> <li>5.3 Treatment of 1</li> <li>5.4 Threahold Lim</li> <li>5.5 Short Term in</li> <li>5.6 Toxicity by in</li> <li>6.7 Late Toxicity:</li> <li>8 Vapor (Gas) in</li> <li>cause eye and</li> <li>6.10 Odor Threahol</li> <li>5.11 IDLH Value: 5</li> </ul>
	)TES	 DTES						

# OILS, MISCELLANEOUS: MOTOR

.

ОМТ

Common Synonyms         Oily liquid         Yellow-brown         Lube oil odor           Crankcase oil Lubricating oil         Floats on water.         Floats on water.         Floats on control state           Stop discharge if posable. Call fire department. Avoid contact with liquid. Isolate and remove discharged material. Notity local health and pollution control agencies.         Notity local health and pollution control agencies.			6. FIRE HAZAROS     6.1 Flash Point: 275—600°F C.C.     6.2 Flammable Limits in Air: Data not available     6.3 Fire Extinguishing Agents: Dry chemcal, foam, or carbon dioxide     6.4 Fire Extinguishing Agents Not to be Used: Water may be ineffective     6.5 Special Mazards of Combustion Products: Not pertnent     6.6 Behavior In Fire: Not pertnent     6.7 Ignition Temperature: 325—625°F	10. HAZARD ASSESSMENT CODE (See Hazard Assessment Handbook) A-T-U 11. HAZARD CLASSIFICATIONS 11.1 Code of Federal Regulations: Not fisted 11.2 NAS Hazard Rating for Bulk Water Transportation: Not fisted	
Fire	Combustible. Extinguish with dry chemica Water may be ineffective or Cool exposed containers wr	I, foarn or carbon dioxode. 5 fre. 5 water.	<ul> <li>6.8 Electrical Hazard: Not pertinent</li> <li>6.9 Burning Rate: 4 mm/min.</li> <li>6.10 Adlabatic Flame Temperature: Data not available</li> <li>6.11 Stolchlometric Air to Fuel Ratio: Data not available</li> <li>6.12 Flame Temperatura: Data not available</li> </ul>	11.3 NFPA Hazard Clessification: Not listed	
Exposure	CALL FOR MEDICAL AID. LIQUID Imating to skin and eyes. Harmbil if swallowed. Flush affected areas with pill for the swallowed of the swallow for the swallow of the swallow of the swallow of the swallow of the swallow of the swallow DO NOT INDUCE VOMITIN	ing and shoes. entry of water. en and fluan with plenty of water. is CONSCIOUS, have victim drink water G.	<ol> <li>CHEMICAL REACTIVITY</li> <li>Reactivity With Water: No reaction</li> <li>Reactivity with Common Materials: No reaction</li> <li>Reactivity of the Common Materials: No reaction</li> <li>Stability During Agents for Acids and Caustics: Not pertnent</li> <li>Inhibitor of Polymerization: Not pertnent</li> <li>Inhibitor of Polymerization: Not pertnent</li> <li>Not pertinent</li> <li>Molar Ratio (Reactant to Product): Data not available</li> <li>Reactivity Group: 33</li> </ol>		
Water Pollution	Effect of low concentrations Fouling to shoreline. May be dangerous if it enter Notly local health and wildli Notry operators of nearby w	on aquatic life is unknown. s water intakes. le officals. vater intakes.		12. PHYSICAL AND CHEMICAL PROPERTIE 12.1 Physical State at 15°C and 1 atm: Liquid 12.2 Molecular Weight Not pertnent 12.3 Bolling Point at 1 atm: Very high 12.4 Freezing Point:	
I. RESPONSE TO DISCHARGE     (See Response Methods Handbook)     Mechanical containment     Should be removed     Chemical and physical treatment			8. WATER POLLUTION     8.1 Aquetic Toxicity: Data not available     8.2 Watertowl Toxicity: Data not available     8.3 Biological Oxygen Demand (BOD):     Data not available     8.4 Food Chain Concentration Potential:     None		
3. CHEMICAL DESIGNATIONS     4. OBSERVABLE CHARACTERISTICS       3.1 CG Compatibility Class: Miscellaneous Hydrocarbon Mixtures     4.1 Physical State (as shipped): Liquid       3.2 Formula: Not applicable     4.2 Color: Yellow fluorescent       3.3 IMO/UN Designation: 3.3/1270     4.3 Odor: Charactenstic       3.4 DOT ID No:: 1270     3.5 CAS Regiatry No:: Data not available				at 20°C 12.10 Vapor (Gas) Specific Gravity: Not perinent 12.11 Ratio of Specific Heats of Vapor (Gas Not perinent 12.12 Latent Heat of Vaporization: Not perinent 12.13 Heat of Combustion: ~18,486 Btu/lb = ~10,270 ca/g = ~429.98 X 10° J/ 12.14 Heat of Decomposition: Not continent	
<ul> <li>5.1 Personal Prot</li> <li>8.2 Symptoms For frequency of but may bec</li> <li>5.3 Treatment of treatment of senal chest</li> <li>5.3 Threshold Lin</li> <li>6.4 Threshold Lin</li> <li>6.5 Short Term in</li> <li>6.6 Toxicity by in</li> <li>6.8 Toxicity by in</li> <li>8.4 Yapor (Cas) in system if p</li> <li>4.5 Uquid or Soli- remain, may</li> </ul>	5. HEAL settive Equipment: Protective e illewing Exposure: INGESTIOI bowei passage may occur ASI bome more severe several hours Exposure: INGESTION: do NC obably not required; delayed de varys. EYES: wash with copiou ter. hit Value: Data not available halaition Limite: Data not available halaition Limite: Data not available halaition Limite: Data not available thalaition Limite: Data not available data not available rithant Characteristics: Min clause smarting and reddening	TH HAZARDS gloves; goggles or face shield. N: minimal gastrointestinal intration; increased PIRATION: pulmonary intration is normally minimal after exposure. DT lavege or induce vomiting. ASPIRATION: velopment of pulmonary imitation can be detected by amounts of water. SKIN: wipe off oil and wash with lable b 15 g/kg t cause a slight amarting of the eyes or respiratory e effect is temporary. imum hazard. If spilled on clothing and allowed to of the skin.	<ol> <li>SHIPPING INFORMATION</li> <li>Grades of Purity: Various viscosities</li> <li>Storage Temperature: Ambient</li> <li>Inert Atmosphere: No requirement</li> <li>Venting: Open (flame arrester)</li> </ol>	12.15 Heat of Solution: Not periodent 12.15 Heat of Solution: Not periodent 12.25 Heat of Polymertzston: Not pertnent 12.25 Heat of Fuelon: Data not available 12.27 Reid Yapor Pressure: Data not available	
5.10 Odor Thread 5.11 IDLH Value: 1	HE: Data not available Deta not available			IU I E S	

## BENZENE

Common Synonyms Watery liquid Coloriess Benzol Benzole Floats on water. Flammable, irritating v point is 42°F.		Coloriess Gasoline-like odor Flammable, irritating vapor is produced Freezing 42°F.	<ol> <li>FIRE HAZARDS</li> <li>Flash Point: 12°F C.C.</li> <li>Flasmmable Limits in Air: 1.3%-7.9%</li> <li>Fire Extinguishing Agents: Dry chemical, loam, or carbon dixude</li> </ol>	10. HAZARD ASSESSMENT CODE (See Hazard Assessment Handbook) A-T-U-V-W	
Avoid contac Wear goggle Shut of ignitu Stop dischar Stay upwind Isolate and r Notify local h	t with liquid and vapor. Keep s and self-contained breathing ion sources and call the depa ge if possible and use water spray to "knoc emove discharged matenal, realth and pollution control ag	people away, apparatus, riment k down" vapor. encies.	<ul> <li>6.4 Fire Extinguishing Agents Not to be Used: Water may be ineffective</li> <li>6.5 Special Hazards of Combustion Products: Not perment</li> <li>6.6 Behavior in Fire; Vapor is heaver than ar and may travel considerable distance to source of igmition and flash back</li> </ul>	11. HAZARD CLASSIFICATIONS 11.1 Code of Federal Regulationa: Flammable liquid 11.2 NAS Hazard Rating for Bulk Weter Trensportation:	
Fire	FLAMMABLE. Flashback along vapor trail may occur. Vapor may explode if ignited in an enclosed area. Waar goggles and self-contained breating apparatus. Extinguist with dry chemical, loam, or carbon dioxide. Water may be ineffective on time Cool exposed containers with water		<ul> <li>5.7 Ignition Tempersture: 1097/F</li> <li>6.8 Electrical Hazard: Class I, Group D</li> <li>6.9 Burning Rete: 5.0 mm/min</li> <li>6.10 Adlabatic Flame Temperature: Data not available</li> <li>6.11 Stoichiometric Air to Fuel Ratio: Data not available</li> <li>5.12 Flame Temperature: Data not available</li> </ul>	Category     Reing       Fre     3       Health     1       Liquid or Solid Irritant     1       Poisons     3       Water Poliution     4       Human Toxicity     3       Aquatic Toxicity     1	
Exposure	CALL FOR MEDICAL AID. VAPOR Infating to eyes, nose and throat. If inhaled, will cause headache, difficult breathing, or loss of conscousness. Move to itresh aik. If breathing has stopped, give artificial respiration If and the stopped give artificial respiration IF IN EYESIED areas with pienty of water. IF IN EYESIED and victim is CONSCIOUS, have victim drink water or muk.		<ol> <li>CHEMICAL REACTIVITY</li> <li>Reactivity With Water: No reaction</li> <li>Reactivity with Common Materials: No reaction</li> <li>Stability During Transport: Stable</li> <li>Neutralizing Agents for Acide end Caustics: Not pertinent</li> <li>Polymerization: Not pertinent</li> <li>Inhibitor of Polymerization: Not pertinent</li> <li>Molar Ratio (Reactant to Product): Data not available</li> <li>Reactivity Group: 32</li> </ol>	Aesthetic Ener	
Water Pollution	MARMFUL TO AQUATIC L May be dangerous if it ent Notify local featin and wild Notify operators of nearby NSE TO DISCHARGE	IFE IN VERY LOW CONCENTRATIONS rs water intakes. ife citicais water intakes 2. LABEL	8. WATER POLLUTION	12.         PHYSICAL AND CHEMICAL PROPERTIES           12.1         Physical State at 15°C and 1 atm: Liquid           12.2         Molecular Weight 78 11           12.3         Boiling Point at 1 stm: 176°F = 80.1°C = 353.3°K           12.4         Freezing Point 42.0°F = 5.5°C = 278.7°K	
(See Response Issue warnin Restrict acce 3. CHENI	e Methods Handbook) ig-high flammability ess cas CAL DESIGNATIONS	2.1 Category: Flammable liquid 2.2 Cless: 3 4. OBSERVABLE CHARACTERISTICS	<ol> <li>Aquatic Toxicity: 5 ppm/6 hr/minnow/lethal/distilled water 20 ppm/24 hr/sunfish/TL_/tap water 8.2 Waterfowl Toxicity: Data not aveilable 8.3 Biological Oxygen Demand (BOD): 1.2 ib/lb, 10 days 8.4 Food Chain Concentration Potential: None</li> </ol>	12.5         Critical remperature:           552 0°F = 288.9°C = 562.1°K           12.6         Critical Pressure:           710 psia = 48.3 atm = 4.89 MN/m²           12.7         Specific Gravity:           0.879 at 20°C (liquid)           12.8         Liquid Surfece Tension:           28.9 dynes/cm = 0.0289 N/m at 20°C           12.9         Liquid Water Interfacial Tension:           35.0 dynes/cm = 0.035 N/m at 20°C	
3.1 CG Compatibili Hydrocarbon 3.2 Formula: C+He 3.3 IMO/UN Design 3.4 DOT ID No.: 11 3.5 CAS Registry h	Ity Class: Aromatic nation: 3.2/1114 14 No.: 71-43-2	<ul> <li>4.1 Physical State (as shipped): Liquid</li> <li>4.2 Color: Coloriess</li> <li>4.3 Odor: Aromatic; rather pleasant aromatic odor: characteristic odor</li> </ul>		12.10 Vapor (Gas) Specific Gravity: 2.7 12.11 Ratio of Specific Hests of Vapor (Gas): 1.061 12.12 Latent Hest of Veporization: 169 Btu/lb = 94 1 cal/g = 3.94 x 10 <sup>5</sup> J/kg 12.13 Hest of Combustion:17.460 Btu/lb = -9698 cal/g =406.0 x 10 <sup>5</sup> J/kg	
<ol> <li>Personal Proti hydrocarbon hydrocarbon</li> <li>Symptome Fo headache, b</li> <li>Trestment of i contaminate INHALATIO alopped, sta</li> <li>Short Term In</li> <li>Short Term In</li> <li>Torkithy by In</li> <li>Late Toxkithy</li> </ol>	5. HE/ ective Equipment: Hydrocari insoluble rubber or plastic glo insoluble apron such as neop libwing Exposure: Dizzness weathlessness, chest construct Exposure: SKIN: flush with v d clothing and wash skin. EY6 W: remove from exposure imm int value: 10 ppm intersuscitation, administer on nit Value: 10 ppm inbalation Limits: 75 pm for gestion: Grade 3, L0xo = 56 Leukema rithant Characteriation: If ore	LLTH HAZARDS son vapor canister, supplied air or a hose mask; wes, chemical goggles or face splash sheek; rene. excitation, pallor, followed by flushing, weakness, on. Coma and possible death. rater followed by soap and water; remove (S): flush with plenty of water until imitation subsides adiately. Call a physician. IF breathing is inegular or gen. 30 min. 0 to 500 mg/kg sent in high concentrations, vapors may cause	<ol> <li>SHIPPING INFORMATION</li> <li>Grades of Purity: Industnal pure</li></ol>	<ul> <li>12.14 Hest of Decomposition: Not pertinent</li> <li>12.15 Hest of Solution: Not pertinent</li> <li>12.16 Hest of Polymerization: Not pertinent</li> <li>12.27 Hest of Fuelor: 30.45 cal/g</li> <li>12.26 Limiting Yalue: Data not available</li> <li>12.27 Reid Vapor Pressure: 3.22 psis</li> </ul>	
mitation of e s.9 Liquid or Solk remain, may 5.10 Odor Thresho 5.11 IDLH Vsiue: 2	eyes or respiratory system. Thi d Irritant Characteristica: M cause smarting and recidenin aid: 4.68 ppm 2,000 ppm	effect is temporary. nimum hazard. If spilled on clothing and allowed to g of the skin		NOTES	

# ETHYLBENZENE

Common Synonyma EB       Liquid       Coloriess       Sweet, gasoline-like odor       6. FIRE HAZARDS       10. HAZARD ASSESSMENT CODE         Phenylethane EB       Floats on water. Flammable, imitating vapor is produced.       6. FIRE HAZARDS       10. HAZARD ASSESSMENT CODE         Avoid contact with liquid and vapor. Koop people away. Wear googles, self-contained breating apparents, and rubber overclothing Shut of lightbon sources).       Avoid contact with liquid and vapor. Koop people away. Wear googles, self-contained breating apparents, and rubber overclothing       6.4 Fire Extinguishing Agents Not to be Used: Not perfinent       11. HAZARD CLASSIFICATIONS         Shut of lightbon sources). Shut of lightbon sources are generated Stay rowind and use writer spray to "knock down" vapor. Isolate and teampre disclaring the more disclaring to Bulk Water       9. Secola Hazard Rating to Bulk Water	Liquid Colorless	Common Panone		
Avoid contact with liquid and vapor. Keep people away. Wear goggles, seli-contained breathing apparatus, and rubber overclottwing (including gloves). Shut off lighton sources and call fre desurtment. Stop discharge it possible Stay towind and use writer spray to "knock down" vapor. Isolate and removed discharged material.	Common Synonyms Liquid Colorless Sweet, gasoline-like Phanylethane EB Floets on water. Flammable, imitating vapor is produced.			
Notify local health and pollution control agencies.         6.6 Behavior in Fire: Vapor is heavier than air         Transportation:           ELANMADIC         and may travel considerable distance to         Category         Ratil	Avoid contact with liquid and vapor. Koep people awey. Wear goglies, sel-contained breating opparatus, and rubber overclothing Shut of instance gives). Shut of instances and call fire decartment. Stay climited and use writer spray to "invok down" vapor. Isolate and remove decharged manenal. Nothy local health and pollution control agencies. FLAMMABLE.			
Floating vacuum of the source of ignition and flash beck.       Free	<ul> <li>Commodel: Fissiback along vapor trail may occur.</li> <li>Vapor may explode if ignited in an enclosed area.</li> <li>Wear goggles, self-contained breathing apparatus, and rubber overclothing (including gioves).</li> <li>Extregular with dry chemical, foam, or carbon dioxide.</li> <li>Walor may be ineffective on fire.</li> <li>Cool exposed containers with water.</li> </ul>			
CALL FOR MEDICAL AID.       7. CHEMICAL REACTIVITY       Aesthetic Effect	CALL FOR MEDICAL AID. VAPOR Infating to eves, nose and throat. If inhaled, will cause dizzness or difficult breathing: More to fresh ar. If breathing is a sincipid, give autificial rospiration. If breathing is cificult, give oxygen. LiQuiD · Will burn skin and eves. Harmful if evailword. Romove contaminated clothing and shoes. Flush affocted areas with plenty of water. IF INE FES, hold evides open and fubah with plenty of water. IF SWALLOWED and victim is CONSCIOUS, have victim donk water or mil. EO NOT INDUCE VOMITING.			
Water         MARIMFUL TO AQUATIC LIFE IN VERY LOW CONCENTRATIONS. Fouring to shoreline. May be dangerous if it enters water intakes.         12. Physical State at 15°C and 1 atm:: Liquid           Pollution         Motify local health and wildlife officials. Notify operators of nearby water intakes.         12. Boiling Point at 1 atm:: Liquid           12. Physical State at 15°C and 1 atm:: Liquid         Liquid           12. Motify local health and wildlife officials. Notify operators of nearby water intakes.         12. Boiling Point at 1 atm:: 277.2°F = 108.2°C = 409.4°K           12.4 Freezing Point         12.4 Freezing Point	UL TO AQUATIC LIFE IN VERY LOW o shoreine. Sangerous if it enters water intakes. cal nealth and wildlife officials. eventors of nearby water intakes.	Water Pollution		
1. RESPONSE TO DISCHARGE     2. LABEL     8. WATER POLLUTION     12.5 Critical Temperature:       (See Response Methods Handbook)     2.1 Cetegory: Flammable liquid     2.2 Class: 3     3.1 Aquatic Toxicity:     12.5 Critical Temperature:       Mechanical containment     2.2 Class: 3     2.2 Class: 3     2.9 pm//96 hr/bluegi//TL_/tresh water     12.5 Critical Temperature:       Should be removed     2.2 Class: 3     2.8 Waterhowl Toxicity: Data not available     5.23 pia = 35.6 atm = 36.1 MN/       Chemical and physical treatment     2.4 Food Chain Concentration Potential:     12.8 Uquid Surface Tension:       None     2.9.2 Opto24 pm = 0.0292 N/m at	1. RESPONSE TO DISCHARGE     2. LABEL       (See Response Methods Handbook)     2.1 Category: Flammable liquid       Mechanical containment     2.2 Classe: 3       Should be removed     Chemical and physical treatment			
3. CHEMICAL DESIGNATIONS     4. OBSERVABLE CHARACTERISTICS     12.9 Liquid Water Interfacial Tension: 35.48 dynes/or = 0.03548 N/m i 20°C       1.1 CG Competibility Class: Aromatic hydrocarbon     4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess     20°C       1.2.5 Formula: C4.16/LisCHa     4.3 Odor: Aromatic     12.10 Vapor (Gas) Specific Gravity: Not pertinent       2.5 Formula: C4.16/LisCHa     4.3 Odor: Aromatic     12.10 Vapor (Gas) Specific Gravity: Not pertinent       3.1 IMO/UN Designation: 3.3/1175     4.3 Odor: Aromatic     12.11 Ratio of Specific Heats of Vapor (G 1.071       1.5 CAS Registry No: 100-41-4     12.12 Latent Heat of Vaportzation: 144 Blu/bl = 80.1 ca/g = 37.5 X 10 / br;	3. CHEMICAL DESIGNATIONS     4. OBSERVABLE CHARACTERISTICS       3.1 CG Compartibility Class: Aromatic hydrocation     4.1 Physical State (as shipped): Liquid       3.2 Formula: CeHicCHaCHa     4.2 Color: Coloriesa       3.3 IMO/UN Designation: 3.3/1175     4.3 Odor: Aromatic       3.4 DOT ID No.: 1175     3.5 CAS Registry No.: 100-41-4			
<ul> <li>5. HEALTH HAZARDS</li> <li>1. Personal Protective Equipment: Self-contained breating apparatus; safety goggles.</li> <li>2. Symptoms Following Exposure: Inhalation may cause initiation of nose, dzzines, depression. Moderate initiation of eye with corneal injury possible. Initiates skin and may cause bisters.</li> <li>3. Treatment of Exposure: INHALATION: If ill effects accur, remove withing itops, give artificial respiration. INGESTION: induce vomting only upon physician's approval; material in lung may cause chemical pneumonitis. SKIN AND EYES: prompty flush with plenty of water (15 min. for eyes) and get medical atention; remove and wash contaminated clothing before neuse.</li> <li>4. Threahold Limit Value: 100 ppm</li> <li>5. Bord Term Inhalation Limits: 200 ppm for 30 min.</li> <li>6. Toxicity: Data not available</li> <li>8. Uquid or Solid Irritant Characteristics: Causes amarting of the skin and frat-degree burns on</li> </ul>	5. HEALTH HAZARDS present: Self-contained breathing app posure: Inhalation may cause initiatio e with corneal injury possible. Initiates INHALATION: If all effects occur, nem it medical help prompty; if breathing is initing only upon physician's approval; iskin AND EYES: prompty fluah with p mines and wash contaminated ch 100 ppm mite: 200 ppm for 30 min. rade 2; LDso = 0.5 to 5 g/kg (rat) valiable acteristica: Vapors cause moderate unpleasant. The effect is temporary.	I Personal Protect     Symptoms Folio     Moderate inits     Treatment of Es     warm and que     INGESTION: is     chemical pneu     and get medic     Short Term Inits     Short Term Inits     Toxicity by Inge     I.7 Late Toxicity: [     S Vapor (Gae) Init     find righ conc     Liquid or Solid		
ahort exposure; may cause secondary burns on long exposure. 5. 10 Odor Threshold: 140 ppm 5. 11 IDLH Value: 2.000 ppm 6. 11 Stoichiometric Air to Fuel Ratio: Data Not Available 6.12 Flame Temperature: Data Not Available	ehort exposure; may cause secondary burns on long exposure. 8.10 Odor Threshold: 140 ppm 8.11 IDLH Value: 2,000 ppm			

TTE

# TETRACHLOROETHYLENE

12.17 SATURATED LIQUID DENSITY		12.18 LIQUID HEAT CAPACITY		12.19 LIQUID THERMAL CONDUCTIVITY		12.20 LIQUID VISCOSITY	
Temperature (degrees F)	Pounds per cubic foot	Temperature (degrees F)	British thermal unit per pound-F	Temperature (degrees F)	British thermal unit-inch per hour- square foot-F	Temperature (degrees F)	Centipois
35	103 400	0	198		N	55	958
40	103.900	10	200			60	.556
45	102,000	20	201		T	65	900
50	102.500	30	202		'	70	873
55	-102.299	40	203		Р	75	848
60	102.000	50	204		F	80	823
65	101 700	60	205		R	85	800
70	101:400	70	206		T	<u>an</u>	.000
75	101.099	80	207			95	756
80	100 799	90	208		Ň	100	736
85	100 500	100	210		F	105	716
90	100 200	110	211		N	110	803
95	99,910	120	212		Ť	115	083
100	99.610	130	213			120	533 533
105	99.320	140	214			125	647
110	99.020	150	215			130	631
115	98,730	160	.216			135	616
120	98.429	170	.217			140	.601
125	98,139	180	.218			145	.588
130	97.839	190	.220			150	.574
135	97,549	200	.221			155	.561
140	97.250	210	.222			160	.549
145	96.959					165	.537
150	96.669					170	.526
155	96.370					175	.515
160	96.080						

-	12.21 SOLUBILITY IN WATER		12.22 SATURATED VAPOR PRESSURE		12.23 SATURATED VAPOR DENSITY		12.24 IDEAL GAS HEAT CAPACITY	
l er (de	nperature egrees F)	Pounds per 100 pounds of water	Temperature (degrees F)	Pounds per square inch	Temperature (degrees F)	Pounds per cubic foot	Temperature (degrees F)	British thermal unit per pound-F
	68.02	016	60	236	60	00702	0	108
	00.02		70	318	70	00929	25	110
			80	425	80	01216	50	113
		•	90	561	90	01575	75	116
			100	732	100	02022	100	118
			110	948	110	02571	125	120
			120	1 217	120	03242	150	122
			130	1.548	130	04055	175	125
			140	1.953	140	.05032	200	.127
			150	2.446	150	.06199	225	.129
			160	3.042	160	.07583	250	.131
	]		170	3.756	170	.09215	275	.132
			180	4.607	180	.11130	300	.134
			190	5.616	190	.13360	325	.136
			200	6.805	200	.15940	350	.138
			210	8.199	210	.18910	375	.139
			220	9.824	220	.22330	400	.141
			230	11.710	230	.26230	425	.142
			240	13.890	240	.30660	450	.143
			250	16.390	250	.35680	475	.144
			260	1 <b>9</b> .260	260	.41330	500	.146
			270	22.520	270	.47680	525	.147
			280	26.230	280	.54790	550	.148
							575	.148
							600	.149

•

# TRICHLOROETHANE

Common Synor 1,1,1-Trichloroethene Methylchloroform Aerothene	путте 🗸 🗸	Vatery liqued	Colories Sweet ador	&1 &2	6. FIRE HAZARDS Finah Point: Dela not available Planmable Limita in Altr. 7%-16%	10. HAZARU ASSESSMENT CODE (Bee Hazard Assessment Handbook)
Chiorothene			u	Fire Extingulating Agents: Dry chemical, foam, or carbon dioxide Fire Extinguishing Agents Not to be		
Stop dischar Avoid contak Call fire depi taolata and r Notity local I	rge if possible. K ct with liquid and artment. remove discharg hearth and politit	Keep people awa d vapor. ged material. rition control agen	y. cies.	6.5 6.5	Used: Not pertnent Special Hearrds of Combustion Products: Toxic and initiating gases are generated in firm. Behavior in Pire: Not pertinent Ignition Temperature: 932°F	12. NAZARD CLASSIFICATIONS 11.1 Code of Federal Regulations: ORM-A 11.2 NAS Hazard Rating for Bulk Water Transportation: Catabory Betting
Fire	Combustible, POISONOUS GASES ARE PRODUCED IN FIPE. Wear gogies and self-contained breathing apparetus. Extinguish with dry chemical, carbon dioxide, of foam.		6.3 6.1 6.1	Electrical Hazaric: Not partnerit Burning Rates (st) 29 mm/min. 2 Adlabatic Flame Temperature: Data not evaluable 1 Stolohomatrio Air to Fuel Ratio: Data not evaluable 2 Rame Temperature: Data not evaluable	The arth     1       Vepor Infrant	
Exposure	CALL FOR MEDICAL AID. VAPOR Initialing to eyes, noise and throat. If inhibited, will clause discusses or difficult breathing. Move to treach are a stopped, give artificial respiration. If breathing a difficult give artificial respiration. If breathing to skin and eyes, If eventories and experimentation of the states of		7.1 7.2 7.3 7.4 7.6 7.6 7.6 7.7 7.8	7. CHEMICAL REACTIVITY Reactivity With Water: Reacts slowly, releasing corrosive hydrochloric acid. Reactivity with Common Materials: Corrodes aluminum, but reaction is not hazardous. Stability During Transport: Stable Heutralizing Agents for Acids and Caustics: Not partiment Polymerization: Not partiment Inhibitor of Polymerization: Not partiment Bioler Retio (Reactant to Product): Dats not evailable Reactivity Group: 36	Assthetic Effect	
Water Pollution	Effect of low concentrations on acustic the is unknown. Nery be deroperous if it enters water intelkes. Notify local hearth and widthe officials. ION Notify operators of nearby water intelkes.					<ol> <li>PHYSICAL AND CHEMICAL PROPERTIES</li> <li>Physical State at 15°C and 1 atm: Liquid</li> <li>Biolecular Weight: 133.41</li> <li>Bolling Power 1 atm: 183°F = 74°C = 347°K</li> </ol>
1. RESPO (Bee Response Should be n Chemical an	1. RESPONSE TO DISCHARGE (See Response Methods Handbook) Should be removed Chamical and physical treatment       2. LABEL         3. ChEMICAL DESIGNATIONS       2.3 Class: Not partiment         3. CHEMICAL DESIGNATIONS       4. OBSERVABLE CHARACTERISTICS         1. CG Compatibility Class: Halogeneted hydrocarbon       4. OBSERVABLE CHARACTERISTICS         2. Calor: Coloreas       4.3 Odor: Chloroform-like; eventish         3. CHEMICAL DESIGNATIONS       4. OBSERVABLE CHARACTERISTICS         4. OBSERVABLE CHARACTERISTICS       4.1 Physical State (as shipped): Liquid 4.2 Color: Coloriess         4.3 Odor: Chloroform-like; eventish       4.3 Odor: Chloroform-like; eventish		4.1 6.2 6.3 6.4	8. WATER POLLUTION Aquastic Toxitoity: 75-150 ppm/*/pinfish/TL_/sailt water *Time period not apacified. Waterfowl Toxicity: Data not available Biological Oxygen Demand (BOD): Data not available Food Chain Concentration Potential:	12.4     Pressing Point:       <-38°F = <-39°C = <234°K       12.5     Critical Temperature: Not partinent       12.8     Gritical Pressure: Not partinent       12.7     Specific Gravity:       1.31 at 20°C (liquid)       12.8     Liquid Surface Tension:       25.4 dynas/cm = 0.0254 N/m at 20°C       12.9     Liquid Writer interfacial Tension: (est.)       45 dynas/cm = 0.045 N/m at 20°C	
3. CHEMI hydrocarbon 5.2 Formula: CH=C 5.3 MBC/UH Desig 5.4 DOT ID No.: 28 5.5 CAS Registry I				None	12.10 Vapor (Gae) Specific Heat of Vapor (Gae): 1.104 12.11 Ratio of Specific Heat of Vapor (Gae): 1.104 12.12 Latent Heat of Vaporization: 100 Bbs/b = 56 cal/g = 2.4 X 10 <sup>5</sup> J/kg 12.13 Heat of Combustion: (est.) 4700 Bbs/b = 2600 cal/g = 110 X 10 <sup>6</sup> J/kg 12.14 Heat of Decomposition: Not pertinent 1.15	
E.1 Personal Prot appartus to and taos shi neoprete or incoordinatic aphysistion tribulation ar SKIN: defatt 6.2 Trestment of exposures. ( INHALATIOI extraminister o thoroughly r with scop ar	Inclive Equipme or emergencies; i polyvinyt acontent allowing Exposus on to loss of com- ing action may or combined with nd may cause so ting action may or ting action action to action the source of the source action of the source acti	5. HEALT ent: Organic vep neoprine or poly satety alose (or it accountees; high cause of consciou ome feeling of ng cause dermatitie, the medical attentio ster adrenalin or 4 in to freeh eit; if in Oht; have victim it remove contain ppm	TH HAZAROS or-acid gas canister; self-contained breathing whyt-acohol-type gloves; chemical safety goggles safter safety shoes plus neoprene footwear); or splate protection. N: symptoms range from loss of equilibrium and concentration can be fattil due to simple ames. IN(ESTON: produces effects amiliar to uses. EYES: slightly initiating and lachtymetory. In for all eye exposures and any other serious over spinephrine; otherwas, imatmant is symptomatic. accessery, apply artificial requirition and/or drink waster and induce vormiting. EYES: flightly initiated clothing and weah exposed area thoroughly		9. SHIPPING INFORMATION Grades of Purity: Uninhibited; inhibited; industral inhibited; white room; cold cleaning Blanage Temperature: Ambient inert Atmosphere: No requirement Ventling: Pressure-vacuum	12.15 Heat of Solution: Not pertinent 12.16 Heat of Polymerization: Not pertinent 12.25 Heat of Fuelon: Data not evaluable 12.26 Limiting Value: Data not evaluable 12.27 Reid Vapor Pressure: 4.0 paia
<ul> <li>L.S. Bhort Term in</li> <li>S.S. Toxicity by in</li> <li>S.T. Late Toxicity;</li> <li>S.G. Vapor (Ges) is system if on</li> <li>S.S. Liquid or Soli remain, may</li> <li>S.10 Odor Threaty;</li> <li>S.11 IDLH Value;</li> </ul>	whaten soo y whaten Limits pestion: Grade rritant Characte seent in high co id inflant Characte y cause smarting old: 100 ppm 1,000 ppm	perm ic 1,000 ppm for a 1; LDss = 5 to able entettos: Vapors ncentrations. The actertatios: Mining g and reddening of	80 min. In man 15 g/kg (net, mouse, rabbit, guines pig) cause a slight emarting of the eyes or respiratory effect is temporary. num hazard. If epilled on clothing and allowed to of the ekin.		N	OTES
			· · · · · · · · · · · · · · · · · · ·			

JANUARY 1991

a a seconda da companya da Presente da companya da com

- WHERGEN WART CONTRACTOR , MADER 12. MILLER CONTRACTOR METAL. Boyden om elher en skull

TERRETER LINE FE

- Thus hubble sector sector is done and the sector sector is the

TOPE AND EXECUTION EXCLUSION

- No five topend is told torne toppen, white such that the set newder, on the form. Novem seake when veind to it product on having mean on trea flate on span s. If it descripts through the tot is top the fire vourself. Seek the help of firefugitury presentes. Containers may cutlibe when coponed to itse
- PUR RUALI FINER: use dry saro, underine draminus addi minus des traves. Or appropriate satul activusiating downers do pet apply water is burnlet superial.

- 产资益,13、13%3、日子学校)(13上)。

Share and The Ly

Short TARM FFRETE: New cause irritation. Asy takes alleroid reactions. Abolticnal opports may include couching, tever, asthma and luca damada, UDAR TARM REFERTER to addition to effects from short term excosure. Deficiention of the nose, runny nose and tack of sense of smell may occur. May also cause cancer.

an a chuirean a

SNORT TERM TREETS: May cause irritation. May cause alleroic reactions Additional effects may include recness and swelling of the eves, request and swelling of free couts, redness and swelling or the curs, rash. itims of fever, as has and lunc damage.

CORRECTERNALS TRACTER Rando effects an enable term considere.

EYE CONTACT:

SHOPE TERM SEPTERAL Has cause projustion.

LINE TERM TEREDYE: No information is available.

A CONTRACTOR OF A DAMA.

A-UR: TERU FRANCE: He is formation shallable on slonchicant arvess of tota-1 MM TERU FRANCES: New cause reproductive effects.

FilsT will because incomprosive area. If preathing has stopped, when the artificial respiration of CPA. Flush skinkeys with water for 15 minutes. If you tipe occurs, hald below hips to the ent application. Set realized attention prescripted, Sec MSDS for purther recommendations.

- A wrie Luger worsel reactively of the teach leadercasts with a reiterral react.

and the second second

Ragel on Colored a conservative state and examine a higher ware service of the service service states.

la companya se substanti de la companya de la compa Le Ulvanez do letro e rior entro a lo tonto a color a verto d

n na seminent egy den sola de rectored en la survival de la solar de la solar de la servición de la filosofic en solar ROMAN Guerri Media de la surviva de la caracteristica de la caracteristica de la survival de la surviv Internetion de la survival de la servición de la caracteristica de la solar de la survival de la servición de

NEEDER EENERAL ERITER DE LEERE EN DE L

.

•

。 1917年,1917年,全国大学研究中国教授公式会议的,让公司大学公司了一次,并不是一部公司人物公开被求人。1949年,1917年,1948年,1947年,1948年,1947年,1948年,1947年,1948年,19 niv by velicer when yorking in nighter whences compariset. At its fBVB com The Wilder And Andrew States of the second of the second second second second second second second second second

Reconstitution Health Services: Inc. for Engrance Source Information contacts (611) Zes-1000

Characteric (FAP) (accurationing area to Carry case) (accord) Accurate Accurate Licensed to: As& Prizeements, Servic : Inc.

The rank concernence is the concernence of the concern of the second second  $\mathcal{A}$  . In ( ),  $\mathcal{A}$ 

and the second n di verigi dun 1448 estemari de Firiti VER MADE TO ROOMED COMPANY 化输送 网络萨莱尔斯斯姓氏 化过敏 医白血病 网络结合 化过去 医结束 医对方形式 计算法 医肉种外外 人名法法法法 医神经炎 医神经炎 医神经炎 化分子 state will depend on the conterpoint that the conter-EPREMENTE LINITER frie cuustance reputated de DEAM and or (DEAM See 1985). F. KE AND LINE HERBOR HOUSERED - Contractions fighted management (section) and the state and the track of the state of the section of the then there of your bar of an established they have the and there itely. In the 化丁酮二丁酸二酸盐酸二丁基酸二丁基乙酸二甲基酸化物盐酸盐 计算机编制 医外外 机输入机械 计算机工具程序学家的过去分词形式的过去分词 FIR litell FileBar less only onemical, teracly highlas woten somey on negative international PERSONAL PERSONAL AND MERSONAL STRUCTURES NOT A STRUCTURE STRUCTURES. (你有意志的) 不能不能不能的。 1 HALAID I BART TERM BEENITE: Max cause screek, nose plans and sufficient uni-The TERM HEFTCOTTS has cause lack of scooting, as weat heodetone weathers na di conserve a SHE'S (ARM EXPERIES Now cause sufficiention. NUMA TACH EFFECTS: New davies mash, itching, lack of spectice, pausea. traderes secondes and distingues. SHIT THE FREETS: New Lause inmitation. New Lause offects as reported on - and norm indestion. Additional affects may include suffication. . LE TIME FRICTS: Us information is available. 3.0000000000 SHET TWO FFECTS: New cause rash. drooling, nausea, vositor, cashr i.s. naspire, arrentian heartbeat, neacache, weaknusa, dixraness, sidiress, northermal a chefre and that an alle analy space. Elimentary budiet and the the of length of the convertion, dereivale, convisioned undoperiousness set WIR STRANDER TO ADDITION IN ACCOUNTS AND A MICH AND A MORE AND A MICH AND A MICH AND A MICH AND A MICH AND A MI ng papéné ana ing pinang pinang AT Alls fernal for excesse areas of Loathan and at period, perform la - 1 alul vestimolium on (PR). Fillet evalveves with water for 15 minutes. If s of the sector as the read below side to prevent appiration. Get modical "theritor isord: "ely. See MSCS for further recommendations. ons margina as el tere terestents meterd, Hav react demographies by providence, ands - a down changed as Bee NSDS for conclete listing.  $\label{eq:constraint} \left\{ \begin{array}{c} e_{1} & e_{2} & e_{3} \\ e_{1} & e_{2} & e_{3} & e_{3} \\ e_{1} & e_{2} & e_{3} & e_{3} \\ e_{2} & e_{3} & e_{3} & e_{3} \\ e_{2} & e_{3} & e_{3} & e_{3} \\ e_{3} e_{3} & e_{3}$ "出现,我们们也也是我们你认真是真是想要,你是我们的主题来了,这是想到了你的我们,还是你的不知道,我们不能说了这个,我们们就是我们的人生。"

.

nt en la companya de la constante de la constante de servición de la constante de la constante de la constante Estate constante de la constant Estate de constante de la constante de la constante de la constante de la tradecimiente de la constante de la c

- Brull Fred Ref Lorden Bon, Burnar den Breve of St. 1986 - Eter St. 1986 Naed Disectore no people avenue facilitie Francia de St. Service Brunck

#### . Is where it is therefore a new second state of the second state of the

- Norre Alexandrean protective plack for all was and terrate vocales for reaches and the new recoverdade was washing in them at the state transformation is a final of the state of the - TRAMPRICES recovered to the state of the state of the state of the state of the protection.

tropatori Realto Secones Iac. For Emergence Electro Information Francis - achieles Paul

le relant 1975 Corrections: Health Berlinder, 1981, ell'Ellante Preen la 1999 - Heal Min Their Health Blancesce 19 Ta case couletto: peter conies for internal restriction and on only (a) a set of the state of the set of the

SUEL MADE MEETERS

CARA ERSPLOYED TOP ALLE ADDRES

DEEDAJET Com E compositor integendates a provisione de la compositor de la c Compositor de la compositor de

TTIS SUBTRANCE FORMATER LA CREATER AND LA CREAT SUBTRANCE.

FIRE AND E-PLOSING HELPEISE

No fire netword. In case of expression norms on the second restriction of the second s

FCS L-MAR FIRES: Dee water screw, too of regular (perc

PERLIE AND FIRE AIDS

and the second second

- 54947 (FAR EFFECTS: Hav cause inmitation. Additions of Mode car include couchy b and chest bais.

1. PAR JERG EFFECTS: In addition to effects incrusions term or substant Serve Nace of accetute, weithe toget difficulty breathers, weathers, black effects tolars luch effects and heart failure may occur. New also cause taster. END: CONTACT:

HART LERM EFFECTS: May cause irritation.

1.0MG TFEN EFFECTS: Same effects as short term excosure.

EYE CONTACT:

BHURT TERM EFFECTS: May cause preitation.

•

ICNE TEPH EFFECTS: Same effects as short term exposure.

A. 승규는 가 지역

-Salk ( TEAM ETFERTS: May cause dastroprotestina) (rottet in Addition) - effects may include didestive disorders.

LONG TERM EFFECTS: No information available on significant bases of ifform. FJRET Albr Prove from exposure area. If breathing has stored, conform analistic respondium of CPR. Fluck skinzekes with water for if mint etc. / stort of occurs, new here below hope to provent asciration. Get menical sciences, modilately. See 4805 for further recompositions.

Simble under correl conditions. See MSDE for complete listard of matched block betondes.

医丁基丁二乙酰胺 医医静脉

Backy up and place to soltable clean. Inv containers non-reclamation and later (isopsal) De not fluch abilled material into seven. Uper unnecessive accele sway

Vive deterration the componente, if spilled new be consultant of der RETOR. Gronte B-44. See MSDS for specific reportable contribues. In applicable.

•

Constant fre carlons. Paars en l'ont concern for the for the 2016 the first school of the first

SATEN CIRC AND CONTRACTOR SOUTHER

You should when protective choices alwaes and safety apples. A reaverage max be reveal when when his is nich on a make on contractance. Each RET for DEGANDER relations takens. If an account occurs, contact satefy perconter.

- Dice enformed large the Hermiteley dense Hos Helenwesian Braile Anglesztich ich and eine Breezber

Convolt 1993 Occuptional Health Bervices. Jon. 411 Richts Perenant Licensed for AAR Environmental Heroices. Inc. - Make unlimited pepar copes for article1 pretribution and per en-

#### 5.0 SITE CONTROL

#### 5.1 ZONATION

The general zonation protocols that should be employed at hazardous waste sites are described in Appendix G. The site-specific zonation that will be used for this project is described as follows:

<u>Exclusion Zone</u>: area within a 15-foot radius of each drilling/sampling location during drilling or sampling (when a borehole or well is open).

<u>Contaminant Reduction Zone</u>: area beyond the 15-foot exclusion zone at each drilling/sampling location where equipment and personnel decontamination will occur, not to occur beyond 50-feet in any area from the drilling/sampling locations.

Support Zone: all areas beyond the contaminant reduction zone.

#### 5.2 COMMUNICATIONS

When radio communication is not used, the following air horn signals will be employed:

HELP	EVACUATION	ALL CLEAR
Three Short Blasts	Three Long Blasts	Alternating Long and
		Short Blasts
()	()	()

#### 5.3 WORK PRACTICES

General work practices to be used during ABB Environmental projects are described in Appendix H. Specific work practices necessary for this project or those that are of significant concern are described as follows:

#### **ABB Environmental Services, Inc.**

5-1

H&SPLAN.REV

All ABB-ES personnel, subcontractors, or third party representatives should be aware of the previously detected presence of mercury. These personnel should also be aware of the potential for the contaminants to exist in airborne form.

#### ABB Environmental Services, Inc.

H&SPLAN.REV

.

#### 6.0 DECONTAMINATION/DISPOSAL

All personnel and/or equipment leaving contaminated areas of the site will be subject to decontamination, which will take place in the contamination reduction zone.

#### 6.1 PERSONNEL DECONTAMINATION

Decontamination procedures are followed by all personnel leaving hazardous waste sites. Under no circumstances (except emergency evacuation) will personnel be allowed to leave the exclusion and contaminant reduction zones prior to decontamination. A typical personnel decontamination station is shown in Appendix L, Figure L-1. Generalized procedures for removal of protective clothing are as follows:

- 1. Drop tools, monitors, samples, and trash at designated drop stations (i.e., plastic containers or drop sheets).
- 2. Step into the designated shuffle pit area and scuff feet to remove gross amounts of dirt from outer boots.
- 3. Scrub outer boots and outer gloves with decon solution or detergent and water. Rinse with water.
- 4. Remove tape from outer boots and remove boots; discard tape and boots in disposal container.
- 5. Remove tape from outer gloves and remove gloves; discard tape and gloves in disposal container.
- 6. If the worker has left the Exclusion Zone to change the air tank on the SCBA or the canister on the air-purifying respirator, this will be the last step in the decontamination procedure. The tank or cartridge should be exchanged, new outer gloves and boot covers donned, and the joints taped; the worker then returns to duty.

#### **ABB Environmental Services, Inc.**
- 7. Remove outer garments and discard in disposal container.
- 8. Remove respirator and place or hang in the designated area.
- 9. Remove inner gloves and discard in disposal container.
- 10. If the site requires use of a decontamination trailer, all personnel must shower before leaving the site at the end of the work day.
- NOTE: Disposable items (i.e., Tyvek coveralls, inner gloves, and latex overboots) will be changed daily unless there is reason to change sooner. Dual respirator canisters will be changed daily, unless more frequent changes are deemed appropriate by site surveillance data or personnel assessment.

Maximum and minimum decontamination layouts for PPE Levels A through C are shown in Figures L-2 through L-6.

Pressurized sprayers or other designated equipment will be available in the decontamination area for washdown and cleaning of personnel, samples, and equipment.

Respirators will be decontaminated daily and taken from the drop area. The masks will be disassembled, the cartridges set aside, and all other parts placed in a cleansing solution. Parts will be pre-coded (e.g., #1 on all parts of Mask #1). After an appropriate time in the solution, the parts will be removed and rinsed with tap water. Old cartridges will be marked to indicate length of use (i.e., if it is possible to evaluate the remaining utility of the cartridge), or discarded in the contaminated trash container for disposal. In the morning, the masks will be reassembled and new cartridges installed, if appropriate. Personnel will inspect their own masks and readjust the straps for proper fit.

### ABB Environmental Services, Inc.

H&SPLAN.REV

### 6.2 SMALL EQUIPMENT DECONTAMINATION

Small equipment will be protected from contamination as much as possible by draping, masking, or otherwise covering the instruments with plastic (to the extent feasible), without hindering operation of the unit. For example, the PI meter can be placed in a clear plastic bag to allow for reading the scale and operating the knobs. The PI meter can be partially wrapped, keeping the sensor tip and discharge port clear.

The contaminated equipment will be taken from the drop area and the protective coverings will be removed and disposed of in appropriate containers. Any dirt or obvious contamination will be brushed or wiped with a disposable paper wipe. The units can then be taken inside in a clean plastic tub, wiped off with damp disposable wipes, and dried. The units will be checked, standardized, and recharged as necessary for the next day's operation, and then prepared with new protective coverings.

### 6.3 HEAVY EQUIPMENT DECONTAMINATION

It is anticipated that drilling rigs and backhoes will become contaminated during borehole and test-pitting activities. They will be cleaned with high-pressure water or steam, followed by a soap and water wash and rinse. Loose material will be removed with a brush. The person performing this activity will usually be at least at the level of protection used during the personnel and monitoring equipment decontamination.

### 6.4 DISPOSAL OF DECONTAMINATED MATERIALS

All protective gear, decontamination fluids (for both personnel and equipment), and other disposable materials will be disposed of at each site.

Decontamination fluids identified to be contaminated by site contaminants (i.e., Liqui-nox, used to decontaminate sampling equipment such as split spoons and groundwater sampling pumps) will be stored in DOT-approved 55-gallon drums. Contaminated disposable materials (e.g., gloves and Tyveks) will be double-bagged and stored as is, or placed in DOT-approved 55-gallon drums.

### **ABB Environmental Services, Inc.**

H&SPLAN.REV

### 7.0 EMERGENCY/CONTINGENCY PLAN

This section identifies the emergency contingency plan that has been developed for operations at this site. Other sections provide further information to be used under emergency conditions. Refer to Appendix D for emergency telephone numbers, routes to emergency medical facilities, and emergency signals.

### 7.1 PERSONNEL ROLES, LINES OF AUTHORITY, AND COMMUNICATION

The site HSO is the primary authority for directing operations under emergency conditions. All communications both on- and off-site will be directed through the HSO. Should the HSO be injured, the Health and Safety Designee (HSD) will assume HSO responsibilities. These responsibilities include:

- Investigate all accidents, illnesses, and incidents occurring on an assigned site and report the findings to the HSM or HSS.
- Assume the role of on-site coordinator during emergency response activities.
- Accompany all USEPA, OSHA, and other government agency, and client personnel visiting an assigned site in response to health and safety issues.
  - The HSO or HSD will immediately notify the HSM after a stop work order is executed. Authorization to resume work, after such a stoppage, will only be issued by the HSO after consultation and approval from the HSM or HSS.

### ABB Environmental Services, Inc.

H&SPLAN.REV

### 7.2 EVACUATION

### 7.2.1 Withdrawal Upwind

The work party will continually note general wind directions while on-site. (A windsock may be set up near the work site for visual determinations.) When conditions warrant moving away from the work site, the crew will relocate upwind a distance of approximately 100 feet or farther, as indicated by site monitoring instruments. Donning an SCBA and a safety harness and line, the HSO and a member of the crew (the buddy system must be used) may return to the work site to determine if the condition noted was transient or persistent. If persistent, an alarm should be activated to notify on-site personnel of the situation and the need to leave the site or don an SCBA. An attempt to decrease emissions should be made only if greater respiratory protection is donned. The HSS and client will be notified of conditions. When site access is restricted, thus hindering escape, the crew may be instructed to evacuate the site rather than move upwind, especially if withdrawal upwind moves the crew away from escape routes.

### 7.2.2 Site Evacuation

When conditions warrant site evacuation, the work party will proceed upwind of the work site and notify the HSO and field office of site conditions. If the decontamination area is upwind and greater than 500 feet from the work site, the crew will pass quickly through decontamination to remove contaminated outer suits. If the hazard is toxic gas, respirators will be retained. The crew will proceed to the field office to assess the situation. If instrumentation indicates an acceptable condition, respirators may be removed. As more information is received from the field crew, it will be relayed to the appropriate agencies. The advisability and type of further response action will be coordinated and carried out by the HSO.

### 7.2.3 Evacuation of Surrounding Area

When the HSO determines that conditions warrant evacuation of downwind residences and commercial operations, local agencies will be notified and assistance requested. Designated on-site personnel will initiate evacuation of the immediate off-site area without delay.

### ABB Environmental Services, Inc.

7-2

H&SPLAN.REV

### 7.3 EMERGENCY MEDICAL TREATMENT/FIRST AID

Any personnel injured on-site will be rendered first aid as appropriate and transported to competent medical facilities for further examination and/or treatment. The preferred method of transport would be through professional emergency transportation means; however, when this is not readily available or would result in excessive delay, other transport will be authorized. Under no circumstances will injured persons transport themselves to a medical facility for emergency treatment.

### ABB Environmental Services, Inc.

H&SPLAN.REV

### **8.0 ADMINISTRATION**

#### 8.1 PERSONNEL AUTHORIZED DOWNRANGE

Personnel authorized to participate in downrange activities at this site have been reviewed and certified for site operations by the PM/CM and the HSS. Certification involves the completion of appropriate training, a medical examination, and a review of this site-specific HASP. All persons entering the site must use the buddy system, and check in with the Field Team Leader and/or HSO before going downrange.

For each field activity, the HSO will complete the "Authorized Personnel" section of the "Task Analysis and Protective Measures" form.

### 8.2 MEDICAL DATA SHEET

The Medical Data Sheet will be completed by all on-site personnel and kept in the Support Zone during site operations. It is not a substitute for the Medical Surveillance Program requirements consistent with the ABB Environmental Corporate Health and Safety Program for Hazardous Waste Sites. This data sheet will accompany any personnel when medical assistance or transport to hospital facilities is required. If more space is required, use the back of this sheet.

### ABB Environmental Services, Inc.

H&SPLAN.REV

	MEDICAL DATA SH	IEET
Task/Activity:		
Name:		
Address:		
Home Telephone: Area Co	de ( )	
Age:	Height:	Weight:
In Case of Emergency conta	act:	
Address:		
Telephone: Area Code (	)	
Do you wear contact lenses	: Yes ( ) No (	)
Allergies:		
List medication(s) taken reg	gularly:	
Particular sensitivities:		
Previous/current medical co	onditions or exposures to l	nazardous chemicals:
Name of Personal Physician	n:	
Telephone: Area Code (	)	

## ABB Environmental Services, Inc.

H&SPLAN.REV

## 8.3 EMERGENCY TELEPHONE NUMBERS

Police Department	
Rescue Service	
Primary Hospital	(716) 464-2130
Alternate Hospital	(716) 275-2100
Fire Department	
Off-site Emergency Services	
Poison Control Center	(800) 962-1253
National Response Center	
NYSDEC Spill Reporting Hotline	(800) 457-7362
Project Manager: Geoff Knight	(716) 424-4700
Construction Manager: Rick Day	(207) 828-3441
Program Manager: Lisa Spahr	
ABB Environmental HSM: C.E. Sundquist	(800) 848-2697 x3309
ABB Kent-Taylor: Bob Halton	(716) 273-6201
Site Security (Ogden Security - Earl Barney)	

## ABB Environmental Services, Inc.

.

H&SPLAN.REV

7198-17

•

### 8.4 ROUTES TO EMERGENCY MEDICAL FACILITIES

The primary source of medical assistance for the site is:

St. Mary's Hospital, 89 Gennesee Street. See Figure 8-1.

DIRECTIONS TO ST. MARY'S:

Left on West Avenue from Hague Street. Follow West Avenue to Intersection of West Avenue and Chili/West Main Street. Hospital is located on the south side of this intersection.

The alternate source of medical assistance for the site is:

Strong Memorial Hospital, Elmwood Avenue. See Figure 8-2.

### DIRECTIONS TO STRONG MEMORIAL:

Right on West Avenue from Hague Street.
Follow West Ave. 8 blocks to Thurston Road, turn left.
Follow Thurston to Chili Avenue, turn right.
Follow Chili to the 390 South.
390 South Approximately 3 miles to the 15 North (South Avenue/Mt. Hope Avenue)
Approximately 1/4 mile to Crittenden Road, turn right.
Hospital 1/10 mile on right side.

### ABB Environmental Services, Inc.

H&SPLAN.REV





## FIGURE 8-2 SECONDARY HOSPITAL ROUTE MAP AMES STREET FACILITY ROCHESTER, NEW YORK

ABB Environmental Services, Inc.-

7198-00

N

## 8.5 HASP APPROVALS

By their signatures, the undersigned certify that this HASP will be used for the protection of the health and safety of all persons entering this site.

Health and Safety Officer	Date
Project Manager	Date
Health and Safety Manager/Supervisor	Date

## ABB Environmental Services, Inc.

•

H&SPLAN.REV

## ATTACHMENT A

### VISITOR LOG AND SAFETY INFORMATION

ABB Environmental Services, Inc.

.

H&SPLAN.REV

## VISITOR LOG AND SIGNATURE FORM Ames Street Site

	General Information
Name:	
Affiliation:	
Date(s) of visit:	
Purpose of visit:	
Site area(s) to be visite Planned activities:	· 
Escort's name:	
	Additional Hazard Information
Nature/locations of act	ve operations:
Summary of other pote	ntial hazards not discussed in Safety Information and Procedures:
"I have read and under briefed by a Combustic abide by the Safety Pro	stand the Safety Information and Procedures for Visitors to the Ames Street Site, and have bee n Engineering representative on potential health and safety hazards. During my visit I agree cedures and will follow direction provided by my escort".
Signature:	Date:

## SAFETY INFORMATION AND PROCEDURES FOR VISITORS TO THE AMES STREET SITE

(Page 1 of 2)

Welcome to the Ames Street Site. You must be aware that this site may contain potential health and safety hazards related to general site conditions, construction activities, or chemicals. The information and procedures provided herein are designed to ensure your visit to the site is a safe one.

### Potential Health and Safety Hazards:

-

- 1. This is a closed facility, and most ground floor windows and doors have been permanently boarded up. Entrance and exit to the buildings is possible only at those locations shown on the attached site plan.
- Utilities have been shut off throughout most of the buildings; many areas or rooms are either dimly lit, or completely dark. Electricity, water and phones are generally available only at the field trailer area shown on the attached site plan.
  - 3. The combination of dim or absent lighting and the potential presence of equipment, demolition or other debris, snow, ice, or water, may lead to a significant "slip, trip and fall" hazard in some areas. These areas may or may not be marked with caution tape or other devices.
- 4. Hazardous materials such as asbestos, mercury, petroleum solvents and other chemicals may be present. These areas may or may not be marked with caution tape or other devices.
- 5. Ongoing activities such as building demolition or environmental sampling may present hazards due to motorized equipment, noise, chemical use, or other factors. Active work areas may or may not be marked with caution tape or other devices.

### Safety Procedures

The following must be observed by all visitors to the Ames Street site.

- Personal protective equipment is required when entering or touring the site for any reason:
  - hard hat
    - suitable footwear (no open-toed shoes, no sneakers or tennis shoes; steel-toed shoes are preferred)
  - Based on your specific activity while at the site, other safety equipment <u>may be required</u>.
  - 2. Before entering the site, you must be briefed by a Combustion Engineering representative or contractor on the hazards specific to the areas you will visit or your planned activities. The briefing will include a description of hazards in the areas you will visit; review of Material Safety Data Sheets (MSDSs) for chemicals of concern; a discussion of current activities occurring on the site; and instructions on how to avoid potential health and safety hazards. It is critical that you pay close attention to the information and instructions provided during this briefing. The individual providing the briefing will be happy to respond to any questions or concerns you may have.

## SAFETY INFORMATION AND PROCEDURES FOR VISITORS TO THE AMES STREET SITE

(Page 2 of 2)

### Safety Procedures (cont'd)

- 3. Each visitor or group will be provided with an escort who is familiar with the site and potential hazards. Visitors must remain within the escort's sight and follow his/her instructions at all times, unless specifically authorized otherwise. Do not enter any room or area of the site or otherwise move about on your own unless authorized to do so.
- 4. Do not approach or interfere with site operations or the persons engaged in them (including talking) for any reason,
  unless specifically authorized to do so by your escort. Doing so may pose a hazard to you, or them.
- In the event you encounter what appears to be an abnormal or dangerous situation, report it immediately to your escort or other Combustion Engineering representative or contractor. Do not attempt to take corrective action on your own.
- 6. Do not operate or attempt to operate any equipment or machinery of any type.
- Do not handle or attempt to handle chemicals or potentially hazardous materials of any type, or handle building materials or other substances from areas which your escort indicates may contain hazardous materials.
- 8. Smoking, eating and drinking are permitted only in the vicinity of the field trailers.

9. Small children or pets are not allowed on the site.

g:\T92\KENT\_TAY\H&S\VISITOR.PLN

## ATTACHMENT B

### TASK ANALYSIS AND PROTECTIVE MEASURES

ABB Environmental Services, Inc.

.

H&SPLAN.REV

\_

TASK ANALYSIS ABB Environmental Services, Inc Ames Street Site	
NOTE: Review Ames Street Facility HASP before completing this form.	
GENERAL TASK INFORMATION	
Task Title:	PAN number:
FOL (name):HSO (name):	
Form prepared by:	Date:
HSS Approval:	Date:
PM Approval:	Date:
Planned activity(s):	
Overall Task Category (from HASP): Other concurrent site activities/tasks:	
Activity/task:	Task Category:
Activity/task:	Task Category:
Activity/task:	Task Category:
Will these activities/tasks impact this task? (describe):	
UTHORIZED PERSONNEL:	······································
* Current First-aid Certification + Current ' of General Information. For each of this task's activities, complete a sub-	t CPR Certification

ACTIVITY ANAL (Page 1)	LYSIS
tivity:	
own or suspected chemical/material hazards (list):	
	· ·
emical/material location(s) (circle): Surface soil Subsur	face soil Groundwater Building Materials
Other (list):	· _
<b>mary exposure route(s)</b> (circle): Dermal Inhalation Inge	estion Puncture
fety hazard(s) (circle): Height Equipment Cold Str	ress Noise Eye Overhead Confined
Spaces Heat Stress Machinery Burns	Lifting Slips/Falls
her (list):	
erall Hazard Estimation: Serious Moderate	Low None Unknown
GINEERING CONTROLS SELECTED (specify):	
Level of Personal Protection	
Level of I cisonal I forection	
Cartridge Respirator Face Shield Gloves Cartridge Type: Safety Glasses inner	
afety Boots/Shoes Ear Protection Coveralls	S
sposable Boot Covers -regular -coated	r
:	

	1
Monie	1
Combustible Gas/Over	1
Jerome Mercury Vapor Apol	1
Other (list): Draeger Tubes (specify two)	1
ergency Equipment (	1
(list): First Aid Kit Fire Extinguisher	1
nant level	1
the vels for modifying protection	1
Tation: 'and/or equipment leaving contaminated sites are subject to decontamination. Under no (except emergency evacuation) will personnel be allowed to leave the site prior to decontamination.	1 1 1 1 1 1 1 1 1
ste is to be placed into plastic bags and then into D.O.T.	1 1 1
edures not covered above:	



## FINAL

## SITE INVESTIGATION WORK PLAN PHASE I

APPENDICES H, I AND J

AMES STREET SITE ROCHESTER, NEW YORK

Prepared by:

ABB Environmental Services, Inc. Rochester, New York

February 1996



ABB Environmental Services, Inc.

## APPENDIX H

## BACKGROUND INFORMATION FOR INDUSTRIAL SOIL INGESTION SCREENING CONCENTRATION

### UNITED STATES ENVIRONMENTAL PROTECTION AGENCY Region III 841 Chestnut Street Philadelphia, Pennsylvania 19107

October 20, 1995

SUBJECT: Risk-Based Concentration Table, July - December 1995

FROM:

Roy L. Smith, Ph.D. Office of RCRA Technical & Program Support Branch (3HW70)

flow

TO: RBC Table mailing list

Attached is the EPA Region III risk-based concentration (RBC) table, which we distribute semi-annually to all interested parties.

### IMPORTANT MESSAGE

EPA Region III has established a homepage on the World Wide Web which you can find at http://earth1.epa.gov:80/ or http://www.epa.gov/. Our homepage will soon include the RBC table in downloadable form. We strongly encourage all RBC table users having Internet access to obtain the table electronically rather than on paper. In this way, users can obtain the most current issue immediately in a form that can be used directly as input for risk assessment calculations. This distribution method will also save large amounts of paper and cost substantially less.

For those lacking Internet access, it's once again time to re-register to receive a paper copy of the RBC table. We need to hear from you periodically to ensure that you still have an interest and that we have your correct address. Please fax your registration request to Vanessa Sizer at 215-597-9890, including your name, address, and phone number. Please don't phone to re-register; we need hard copy to document your continued interest. If we don't hear from you by March 30, 1996, we'll assume you no longer need a paper copy. Thanks for your cooperation.

CONTENTS, USES, AND LIMITATIONS OF THE RBC TABLE

The table contains reference doses and carcinogenic potency slopes (obtained from IRIS through September 1, 1995, HEAST through May 1995, the EPA-NCEA Superfund Health Risk Technical Support Center, and other EPA sources) for nearly 600 chemicals. These toxicity constants have been combined with "standard" exposure scenarios to calculate RBCs--chemical concentrations corresponding to fixed levels of risk (*i.e.*, a hazard quotient of 1, or lifetime cancer risk of 10<sup>-6</sup>, whichever occurs at a lower concentration) in water, air, fish tissue, and soil.

The RBC table also includes soil screening levels (SSLs) for protection of groundwater

EPA Region III Risk-Based Concentrations: R.L. Smith (10/20/95)

### ANSWERS TO FREQUENTLY ASKED QUESTIONS

To help you better understand the RBC table, here are answers to our most often-asked questions:

1. How can the age-adjusted inhalation factor (11.66) be less than either the inhalation rate for a child (12) or for an adult (20)?

Age-adjusted factors are not intake rates, but rather partial calculations which have different units than intake rates do. The fact that these partial calculations have values similar to intake rates is really coincidental, an artifact of the similar magnitude of years of exposure and time-averaged body weight.

2. Why does arsenic appear in the RBC table separately as a carcinogen and a noncarcinogen, while other contaminants do not?

Arsenic is double-entered to ensure that the risk assessor realizes that non-carcinogenic concerns are significant for arsenic. Otherwise, one might be tempted to accept a 1e-4 risk (37 ppm in residential soil), when the oral reference dose would be exceeded at 23 ppm.

Also, EPA has a little-known risk management policy for arsenic (dating from 1988) that suggests that arsenic-related cancer risks of up to 1e-3 can be accepted because the cancers are squamous cell carcinomas with a low mortality rate. Thus, noncarcinogenic RBCs represent an important limitation on acceptable arsenic concentrations.

3. Many contaminants have no inhaled reference dose or carcinogenic potency slope in IRIS, yet these numbers appear in the RBC table with IRIS given as the source. Where did the numbers come from?

Most inhaled reference doses and potency slopes in the RBC table are converted from reference concentrations and unit risk values which do appear in IRIS. These conversions assume 70-kg persons inhaling 20 m<sup>3</sup>/d. For example, the inhalation unit risk for arsenic (4.3e-3 risk per  $\mu$ g/m<sup>3</sup>) is divided by 20 m<sup>3</sup>/d and multiplied by 70 kg times 1000  $\mu$ g/mg, yielding a CPSi of 15.1 risk per mg/kg/d.

4. Why does the RBC table base soil RBCs for cadmium and manganese on reference doses that apply only to drinking water?

The RBC table's use of the drinking water RfDs for cadmium and manganese reflects (1) the limited space available in the already-crowded table, and (2) the intended use of the table as a screening tool rather than a source of cleanup levels (thereby making false positives acceptable). For a formal risk assessment, Region III would use the food RfDs for soil ingestion.

At this time, only two substances (as far as we know) have distinct oral RfDs for water and food--cadmium and manganese. Adding the two food RfDs to the table would require an entire column, which would be about 99.9% blank. The table has become so crowded that it would be difficult to accommodate another column. Also, we given this problem a relatively low

ŧ,

EPA Region III Risk-Based Concentrations: R.L. Smith (10/20/95)

reference dose or potency slope for inorganic lead, so it wasn't possible to calculate risk-based concentrations. EPA considers lead a special case because:

- (1) Lead is ubiquitous in all media, so human exposure comes from multiple sources. Comparing single-medium exposures with a reference dose would be misleading.
- (2) If EPA did develop a reference dose for lead by the same methods other reference doses, we would probably find that most people already exceed it. Since EPA already knows this and is moving aggressively to lower lead releases nationally, such findings at individual sites would be irrelevant and unduly alarming.
- (3) EPA decided to take a new approach to separate important lead exposures from trivial ones. EPA developed a computer model (the IEUBK model) which predicts children's blood lead concentrations using lead levels in various media as inputs. The idea is to evaluate a child's entire environment, and reduce lead exposures in the most cost-effective way.

On the practical side, there are several EPA policies which effectively substitute for RBCs. The EPA Office of Solid Waste has released a detailed directive on risk assessment and cleanup of residential soil lead. The directive recommends that soil lead levels less than 400 ppm be considered safe for residential use. Above that level, the document suggests collecting certain types of data and modeling children's blood lead with the IEUBK model. For the purposes of the RBC table, the de facto residential soil number would be 400 mg/kg. For water, we suggest 15 ppb (from the national EPA Action Level), and for air, the National Ambient Air Quality Standard.

### 9. Where did the potency slopes for carcinogenic PAHs come from?

The source of the potency slopes for PAHs is "Provisional Guidance for Quantitative Risk Assessment of Polycyclic Aromatic Hydrocarbons", Final Draft, EPA Environmental Criteria and Assessment Office, Cincinnati, OH. It's available from NTIS as document number ECAO-CIN-842 (March, 1993). The slopes are expressed in terms of order-of-magnitude equivalence factors relating the compounds to benzo[a]pyrene; we have converted these TEQs to potency slopes to fit the format of the table.

### 10. May I please have a copy of the January 1991 RBC table?

We're sorry, but no. The RBC table doesn't represent regulation or guidance, so past issues should have no legal importance. Each time we update the table we destroy all obsolete copies, electronic and paper. We do this to ensure that only one set of RBCs, that based on current information, exists at any time.

11. I've noticed that some soil RBCs are 1 million parts per million. Since some of these substances are liquids, that's obviously ridiculous. What is that basis for these calculations?

A soil RBC of 1 million parts per million means that no amount of the contaminant in soil will cause a receptor to exceed the oral reference dose by incidental ingestion of soil. In

EPA Region III Risk-Based Concentrations: R.L. Smith (10/20/95)

It has been NCEA's policy to deny requests for documentation of interim toxicity constants. Although Region 3 has sometimes provided this documentation on request, for the above-stated reasons we have no assurance that the documentation, or even the interim numbers, are current. We've decided to discontinue distributing information that may be misleading. If one of the "E"-coded contaminants is a major risk contributor at your site, we strongly suggest that you work with EPA to develop an up-to-date reference dose or slope factor.

### CHANGES IN THIS ISSUE OF THE RBC TABLE

New or revised EPA toxicity constants are now marked with "\*\*" before the contaminant name. This is to help users quickly pick out substances with new RBCs. Formerly these contaminants were printed in underlined boldface type that copied badly. A new basis code, "M" for MCL, has been added to the upper right corner of each page. This code denotes soil screening levels for groundwater protection that are based on EPA Maximum Contaminant Levels.

If you want to raise issues or get answers to questions about the RBC table, please call the Technical Support Help Line at 215-597-1116. The line has a voice mail system to take your calls if we're not available. We'll return your call as soon as we can. Please limit calls to RBC issues; if you have a question about applying RBCs to a site, please call the EPA Regional office handling the project. Thanks for your help and cooperation, and we hope the RBC table continues to be a useful resource.

Attachment

EPA Region III Risk-Based Concentration Table: R.L. Smith (October 4, 1995)

Exposure variables	Value	Symbol
Occupational:		
Exposure frequency (d/y):	250	EFo
Exposure duration (y):	25	EDo
Fraction of contaminated soil ingested (unitless)	0.5	FC

": Contaminant-specific toxicological constants. The priority among sources of toxicological constants was as follows: (1) IRIS, (2) HEAST, (3) HEAST alternative method, (4) EPA-NCEA Superfund Health Risk Technical Support Center, (5) withdrawn from IRIS or HEAST, and (6) other EPA documents. Each source was used only if numbers from higher-priority sources were unavailable. The EPA Superfund Health Risk Technical Support Center, part of the EPA National Center for Environmental Assessment in Cincinnati, develops provisional RfDs and CPSs on request for contaminants not in IRIS or HEAST. These provisional values are labeled "E = EPA-NCEA provisional" in the table. It is possible they may be obsolete. If one of the "E" constants is important to a Superfund risk assessment, consider requesting, through a Regional risk assessor, a new provisional value.

### Age-adjusted factors

Because contact rates with tap water, ambient air, and residential soil are different for children and adults, carcinogenic risks during the first 30 years of life were calculated using age-adjusted factors. These factors approximated the integrated exposure from birth until age 30 by combining contact rates, body weights, and exposure durations for two age groups - small children and adults. The age-adjusted factor for soil was obtained from RAGS IB; the others were developed by analogy.

Air inhalation

$$IFAadj \quad \frac{m^3 \cdot y}{kg \cdot d} = \frac{EDc \cdot IRAc}{BWc} + \frac{(EDtot - EDc) \cdot IRAa}{BWa}$$

Tap water ingestion

$$IFWadj \quad \frac{L \cdot v}{kg \cdot d} = \frac{EDc \cdot IRWc}{BWc} + \frac{(EDtot - EDc) \cdot IRWa}{BWa}$$

Soil ingestion

$$IFSadj \quad \frac{mg \, y}{kg \cdot d} = \frac{EDc \cdot IRSc}{BWc} + \frac{(EDtot - EDc) \cdot IRSa}{BWa}$$

#### **Residential water**

Volatilization terms were calculated only for compounds with a mark in the "VOC" column. Compounds having a Henry's Law constant greater than 10<sup>-5</sup> were considered volatile. The list may be incomplete, but is unlikely to include false positives. The equations and the volatilization factor (K, above) were obtained from RAGS IB. Oral potency slopes and reference doses were used for both oral and inhaled exposures for volatile compounds lacking inhalation values. Inhaled potency slopes were substituted for unavailable oral potency slopes only for volatile compounds; inhaled RfDs were substituted for unavailable

### Commercial/industrial soil ingestion

RBCs were based on adult occupational exposure, including an assumption that only 50% of total soil ingestion is work-related.

Carcinogens

$$RBC \frac{mg}{kg} = \frac{TR \ BWa \ ATc}{EFo \ EDo \cdot \frac{IRSa}{10^6 \ \frac{mg}{kg}} \cdot FC \cdot CPSc}$$

Non-carcinogens

$$RBC \frac{mg}{kg} = \frac{THQ \cdot RfDo \cdot BWa \cdot ATn}{EFo \cdot EDo \cdot \frac{IRSa}{10^{6} \frac{mg}{kg}} \cdot FC}$$

### Residential soil ingestion

RBCs for carcinogens were based on combined childhood and adult exposure; RBCs for non-carcinogens were based on childhood exposure only.

Carcinogens

$$RBC \frac{mg}{kg} = \frac{TR \cdot ATc}{EFr \cdot \frac{IFSadj}{10^{6} \frac{mg}{kg}} \cdot CPSo}$$

Non-carcinogens

$$RBC \frac{mg}{kg} = \frac{THQ \cdot RfDo \cdot BWc \cdot ATn}{EFr \cdot EDc \cdot \frac{IRSc}{10^{6} \frac{mg}{kg}}}$$

### **Development of Soil Screening Levels**

### General

In December 1994 the EPA Office of Solid Waste and Emergency Response proposed Soil Screening Guidance (Document 9355.4-1, PB95-963530, EPA540/R-94/101, available through NTIS at 703-487-4650). This draft document provides (1) a framework in which soil screening levels are to be used, (2) a detailed methodology for calculating soil screening levels, and (3) soil screening levels for 107 substances.

Consistent with this new guidance, the risk-based concentration table now includes two columns of generic soil screening levels (SSLs). OSWER's 107 proposed soil screening levels have been added verbatim. In addition, the proposed SSL methodology has been used to calculate soil screening levels for more substances, which are also included in the

EPA Region III Risk-Based Concentration Table: R.L. Smith (October 4, 1995)

consulting that document. The "unofficial" SSLs were developed under the following conditions:

#### Soil Screening Levels for Inhalation

Inhaled reference doses and potency slopes were used if available. If inhalation values were not available, oral RfDs and potency slopes were substituted. SSLs were calculated only for substances for which aqueous solubility, Koc, Henry's Law constant, and diffusivity in air were available. SSLs were calculated only for substances for which a volatilization factor could be calculated. This was done because OSWER's large proposed particulate emission factor rendered it pointless to estimate SSLs for particulate emissions alone. The final calculated SSL shown in the RBC table is the smaller of the risk-based SSL and the soil saturation concentration. All calculated SSLs were rounded to 2 significant figures.

The OSWER risk algorithms for inhalation were revised in order to be consistent with the rest of the RBC table. Only calculated SSLs were affected by this; SSLs proposed by OSWER are presented verbatim. Calculated SSLs for inhalation of carcinogens were based on an integrated lifetime exposure rather than adult exposure. SSLs for inhalation of noncarcinogens were based on adult exposure for 350 days per year rather than 365 days per year. The following algorithms were used to calculate inhalation SSLs:

Carcinogens

$$SSL \quad \frac{mg}{kg} = \frac{TR \cdot ATc}{EFr \cdot IFAadj \cdot \left(\frac{1}{VF} + \frac{1}{PEF}\right) \cdot CPSi}$$

Non-carcinogens

$$SSL \quad \frac{mg}{kg} = \frac{THQ \cdot BWa \cdot ATn \cdot RfDi}{EFr \cdot EDtot \quad \cdot IRAa \cdot \left(\frac{1}{VF} + \frac{1}{PEF}\right)}$$

### Soil Screening Levels for Groundwater Use

All algorithms were as proposed by OSWER. MCLs were used as target groundwater concentrations if available. If MCLs were unavailable the risk-based concentration in the "tap water" column of the RBC table was used as the target groundwater concentration. All SSLs for groundwater are based on a dilution-attenuation factor (DAF) of 10. Since these SSLs scale linearly with DAF, the SSLs for DAF=1 would be ten times lower. They were omitted to conserve space. All groundwater SSLs were rounded to 2 significant figures and capped at unity.

Ĩ.

t

Sources: I=IRIS II=IIEAST A=IIEAST alternate W=Withdrawn from IRIS or IIEAST							Basis : C=carcinogenic effects N=noncarcinogenic effects E=EPA draft Soil Screening Level								
E=EPA-NCEA Regional Support provisional value O=Other EPA documents.							S=soil saturation concentration M=EPA MCL.								
	í í	[	· [	[			۱ 	Risk-Bas		Soil Screening Levels-					
		1				V	Tap	Ambient		Soil Ir	ngestion	Transfers	from Soil to:		
		RíDo	RIDi	CPSo	CPSI	0	Water	Air	Fish	Industrial	Residential	Air	Groundwater		
Contaminant	CAS	mg/kg/d	mg/kg/d	kg d/mg	kg d/ing	C	με/L	μg/m3	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg		
Arsine	7784421		1.43E-05 i				0.52 N	0.052 N							
Assure	76578148	9.0015-03					330 N	33 N	12 N	18000 N	700 N				
Asulanı	3337711	5.0015-02 (					1800 N	180 N	68 N	100000 N	3900 N		· 1		
Atrazine	1912249	3.50E-02 I		2.22Е-01 н			0.3 с	0.028 c	0.014 c	26 c	<b>2.9</b> ε				
Avermectin B1	65195553	4.00E-04 i					15 N	1.5 N	0.54 N	́ 820 м	31 N				
Azobenzene	103333			1.10E-01 i	1.08E-01		0.61 c	0.058 c	0.029 c	52 c	5.8 c		h		
Barium and compounds	7440393	7.00E-02 i	1.43E-04 🗚				2600 N	0.52 N	95 N	140000 N	. 5500 N	350000 e	32 1		
Baygon	114261	4.00E-03 ı					150 N	15 N	5.4 N	8200 N	310 N				
Bayleton	43121433	3.00E-02 i			·		1100 N	110 N	41 N	61000 N	. 2300 N				
Baythroid	68359375	2.50E-02 +					910 N	91 N	34 н	51000 N	2000 N				
Benefin	1861401	3.00E-01 1					11000 N	1100 N	410 N	610000 N	. 23000 N		· (		
Benomyl	17804352	5.00E-02 +					1800 N	180 N	68 N	100000 N	3900 N				
Bentazon	25057890	2.50E-03 +					• '91 N	9.1 N	3.4 N	5100 N	200 N				
Benzaldehyde	100527	1.00E-01 +				021	610 N	370 N	140 N	200000 N	7800 N				
Benzene	71432		1.71E-03 E	2.90E-02 +	2.90E-02	Ø	0. <b>36</b> c	0.22 c	0.11 c	200 c	22 c	0.5 E	0.02 e		
Benzenethiol	108985	1.00E-05 H					0.37 N	0.037 N	0.014 N	20 N	0.78 N				
Benzidine	92875	3.00E-03 I		2.30E+02 i	2.35E+02		0.00029 c	0.00003 c	0.00001 c	0.025 c	0.0028 c	1.3 c	1.100E-06 c		
Benzoic acid	65850	4.00E+00 +					150000 N	15000 N	5400 N	1E+06 N	310000 N	320 s	280 f		
Benzotrichloride	98077			1.30E+01 i		_	0.0052 c	0.00048 c	0.00024 c	0.44 c	0.049 c	0.012 c	0.000073 c		
Benzyl alcohol	100516	3.00Е-01 н					11000 N	1100 N	410 N	610000 N	23000 N				
Benzyl chloride	100447			1.70E-01 I		Ø	0.062 c	0.037 c	0.019 c	34 c	3.8 c	0.5 c	0.0003/		
Beryllium and compounds	7440417	5.00E-03 i		4.30E+00 I	8.40E+00 +		0.016 c	0.00075 c	0.00073 c	1.3 c	0.15 c	690 E	180 L[		
Bidrin	141662	1.00E-04 I					3.7 N	0.37 N	0.14 N	200 N	7.8 N				
Biphenthrin (Talstar)	82657043	1.50E-02 +					<u>550 n</u>	55 N	20 N	31000 N	· 1200 N				
1,1-Biphenyl	92524	5.00E-02 (					1800 N	180 M	68 N	100000 N	3900 N	9000 .	110 N		
Bis(2-chloroethyl)ether	111444			1.10E+00 +	1.16E+00 i	Ø	0.0092 c	0.0054 c	0.0029 c	5.2 c	0.58 c	0.3 E	0.0003 E		
Bis(2-chloroisopropyl)ether	39638329	4.00E-02		7.00E-02 #	3.50E-02 +	Ø	0.26 c*	0.18 c	0.045 c	82 c	9.1 c				
Bis(chloromethyl)ether	542881			2.20E+02 +	2.17E+02 i	Ø	0.00005 c	0.00003 c	0.00001 c	0.026 c	0.0029 c	0.00004 c	1.000E-07 c		
Bis(2-chloro-1-methylethyl)ether	1			7.00E-02 w	7.00E-02 v	•	0.96 c	0.089 с	0.045 c	82 c	9.1 c	,			
Bis(2-ethylhexyl)phthalate (DEIIP)	117817	2.00E-02 I		1.40E-02 i			4.8 c	0.45 c	0.23 c	410 c	46 c	210 e	11 E		
Bisphenol A	80057	5.00E-02 i					1800 N	180 N	68 N	100000 N	3900 N				
Boron (and borates)	7440428	9.00E-02 i	5.71E-03 H				3300 N	21 N	120 N	180000 N	7000 1				
Boron trifluoride	7637072		2.00Е-04 н				7.3 н	0.73 N			]				
Bromodichloromethane	75274	2.00E-02 i		6.20E-02 i		∞1	0.17 c	0.1 c	0.051 c	92 c	10 c	1800 E	0.3 E		
Bromoethene	593602				1.10E-01 H	ω	0.096 с	0.057 с					1		
Bromoform (tribromomethane)	75252	2.00E-02		7.90E-03 I	3.85E-03 +	<b>X</b>	2.4 c	<u>1.6 c</u>	<u>0,4 c</u>	720 c	<b>81</b> c	46 E	0.5 ε		
Bromomethane	74839	1.40E-03 i	1.43E-03 i			Ø	8.7 N	5.2 N	1.9 N	`2900 н	110 M	2 €	0,1 ε		
4-Bromophenyl phenyl ether	101553	5.80E-02 o					2100 N	210 N	. 78 N	120000 N	4500 N		·		
Bromophos	2104963	5.00E-03 H					180 N	<u>18 н</u>	6.8 N	<u>н 00001</u>	390 N				

.

Sources: I=IRIS II=IIEAST A=IIEAST alternate W=Withdrawn from IRIS or IIEAST								Basis : C=carcinogenic effects N=noncarcinogenic effects E=EPA draft Soil Screening Level								
E=EPA NCEA Regional Support provisional value O=Other EPA documents.								S=soil saturation concentration M=EPA MCL.								
								Risk-Bas	Soil Screening Levels-							
						V	Тар	Ambient		Soit l	ngestion	Transfers	from Soil to:			
		RíDo	RÆDi	Cl'So	CPSI	0	Water	Air	Fish	Industrial	Residential	Air	Groundwater			
Contaminant	CAS	mg/kg/d	mg/kg/d	kg·d/mg	kg·d/mg	C	JIGL	jig/m3	mg/kg	nig/kg	mg/kg	mg/kg	nig/kg			
Chlorodifluoromethane	75456		1.43c+01 +			X	87000 N	52000 N					<b>_</b>			
Chloroethane	75003	4.00E-01 E	2.86E+00 +			<b>X</b>	8600 N	10000 N	540 N	820000 N	і 31000 м	2600 .	33 N			
2-Chloroethyl vinyl ether	110758	2.50E-02 o				Ø	150 N	91 N	34 N	51000 N	1 2000 N		•			
Chloroform	67663	1.00E-02 i		6.10E-03 i	8.05E-02 (	IX)	0.15 c	0.078 c	0.52 c	940 c	: 100 c	0.2 ε	0			
Chloromethane	74873			1.30Е-02 н	6.30E-03 н	120	1.4 c	0.99 c	0.24 c	440 c	· 49 c	0.063 c	0.0066 c			
4-Chloro-2,2-nicthylaniline hydrochloride	3165933			4.60E-01 H			0.15 c	0.014 c	0.0069 c	12 a	. 1.4 c	•				
4-Chloro-2-methylaniline	95692			5.80E-01 н			0.12 c	0.011 c	0.0054 c	9.9 c	1.l c					
beta-Chloronaphthalene	91587	8.00E-02					2900 N	290 N	110 N	160000 N	6300 N	2.8 s	140 N			
o-Chloronitrobenzene	88733			2.50E-02 #		ß	0.42 c	0.25 c	0.13 c	230 c	: 26 с					
p-Chloronitrobenzene	100005		·	1.80Е-02 н		(X)	0.59 c	0.35 c	0.18 c	320 c	35 c					
2-Chlorophenot	95578	5.00E-03 +					180 N	18 N	6.8 N	10000 N		53000 €				
2-Chloropropane	75296		2.8612-02 н	٠		ß	170 N	100 M				22 N	0.64 N			
Chlorothalouil	1897456	1.50E-02 i		1.10E-02 H			6.1 c	0.57 c	0.29 c	520 c	58 c					
o-Chlorotoluenc	95498	2.00E-02 +				ß	120 N	73 N	27 N	41000 N	1600 N	1200 N	5.6 N			
Chlorpropham	101213	2.00E-01 I	_			_	7300 N	730 N	270 N	410000 N	16000 N					
Chlopyrifos	2921882	3.00E-03 I					110 N	11 N	4.1 N	6100 N	230 14					
Chlorpyrifos-methyl	5598130	1.00E-02 н					370 N	37 N	14 N	20000 N	780 N					
Chlorsulfuron	64902723	5.00E-02 i		_			1800 N	180 N	68 N	100000_N	3900 N					
Chlorthiophos	60238564	8.00E-04 н					29 N	2.9 N	 1.1 м	1600 N	63 N					
Chromium III and compounds	16065831	1.00E+00 i	5.71E-07 w			Í	37000 N	0.0021 N	1400 N	1E+06 N	78000 N		ſ			
Chromium VI and compounds	18540299	5.00E-03 i			4.20E+01 i		180 N	0.00015 c	6.8 N	10000 N	. 390 N	140 e	I′_			
Coal tar	8001589				2.20E+00 w	,		0.0028 c					 			
Cobalt	7440484	6.00E-02 E					2200 N	220 N	81 N	120000 N	4700 N					
Coke Oven Emissions	8007452				2.17E+00 i			0. <b>0029</b> c			•					
**Copper and compounds	7440508	4.00E-02 e					1500 N	150 N	54 N	82000 N	13100 N					
Crotonaldehyde	123739	1.00E-02 w	۱	1.90Е+00 н	1.90E+00 w	,	0.035 c	0.0033 c	0.0017 c	3 c	0.34 c					
Cumene	98828	4.00E-02 i	2.57Е-03 н				1500 H	9.4 N	54 N	82000 N	3100 N	81 N	65 N			
Cvanides:																
Barium cyanide	542621	1.00E-01 w					3700 N	370 N	140 N	200000 N	7800 N		1			
Calcium cyanide	592018	4.00E-02 1				_ (	_1500 N	150 N	54 N	82000 N	3100 N					
Copper cyanide	544923	5.00E-03 i					180 N	18 N	6.8 N	10000 N	390 N		-			
Cyanazine	21725462	. 2.00E-03 H		8.40Е-01 н			0.08 c	0.0075 c	0.0038 c	6.8 с	0.76 c					
Cvanogen	460195	4.00E-02 i					1500 N	150 N	54 N	82000 N	3100 N					
Cyanogen bromide	506683	9.00E-02 i					3300 N	330 N	120 N	180000 N	7000 N					
Cyanogen chiloride	506774	5.00E-02 i				- {	1800 N	180 N	68 N	100000 N	3900 N					
Free cvanide	57125	2.00E-02 i					730 N	73 н	27 н	41000 N	1600 N					
llvdrogen cvanide	74908	2.00E-02 I	8.57E-04 i				730 N	3.1 N	27 N	41000 N	1600 N					
Potassium cyanide	151508	5.00E-02 (					1800 N	- 180 N	68 N	100000 N	3900 N		Í			
Potassium silver cvanide	506616	2.00E-01 (				- 1	7300 N	730 N	270 N	410000 N	16000 N					

ł

ŝ.

L

• 

# EPA Region III Risk-Based Concentrations: R.L. Smith (10/04/95)

· · · ·

.

		( 1010					·							
Sources: I=IRIS H=HEAST A=HEAST alternal	ie W=Wilhdro	wn Jrom IKIS	or HEAST			Basis : C=carcinogenic effects N=noncarcinogenic effects E=EPA druft Soil Screening Level								
E=EPA-NCEA Regional Support provisio	onal value O=	Other El'A do	cuments.			S=so	Il saturation co	ncentrution	_M=EPA MO	<u>, l</u>	<u> </u>			
						Kisk-Based Concentrations						ening Levels.		
	í .	0.00	· .	010		Tap	Ambient	191 A	Son Ir	gestion	Iransfers	from Soil to:		
	0.0	KID0	<u></u>	CPSo	CPSI U	Water	Air	Fish	Industrial	Residential	Air	Groundwater		
		mg/kg/d	mg/kg/d	kg·d/nig	kg·d/mg_[C	<u>µg/L</u>	_µg/m3	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg		
4-(2,4-Dichlorophenoxy)butyric Acid	94826	8.00E-03 I				290 N	29 N	11 N	16000 N	630 N				
1,2-Dichloropropane	78875		1.14E-03 +	6.80Е-02 н	123	0.16 c	0.092 c	0.046 c	84 c	9.4 c	11 E	0.02		
2,3-Dichloropropanol	616239	3.00E-03 I				110 N	<u>11 м</u>	4.1 N	6100 N	230 N				
1,3-Dichloropropene	542756	3.00E-04 i	5.71E-03 i	1.75E-01 II	1.30E-01 mIX	0.077 c	0.048 c	0.018 c	33 c	3.7 c	0.1 £	0.001		
Dichlorvos	62737	5.00c-04 +	1.43E-04 +	2.90E-01 i		0.23 c	0.022 c	0.011 c	· 20 c	2.2 c	3.5 c	0.00072		
Dicofol	115322			4.40E-01 w	·	0.15 c	0. <u>014 c</u>	0.0072 c	13 c	<u>· 1.5 c</u>				
Dicyclopentadiene	77736	3.00Е-02 н	5.71E-05 A		(X)	0.42 N	0.21 N	41 N	61000 N	2300 N				
Dieldrin	60571	5.00E-05 i		1.60E+01 +	1.61E+01 i	0.0042 c	0.00039 c	0.0002 c	0.36 c	0.04 c	2 €	0.001		
Diesel emissions			1.43E-03 i			<u>52 н</u>	<u>5.2 н</u>							
Diethyl phthalate	84662	8.00E-01 +				29000 N	2900 N	1100 N	1E+06 N	63000 N	520 e	110		
Diethylene glycol, monobutyl ether	112345		5.71E-03 #			210 N	21 N					•		
Diethylene glycol, monoethyl ether	111900	2.00Е+00 н				73000 N	7 <u>3</u> 00 N	2700 H	1E+06 N	160000 N				
Diethylforainide	617845	1.10Е-02 н				400 N	40 N	15 N	22000 м	860 N				
Di(2-ethylliexyl)adipate	103231	6.00E-01 +		1.20E-03 i		56 c	5.2 c	2.6 c	4800 c	530 c				
Diethylstilbestrol	56 <u>53</u> 1			4.70E+03 н		0.00001 c	1E-06 c	7E-07 c	0.0012 c	0.00014 c				
Difenzoquat (Avenge)	43222486	8.00E-02 I				2900 N	290 N	110 N	160000 м	. 6300 N				
Diflubenzuron	35367385	2.00E-02 +				730 N	73 N	27 N	41000 N	1600 N				
1,1-Difluoroethane	75376		1.14E+01 i			69000 N	42000 N							
Diisopropyl methylphosphonate (DIMP)	1445756	8.00E-02 +				2900 N	290 N	, 110 N	160000 N	6300 N	_			
Dimethipin	55290647	2.00E-02 I				730 N	73 N	27 н	41000 N	1600 N				
Dimethoate	60515	2.00E-04 +				7.3 N	0.73 N	. 0.27 N	410 N	16 N	_	_		
3.3'-Dimethoxybenzidine	119904			1.40Е-02 н		4.8 c	0.45 c	0.23 c	410 c	46 c				
Dimethylamine	124403		5.71E-06 w			0.21 H	0.021 N							
2.4-Dimethylaniline hydrochloride	21436964	•		5.80E-01 н		0.12 c	0.011 c	0.0054 c	9.9 с	1.1 c				
2.4-Dimethylaniline	95681			7.50E-01 #		0.09 c	0.0083 c	0.0042 c	7.6 c	0.85 c				
N-N-Dimethylaniline	121697	2.00E-03 i				73 н	7.3 N	2.7 н	4100 N	160 N				
3.3'-Dimethylbenzidine	119937			9.20E+00 н		0.0073 c	0.00068 c	Ó.00034 c	0.62 c	0.069 c	29 c	0.00039		
N.N-Dimethylformamide	68122	1.00E-01 #	8.57E-03	,		3700 N	31 N	140 N	200000 N	7800 N				
1.1-Dimethylhydrazine	57147		•	2.60E+00 w	3.50E+00 w	0.026 c	0.0018 c	0.0012 c	2.2 с	0.25 c				
1.2-Dimethylhydrazine	540738			3.70E+01 w	3.70E+01 w	0.0018 c	0.000 <b>17 c</b>	0.00009 c	0.15 c	0.017 c		_		
2.4-Dimethylphenol	105679	2.00E-02 (				. 730 N	73 N	27 H	41000 N	1600 N	5400 s	3		
2.6-Dimethylphenol	576261	6.00E-04 i				22 N	2.2 N	0.81 N	1200 N	47 N				
3.4-Dimethylphenol	95658	1.00E-03 +				37 N	3.7 N	1.4 N	2000 N	78 N				
Dimethyl ohthalate	131113	1.00E+01 H				370000 N	37000 N	14000 N	1E+06 N	780000 N	- 1600 E	1200		
Dinethyl terenbthalate	120616	1.00E-01			•	3700 N	370 N	140 N	200000 м	7800 №				
1.2-Dinitrohenzene	528290	4.00E-04 m				15 N	1.5 N	0.54 N	820 N	31 .				
1.3. Dinitrohenzene	99650	1.00F-04 +				3.7 N	0.37 N	0.14 N	200 N	7.8 N				
1 A Divitrohenzene	100254	4 00F.04 -	•			15 н	1.5 H	0.54 H	820 H	31				
1,4-Dillitopenzene	121904	2000 02 .				71	73,	2.2.1	4100	160 .				
(4 6-1) INTERO-O-CVCIONEXVI DIICHUI	1 131073	2.000-031			_			A.7 M	4100 N	100 1				

1

12

ł

.

Sources: I=IRIS II=IIEAST A=IIEAST alternate W=Withdrawn from IRIS or IIEAST Be							Basis : C=carcinogenic effects N=noncarcinogenic effects E=EPA druft Soil Screening Level								
E=EPA NCEA Regional Support provisional value O=Other EPA documents.							S=soil saturation concentration M=EPA MCL.								
							Risk-Based Concentrations						ening Lovels-	7	
· · · ·						V	Tap	Ambient		Soil In	gestion	Transfers	from Soil to:		
		RíDo	RíDi	CPSo	CPSi	0	Waler	Air	Fish	Industrial	Residential	Air	Groundwater	. <b>1</b>	
Contaminant	CAS	mg/kg/d	mg/kg/d	kg d/mg	kg·d/mg	C	μg/L	µg/m3	mg/kg	nig/kg	mg/kg	mg/kg	nig/kg	1	
Ethyl p-nitrophenyl phenylphosphorothioate	2104645	1.00E-05 +					0.37 N	0.037 H	0.014 N	20 N	0.78 H			-	
Ethylnitrosourea	759739			1.40E+02 w			0.00048 c	0.00005 c	0.00002 c	0.041 c	0.0046 c				
Ethylphthalyl ethyl glycolate	84720	3.00E+00 (					110000 N	11000 N	4100 N	1E+06 N	230000 N			1	
Express	10120	8.00E-03 +					290 N	29 H	• 11 H	16000 N	630 N			J	
Fenamiphos	22224926	2.50E-04 I					9.1 N	0.91 N	0.34 N	· 510 N	20 N				
Fluometuron	2164172	1.30E-02 +					470 N	47 N	18 N	27000 N	1000 N				
Fluoride	7782414	6.00E-02 I					2200 N	220 N	81 N	120000 N	4700 N			1	
Fluoridone	59756604	8.00E-02 /					2900 N	290 N	110 N	160000 N	6300 N			1	
Flurprimidol	56425913	2.00E-02 1					730 N	73 N	27 н	41000 N	1600 N				
Flutolanil	66332965	6.00E-02 I					2200 N	220 N	81 N	120000 N	4700 N			7	
Fluvalinate	69409945	1.00E-02 i					370 N	37 м	14 н	20000 N	780 N		•		
Folpet	133073	1.00E-01 +		3.50E-03 i			19 c	1.8 c	0.9 c	1600 c	180 c				
Foinesafen	72178020			1.90E-01 i			0.35 c	0.033 c	0.017 c	30 c	- 3.4 с				
Fonofos	944229	2.00E-03 i					73 N	7.3 N	2.7 н	4100 N	160 N			·	
Formaldehyde	50000	2.00E-01 I			4.55E-02 +		7300 N	0.14 c	270 N	410000 N	16000 N				
Formic Acid	64186	2.00Е+00 н	_		_		73000 N	7300 N	2700 N	1E+06 N	160000 N			7	
Fosetyl-al	39148248	3.00E+00 i					110000 N	11000 N	4100 N	1E+06 N	230000 N				
Furan	110009	1.00E-03 +					37 N	3.7 N	<u>1,4 н</u>	2000 N	78 N				
Furazolidone	67458			3.80E+00 H			0.018 c	0.0016 c	0.00083 c	1.5 c	0.17 c			1	
Furfural	98011	3.00E-03 i	1.43E-02 🔺				110 N	52 N	4.1 н	6100 N	- 230 N		· · ·		
Furium	531828			5.00E+01 н		_	0.0013 c	0.00013 c	0.00006 c	0.11 c	0.013 c			1	
Furmecyclox	60568050			3.00E-02 i			· 2.2 c	0.21 c	0.11 c	190 c	<b>21</b> c			Ľ	
Glufosinate-ammonium	77182822	4.00E-04 i					15 N	1.5 н	0.54 N	820 N	31 M				
Glycidaldehyde	765344	4.00E-04 I	2.86Е-04 н				15 N	<u> </u>	0.5,4 н	820 N	<u>31 n</u>			_	
Glyphosate	1071836	1.00E-01 i					3700 N	370 N	140 N	200000 N	7800 N				
Haloxyfop-methyl	69806402	5.00E-05 I					· 1.8 N	0.18 N	0.068 N	100 N	3.9 N				
llarmony	79277273	1.30E-02 i					470 N	47 N	<u>18 n</u>	27000 N	1000 N			_	
11C11 (alpha)	319846			6.30E+00 i	6.30E+00 i		0.011 c	0.00099 c	0.000 <b>5</b> c	0.91 c	0.1 ε	0.9 e	0.0004 e	É	
IICII (beta)	319857			1.80E+00 +	1.80E+00 i		0.037 c	0.0035 c	0.0018 c	3.2 c	0.35 c	16 E	0.002 e	ŧ	
11C11 (gamma) Lindane	58899	3.00E-04 I		1.30Е+00 н			0.052 c	0.0048 c	0.0024 c	4.4 c	0,49 c	<u>4.2 c</u>	0.006 E	£	
HCH-technical	608731			1.80E+00 i	1.79E+00 i		0.037 c	0.0035 c	0.0018 c	3.2 с	0.35 c				
Heptachlor	76448	5.00E-04 i		4.50E+00 i	4.55E+00 i	Ø	0.0023 c	0.0014 c	0.0007 c	. I.3 c	0.14 c	0.3 e	0.06 E	:	
Heptachlor epoxide	1024573	1.30E-05 i		9.10E+00 i	9.10E+00 i	B	0.0012 c	<u>0.00069 c</u>	0.00035 c	0.63 c	<u> </u>	<u> </u> E	0.03 E	·	
llexabromobenzene	87821	2.00E-03 i				∞	12 N	7.3 м	2.7 N	4100 N	160 w				
Hexachlorobenzene	118741	8.00E-04 +		1.60E+00 i	1.61E+00 i	ß	0.0066 c	0.0039 c	0.00 <b>2</b> c	3.6 c	0.4 c	1 6	0.8 E		
[lexachlorobutadiene	87683	2.00E-04 н		7.80E-02 +	7.70E-02 I	<u> </u>	0.14 c	0.081 c	0.04 c	<u>73 c</u>	8.2 c	<b>i</b> e	0.1 ε		
llexachlorocyclopentadiene	77474	7.00E-03 I	2.00E-05 ii			<b>1</b> 20	0.15 N	0.073 N	9.5 N	14000 N	550 N	2ε	10 E		
Hexachtorodibenzo-p-dioxin mixture	19408743			6.20E+03 i	4.55E+03 i	·	0.00001 c	. 1E-06 c	5E-07 c	0.0009 c	0.0001 c				
Hexachloroethane	67721	1.00E-03 (		1.40E-02 I	1.40E-02 (	(X)	<u>0.75 c</u>	0.45 c	0.23 c	410 c	46 c	· 49 E	<u>0.2 є</u>	]	

۱

ł,

.

ł

.

.

14

. 1

۱

1

Sources: I=IRIS II=IIEAST _ A=IIEAST alternate W=Withdrawn from IRIS or IIEAST						Basis : C=carcinogenic effects N=noncarcinogenic effects E=EPA draft Soil Screening Level									
E=EPA-NCEA Regional Support provisional value O=Other EPA documents.							S=soll saturation concentration M=EPA MCL.								
							Risk-Based Concentrations Soil Screening I								
						V	Tap	Ambient		Soil Iı	gestion	Transfers	from Soil to:		
		RíDo	RfDi	CPSo	CPSi	0	Water	Air	Fish	Industrial	Residential	Air	Groundwater		
Contaminant	CAS	mg/kg/d	mg/kg/d	kg·d/mg_	kg·d/mg	C	µg/L	µg/m3	mg/kg	mg/kg	mg/kg	mg/kg	nıg/kg		
Methamidophos	10265926	5.00E-05 i					1.8 N	0.18 N	0.068 N	100 N	3.9 N				
Methanol	67561	5.00E-01 i					18000 N	1800 N	680 N	1E+06 N	39000 H				
Methidathion	950378	1.00E-03 I					· 37 N	3.7 N	1.4 н	2000 N	78 N				
Methomyl	16752775	2.50E-02 I					910 N	91 N	34 N	51000 N	2000 N		·		
Methoxychlor	72435	5.00E-03 I					180 N	18 N	6.8 н	, 10000 N	390 N	41 .	62		
2-Methoxyethanol acetate	110496	2.00E-03 🗚					73 N	7.3 N	2.7 N	4100 N	. 160 N				
2-Methoxyethanol	109864	1.00E-03 H	5.71E-03 i				37 N	21 N	t.4 n	2000 N	78 N		,		
2-Methoxy-5-nitroaniline	99592			4.60Е-02 н			1.5 c	0.14 c	0.069 c	120 c	14 c				
Methyl acetate	79209	1.00E+00 н					37000 N	3700 N	1400 N	1E+06 N	78000 N				
Methyl acrylate	96333	3.00E-02 A					1100 N	110 N	41 N	61000 N	2300 N				
2-Methylaniline hydrochloride	636215			1.80E-01 н			0.37 c	0.035 c	0.018 c	32 c	3.5 c		•		
2-Methylaniline	95534	_		2.40Е-01 н			0.28 c	0.026 c	0.013 c	24 c	2.7 c				
Methyl chlorocarbonate	79221	1.00E+00 w		-			37000 N	3700 N	1400 N	1E+06 N	78000 N				
4-(2-Methyl-4-chlorophenoxy) butyric acid	94815	1.00E-02 I					370 N	37 N	14 N	20000 N	780 N				
2-Methyl-4-chlorophenoxyacetic acid	94746	5.00E-04 +					18 N	1.8 N	0.68 N	1000 N	39 N				
2-(2-Methyl-14-chlorophenoxy)propionic acid	93652	1.00E-03 i					37 N	3.7 N	1.4 н	2000 N	78 N				
Methylcyclohexane	108872		8.57Е-01 н				31000 N	3100 N				60 s	1500 N		
Methylene bromide	74953	1.00E-02 A				Ø	61 N	37 N	14 N	20000 N	780 N				
Methylene chloride	75092	6.00E-02 I	8.57E-01 #	7.50E-03 i	1.64E-03	ß	4.1 c	3.8 c	0.42 c	760 c	85 c	7 €	0.01 E		
4,4'-Methylene bis(2-chloroaniline)	101144	7.00Е-04 н		1.30E-01 H	1.30E-01 i	н	0.52 c	0.048 c	`0.024 c	44 c	4.9 c				
4,4'-Methylenebisbenzeneamine	101779			2.50E-01 w			0.27 c	0.025 c	0.013 c	23 c	<b>2</b> .6 c				
4,4'-Methylene bis(N,N'-dimethyl)aniline	101611			4.60E-02 +			1.5 c	0.14 c	0.069 c	120 c	· 14 c		· .		
4,4'-Methylenediphenyl isocyanate	101688		5.71E-06 i			80	0.035 N	0.021 N							
Methyl ethyl ketone	78933	6.00E-01 i	2.86E-01 i			Ø	1900 N	1000 N	810 N	1E+06 N	47000 N				
Methyl hydrazine	60344			1.10E+00 w			0.061 c	0.0057 c	0.0029 c	5.2 c	0.58 c	_			
Methyl isobutyl ketone	108101	8.00E-02 н	2.29E-02 A				2900 N	84 N	110 N	160000 N	. 6300 N				
Methyl methacrylate	80626	8.00E-02 н					2900 N	290 N	110 N	160000 м	6300 N				
2-Methyl-5-nitroaniline	99558			3.30E-02 #			2 c	0.19 c	0.096 c	170 c	19 c				
Methyl parathion	298000	2.50E-04 i					9.1 N	0.91 N	0.34 N	510 N	20 N	28 s	0.041 N		
2-Methylphenol (o-cresol)	95487	5.00E-02 I					1800 N	180 N	68 N	100000 N	3900 N	12000 s	6 E		
3-Methylphenol (m-cresol)	103394	5.00E-02 i					1800 N	180 N	68 N	100000 N	3900 N				
4-Methylphenol (p-cresol)	106445	5.00E-03 н					180 N	18 N	<sup>.</sup> 6.8 м	10000 N	390 N				
Methyl styrene (mixture)	25013154	6.00E-03 🔺	1.14E-02 A			ß	60 N	42 N	8.1 N	<u>12000 н</u>	470 N	м 001	<u> </u>		
Methyl styrene (alpha)	98839	7.00E-02 A				ß	430 N	260 N	95 N	140000 N	5500 N	8.8 s	7.5 N		
Methyl tertbutyl ether (MTBE)	1634044	5.00E-03 E	8.57E-01 i			8	180 M	3100 N	6.8 N	10000 N	390 N				
Metolaclor (Dual)	51218452	1.50E-01 H					5500 N	550 N	200 N	310000 N	12000 N				
Metribuzin	21087649	2.50E-02 I					910 N	91 N	34 N	51000 N	2000 N				
Mirex	2385855	2.00E-04 I		1.80E+00 w			0.037 c	0.0035 c	0.0018 c	3.2 c	0.35 c				
Molimate	2212671	2.00E-03 I					73 N	7.3 N	2.7 н	4100 N	160 N				

.

.

Era Region III Risk-Based Concentrations: R.L. Smith (10/04/95)

ŧ

Sources: I=IRIS II=IIEAST A=IIEAST alter	mate W=Withdra	wn from IRIS	or IIEAST				Basis : C≠ca	rcinogenic effe	ects N=nonc	arcinogenic e	fects E=EP	A draft Soll So	reening Level	
E=EPA-NCEA Regional Support prov	isional value 0=	Other EPA do	cuments,	·		<b>.</b>	S=so	il saturation co	oncentration	M=EPA MC	<u>L.</u>		<u> </u>	
								Risk-Ba	sed Concent	rations		Soil Screening Levels-		
			•			V	Тар	Ambient		<u>Soil In</u>	gestion	Transfers	from Soil to:	
		RíDo	<u> </u>	CPSo	<u>CPSi</u>	0	Water	<u> </u>	Fish	Industrial	Residential	Air	Groundwater	
Contaminant	CAS	_mg/kg/d	mg/kg/d	kg·d/mg	kg·d/mg	C	µg/L	µg/m3	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	
Oxadiazon	19666309	5.00E-03 +					180 N	18 N	6.8 N	10000 N	390 N			
Oxamyl	23135220	2.50E-02 i					910 N	91 N	34 N	- 51000 н	2000 N			
Oxyfluorfen	42874033	3.00F-03 i					110 N	<u>— 11 н</u>	<u>4.1 n</u>	6100 N	230 N		-	
Paclobutrazol	76738620	1.30E-02 i					470 N	47 н	18 N	27000 м	1000 N		·	
Paraquat	1910425	4.50E-03 i					160 N	16 N	б.1 н	́ 9200 м	· 350 N			
Parathion	56382	6.00Е-03 н		·			220 N	22 N	8.1 N	12000 N	470 N	110 .	3.9 •	
Pebulate	1114712	5.00E-02 H					1800 N	180 N	68 N	100000 н	3900 N			
Pendimethalin	40487421	4.00E-02 i					1500 N	150 N	54 N	82000 N	3100 N			
Pentabromo-6-chloro cyclohexane	87843			2.30E-02 H			2.9 c	0.27 с	0.14 c	250 с	<b>28</b> c			
Pentabromodiphenyl ether	32534819	2.00E-03 1					73 N	7.3 N	2.7 N	4100 N	160 N			
Pentachlorobenzene	608935	8.00E-04 i				ß	4.9 н	2.9 N	1.1 н	1600 N	. 63 N	570 N	48 M	
Pentachloronitrobenzene	82688	3.QOE-03 i		2.60E-01 н		ß	0.041 c	0.024 c	0.012 c	22 c	<b>2.5</b> с			
Pentachlorophenol	87865	3.00E-02 +		1.20E-01 I			- 0.56 c	0.052 c	0.026 c	48 c	5.3 c	7.9 c	0.2 6	
Permethrin	52645531	5.00E-02 i					1800 N	180 N	68 N	100000 N	3900 N			
Phenmedipham	13684634	2.50E-01 i					9100 N	́ 910 н	340 N	510000 N	20000 N			
Phenol	108952	6.00E-01					22000 N	2200 N	810 N	1E+06 N	47000 N	21000 #	49 e	
m-Phenylenediamine	108452	6.00E-03 i		•			220 H	22 N	8.1 N	12000 N	470 N			
p-Phenylenediamine	106503	1.90Е-01 н					6900 N	690 N	260 N	390000 N	15000 N			
Phenylmercuric acetate	62384	8.00E-05 i					2.9 N	0.29 N	.0.11 H	160 N	6.3 N			
2-Phenylohenol	90437			1.94Е-03 н		i	35 c	3.2 c	1.6 c	3000 c	330 c			
Phorate	298022	2.00E-04 н					7.3 N	0.73 N	0.27 H	410 N	16 N			
Phosmet	732116	2.00E-02					730 N	73 H	27 N	41000 N	1600 N		· .	
**Phosphine	7803512	3.00E-04	8.57E-05 н				11 11	0.31 H	- 0.41 N	610 N	23 N			
**Phosphoric acid	7664382		2.86E-03				100 M	10 N	•		· · · ·			
Phosphorus (white)	7723140	2.00E-05 i					0.73 H	0.073 N	0.027 N	41 N	L6 N			
n-Phthalic acid	100210	1.00E+00 H					37000 M	3700 N	1400 N	1E+06 N	78000 N			
Phthalic anhydride	85449	2.00E+00	3.43Е-02 н				73000 N	130 N	2700 N	1E+06 N	160000 N			
Picloram	1918021	7.00E-02 1					2600 N	260 N	95 m	140000 N	5500 N			
Picininhos-methyl	29232937	1.00E-02 1				•	370 N	37 N	14 n	20000 N	780 N			
Polyhrominated hinhenvis		7.00E-06 u		8.90E+00 H			0.0076 c	0.0007 c	0.00035 c	0.64 c	0.072 c			
Polychlorinsted binbenyls (PCBs)	1336363	7,002.001		7.708+00 +			0.0087 c	0.00081 c	0.00041 c	0.74 c	0.083 c			
A malor 1016	12674112	7.006.05 (		11102 00 1	<b>.</b>		26	0.26 м	0.095	140	55.4			
	11007601	2.000.05					0.73	0.073	0.027	41 u	16			
Alugitarianted tembersule (DCTe)	1107/091	2.002-03 1		4 502+00 -			0.015 -	0.075 N	0.0007 ~	<u> </u>	014			
Polychiorinated terpitenyis (PC 18)	l l			-1.005100 F			0.015 C	0.0014 6	0.0007 C	1.3 C	0.14 0	110000 -		
rorynuclear aromatic hydrocardons		4 005 00					1100	220	0 f	120000	4200	130 -	- <b>2</b> 00 -	
Acenaphthene		0.0012-02 1					N	1100 ···	BIN	120000 N	32000 N	120 S	<u></u>	
Anthracene	120127	3.008-01 1		7 205 01 -	6 105 01		11000 N	1100 N	410 N 	010000 N	V 000 N	0.5 \$	4300 €	
Benzlajanthracene	50553			7.30E-01 E	0.IUE-UI E		0.092 c	0.01 C	U.UU43 C	7.8 C	U.88 C	2/ 3	U.7 €	
Benzo b fluoranthene	205992			<u></u>	0.1015-01 E		<u>0.092 c</u>	U.UI C	<u>0.0043 c</u>	<u>/.8_</u> c	<u> </u>	23 \$	4	

ł.

.

1

ŧ

.

ł

ŧ

1

18

1

ł.
EPA Region III Risk-Based Concentrations: R.L. Smith (10/04/95)

I,

Sources: I=IRIS II=IIEAST A=IIEAST alternat E=El'A-NCEA Regional Support provisia	e W≈Withdra nalvalue O=	wn from IRIS Other EPA do	or IIEAST cuments,				Basis: C=ca S=sol	rcinogenic effe Il saturation co	ects N≠nonce oncentration	arcinogenic M≠EPA M(	e∬ects E=EP CL.	A draft Soil S	creening Level
······································		I				1		Risk-Bas	ed Concent	rations		Soil Scree	ning Levels.
					Tan Amhlent Soil Invest			restion	n Transfers from Soil to:				
	•	RſD₀	RíDi	CPSo	CPSi	0	Water	Air	Fish	Industrial	Residential	Air	Groundwater
Contaminant	CAS	mg/kg/d	mg/kg/d	kg d/mg	kg·d/mg	C	µg/L	µg/m3	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Silver and compounds	7440224	5.00E-03 I					180 N	18 N	6.8 N	10000 N	390 N		
Simazine	122349	5.00E-03 1		1.20Е-01 н			0.56 c	0.052 c	0.026 c	48 c	- 5.3 c		
Sodium azide	26628228	4.00E-03 I			. •		150 н	15 N	5.4 N	8200 N	310 N		
Sodium diethyldithiocarbamate	148185	3.00E-02 I		2.70Е-01 н			0.25 c	0.023 c	0.012 c	21 c	2.4 c		
Sodium fluoroacetate	62748	2.00E-05 +					0.73 N	0.073 N	0.027 N	· 41 н	. 1.6 N		
Sodium metavanadate	13718268	1.00Е-03 н					37 N	3.7 N	1.4 м	2000 N	78 N		
Strontium, stable	7440246	6.00E-01 +		· · ·			22000 N	2200 N	810 N	1E+06 N	47000 N		
Strychnine	57249	3.00E-04 i	,				11 м	1.1 n	0.41 N	610 N	23 N		
Styrene	100425	2.00E-01 i	2.86E-01 i				1600 N	1000 N	270 N	410000 N	16000 N	1400 E	2 €
Systhane	88671890	2.50E-02 +					910 N	91 N	34 N	51000 N	2000 N		
2,3,7,8-TCDD (dioxin)	1746016			1.56E+05 II	1.16E+05 #	1	4E-07 c	SE-08 c	c	4E.05 c	4E-06 c		
Tebuthiuron	34014181	7.00E-02 I					2600 N	260 N	95 N	140000 N	5500 M		
Temephos	3383968	2.00E-02 H					730 N	73 н	27 N	41000 N	1600 N		
Terbacil	5902512	1.30E-02 i	•				470 N	47 N	18 N	27000 N	1000 N		
Terbufos	13071799	2.50Е-05 н					0.91 N	0.091 N	0.034 N	51 N	2 N		
Terbutryn	886500	1.00E-03 +					37 N	3.7 N	1.4 N	2000 N	78 N		
1,2,4,5-Tetrachlorobenzene	95943	3.00E-04 i				8	1.8 и	1.1 м	0.41 N	610 N	23 N	91 N	0.69 .
1,1,1,2-Tetrachloroethane	630206	3.00E-02 I		2.60E-02	2.59E-02		0.41 c	0.24 c	0.12 c	220 c	25 c		
1,1,2,2-Tetrachloroethane	79345			2.00E-01 )	2.03E-01	8	0.052 c	0.031 c	0.016 c	29 c	3.2 c	0.4 e	0.001 E
Tetrachloroethylene (PCE)	127184	1.00E-02 1		5.20E-02 ∉	2.03E-03		1.1 c	3.1 c	0.061 c	110 c	12 c	11 E	0.04 E
2,3,4,6-Tetrachlorophenol	58902	3.00E-02 I					1100 N	_ 110 N	_41 N	61000 N	2300 N		
p, s, a, a-Tetrachlorotoluene	5216251			2.00Е+01 н			0.00053 c	0.00031 c	0.00016 c	0.29 c	0.032 c		
Tetrachlorovinphos	961115	3.00E-02 I		2.40E-02 н			2.8 с	0.26 c	0.13 c	240 с	27 c		
Tetraethyldithiopyrophosphate	3689245	5.00E-04 I					18 N	1.8 N	0.68 N	1000 N	39 N		•
Tetraethyl lead	78002	1.00E-07 I	•				0.0037 N	0.00037 N	0.00014 N	0.2 N	0.0078 N	0.00068 N	0.000034 N
**1,1,1,2-Tetrafluoroethane	811972		2.29E+01			Ø	140000 N	84000 N			•		
Thallic oxide	1314325	7.00E-05 w					2.6 N	0.26 N	0.095 N	140 N	5.5 N		
Thallium													0.4 ε
Thallium acetate	563688	9.00E-05 i					· 3.3 N	0.33 N	0.12 N	180 N	7 N		
Thallium carbonate	6533739	8.00E-05					2.9 N	0.29 N	0.11 N	<u>160 n</u>	6.3 N		
Thallium chloride	7791120	8.00E-05 1					2.9 N	0.29 H	0.11 N	160 N	6.3 N		
Thallium nitrate	10102451	.9.00E-05 i					3.3 N	0.33 N	0.12 N	180 N	7 N		
Thallium selenite	12039520	9.00E-05 w					3.3 N	0.33 N	0.12 N	180 N	7 N		
Thallium sulfate	7446186	8.00E-05 i					2.9 н	0.29 N	0.11 N	160 N	6.3 N		
Thiobencarb	28249776	1.00E-02 i		•			370 N	37 н	14 н	20000 N	780 N		
2-(Thiocyanomethylthio)-benzothiazole	21564170	3.00Е-02 н					1100 м	<u>110 n</u>	41 N	61000 N	2300 N		
Thiofanox	39196184	3.00E-04 II					11 н	1.1 н	0.41 N	610 N	23 N		
Thiophanate-methyl	23564058	8.00E-02 +					2900 N	- 290 н	110 N	160000 H	6300 N		
Thiram	137268	5.00E-03 I					180 N	18 N	6.8 N	10000 N	390 N		

1

ł

20

1

.

ŧ.

1

1

I.

1

â

EPA Region in Risk-Based Concentrations: R.L. Smith (10/04/95)

.

.

.

Sources: 1=1RIS H=11EAST A=HEAST alternate	W=Withdra	wn from IRIS	or HEAST	•			Basis : C=ca	rcinogenic effe	cts N=nonc	arcinogenic	effects E=EP	A draft Soil S	creening Level	
E=EPA-NCEA Regional Support provisional value O=Other EPA documents.						S=soll saturation concentration M=EPA MCL.								
								Risk-Based Concentrations					Soil Screening Lovels-	
	· ·				· .	V.	Tap	Amblent	•.	Soil In	gestion	Transfers	from Soil to:	
		RíDo	RfDi	CPSo	CPSI	0	Water	Air	Fish	Industrial	Residential	Air	Groundwater	
Contaminant	CAS	ung/kg/d	mg/kg/d	kg·d/mg	kg d/mg	]C	jig/L	jig/m3	mg/kg	nig/kg	mg/kg	mg/kg	mg/kg	
Vanadium sulfate	36907423	2.00Е-02 н					730 n <sup>i</sup>	73 N	27 N	41000 N	1600 N			
Vernam	1929777	1.00E-03 i					37 н	3.7 N	1.4 N	2000 N	78 N			
Vinclozolin	50471448	2.50R-02 I					910 N	91 н	34 N	51000 N	2000 N		. I	
Vinyl acetate	108054	1.00Е+00 н	5.71E-02 i		· ·		37000 N	210 N	1400 N	1E+06 N	78000 N	370 E	8.	
Vinyl bromide	593602	•	8.57E-04 i			X	5.2 н	3.1 N				2 N	0.018 N	
Vinyl chloride	75014			1.908+00 н	3.00E-01	1021	0.019 c	0.021 c	0.0017 c	3 с	0.34 c	0.002 ∉	0.01 E	
Warfarin	81812	3.00E-04 (					11 м	1.1 и	0.41 N	610 N	23 N	0.046 N	1800 N	
m-Xylene	108323	2.00Е+00 н	2.00E-01 w			Ø	1400 N	730 N	2700 N	1E+06 N	160000 N	950 s	240 M	
o-Xylene	9.55E+04	2.00Е+00 н	2.00E-01 w			(8)	1400 N	730 N	2700 N	1E+06 N	160000 N	730 s	1.50E+02 u	
p-Xylene	1.06E+05	·	8.57E-02 w				520 N	310 N			2	1000 #	2.20E+02 M	
Xylene (mixed)	1.33E+06	2.00E+00 +					12000 N	7300 н	2700 N	1E+06 N	и 0000а1	<b>320 €</b>	7.40E+01 €	
<u>Zinc</u>	7.44E+06	3.00E-01 (					11000 N	1100 N	410 N	610000 N	23000 N		4.20E+04 €	
Zinc phosphide	1.31E+06	3.00E-04 i					· 11 N	1.1 N	0.41 N	610 N	23 N			
Zineb	1.21E+07	5.00E-02 1				i	1800 N	180 N	68 N	100000 N	3900 N			
												•	_	

L

ł

.

I.

.

1

.

22

L

4

ŧ.

.

1

ŧ

# APPENDIX I

# DERIVATION OF CONSTRUCTION WORKER SOIL INGESTION SCREENING CONCENTRATION

.

.

# DERIVATION OF CONSTRUCTION WORKER SOIL INGESTION SCREENING CONCENTRATION

The construction worker soil ingestion screening concentration has been derived using the same basic approach as the industrial soil screening value, except that a construction worker soil ingestion rate has been utilized and a one year exposure duration was assumed for construction activities related to site redevelopment. A target hazard quotient of 1 is the basis of the screening concentration. Since a one year exposure is considered a subchronic exposure scenario, a subchronic Reference Dose (RfD) for mercury has been employed. However, the conservative screening value is based on the subchronic RfD for methyl mercury (0.0001 mg/kg/day), the mercury species with the lowest RfD. The USEPA has adopted the chronic RfD as the subchronic RfD as well. The calculation of the soil ingestion screening concentration for the construction worker is shown in Table B-1.

The construction worker soil ingestion rate (118 mg soil per day) has been calculated based on a series of assumptions previously made by the USEPA. The soil ingestion rate represents a recalculation of work previously conducted by Hawley, with an updated skin soil adherence rate. The soil ingestion rate has been calculated as follows. Hawley has assumed that an adult working outdoors ingests twice daily a quantity of soil corresponding to one-half the covering of the inside surface of the fingers and thumbs of both hands. According to USEPA (1992), the inside surface of the fingers and thumbs of both hands is 14% of the surface area of the hands or 118 cm<sup>2</sup> and the upper bound estimate of soil adherence rate is 1.0 mg/cm<sup>2</sup>. Based on this information, the daily soil intake rate is:  $2 \times 0.5$  (118 cm<sup>2</sup>) x 1.0 mg soil/cm2 = 118 mg soil / day.

Hawley, J.K., 1985. Assessment of Health Risk from Exposure to Contaminated Soil, Risk Analysis, Vol. 5, No. 4, pp.289-302.

USEPA, 1992. Dermal Exposure Assessment: Principles and Applications, Interim Report, EPA/600/8-91/011B, January

# ABB ENVIRONMENTAL SERVICES

G:\T92\KENT\_TAY\PHASE1\SEC55##2.DOC

#### TABLE B-1 - DERIVATION OF CONSTRUCTION WORKER SOIL INGESTION SCREENING CONCENTRATION

CONSTRUCTION WORKER SCENARIO	UNITS	VALUE
SOIL INGESTION RATE	MG/DAY	118
FREQUENCY OF EXPOSURE	DAYS/WK	5
DURATION OF EXPOSURE	WEEKS	50
FRACTION OF SOIL INGESTION AT SITE	NA	0.5
BODYWEIGHT	KG	70
RELATIVE ABSORPTION FACTOR	NA	1
UNITS CONVERSION FACTOR (CF1)	KG/MG	1.00E-06
UNITS CONVERSION FACTOR (CF2)	DAY/WK	7.00E+00
ORAL RfD	MG/KG/DAY	1.00E-04 METHYL MERCURY
TARGET HAZARD INDEX	NA	1.00E+00

TARGET CONCENTRATION (SOIL INGESTION) MG/KG 166

TARGET SOIL CONC (INGESTION ONLY) = TARGET HI X RED X BW X DURATION X CF2 / (SOIL INGESTION RATE X FREQUENCY X DURATION X RELATIVE ABSORPTION FACTOR X FRACTION FROM SITE X CF1)

ŝ,

# APPENDIX J

# MERCURY SPECIATION AND BIOAVAILABILITY TESTING BACKGROUND INFORMATION AND TECHNIQUES

# **TECHNICAL BACKGROUND AND RATIONALE**

### MERCURY SPECIATION

Mercury can occur in soils as elemental mercury in liquid or vapor form, organic mercury compounds, mercuric chloride, or one of several different mineral species, including mercuric oxides, carbonates, and sulfides. In general, organic mercury, mercuric chloride, and elemental mercury in the vapor phase are very soluble and bioavailable, mercuric oxides and carbonates are less soluble, and liquid elemental mercury and mercuric sulfides are insoluble and non-bioavailable. Furthermore, mercury speciation may vary with depth in soils. The chemical form of mercury controls its mobility in the soil, its bioavailability when ingested, and its response to specific remedial actions. Therefore, an understanding of mercury speciation in soils at the Ames Street site will be critical for determining the bioavailability of mercury, evaluating risk, and selecting appropriate remedial actions.

The importance of mercury speciation can be illustrated with two examples. If a soil contains only organic mercury compounds, which generally are highly soluble, the mercury is likely to be highly bioavailable. In addition, a relatively simple technology—such as soil washing—may be a viable means of remediating the soil. In contrast, if all the mercury is present as insoluble mercuric sulfide, the bioavailability will be low and will result in a less stringent site-specific cleanup standard for soil across the site. However, due to the same physical properties, mercuric sulfide may be more difficult to remove from the soils, and a more aggressive remedial technology may be required to meet the cleanup standard.

Mercury speciation in soils can be evaluated using three general methods:

- Sequential extractions
- Electron microprobe analysis
- Heavy mineral separations.

Recently, several investigators have focused on developing sequential extraction procedures to quantitatively evaluate the speciation of mercury in soils (Revis et al. 1989; Miller 1993; Sakamoto et al. 1992). Application of the procedures of each investigator to the same samples from Oak Ridge, Tennessee showed mercury occurring predominantly as elemental mercury and mercuric sulfide minerals (Barnett et al. 1994). However, the relative proportions of the two species did not agree among procedures, indicating that the extractions were either not fully effective in removing specific mercury compounds or not fully specific in extracting individual mercury species. This problem is common to sequential extraction methods (Belzile et al. 1989). All the extraction techniques gave similar levels of organic mercury in soils. However, the method of Miller (1993), developed by the EPA, generally found much less elemental mercury and mercuric sulfide

than the other two extraction procedures. The method of Sakamoto et al. (1992) tended to have poor recovery for elemental mercury. The method of Revis et al. (1993) showed higher recoveries of mercuric sulfide and elemental mercury, but it does not include a procedure for mercuric oxides and carbonates (acid-soluble mercury). Given the drawbacks of all the methods, a procedure combining the most effective aspects of each is likely to produce the most reliable results.

Electron microprobe analysis is a mineralogical technique that provides direct visual evidence of the mercury phases present in soil. The microprobe is used to determine the distribution of the specific mercury-bearing phases in the soil and can be used to qualitatively, rather than quantitatively, confirm the visible amounts of these phases. Microprobe analysis is particularly useful for documenting the morphology and composition of the metal-bearing grains and photographing these relations. This information can then be used to assess the bioavailability of the metal in the soil (Davis et al. 1993). The microprobe technique, however, is not without limitations. It is difficult to quantitatively determine the entire mass of mercury in the soil, because some phases may be distributed throughout the soil at low concentrations that are difficult to quantify. Also, it may be difficult to detect mercury-bearing phases in soils with very low levels of mercury. Finally, in preparing the samples for microprobe analysis, some of the organic and elemental mercury may be lost due to volatilization, potentially skewing the results.

Heavy mineral separation of mercury-bearing phases from soil is an additional mineralogical technique to provide information on the distribution of mercury species in the soil. This technique involves grinding a soil sample and mixing in a high-density liquid such as methylene iodide (specific gravity 3.325). In this liquid, silicate minerals and organic materials will float, and heavy mercury-bearing phases will settle out, along with other heavy minerals. This heavy mineral concentrate can then be analyzed visually, by microprobe, and by x-ray diffraction to detect mercury-bearing phases. The results of heavy mineral separations provide visual confirmation of the mercury speciation results. More importantly, mineral separations provide conclusive evidence of the presence or absence of significant concentrations of all mercury phases in site soils.

The interpretation of mercury speciation data will focus on determining the internal consistency and applicability of the sequential extraction results. Total mercury concentrations will be compared to the sum of the individual mercury species determined in each soil. Also, duplicate analyses will be compared. Speciation results from soils spiked with known quantities of mercury species will be evaluated to determine the portion of mercury re-extracted by the speciation procedures. Once the sequential extraction data are analyzed, these results will be compared to microprobe and heavy mineral separation data, to evaluate whether the different speciation techniques provide consistent results. If the results are different, the discrepancies will be evaluated in light of the known limitations of the analytical methods, to develop a realistic assessment of the distribution of mercury species in soils from the Ames Street site.

# MERCURY BIOAVAILABILITY

In humans, an orally administered dose of a compound is seldom completely absorbed, and differences in the extent of absorption of orally administered compounds exist among different exposure media. For most compounds, the toxicity values derived by the U.S. Environmental Protection Agency (EPA) are not adjusted to absorbed dose (i.e., the dose response evaluation is based on the administered dose). This procedure can lead to errors in assessing the risks of exposure to a particular chemical in a medium other than the one used in the toxicity or epidemiology studies on which the toxicity values are based. For example, the EPA's oral toxicity value, or reference dose (RfD), for inorganic mercury was derived from studies in which mercuric chloride dissolved in water was administered to laboratory animals. Because it is likely that most of the mercury at the Ames Street site is present in forms that are less soluble than mercuric chloride, absorption of mercury from ingested site soils will be reduced compared to mercuric chloride. If these differences in mercury bioavailability are not accounted for, risks associated with ingestion of mercury in site soils will be overestimated. The adjustment factor to correct for differences in absorption from different exposure media is termed the bioavailability adjustment factor (BAF). This fractional value is used to adjust the dose or intake so that it is expressed in the same terms as the doses used to generate the toxicity values.

Substantial evidence exists that mercury solubility and bioavailability vary with mercury species. Studies in rodents suggest that 10 to 20 percent of mercuric chloride is absorbed from single oral doses. Several studies comparing tissue levels in rodents after single or repeated doses of mercuric chloride and mercuric sulfide have concluded that mercuric sulfide is very poorly absorbed. In 1993, the EPA reviewed available studies on the toxicity and bioavailability of mercuric sulfide in response to a petition for a provisional mercuric sulfide reference dose for an Oak Ridge, Tennessee, site. At that time, the EPA concluded that insufficient information was available to derive a separate RfD for mercuric sulfide, but they did note that comparison of relative tissue levels of mercury in animal studies suggested that mercuric sulfide was 30 to 80 times less bioavailable than mercuric chloride. Thus, a relative BAF of 1/30 to 1/80 (0.03–0.01) may be appropriate when

applying toxicity values for mercuric chloride to mercuric sulfide. Little or no information is available on the oral absorption of other mercury compounds or elemental mercury relative to mercuric chloride; however, other mercury species are likely to be more bioavailable than mercuric sulfide. The bioavailability of mercury species in soil may be further reduced due to interactions with soil constituents. Thus, site-specific BAFs will vary, depending on the mix of mercury species present at the site and the composition of other soil constituents. Because a variety of mercury species may be present in soils at the Ames Street site, site-specific mercury BAF(s) will be determined based on a study of site soil samples.

For the purpose of this study, bioaccessible mercury is defined as the fraction of mercury that is soluble in the gastrointestinal (GI) tract and is available for absorption, while bioavailability is defined as the fraction of mercury that is absorbed into the bloodstream. Because mercury in soil must be solubilized in order to become bioavailable, mercury bioaccessibility is a precursor to, and provides an upper-bound estimate of, mercury bioavailability.

The PTI *in vitro* test has been utilized to assay the bioavailability of lead and arsenic in soils, and has been validated in several animal models (Ruby et al. 1993, 1995; Appendix A, Attachment D). For this study, the standard PTI *in vitro* test has been modified to provide a test system appropriate for mercury bioaccessibility evaluation (see Methodology section, below).

In vitro assays similar to the PTI test have been employed at several other sites to estimate site-specific bioavailability of mercury in soil. At the Almaden Quick Silver County Park in Los Gatos, California, the form of mercury present in site soils, which resulted from mining and ore processing (predominantly mercuric sulfide), was experimentally measured to be from 0.03 to 9.4 percent as soluble as mercuric chloride, in a simulated gastrointes-tinal environment (CDM 1992). The Los Gatos site samples were tested using a leaching procedure designed to emulate the human gastrointestinal system. Two-hundred milligrams (mg) of sample (sieved to <2 mm) was added to 480 milliliters (mL) of a pH-2.5 solution of dilute hydrochloric acid (HCl) in 500-mL bottles, and the bottles were agitated for 4 hours to simulate conditions in the human stomach. The human intestine was emulated by adjusting the pH of the solution to 6.5 using sodium hydroxide, and agitating for an additional 4 hours. At the end of the simulated stomach and intestinal phases, aliquots of the solutions were filtered (0.45  $\mu$ m) and analyzed for their mercury content. Based on the results of this *in vitro* assay, the Santa Clara County Parks and Recreation Department and California state regulatory authorities agreed to use a BAF of 0.3 for the Los Gatos site.

An *in vitro* procedure nearly identical to the one above was used to evaluate the solubility of mercury in soil samples collected at Oak Ridge National Laboratory in Tennessee (Barnett and Turner 1995). The experimental procedure was altered in that the soil samples were pulverized after sieving, and only the < 180-µm size fraction was subjected to the leaching procedure. For 19 of the 20 samples, the mercury in soils was determined to be from 0.3

to 14.2 percent soluble (average of 3.2 percent). One sample, the only sample with detectable mercury vapor in the sample headspace, contained 45.9 percent soluble mercury by this *in vitro* method. Mercuric chloride was determined to be 100 percent soluble in the *in vitro* test system. Based on these analyses, the EPA accepted a site-specific BAF of 0.1 for mercury in soils (DOE 1995).

# METHODOLOGY

#### **OVERVIEW**

Mercury speciation analysis will be conducted on selected soil samples to determine the forms of mercury present. The speciation data will indicate the predicted solubility of mercury in the soil samples and will provide a mechanistic explanation for the estimated bioavailability of mercury from the Ames Street site soils. An *in vitro* test that replicates human gastrointestinal tract chemistry and function will be performed on selected samples following speciation to determine the fraction of mercury in soil samples that is soluble and available for absorption in the gastrointestinal tract (i.e., the fraction that is bioaccessible). Because the bioaccessible fraction of mercury provides an upper-bound estimate on the bioavailability of ingested mercury, the *in vitro* test data for the Ames Street site soil samples can be used to develop conservative site-specific BAF(s). The resulting BAF(s) can then be used to adjust the soil mercury intake estimates and to develop revised site-specific soil remediation goals.

#### MERCURY SPECIATION ANALYSIS

As described above, both sequential extractions and mineralogical techniques for determining the speciation of mercury in soils are not without limitations. In order to address these limitations, PTI will conduct a coupled study of mercury speciation that combines sequential extractions and mineralogical techniques. The use of more than one method will allow for data cross-checking and validation, which will increase the reliability of study results. Also, the combined approach will allow for better quantification of mercury species distribution, especially organic and elemental mercury in soils, and the mineralogical photographs will provide visual evidence of mercury distribution.

Total mercury concentrations will be measured in all of the soil samples. Speciation analysis will then be conducted on selected samples with total mercury concentration >10 mg/kg. Prior to speciation analysis, mercury in the headspace of the sample bottles will be determined in the laboratory at room temperature using a Jerome mercury vapor analyzer, Model 431X. Speciation analysis will be performed on dry samples. Because of the volatile nature of mercury, the samples will be air dried at room temperature, instead of oven dried. The speciation analysis will first be performed using a sequential extraction procedure, whereby samples are extracted with chloroform to analyze for organic mercury, and then treated with 0.1 M H<sub>2</sub>SO<sub>4</sub> to extract mercuric oxide and carbonate minerals. The remaining sample will be analyzed for total mercury (i.e., elemental mercury + mercuric sulfide) and then heated to extract elemental mercury. Mercuric sulfide will be the concentration of total mercury left after heating. Elemental mercury will be determined by subtracting the mercuric sulfide concentration from the total mercury concentration prior to heating. In addition to the sequential extractions, speciation will be determined by microprobe and heavy mineral separations. These results will provide corroborative visual evidence of various mercury phases in site soils. Quality assurance and quality control (QA/QC) procedures will be implemented, including collection and analysis of sample duplicates, spiked soils, and sample blanks. The results of the various mercury speciation studies will be used to identify samples for further *in vitro* studies to assess mercury bioavailability.

# IN VITRO BIOAVAILABILITY TESTING

The *in vitro* procedure is described in detail in the SAP (Appendix A). Extracts from the *in vitro* procedure will be submitted to Columbia Analytical Services (Kelso, Washington) for mercury analysis. Analytical methods and laboratory quality assurance measures are described below. The *in vitro* test results for selected Ames Street site samples will be used to develop a site-specific BAF for mercury, based on the average fraction of mercury solubilized from the soils, corrected for recovery of mercuric chloride in the assay. Multiple BAFs may be developed for different areas of the site or different mercury forms in soil, based on the speciation data, if the testing results support this approach to data interpretation.

QA/QC procedures will be implemented by spiking two stomach solution samples. Instead of adding a soil sample to the reaction vessel, a known amount of an aqueous solution of reagent-grade mercuric chloride (HgCl<sub>2</sub>) will be added as a spike. The QA/QC procedure will then follow the *in vitro* test method as described in the SAP (Appendix A). The samples will be spiked with a low concentration of HgCl<sub>2</sub>, relative to total soil mercury concentrations. The duplicate spike solutions will be evaluated to determine recovery of mercuric chloride in the *in vitro* test.

# MERCURY SPECIATION ANALYSIS

# SAMPLE PREPARATION

Based on the total mercury results, samples will be selected from locations at the Ames Street site for speciation analysis. The samples selected must contain enough mercury (i.e., > 10 mg/kg) to perform the speciation analyses and to quantify the spatial distribution of mercury in Ames Street site soil. Speciation will be performed to determine organic mercury, mercury oxide, elemental mercury, and mercuric sulfide. Personnel at the PTI laboratory in Boulder, Colorado will perform the sequential extractions.

# MERCURY VAPOR ANALYSIS

To measure the headspace mercury in the 16-oz soil sampling bottles, attach the Jerome Model 431X mercury vapor analyzer to the septum. If the Jerome Model 431X mercury vapor analyzer reads the upper detection limit of  $1 \text{ mg/m}^3$ , then a Jerome Dilution Module can be used. This device will dilute the headspace mercury so that a percentage of mercury vapor can be detected. After reading the headspace mercury, the samples will be air dried.

# SPECIATION EXTRACTION METHODS

The methods of Revis et al. (1989) and Sakamoto et al. (1992) will be followed for the extraction of mercury species from the Ames Street site soils. Modifications have been made to both procedures in order to combine the two methods.

The PTI laboratory will be set up to perform extractions of mercury species. All procedures will be performed under a vapor hood. Sample extracts for each mercury species will be sent to Columbia Analytical Services (CAS) for mercury analyses.

# Organic Mercury and Acid-Soluble Mercury

Sakamoto et al. (1992) developed a method for differential determination of organic mercury and acid-soluble mercury, which includes mercury(I) oxide, mercury(II) oxide, mercury carbonates, and mercuric chloride, based on the successive extraction of these mercury compounds with chloroform and sulfuric acid. The mercury in each extract is determined by cold vapor atomic adsorption spectroscopy (CVAAS).

The method for extracting organic mercury from soils is as follows:

- Place 20 mL of chloroform and 1-5 g of sediment sample in a 50-mL glass centrifuge tube
- Stopper the tube and shake in a shaker for 2 minutes
- Centrifuge at 3000 rpm for 2 minutes
- Transfer the chloroform phase into a separatory funnel
- Repeat the extraction with another 20 mL of chloroform
- Add 3 mL of 0.01 M sodium thiosulfate solution to the combined chloroform extract in the separatory funnel and shake for 2 minutes
- Send the aqueous solution to CAS to determine the mercury concentration by CVAAS.

The method for extracting acid-soluble mercury from soils is as follows:

- After completion of the organic mercury extraction, leave the 50-mL glass centrifuge tube unstoppered to evaporate the residual chloroform to dryness
- Add 10 mL of 0.1 M sulfuric acid to the residue
- Stopper the centrifuge tube and shake it in the shaker for 2 minutes
- Centrifuge it at 3000 rpm for 2 minutes
- Send the supernatant to CAS to determine the mercury concentration by CVAAS

Air dry and save the residue in the centrifuge tube for the elemental mercury extraction.

# Elemental Mercury (Hg<sup>0</sup>) and Mercuric Sulfide

To separate and determine elemental mercury  $(Hg^0)$  and mercuric sulfide (HgS), use the residue remaining after extracting organic mercury and mercury oxide, and follow the method of Revis et al. (1989):

- Send a residue split to CAS to determine ΣHg (i.e., Hg<sup>0</sup> + HgS) using CVAAS
- Thinly spread a 5-g sample of homogenized residue on a stainless steel tray
- Place the tray in a continuously aerated oven at 150 °C for 5 days
- Digest the sample with aqua regia acid
- Send the sample to CAS to determine  $\Sigma$ Hg by CVAAS
- The amount of HgS in the sample is the amount of  $\Sigma$ Hg after roasting.
- The amount of  $Hg^0$  in the sample is the difference between the  $\Sigma Hg$  prior to roasting and the amount of HgS after roasting.

### **MICROPROBE ANALYSIS**

Polished sample "pucks" will be prepared at the Laboratory for Geological Studies, University of Colorado, Boulder, for electron microprobe analysis by embedding 4 grams of sample in epoxy within a sample mold, setting the mold to cure at room temperature, and grinding a flat surface on the sample side to expose as much sample as possible. Successive polishing steps will employ a 600-grit wet/dry abrasive paper stretched across a glass plate, 15- $\mu$ m and 6- $\mu$ m diamond on a cloth pad fixed to a steel lap, and finally 0.1- $\mu$ m diamond on a felt pad fixed to a steel lap. All polishing steps will use kerosene to avoid dissolution of water-soluble Hg phases, and all polishing will be performed at low speeds to avoid plucking of the sample grains. Finally, sample pucks will be cleaned in an ultrasonic cleaner with isopropyl alcohol, air dried, and placed in a carbon coater, where a thin layer of carbon will be sputtered onto the surface of each puck. Electron microprobe analysis (EMPA) will also be conducted at the Laboratory for Geological Studies, University of Colorado, Boulder, on a JEOL 8600 electron microprobe operating at 15 kV with a 20-nA specimen current and a 1-um beam, according to the methods described in Attachment C, as adapted for mercury speciation. Quantitative mineralogic data will be collected using wavelength dispersive spectrometers and mineral standards, and corrected using Phi Rho Z parameters. The Hg-bearing particles will be identified using a combination of energy dispersive detection (EDS), wavelength dispersive detection (WDS), and backscatter electron image detection (BEI). Initially, spectra are generated for each grain that allow identification of all elements with an atomic mass greater than or equal to that of carbon. Subsequently, the elemental proportions are quantified using standards, and the mineral proportions are identified based on the equivalent weight of the oxide. Therefore, the identifications provide quantitative stoichiometric ratios from which the mineral identity can be calculated. The relations between Hg-bearing phases will be established from BEI images and WDS/EDS analyses as necessary. Representative BEI photomicrographs of identified phases and their associations will be produced, with scale bar, magnification, sample identification, and phase identification recorded on each photomicrograph.

Individual Hg-bearing particles will be analyzed (representing one point count each) until a minimum of 100 particles has been evaluated, or 5 hr of machine time has been spent on the analysis. Point counts will be made by traversing each sample from left to right and top to bottom in a grid pattern, with each vertical displacement moving only to the adjacent field of view. Magnification settings of 40 to  $100 \times$  and 300 to  $600 \times$  will be used; the latter magnification allows analysis of the smallest identifiable (1–2 µm) phases. The grain size of each Hg carrier will be determined by measuring the dimension of the long axis. Percent compositions of Hg phases in each sample will be determined by summing the total area of all Hg grains and dividing the area for each phase by the total area.

## HEAVY MINERAL SEPARATIONS

Heavy minerals will be identified in the PTI laboratory in Boulder, Colorado by shaking 5 g of ground and sieved soil in 100 mL of methylene iodide (specific gravity 3.325) in a separatory funnel. The samples will be allowed to settle until the liquid clears. The heavy fraction will be dispensed into a beaker and triple washed with acetone. The heavy fraction will be collected, then examined and photographed under a binocular microscope. This heavy fraction will also be analyzed by powder x-ray diffraction at the Laboratory for Geological Studies, University of Colorado, Boulder. Finally, the heavy mineral fraction will be analyzed by electron microprobe in a fashion similar to the bulk soil samples. Results of this visual observation of mercury species in soils will be tabulated and used to assist in evaluating the sequential extraction mercury speciation.

# **ANALYTICAL PROCEDURES**

Sample extracts will be shipped on ice under strict chain of custody, in accordance with SOP-5, to CAS and the PTI laboratory in Boulder, Colorado. Soil samples for total mercury analysis will be analyzed by CVAAS (Method 7471A, U.S. EPA 1991), which includes acid digestion. Sample results will be reported on a dry-weight basis. Aqueous-phase extracts will be analyzed for mercury by a similar CVAAS methodology (Method 7470A, U.S. EPA 1991). Samples also will be analyzed for total sulfides, total carbonates, and TOC.

# IN VITRO BIOAVAILABILITY TESTING

#### SAMPLE PREPARATION

Selected soil samples will undergo the *in vitro* procedure to estimate relative mercury bioavailability. The samples will be prepared in PTI's Boulder, Colorado laboratory by air drying and sieving to  $<250 \,\mu\text{m}$ . The  $<250 -\mu\text{m}$  size fraction has been selected for this study because this particle size has been observed to adhere to children's hands, and is the fraction of soil most likely to be ingested (Duggan and Inskip 1985).

A split of each sieved sample ( $<250 \,\mu$ m) also will be submitted for determination of total mercury and sulfur, and total organic carbon (TOC), by the analytical method described below.

# **IN VITRO TEST METHOD**

The *in vitro* test is designed to determine the fraction of mercury that is solubilized and available for absorption in the gastrointestinal tract. Development of the test, and the rationale for selection of representative parameters, are described in detail in the literature included in Attachment D. The *in vitro* method was designed to replicate gastrointestinal-tract parameters for a human child, including stomach and small intestinal pH and chemistry, soil-to-solution ratio, stomach mixing, and stomach emptying rate. The method is implemented in two phases, simulating the passage of ingested soil from the acidic environment of the stomach to the near-neutral conditions of the small intestine.

Because of the concern for potential loss of volatile mercury from the reaction vessel, the *in vitro* test methodology used to estimate the bioavailability of arsenic and lead has been altered for mercury bioavailability testing. The reaction will be carried out in a sealed container, to minimize potential loss of volatile mercury. Argon gas will be introduced into the reaction vessel at the beginning of the *in vitro* assay to purge it of atmospheric oxygen, to simulate the anoxic conditions present in the gastrointestinal tract. A gold trap will be placed on the inflowing argon gas to remove mercury from the inflowing gas.

The *in vitro* test will be conducted according to the following method (all chemicals from Sigma Chemical Company, unless otherwise noted):

Prepare the stomach solution by adding the following compounds to 1 L of deionized water (stirred continually on a stir plate):

1.25 g pepsin (50 mg, activity of 800-2,500 units/mg)
0.50 g citrate (Fisher Chemical Co.)
0.50 g malate (Aldrich Chemical Co.)
420 μL lactic acid (synthetic syrup 85 percent w/w)
500 μL acetic acid (97 percent w/w; Fisher Chemical Co.).

- Adjust the pH of the stomach solution to 2.5 by adding a measured volume of concentrated HCl.
- Add 150 mL of stomach solution to the 200-mL acrylic reaction vessel (see Attachment D).
- Sparge the stomach solution with argon for 5 minutes to remove oxygen.
- Measure the Eh of the stomach solution.
- Sparge the stomach solution with argon for an additional 2 minutes.
- Add 1.5 g of soil and seal the reaction vessel.
- Submerge the reaction vessel approximately half-way into a temperature-controlled water bath heated to maintain a constant 37 °C in the reaction vessel (Attachment D)
- Allow the soil/stomach solution to stand (no agitation) for 10 minutes.
- Stir the mixture with a plastic propeller stir rod mounted in a rheostatcontrolled motor (Arrow Engineering Model 1750 motor on a rheostat setting of 2, resulting in approximately 150 rpm for the stir rod).
- Check the pH at 5-minute intervals, and readjust to pH 2.5 with HCl if necessary.
- Collect 5-mL samples at 30 and 60 minutes, using a stainless-steel hypodermic syringe to pierce the sampling septum. Centrifuge the 5-mL samples at

approximately 2500 xg for 25 minutes and decant the supernatant for analysis.

- At 1 hour, titrate the solution to pH 7.0 by adding a 5-in length of dialysis tubing containing approximately 1 g of NaHCO<sub>3</sub> to each reaction vessel. The dialysis tubing is added without exposing the reaction vessel to atmospheric oxygen.
- Allow the pH of the reaction vessel solution to increase slowly to  $7.0 \pm 0.2$  before removing the dialysis bag.
- Dissolve 260 mg of bile salts and 75 mg of pancreatin in 10 mL of deionized water and inject the fluid into the reaction vessel through the septum.
- Using a stainless-steel hypodermic syringe, obtain 5 mL of intestinal-phase sample through the septum at 1.0 and 3.0 hours after the reaction fluid reaches equilibrium at pH 7. Centrifuge each sample at approximately 2500 xg for 25 minutes and decant the supernatant for analysis.
- Measure and record the concentration of mercury vapor in the headspace of the reaction flask by connecting a mercury vapor analyzer (Jerome Model 431X) to the reaction vessel, and opening the sealed sampling septum to allow air flow through the reaction vessel.
- After the final sample is collected, measure and record the pH and Eh of the flask contents.
- Measure and record the final volume of the flask contents in a graduated cylinder.
- Analyze each of the two stomach-phase and the two small-intestinal-phase samples for mercury concentration, by the analytical method described below.

# IN VITRO TEST SYSTEM EVALUATION

Prior to analyzing samples for the purpose of developing a site-specific bioavailability adjustment factor (BAF), site soil samples will be evaluated using the *in vitro* test to determine the potential for loss of mercury during the test (e.g., from volatilization, or mercury adhering to the test cell walls). In a mass balance experiment, two site soil samples will be tested in triplicate in the assay. The absolute quantity of mercury recovered in the fluid, solid, and vapor phases from the reaction vessel after the assay is completed (analytical procedures described below) will be compared to the quantity of mercury determined to be present in the soil sample before the *in vitro* assay (estimated from analysis of a split of the soil sample), to evaluate recovery of mercury from the test system.

# ANALYTICAL PROCEDURES

All *in vitro* test samples will be shipped to CAS under strict chain of custody. Soil samples for total mercury analysis will be analyzed by CVAAS (Method 7471A, U.S. EPA 1991), which includes acid digestion. Sample results will be reported on a dry-weight basis. *In vitro* extracts will be analyzed for mercury by a similar CVAAS methodology (Method 7470A, U.S. EPA 1991).

# QUALITY CONTROL PROCEDURES

### MERCURY SPECIATION ANALYSIS

Quality Assurance and Quality Control (QA/QC) samples will be collected in accordance with SOP-6 to provide checks on sample collection and handling procedures, and analytical accuracy and precision. Field quality control will include field duplicates, external contamination blanks (ECBs), cross-contamination blanks (CCBs), and standard reference materials (SRMs). Laboratory quality control will include blank, spike, and duplicate samples. PTI laboratory control samples will be prepared as specified below.

# IN VITRO TESTING

*In vitro* test quality control samples will include two soils that will be run through the procedure in triplicate. In addition, two *in vitro* tests with a soluble mercury spike will be performed to evaluate matrix spike recovery. Finally, a blank stomach solution spiked with a known amount of soluble mercury will be submitted as a blind laboratory control sample.

# LABORATORY QUALITY CONTROL

The specific quality control procedures to be performed for the analyses of mercury and other metals are cited in U.S. EPA (1991). The laboratory quality control samples will include a preparation blank, laboratory control sample, laboratory duplicate, and matrix spike sample for each batch of 20 samples or each digestion group, whichever is more frequent.

For every 20 or fewer samples of a similar matrix analyzed by a particular method, the laboratory will submit a complete data package containing the following data and supporting information:

- A cover letter discussing the analytical procedures used and the problems encountered during sample analysis (if any).
- Sample log listing the identifying sample numbers and corresponding laboratory numbers (if applicable) for all samples included in the data package.
- Chain-of-custody forms for all samples included in the data package.
- Analyte concentrations with reporting units identified.

- The original raw laboratory data, bench sheets, and instrument printouts for all samples, including all laboratory quality control samples and blanks.
- Final dilution volumes, sample sizes, wet-to-dry ratios, and any other information—including formulas—required to derive the final reported sample concentration from the raw laboratory data.
- Final analytical results, with appropriate concentration units, for all *in vitro* and quality control *in vitro* samples, as well as laboratory quality control samples when required (i.e., laboratory method blanks, laboratory control samples [LCSs], and matrix spike samples).
- Instrument detection limits for each analyte in each package.
- A summary form indicating which method blanks are associated with each batch of samples for every analysis.
- Summarized recovery and/or relative percent difference (RPD) results for all laboratory quality assurance and quality control (QA/QC) checks, including all laboratory spike samples, calibration check samples, laboratory duplicate samples, method blanks, and LCSs for each analysis.
- Appropriate laboratory data qualification codes and their definitions.
- Summary forms for all initial and continuing instrument calibrations performed that apply to the project samples in each data package. These summaries must include the exact concentrations for the calibration standards and the acceptable linear calibration ranges for each instrument used. Some measure of the linearity of the initial calibration curve also must be determined and reported, as specified in the method.

## REFERENCES

Barnett, M.O., and R.R. Turner. 1995. Bioavailability of mercury in East Fork Poplar Creek soils. Environmental Sciences Division, Oak Ridge National Laboratory, Oak Ridge, TN. Y-12 ER Report.

Barnett, M.O., L.A. Harris, R.R. Turner, T.J. Henson, R.E. Melton, and R.J. Stevenson. 1994. Characterization of mercury species in contaminated floodplain soils. Water Air Soil Pollut.

Belzile, N., P. Lecomte, and A. Tessler. 1989. Testing readsorption of trace elements during partial chemical extractions of bottom sediments. Environ. Sci. Technol. 23:1015–1020.

CDM. 1992. Final report. Risk assessment, Almaden Quicksilver County Park, Volume II - Appendices. Appendix E. Camp Dresser & McKee, Inc., Denver, CO, May 29, 1992.

Davis, A., J.W. Drexler, M.V. Ruby, and A. Nicholson. 1993. Micromineralogy of mine wastes in relation to lead bioavailability, Butte, Montana. Environ. Sci. Technol. 27(7):1415-1425.

DOE. 1995. Record of Decision for Lower East Fork Poplar Creek. U.S. Department of Energy, Office of Environmental Restoration and Waste Management. Prepared by Jacobs ER Team, Oak Ridge, TN. DOE/OR/02-1370&D1. May.

Miller, E.L. 1993. Speciation of mercury in soil. EPA Contract 68-CO-0049. U.S. Environmental Protection Agency, Quality Assurance and Methods Development Division, Environmental Monitoring Systems Laboratory, Office of Research and Development, Lockheed Environmental Sciences and Technology Company, Las Vegas, NV.

Revis, N.W., T.R. Osborne, D. Sedgley, and A. King. 1989. Quantitative method for determining the concentration of mercury(II) in soils and sediments. Analyst 114:823-825.

Ruby, M.V., A. Davis, T.E. Link, R. Schoof, R.L. Chaney, G.B. Freeman, and P. Bergstrom. 1993. Development of an *in vitro* screening test to evaluate the *in vivo* bioaccessibility of ingested mine-waste lead. Environ. Sci. Technol. 27(13):2870–2877.

Ruby, M.V., A. Davis, T.E. Link, R. Schoof, S. Eberle. 1995. Development of a physiologically based test to estimate lead bioavailability. The Toxicologist 15(1):135.

Sakamoto, H.T., T. Tomiyasu, and N. Yonehara. 1992. Differential determination of organic mercury, mercury(II) oxide and mercury(II) sulfide in sediments by cold vapor atomic adsorption spectrometry. Anal. Sci. 8:35-39.

U.S. EPA. 1991. Test methods for evaluating solid waste. Physical/chemical methods. SW-846. Third ed. U.S. Environmental Protection Agency, Washington, DC.